Table of Contents

First Revised Sheet No. 0	14
Seventh Revised Sheet No. 1	15
Second Revised Sheet No. 1A	16
Third Revised Sheet No. 1B	17
First Revised Sheet No. 1C	18
First Revised Sheet No. 2	19
Fifth Revised Sheet No. 3	20
Fourth Revised Sheet No. 3A	21
Sixth Revised Sheet No. 3B	22
Seventieth Revised Sheet No. 4	23
Seventieth Revised Sheet No. 5	25
Seventieth Revised Sheet No. 6	27
Seventy-Third Revised Sheet No. 7	28
Seventy-Third Revised Sheet No. 8	29
Thirteenth Revised Sheet No. 9	30
Ninth Revised Sheet No. 10	31
Fifteenth Revised Sheet No. 11	32
Fifteenth Revised Sheet No. 12	33
Eleventh Revised Sheet No. 13	34
Fourteenth Revised Sheet No. 14	35
Forty-Ninth Revised Sheet No. 15	36
Second Revised Sheet No. 16	38
Fourteenth Revised Sheet No. 17	39
Second Revised Sheet No. 18	41
Twenty-Seventh Revised Sheet No. 19	42
First Revised Sheet No. 20	43
Original Sheet No. 21	44
Original Sheet No. 22	45
Original Sheet No. 23	46
Sub First Revised Sheet No. 24	47
Original Sheet No. 25	48
Original Sheet No. 25A	49
Original Sheet No. 25B	50
Original Sheet No. 25C	51
Original Sheet No. 25D	52
Original Sheet No. 25E	53
Original Sheet No. 25F	54
Original Sheet No. 25G	55
Original Sheet No. 25H	56
Original Sheet No. 25I	57
Original Sheet No. 25J	58
Original Sheet No. 25K	59
Original Sheet No. 25L	60
Original Sheet No. 25M	61
Original Sheet No. 25N	62

Original Sheet No. 250	63
Original Sheet No. 25P	64
Original Sheet No. 25Q	65
Original Sheet No. 25R	66
First Revised Sheet No. 26	67
Original Sheet No. 27	68
Sheet Nos. 28-29	69
Original Sheet No. 30	70
First Revised Sheet No. 31	71
Original Sheet No. 32	72
Substitute Original Sheet No. 33	73
Fourth Revised Sheet No. 34	74
Second Revised Sheet No. 35	75
Sub Original Sheet No. 35A	76
Third Revised Sheet No. 35B	77
First Revised Sheet No. 36	78
Original Sheet No. 37	79
First Revised Sheet No. 38	80
Substitute Original Sheet No. 39	81
Substitute Original Sheet No. 40	82
Second Revised Sheet No. 41	83
Fourth Revised Sheet No. 42	84
Second Revised Sheet No. 42A	85
Third Revised Sheet No. 42B	86
Fourth Revised Sheet No. 43	87
Original Sheet No. 44	88
Original Sheet No. 45	89
First Revised Sheet No. 46	90
Second Revised Sheet No. 47	91
Fourth Revised Sheet No. 48	92
First Revised Sheet No. 49 Second Revised Sheet No. 50	93
	94
Original Sheet No. 51	95
Original Sheet No. 52	96
Substitute Original Sheet No. 53	97
Original Sheet No. 54	98
Original Sheet No. 55	99
Original Sheet No. 56	100
Original Sheet No. 57	101
Substitute Original Sheet No. 58	102
Third Revised Sheet No. 59	103
Second Revised Sheet No. 60	104
Original Sheet No. 61	105
Original Sheet No. 62	106
Fifth Revised Sheet No. 63	107
Third Revised Sheet No. 64	108

Original Sheet No. 65	109
Original Sheet No. 66	110
Original Sheet No. 67	111
Substitute Original Sheet No. 68	112
Substitute Original Sheet No. 69	113
Third Revised Sheet No. 70	114
Third Revised Sheet No. 71	115
Sub Original Sheet No. 71A	116
Substitute Original Sheet No. 72	117
Substitute Original Sheet No. 73	118
Original Sheet No. 74	119
Third Revised Sheet No. 75	120
Third Revised Sheet No. 76	121
Original Sheet No. 76A	122
Substitute Original Sheet No. 77	123
Substitute Original Sheet No. 78	124
Original Sheet No. 79	125
Original Sheet No. 80	126
First Revised Sheet No. 81	127
Fourth Revised Sheet No. 82	128
Fourth Revised Sheet No. 83	129
Fourth Revised Sheet No. 84	130
Fourth Revised Sheet No. 85	131
Second Revised Sheet No. 86	132
Original Sheet No. 87	133
First Revised Sheet No. 88	134
Original Sheet No. 89	135
Original Sheet No. 90	136
Fourth Revised Sheet No. 91	137
Fourth Revised Sheet No. 92	138
Sub Original Sheet No. 92A	139
Substitute Original Sheet No. 93	140
First Revised Sheet No. 94	141
Substitute Original Sheet No. 95	142
Substitute Original Sheet No. 96	143
Second Revised Sheet No. 97	144
Second Revised Sheet No. 98	145
Fourth Revised Sheet No. 99	146
Second Revised Sheet No. 100	147
First Revised Sheet No. 100A	148
Sixth Revised Sheet No. 101	149
Original Sheet No. 101A	150
Sub Second Revised Sheet No. 102	151
First Revised Sheet No. 102A	152
Second Revised Sheet No. 103	153
Sub First Revised Sheet No. 104	154

1 at Day Original Shoot No. 105	155
1st Rev Original Sheet No. 105	155
1st Rev Original Sheet No. 106	156 157
1st Rev Original Sheet No. 107	157
1st Rev Original Sheet No. 108	158
First Revised Sheet No. 109	159
Original Sheet No. 110	160
Original Sheet No. 111	161
Original Sheet No. 112	162
Original Sheet No. 113	163
Original Sheet No. 114	164
Original Sheet No. 115	165
Original Sheet No. 116	166
Original Sheet No. 117	167
Sheet Nos. 118-199	168
Original Sheet No. 200	169
First Revised Sheet No. 201	170
Third Revised Sheet No. 202	171
Third Revised Sheet No. 203	172
Sixth Revised Sheet No. 203	173
Third Revised Sheet No. 204	174
Fourth Revised Sheet No. 205	175
Seventh Revised Sheet No. 206	176
Seventh Revised Sheet No. 207	177
Sixth Revised Sheet No. 208	178
Eighth Revised Sheet No. 209	179
Second Revised Sheet No. 210	180
Second Revised Sheet No. 211	181
Third Revised Sheet No. 212	182
Original Sheet No. 213	183
Original Sheet No. 214	184
Original Sheet No. 215	185
Original Sheet No. 216	186
Original Sheet No. 217	187
First Revised Sheet No. 218	188
Fifth Revised Sheet No. 219	189
Original Sheet No. 220	190
First Revised Sheet No. 221	191
Third Revised Sheet No. 222	192
First Revised Sheet No. 223	193
First Revised Sheet No. 224	194
Second Revised Sheet No. 225	195
Original Sheet No. 226	196
Original Sheet No. 227	197
First Revised Sheet No. 228	198
First Revised Sheet No. 229	199
First Revised Sheet No. 230	200
i iist Nevised Sileet NO. 250	200

First Revised Sheet No. 231	201
Second Revised Sheet No. 232	202
Original Sheet No. 233	203
Second Revised Sheet No. 234	204
Original Sheet No. 234A	205
Fifth Revised Sheet No. 235	206
Fourth Revised Sheet No. 236	207
Third Revised Sheet No. 237	208
Second Revised Sheet No. 238	209
Fifth Revised Sheet No. 239	210
Third Revised Sheet No. 239A	211
First Revised Sheet No. 239B	212
First Revised Sheet No. 239C	213
Second Revised Sheet No. 240	214
Third Revised Sheet No. 241	215
Fifth Revised Sheet No. 242	216
Fifth Revised Sheet No. 243	217
Fifth Revised Sheet No. 244	218
Sub Second Revised Sheet No. 245	219
Original Sheet No. 246	220
First Revised Sheet No. 246	221
Original Sheet No. 247	222
First Revised Sheet No. 248	223
First Revised Sheet No. 249	224
First Revised Sheet No. 250	225
Third Revised Sheet No. 251	226
Fifth Revised Sheet No. 252	227
Second Revised Sheet No. 253	228
Second Sub Original Sheet No. 254	229
First Revised Sheet No. 254A	230
Fourth Revised Sheet No. 255	231
Third Revised Sheet No. 255A	232
Original Sheet No. 256	233
Original Sheet No. 257	234
Sixth Revised Sheet No. 258	235
Original Sheet No. 259	236
Original Sheet No. 260	237
Original Sheet No. 261	238
Original Sheet No. 262	239
Second Revised Sheet No. 263	240
Original Sheet No. 263A	241
First Revised Sheet No. 263B	242
Substitute Original Sheet No. 264	243
Fifth Revised Sheet No. 265	244
Fifth Revised Sheet No. 266	245
First Revised Sheet No. 267	246

Substitute First Revised Sheet No. 268	247
Original Sheet No. 268A	248
First Revised Sheet No. 269	249
First Revised Sheet No. 270	250
Original Sheet No. 271	251
Second Revised Sheet No. 272	252
Third Revised Sheet No. 272A	253
First Revised Sheet No. 272B	254
Original Sheet No. 272C	255
Original Sheet No. 272D	256
Original Sheet No. 272E	257
Original Sheet No. 272F	258
Original Sheet No. 272G	259
Second Revised Sheet No. 273	260
Second Revised Sheet No. 273A	261
First Revised Sheet No. 274	262
Second Revised Sheet No. 275	263
Eighth Revised Sheet No. 276	264
Fourth Revised Sheet No. 277	265
Ninth Revised Sheet No. 278	266
Sixth Revised Sheet No. 278A	267
Second Revised Sheet No. 278B	268
Sixth Revised Sheet No. 279	269
Fourth Revised Sheet No. 280	270
Fifth Revised Sheet No. 281	271
Ninth Revised Sheet No. 282	272
First Revised Sheet No. 282A	273
Fourth Revised Sheet No. 283	274
First Revised Sheet No. 284	275
First Revised Sheet No. 285	276
Sub Second Revised Sheet No. 286	277
Original Sheet No. 286A	278
First Revised Sheet No. 287	279
Original Sheet No. 288	280
Fourth Revised Sheet No. 289	281
Substitute First Revised Sheet No. 290	282
First Revised Sheet No. 290A	283
Fourth Revised Sheet No. 291	284
First Revised Sheet No. 292	285
Original Sheet No. 293	286
Original Sheet No. 294	287
Third Revised Sheet No. 295	288
Third Revised Sheet No. 296	289
First Revisd Sheet No. 297	290
First Revised Sheet No. 298	291
First Revised Sheet No. 298A	292

Third Davised Cheet No. 200	202
Third Revised Sheet No. 299	293
Original Sheet No. 300	294
First Revised Sheet No. 301	295
Third Revised Sheet No. 302	296
Substitute First Revised Sheet No. 303	297
Substitute First Revised Sheet No. 304	298
First Revised Sheet No. 305	299
First Revised Sheet No. 306	300
Substitute First Revised Sheet No. 307	301
Original Sheet No. 308	302
Original Sheet No. 309	303
First Revised Sheet No. 310	304
Fourth Revised Sheet No. 311	305
Fourth Revised Sheet No. 312	306
First Revised Sheet No. 313	307
Third Revised Sheet No. 314	308
Fourth Revised Sheet No. 315	309
First Revised Sheet No. 316	310
Original Sheet No. 317	311
Second Revised Sheet No. 318	312
Second Revised Sheet No. 319	313
First Revised Sheet No. 320	314
Original Sheet No. 321	315
Second Revised Sheet No. 321A	316
Third Revised Sheet No. 321B	317
Second Revised Sheet No. 321C	318
Second Revised Sheet No. 321D	319
Second Revised Sheet No. 321E	320
First Revised Sheet No. 321F	321 322
Original Sheet No. 321G	
First Revised Sheet No. 321H	323
Original Sheet No. 3211	324
First Revised Sheet No. 321J	325
First Revised Sheet No. 321K	326
First Revised Sheet No. 321L	327
Third Revised Sheet No. 321M	328
Second Revised Sheet No. 321N	329
Substitute First Revised Sheet No. 3210	330
Substitute First Revised Sheet No. 321P	331
Third Revised Sheet No. 321Q	332
Third Revised Sheet No. 321R	333
Fourth Revised Sheet No. 321S	334
Third Revised Sheet No. 321T	335
Third Revised Sheet No. 321U	336
Fourth Revised Sheet No. 321V	337
First Revised Sheet No. 321W	338

Original Sheet No. 322 Original Sheet No. 323 Original Sheet No. 324 Original Sheet No. 324 Original Sheet No. 325 Original Sheet No. 325 Substitute Original Sheet No. 326 Fourth Revised Sheet No. 327 Fourth Revised Sheet No. 328 Fourth Revised Sheet No. 328 Fourth Revised Sheet No. 329 Original Sheet No. 330 Original Sheet No. 330 Original Sheet No. 330A Original Sheet No. 330A Original Sheet No. 330B Original Sheet No. 330B Original Sheet No. 331 Sibrith Revised Sheet No. 332 Sibrith Revised Sheet No. 333 Sibrith Revised Sheet No. 334 Sibrith Revised Sheet No. 335 Sibrith Revised Sheet No. 336 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 337 Sibrith Revised Sheet No. 338 Sibrith Revised Sheet No. 309 Sibrith Revised Sheet No. 300 Sibrith Revised Sheet No. 500 Sibrith Revised Sheet No. 500 Sibrith Revised Sheet No. 500 Sibrith Revised Sheet No. 501 Sibrith Revised Sheet No. 502 Sibrith Revised Sheet No. 503 Sibrith Revised Sheet No. 504 Sibrith Revised Sheet No. 505 Sibrith Revised Sheet No. 507 Sibrith Revised Sheet No. 508 Sibrith Revised Sheet No. 509 Sibrith Revised Sheet No. 509 Sibrith Revised Sheet No. 501 Sibrith Revised Sheet No. 503 Sibrith Revised Sheet No. 504 Sibrith Revised Sheet No. 505 Sibrith Revised Sheet No. 506 Sibrith Revised Sheet No. 507 Sibrith Revised Sheet No. 508 Sibrith Revised Sheet No. 509 Sibrith Revis	Original Sheet No. 321X	339
Original Sheet No. 324 341 Original Sheet No. 325 343 Fourth Revised Sheet No. 326 344 Fifth Revised Sheet No. 327 345 Fourth Revised Sheet No. 328 346 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Fourth Revised Sheet No. 330 349 Original Sheet No. 330A 349 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Full Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 Fifth Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 358 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 362 Sheet No. 341-499 363 First Revised Sheet No. 500 364 First Revised Sheet No. 501 365 First Revised Sheet No. 504 368 Second Revised	<u> </u>	
Original Sheet No. 324 342 Original Sheet No. 325 344 Fourth Revised Sheet No. 327 345 Fourth Revised Sheet No. 328 346 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Original Sheet No. 330A 348 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 353 Fifth Revised Sheet No. 335 352 Fifth Revised Sheet No. 336 353 Fourth Revised Sheet No. 336 355 Fifth Revised Sheet No. 336 355 Fifth Revised Sheet No. 336 356 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 339 361 Second Revised Sheet No. 340 360 Sheet No. 341-499 363 First Revised Sheet No. 500 365 Firth Revised Sheet No. 501 365 Firth Revised Sheet No. 505 366 Firth Revis	· ·	
Original Sheet No. 325 343 Fourth Revised Sheet No. 326 3445 Fifth Revised Sheet No. 328 346 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Fourth Revised Sheet No. 330 348 Original Sheet No. 330A 349 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 353 Furth Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 359 Original Sheet No. 338 359 Original Sheet No. 338 360 Twelfth Revised Sheet No. 339 361 Second Revised Sheet No. 500 364 Original Sheet No. 501 365 First Revised Sheet No. 502 366 Firth Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised	-	
Fourth Revised Sheet No. 326 344 Fifth Revised Sheet No. 327 345 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Original Sheet No. 330A 349 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 334 354 Third Revised Sheet No. 334 354 Third Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 339 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 Sirst Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 505 369 First Revised Sheet No. 505 369	· ·	
Fifth Revised Sheet No. 328 345 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Original Sheet No. 330A 349 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 362 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 501 365 First Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 508 372 Substitute Original Sheet No. 509 373 Fi		
Fourth Revised Sheet No. 328 Fourth Revised Sheet No. 329 347 Fourth Revised Sheet No. 330 348 Original Sheet No. 330A 349 Original Sheet No. 330B Fifth Revised Sheet No. 331 Third Revised Sheet No. 332 Fifth Revised Sheet No. 332 Fifth Revised Sheet No. 333 Fourth Revised Sheet No. 333 Fourth Revised Sheet No. 334 Third Revised Sheet No. 335 Fifth Revised Sheet No. 336 First Revised Sheet No. 336 First Revised Sheet No. 337 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 339 Second Revised Sheet No. 340 Second Revised Sheet No. 340 Second Revised Sheet No. 501 First Revised Sheet No. 502 Third Revised Sheet No. 503 Original Sheet No. 503 Original Sheet No. 504 Fifth Revised Sheet No. 505 Fifth Revised Sheet No. 505 Fifth Revised Sheet No. 505 First Revised Sheet No. 507 Original Sheet No. 508 First Revised Sheet No. 508 First Revised Sheet No. 509 First Revised Sheet No. 509 First Revised Sheet No. 507 Original Sheet No. 508 Signal Sheet No. 509 First Revised Sheet No. 501 Substitute Original Sheet No. 511 Substitute Original Sheet No. 512 Second Revised Sheet No. 513 Fourth Revised Sheet No. 514 Second Revised Sheet No. 515 Second Revised Sheet No. 516 Signal Sheet No. 517		
Fourth Revised Sheet No. 329 Fourth Revised Sheet No. 330 Signal Sheet No. 330A Original Sheet No. 330B Original Sheet No. 330B Signal Sheet No. 330B Signal Sheet No. 330B Signal Sheet No. 331 Signal Sheet No. 331 Signal Sheet No. 332 Signal Sheet No. 333 Signal Sheet No. 333 Fourth Revised Sheet No. 334 Signal Sheet No. 335 Fifth Revised Sheet No. 335 Fifth Revised Sheet No. 336 Signal Sheet No. 337 Signal Sheet No. 338 Signal Sheet No. 339 Signal Sheet No. 340 Sheet Nos. 341-499 Signal Sheet No. 500 Signal Sheet No. 501 Signal Sheet No. 501 Signal Sheet No. 502 Signal Sheet No. 503 Signal Sheet No. 504 Signal Sheet No. 504 Signal Sheet No. 505 Signal Sheet No. 507 Signal Sheet No. 508 Signal		
Fourth Revised Sheet No. 330A Original Sheet No. 330A Original Sheet No. 330B Signal Sheet No. 330B Fifth Revised Sheet No. 331 Third Revised Sheet No. 332 Fifth Revised Sheet No. 332 Fifth Revised Sheet No. 333 Fourth Revised Sheet No. 333 Fourth Revised Sheet No. 334 Third Revised Sheet No. 335 Fifth Revised Sheet No. 335 Fifth Revised Sheet No. 336 First Revised Sheet No. 337 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 338 Signal Signal Signal Signal Signal Signal Signal Sheet No. 338 Fifth Revised Sheet No. 338A Original Sheet No. 338B Original Sheet No. 339 Signal Sheet No. 340 Sheet Nos. 341-499 First Revised Sheet No. 500 Sheet Nos. 341-499 First Revised Sheet No. 500 Third Revised Sheet No. 501 Signal Sheet No. 501 Signal Sheet No. 501 Signal Sheet No. 504 Second Revised Sheet No. 505 First Revised Sheet No. 505 First Revised Sheet No. 506 Original Sheet No. 507 Original Sheet No. 508 Signal Sheet No. 508 Signal Sheet No. 508 Signal Sheet No. 509 First Revised Sheet No. 501 Signal Sheet No. 508 Signal Sheet No. 508 Signal Sheet No. 508 Signal Sheet No. 508 Signal Sheet No. 509 First Revised Sheet No. 510 Substitute Original Sheet No. 511 Substitute Original Sheet No. 512 Second Revised Sheet No. 514 Second Revised Sheet No. 515 Second Revised Sheet No. 516 Signal Sheet No. 516A Sub Original Sheet No. 516B Signal Sheet No. 516B Signal Sheet No. 516B Signal Sheet No. 516B Signal Sheet No. 517 Signal Sheet No. 516B Signal Sheet No. 517		
Original Sheet No. 330A 349 Original Sheet No. 330B 350 Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 501 365 First Revised Sheet No. 505 369 First Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original S		
Original Sheet No. 330B 350 Fifth Revised Sheet No. 332 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 Fifth Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338B 360 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Fifth Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 514 378 Seco		
Fifth Revised Sheet No. 331 351 Third Revised Sheet No. 332 352 Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 339 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 366 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 368 Second Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 514 376 Second Revised Sheet No. 514 378	<u> </u>	
Third Revised Sheet No. 332 Fifth Revised Sheet No. 334 Fourth Revised Sheet No. 334 Third Revised Sheet No. 335 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338A Original Sheet No. 338B Second Revised Sheet No. 340 Second Revised Sheet No. 540 Sheet Nos. 341-499 Sirst Revised Sheet No. 500 Sirst Revised Sheet No. 501 Fifth Revised Sheet No. 502 Third Revised Sheet No. 503 Fifth Revised Sheet No. 503 Fifth Revised Sheet No. 503 Fifth Revised Sheet No. 505 Second Revised Sheet No. 505 Signal Sheet No. 504 Second Revised Sheet No. 505 Signal Sheet No. 506 Signal Sheet No. 507 Original Sheet No. 507 Original Sheet No. 507 Signal Sheet No. 507 Signal Sheet No. 508 Substitute Original Sheet No. 510 Substitute Original Sheet No. 511 Substitute Original Sheet No. 512 Second Revised Sheet No. 513 Fourth Revised Sheet No. 514 Second Revised Sheet No. 515 Second Revised Sheet No. 516 Signal Sheet No. 516	<u> </u>	
Fifth Revised Sheet No. 333 353 Fourth Revised Sheet No. 334 354 Third Revised Sheet No. 335 355 Fifth Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 339 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 381		
Fourth Revised Sheet No. 334 Third Revised Sheet No. 335 Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 337 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338A Original Sheet No. 338B Twelfth Revised Sheet No. 339 Second Revised Sheet No. 340 Sheet No. 341-499 Second Revised Sheet No. 500 Triginal Sheet No. 501 Fifth Revised Sheet No. 502 Third Revised Sheet No. 502 Third Revised Sheet No. 503 Second Revised Sheet No. 503 First Revised Sheet No. 505 Fifth Revised Sheet No. 505 Fifth Revised Sheet No. 505 Fifth Revised Sheet No. 507 Original Sheet No. 506 Second Revised Sheet No. 506 Triginal Sheet No. 507 Signal Sheet No. 507 Signal Sheet No. 508 Second Revised Sheet No. 509 First Revised Sheet No. 509 First Revised Sheet No. 509 First Revised Sheet No. 510 Substitute Original Sheet No. 510 Substitute Original Sheet No. 511 Substitute Original Sheet No. 514 Second Revised Sheet No. 514 Second Revised Sheet No. 514 Second Revised Sheet No. 516 Second		
Third Revised Sheet No. 335 Fifth Revised Sheet No. 336 First Revised Sheet No. 337 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338A Original Sheet No. 338B Twelfth Revised Sheet No. 339 Second Revised Sheet No. 340 Sheet Nos. 341-499 Sirst Revised Sheet No. 500 Original Sheet No. 501 Sifth Revised Sheet No. 500 Fifth Revised Sheet No. 502 Third Revised Sheet No. 502 Third Revised Sheet No. 503 Original Sheet No. 504 Second Revised Sheet No. 505 First Revised Sheet No. 505 First Revised Sheet No. 505 Sirst Revised Sheet No. 506 Original Sheet No. 507 Original Sheet No. 507 Original Sheet No. 507 Substitute Original Sheet No. 509 First Revised Sheet No. 509 First Revised Sheet No. 510 Substitute Original Sheet No. 511 Substitute Original Sheet No. 512 Second Revised Sheet No. 514 Second Revised Sheet No. 515 Second Revised Sheet No. 516 Se		
Fifth Revised Sheet No. 336 356 First Revised Sheet No. 337 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516B 381 Sub Original Sheet No. 516B 382 First R		
First Revised Sheet No. 338 357 Fifth Revised Sheet No. 338 358 Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516B 381 Sub Original Sheet No. 516B 382		
Fifth Revised Sheet No. 338A 358 Fifth Revised Sheet No. 338B 360 Twelfth Revised Sheet No. 339 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Fifth Revised Sheet No. 338A 359 Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 516A 380 Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Original Sheet No. 338B 360 Twelfth Revised Sheet No. 340 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 514 377 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Twelfth Revised Sheet No. 339 361 Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 379 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Second Revised Sheet No. 340 362 Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	9	
Sheet Nos. 341-499 363 First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
First Revised Sheet No. 500 364 Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Original Sheet No. 501 365 Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Fifth Revised Sheet No. 502 366 Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Third Revised Sheet No. 503 367 Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Original Sheet No. 516B 382 First Revised Sheet No. 517 383	9	
Original Sheet No. 504 368 Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Second Revised Sheet No. 505 369 First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516A 381 Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
First Revised Sheet No. 506 370 Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	9	
Original Sheet No. 507 371 Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Original Sheet No. 508 372 Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Substitute Original Sheet No. 509 373 First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	· ·	
First Revised Sheet No. 510 374 Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	<u> </u>	
Substitute Original Sheet No. 511 375 Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	-	
Substitute Original Sheet No. 512 376 Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Second Revised Sheet No. 513 377 Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Fourth Revised Sheet No. 514 378 Second Revised Sheet No. 515 379 Second Revised Sheet No. 516 380 Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383	•	
Second Revised Sheet No. 515379Second Revised Sheet No. 516380Original Sheet No. 516A381Sub Original Sheet No. 516B382First Revised Sheet No. 517383		
Second Revised Sheet No. 516380Original Sheet No. 516A381Sub Original Sheet No. 516B382First Revised Sheet No. 517383		
Original Sheet No. 516A 381 Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
Sub Original Sheet No. 516B 382 First Revised Sheet No. 517 383		
First Revised Sheet No. 517 383	<u> </u>	
	•	
Original Sheet No. 518 384		
	Original Sheet No. 518	384

Fifth Deviced Cheet No. 540	205
Fifth Revised Sheet No. 519	385
Fourth Revised Sheet No. 520	386
Original Sheet No. 521	387
Second Revised Sheet No. 522	388
First Revised Sheet No. 523	389
Original Sheet No. 524	390
Substitute Original Sheet No. 525	391
Substitute Original Sheet No. 526	392
First Revised Sheet No. 527	393
Substitute Original Sheet No. 528	394
Substitute Original Sheet No. 529	395
Second Revised Sheet No. 530	396
Fourth Revised Sheet No. 531	397
Second Revised Sheet No. 532	398
Second Revised Sheet No. 532 Second Revised Sheet No. 533	399
Original Sheet No. 533A	400
Sub Original Sheet No. 533B	401
First Revised Sheet No. 534	402
Original Sheet No. 535	403
Third Revised Sheet No. 536	404
First Revised Sheet No. 537	405
Original Sheet No. 538	406
Second Revised Sheet No. 539	407
First Revised Sheet No. 540	408
Original Sheet No. 541	409
Substitute Original Sheet No. 542	410
Substitute Original Sheet No. 543	411
First Revised Sheet No. 544	412
Substitute Original Sheet No. 545	413
Substitute Original Sheet No. 546	414
Original Sheet No. 547	415
First Revised Sheet No. 548	416
Original Sheet No. 549	417
Third Revised Sheet No. 550	418
First Revised Sheet No. 551	419
Original Sheet No. 552	420
Second Revised Sheet No. 553	421
First Revised Sheet No. 554	422
Original Sheet No. 555	423
Original Sheet No. 556	424
Original Sheet No. 557	425
<u> </u>	
First Revised Sheet No. 558	426 427
Original Sheet No. 559	427
Original Sheet No. 560	428
Original Sheet No. 561	429
First Revised Sheet No. 562	430

Original Sheet No. 563	431
Third Revised Sheet No. 564	432
First Revised Sheet No. 565	433
Original Sheet No. 566	434
Second Revised Sheet No. 567	435
First Revised Sheet No. 568	436
Original Sheet No. 569	437
Original Sheet No. 570	438
Original Sheet No. 571	439
	440
First Revised Sheet No. 572	
Original Sheet No. 573	441
Original Sheet No. 574	442
Original Sheet No. 575	443
First Revised Sheet No. 576	444
Original Sheet No. 577	445
Fifth Revised Sheet No. 578	446
Second Revised Sheet No. 579	447
Original Sheet No. 580	448
Second Revised Sheet No. 581	449
First Revised Sheet No. 582	450
Second Revised Sheet No. 583	451
Second Revised Sheet No. 584	452
Second Revised Sheet No. 585	453
Second Revised Sheet No. 586	454
Sub Original Sheet No. 586A	455
First Revised Sheet No. 587	456
Original Sheet No. 588	457
Second Revised Sheet No. 589	458
First Revised Sheet No. 590	459
Second Revised Sheet No. 591	460
First Revised Sheet No. 592	461
Original Sheet No. 593	462
First Revised Sheet No. 594	463
Substitute Original Sheet No. 595	464
Sixth Revised Sheet No. 596	465
Original Sheet No. 597	466
Second Revised Sheet No. 598	467
Second Revised Sheet No. 599	468
First Revised Sheet No. 600	469
Second Revised Sheet No. 601	470
Second Revised Sheet No. 602	471
Second Revised Sheet No. 603	472
Second Revised Sheet No. 604	473
Sub Original Sheet No. 604A	474
First Revised Sheet No. 605	475
Substitute Original Sheet No. 606	476
Substitute Stigitial Officer No. 000	710

Third Revised Sheet No. 607 First Revised Sheet No. 608 A78 Original Sheet No. 609 Second Revised Sheet No. 610 A80 Original Sheet No. 611 A81 Original Sheet No. 611 A81 Original Sheet No. 612 First Revised Sheet No. 613 A83 Substitute Original Sheet No. 614 A84 Substitute Original Sheet No. 615 A85 Fifth Revised Sheet No. 616 A86 First Revised Sheet No. 616 A87 Second Revised Sheet No. 617 A87 Second Revised Sheet No. 618 Second Revised Sheet No. 619 First Revised Sheet No. 620 Second Revised Sheet No. 621 Second Revised Sheet No. 621 Second Revised Sheet No. 622 Second Revised Sheet No. 624 Sub Original Sheet No. 624 A93 Second Revised Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 626 First Revised Sheet No. 627 Second Revised Sheet No. 628 Second Revised Sheet No. 629 Second Revised Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 627 Sirst Revised Sheet No. 627 Sirst Revised Sheet No. 628 Original Sheet No. 628 Original Sheet No. 629 Socond Revised Sheet No. 630 First Revised Sheet No. 631 Second Revised Sheet No. 631 Second Revised Sheet No. 633 First Revised Sheet No. 633 First Revised Sheet No. 634 Original Sheet No. 635 First Revised Sheet No. 636 Socond Revised Sheet No. 637 First Revised Sheet No. 638 Socond Revised Sheet No. 639 Original Sheet No. 640 Second Revised Sheet No. 641 Sirst Revised Sheet No. 642 Socond Revised Sheet No. 645 First Revised Sheet No. 646 First Revised Sheet No. 647 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 649 First Revised Sheet No. 64	First Revised Sheet No. 608 478 Original Sheet No. 609 478 Second Revised Sheet No. 610 480 First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 631 502		
Original Sheet No. 609 479 Second Revised Sheet No. 610 480 First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 First Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 635 505	Original Sheet No. 609 479 Second Revised Sheet No. 611 481 Crist Revised Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 626 497 First Revised Sheet No. 628 498 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 <		
Second Revised Sheet No. 610 480 First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 622 492 Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 634 505	Second Revised Sheet No. 610 480 First Revised Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Original Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 503 First Revised Sheet No. 636 507	First Revised Sheet No. 608	478
First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 First Revised Sheet No. 633 504 Original Sheet No. 635 505 First Revised Sheet No. 636 507 Original Sheet No. 635 506 <t< td=""><td>First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 486 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 629 489 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 635 506 First Revised Sheet No. 636 507 <td>Original Sheet No. 609</td><td>479</td></td></t<>	First Revised Sheet No. 611 481 Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 615 486 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 629 489 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 635 506 First Revised Sheet No. 636 507 <td>Original Sheet No. 609</td> <td>479</td>	Original Sheet No. 609	479
Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 638 509 <t< td=""><td>Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 630 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Se</td><td>Second Revised Sheet No. 610</td><td>480</td></t<>	Original Sheet No. 612 482 First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 630 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Se	Second Revised Sheet No. 610	480
First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 624 494 First Revised Sheet No. 626 497 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 First Revised Sheet No. 632 503 First Revised Sheet No. 635 506 Original Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 640 511 <td< td=""><td>First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 Fifth Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 624 496 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 635 506 First Revised Sheet No. 636 507</td><td>First Revised Sheet No. 611</td><td>481</td></td<>	First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 Fifth Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 624 496 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 635 506 First Revised Sheet No. 636 507	First Revised Sheet No. 611	481
First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 624 494 First Revised Sheet No. 626 497 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 First Revised Sheet No. 632 503 First Revised Sheet No. 635 506 Original Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 640 511 <td< td=""><td>First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 Fifth Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 635 506 First Revised Sheet No. 636 507</td><td>Original Sheet No. 612</td><td>482</td></td<>	First Revised Sheet No. 613 483 Substitute Original Sheet No. 615 485 Fifth Revised Sheet No. 616 486 Fifth Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 624 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 635 506 First Revised Sheet No. 636 507	Original Sheet No. 612	482
Substitute Original Sheet No. 614 484 Substitute Original Sheet No. 616 485 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511	Substitute Original Sheet No. 615 484 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 635 505 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 639 510 <t< td=""><td>•</td><td>483</td></t<>	•	483
Substitute Original Sheet No. 616 486 Fifth Revised Sheet No. 617 486 First Revised Sheet No. 618 488 Second Revised Sheet No. 619 489 First Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512	Substitute Original Sheet No. 616 485 Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 620 480 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 625 496 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 633 504 Original Sheet No. 636 507 Original Sheet No. 638 509 Second Revised Sheet No. 640 511 <		
Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 633 504 Original Sheet No. 635 503 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original S	Fifth Revised Sheet No. 616 486 First Revised Sheet No. 617 487 Second Revised Sheet No. 618 488 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 630 501 First Revised Sheet No. 633 504 Original Sheet No. 633 504 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 639 500 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No.	•	
First Revised Sheet No. 617 Second Revised Sheet No. 618 Second Revised Sheet No. 619 First Revised Sheet No. 620 Second Revised Sheet No. 621 Second Revised Sheet No. 621 Second Revised Sheet No. 622 Second Revised Sheet No. 622 Second Revised Sheet No. 623 Second Revised Sheet No. 624 Second Revised Sheet No. 624 Second Revised Sheet No. 625 First Revised Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 627 Original Sheet No. 627 Original Sheet No. 628 Original Sheet No. 629 Fourth Revised Sheet No. 630 Fourth Revised Sheet No. 630 Fourth Revised Sheet No. 631 Second Revised Sheet No. 632 First Revised Sheet No. 633 First Revised Sheet No. 633 First Revised Sheet No. 635 First Revised Sheet No. 636 Original Sheet No. 635 First Revised Sheet No. 636 First Revised Sheet No. 636 First Revised Sheet No. 637 First Revised Sheet No. 638 Second Revised Sheet No. 639 Second Revised Sheet No. 639 Second Revised Sheet No. 639 Second Revised Sheet No. 640 Second Revised Sheet No. 641 Original Sheet No. 642 Second Revised Sheet No. 645 First Revised Sheet No. 646 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 649 First Revised Sheet No. 648 First Revised Sheet No. 649 First Revised Sheet No. 649 First Revised Sheet No. 648 First Revised Sheet No. 649	First Revised Sheet No. 617 Second Revised Sheet No. 618 Second Revised Sheet No. 619 First Revised Sheet No. 620 Second Revised Sheet No. 621 Second Revised Sheet No. 621 Second Revised Sheet No. 622 Second Revised Sheet No. 623 Second Revised Sheet No. 623 Second Revised Sheet No. 624 Second Revised Sheet No. 624 Second Revised Sheet No. 625 Second Revised Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 627 Sorginal Sheet No. 628 Original Sheet No. 628 Original Sheet No. 629 Original Sheet No. 630 First Revised Sheet No. 631 Second Revised Sheet No. 632 First Revised Sheet No. 633 First Revised Sheet No. 634 Sorginal Sheet No. 635 First Revised Sheet No. 636 Original Sheet No. 635 First Revised Sheet No. 636 Original Sheet No. 636 Original Sheet No. 636 Sorginal Sheet No. 637 First Revised Sheet No. 638 Second Revised Sheet No. 638 Second Revised Sheet No. 639 Original Sheet No. 640 Sorginal Sheet No. 640 Sorginal Sheet No. 642 Sorginal Sheet No. 643 Sorginal Sheet No. 643 Sorginal Sheet No. 645 Sorginal Sheet No. 646 Sorginal Sheet No. 646 Sorginal Sheet No. 646 Sorginal Sheet No. 646 Sorginal Sheet No. 647 Sorginal Sheet No. 648 Sorginal Sheet No. 649 Sorginal Sheet No. 650	· · · · · · · · · · · · · · · · · · ·	
Second Revised Sheet No. 619 488 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 631 502 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 644 515 Second Revised Sheet No. 645 516 Original Sheet N	Second Revised Sheet No. 619 488 Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised S		
Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 631 502 Second Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 6	Second Revised Sheet No. 620 490 Second Revised Sheet No. 621 491 Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 635 505 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 645 516 Original		
First Revised Sheet No. 620 Second Revised Sheet No. 621 Second Revised Sheet No. 622 Second Revised Sheet No. 623 Second Revised Sheet No. 623 Second Revised Sheet No. 623 Second Revised Sheet No. 624 Sub Original Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 627 Original Sheet No. 627 Original Sheet No. 628 Original Sheet No. 629 Fourth Revised Sheet No. 630 Fourth Revised Sheet No. 631 Second Revised Sheet No. 632 First Revised Sheet No. 633 First Revised Sheet No. 633 First Revised Sheet No. 633 First Revised Sheet No. 636 Original Sheet No. 635 First Revised Sheet No. 636 Original Sheet No. 636 First Revised Sheet No. 636 Original Sheet No. 636 First Revised Sheet No. 638 Second Revised Sheet No. 639 First Revised Sheet No. 640 Second Revised Sheet No. 641 Original Sheet No. 642 Original Sheet No. 642 Original Sheet No. 643 Second Revised Sheet No. 644 First Revised Sheet No. 645 First Revised Sheet No. 646 First Revised Sheet No. 646 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 649	First Revised Sheet No. 620 Second Revised Sheet No. 621 Second Revised Sheet No. 622 Second Revised Sheet No. 623 Second Revised Sheet No. 623 Second Revised Sheet No. 624 Sub Original Sheet No. 624 Sub Original Sheet No. 625 First Revised Sheet No. 625 First Revised Sheet No. 626 First Revised Sheet No. 626 First Revised Sheet No. 627 Second Revised Sheet No. 627 First Revised Sheet No. 628 Original Sheet No. 628 Original Sheet No. 629 Fourth Revised Sheet No. 630 First Revised Sheet No. 631 Second Revised Sheet No. 632 First Revised Sheet No. 633 First Revised Sheet No. 633 First Revised Sheet No. 634 Original Sheet No. 635 First Revised Sheet No. 636 First Revised Sheet No. 636 First Revised Sheet No. 636 First Revised Sheet No. 637 First Revised Sheet No. 638 Second Revised Sheet No. 639 Original Sheet No. 640 Second Revised Sheet No. 641 Second Revised Sheet No. 642 Original Sheet No. 643 Second Revised Sheet No. 645 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 649 First Revised Sheet No. 650		
Second Revised Sheet No. 621 491 Second Revised Sheet No. 623 492 Second Revised Sheet No. 624 493 Second Revised Sheet No. 624A 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 640 501 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 Original Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647	Second Revised Sheet No. 621 491 Second Revised Sheet No. 623 492 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 641 511 Original Sheet No. 642 511 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No.		
Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Original Sheet No. 640 511 Original Sheet No. 643 512 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No.	Second Revised Sheet No. 622 492 Second Revised Sheet No. 623 493 Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 643 514 Second Revised Sheet No. 645 515 First Revised Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised She		
Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 648 <td>Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No.</td> <td></td> <td></td>	Second Revised Sheet No. 624 494 Second Revised Sheet No. 624 494 Sub Original Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No.		
Second Revised Sheet No. 624A 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 515 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 648 </td <td>Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 649<!--</td--><td></td><td></td></td>	Second Revised Sheet No. 624 494 Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 649 </td <td></td> <td></td>		
Sub Original Sheet No. 624A 495 First Revised Sheet No. 626 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 639 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Sub Original Sheet No. 624A 495 First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 <td>Second Revised Sheet No. 623</td> <td></td>	Second Revised Sheet No. 623	
First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 517 First Revised Sheet No. 648 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 <td>First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 649 520 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521</td> <td>Second Revised Sheet No. 624</td> <td></td>	First Revised Sheet No. 625 496 First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 649 520 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	Second Revised Sheet No. 624	
First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 626 497 First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	Sub Original Sheet No. 624A	495
First Revised Sheet No. 627 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 646 517 First Revised Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 628 498 Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	First Revised Sheet No. 625	496
Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 628 499 Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	First Revised Sheet No. 626	497
Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 649 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	First Revised Sheet No. 627	498
Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 629 500 Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	Original Sheet No. 628	499
Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 640 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Fourth Revised Sheet No. 630 501 First Revised Sheet No. 631 502 Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	•	500
Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 649 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	· ·	501
Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Second Revised Sheet No. 632 503 First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 649 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	First Revised Sheet No. 631	502
First Revised Sheet No. 633 504 Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 634 504 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	Second Revised Sheet No. 632	503
Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 634 505 Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 635 506 First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 636 507 Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	-	
Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 637 508 First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 638 509 Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Second Revised Sheet No. 639 510 Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 640 511 Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Second Revised Sheet No. 641 512 Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 642 513 Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 643 514 Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Second Revised Sheet No. 644 515 First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	-	
First Revised Sheet No. 645 Original Sheet No. 646 First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 648 First Revised Sheet No. 649 510	First Revised Sheet No. 645 516 Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	Original Sheet No. 646 517 First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
First Revised Sheet No. 647 518 First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 647 First Revised Sheet No. 648 First Revised Sheet No. 649 First Revised Sheet No. 650 518 519 520 521		
First Revised Sheet No. 648 519 First Revised Sheet No. 649 520	First Revised Sheet No. 648 519 First Revised Sheet No. 649 520 First Revised Sheet No. 650 521	U	
First Revised Sheet No. 649 520	First Revised Sheet No. 649 520 First Revised Sheet No. 650 521		
	First Revised Sheet No. 650 521		
First Revised Sheet No. 650 521		First Revised Sheet No. 649	
1 1131 1132 1132 1132 1132 1132 1132 11	First Revised Sheet No. 651 522	First Revised Sheet No. 650	521
First Revised Sheet No. 651 522	v 	First Revised Sheet No. 651	522

First Revised Sheet No. 652	523
First Revised Sheet No. 653	524
Second Revised Sheet No. 654	525
Original Sheet No. 655	526
Fifth Revised Sheet No. 656	527
Second Revised Sheet No. 657	528
Second Revised Sheet No. 658	529
Second Revised Sheet No. 659	530
First Revised Sheet No. 660	531
Second Revised Sheet No. 661	532
Second Revised Sheet No. 662	533
Second Revised Sheet No. 663	534
Second Revised Sheet No. 664	535
Sub Original Sheet No. 664A	536
First Revised Sheet No. 665	537
Substitute Original Sheet No. 666	538
Fourth Revised Sheet No. 667	539
Third Revised Sheet No. 668	540
Substitute Original Sheet No. 669	541
Second Revised Sheet No. 670	542
First Revised Sheet No. 671	543
Substitute Original Sheet No. 672	544
Substitute Original Sheet No. 673	545
Substitute Original Sheet No. 674	546
First Revised Sheet No. 675	547
Substitute Original Sheet No. 676	548
Substitute Original Sheet No. 677	549
Third Revised Sheet No. 678	550
Third Revised Sheet No. 679	551
First Revised Sheet No. 680	552
Original Sheet No. 680A	553
Sub Original Sheet No. 680B	554
Second Revised Sheet No. 681	555
Second Revised Sheet No. 682	556
Original Sheet No. 682A	557
<u> </u>	
First Revised Original Sheet No. 683	558 550
Second Revised Sheet No. 684	559
First Revised Sheet No. 685	560
First Revised Sheet No. 686	561
First Revised Sheet No. 687	562
First Revised Sheet No. 688	563
Second Revised Sheet No. 689	564
1st Rev Original Sheet No. 690	565
First Revised Sheet No. 691	566
First Revised Sheet No. 692	567
First Revised Sheet No. 693	568

First Revised Sheet No. 694	569
1st Rev Original Sheet No. 695	570
1st Rev Original Sheet No. 696	571
Original Sheet No. 697	572
Original Sheet No. 698	573
Original Sheet No. 699	574
First Revised Sheet No. 700	575
Original Sheet No. 701	576
Original Sheet No. 702	577
Original Sheet No. 703	578
Original Sheet No. 704	579
Original Sheet No. 705	580
Original Sheet No. 706	581
Original Sheet No. 707	582
Original Sheet No. 708	583
Original Sheet No. 709	584
Original Sheet No. 710	585
Original Sheet No. 711	586
Original Sheet No. 712	587
First Revised Sheet No. 713	588
Original Sheet No. 714	589
Original Sheet No. 715	590

FERC Docket: RP99-506-000

First Revised Sheet No. 0 Panhandle Eastern Pipe Line Company: First Revised Volume No. 1

First Revised Sheet No. 0 : Superseded

Superseding: Original Sheet No. 0

FERC GAS TARIFF

FIRST REVISED VOLUME NO. 1 (Supersedes Original Volume No. 1)

of

PANHANDLE EASTERN PIPE LINE COMPANY, d.b.a. CMS PANHANDLE EASTERN PIPE LINE COMPANY

Filed With The

FEDERAL ENERGY REGULATORY COMMISSION

Communications Concerning this Tariff Should be Addressed To:

William W. Grygar Vice President Rates and Regulatory Affairs Telephone: (713) 989-7000

Facsimile: (713) 989-1180

PANHANDLE EASTERN PIPE LINE COMPANY
P. O. Box 4967
Houston, Texas 77210-4967

5444 Westheimer Road Houston, Texas 77056-5306

FERC Docket: RP02-466-000

Seventh Revised Sheet No. 1 Seventh Revised Sheet No. 1: Superseded Superseding: Sixth Revised Sheet No. 1

TABLE OF CONTENTS

21.12.22 02 00.1.12.1.10	Sheet No.
PRELIMINARY STATEMENT	2
SYSTEM MAP	3
CURRENTLY EFFECTIVE RATES	4
RATE SCHEDULES	
FT - Firm Transportation Service	30
EFT - Enhanced Firm Transportation Service	37
SCT - Small Customer Transportation Service	44
IT - Interruptible Transportation Service	51
EIT - Enhanced Interruptible Transportation Service	56
<pre>IOS - In/Out Storage Service</pre>	61
<pre>IIOS - Interruptible In/Out Storage Service</pre>	65
WS - Winter Storage Service	68
PS - Peaking Storage Service	72
IWS - Interruptible Winter Storage Service	77
GDS - General Delivery Service	81
TBS - Transportation Balancing Service	86
FS - Flexible Storage Service	89
LFT - Limited Firm Transportation Service	93
GPS - Gas Parking Service	102
HFT - Hourly Firm Transportation	105
DVS - Delivery Variance Service	114
GENERAL TERMS AND CONDITIONS	200
INDEX	201
1. Definitions	203
2. Electronic Communication System	210
3. Quality of Gas	213
4. Measurement	215
5. Responsibility During Transportation or Storage	218

FERC Docket: RP03-479-000

Second Revised Sheet No. 1A Second Revised Sheet No. 1A: Superseded Superseding: First Revised Sheet No. 1A

TABLE OF CONTENTS
(Continued)

	, and a state,	Sheet No.
6.	Procedures for Requesting Service Rights	219
7.	Contracting for Service Rights	232
8.	Nomination and Scheduling of Service	238
9.	Curtailment and Interruption	245
10.	Points of Receipt	249
11.	Points of Delivery	253
12.	Conditions of Receipt and Delivery	256
13.	Policy for Construction of New Receipt or Delivery Facilities	273
14.	Pooling and In-field Transfers of Storage Inventory	274
15.	Procedures For Assignment of Service Rights	276
16.	Assignment of Upstream Service Rights	284
17.	Statements and Payments	285
18.	Periodic Rate Adjustments and Surcharges \dots	288
19.	Warranty and Assignments	322
20.	Force Majeure	323
21.	Termination for Default	324
22.	Regulations	325
23.	Operation of Rate Schedules in Conjunction with Marketing and Gathering Affiliates	326
24.	Fuel Reimbursement Adjustment	327
25.	Flow Through of Cash-out Revenues and Penalties in Excess of Costs	329
26.	Reserved for Future Use	331
27	Miscellaneous	337

FERC Docket: RP00-395-003

Third Revised Sheet No. 1B Third Revised Sheet No. 1B: Superseded

Superseding: Second Revised Sheet No. 1B $\,$

TABLE OF CONTENTS (Continued)

Sheet No. FORM OF SERVICE AGREEMENTS - Firm Transportation Service EFT - Enhanced Firm Transportation Service 517 SCT - Small Customer Firm Transportation Service . . . 534 ΤТ - Interruptible Transportation Service 548 EIT - Enhanced Interruptible Transportation Service . 562 IOS 576 IIOS - Interruptible In/Out Storage Service 587 WS 594 IWS - Interruptible Winter Storage Service 605 - Peaking Storage Service PS 613 GDS - General Delivery Service 62.5 TBS - Transportation Balancing Service 639 647 FS - Flexible Storage Service 654 - Limited Firm Transportation Service LFT 665 GPS 681 686 - Hourly Firm Transportation HFT 695 - Delivery Variance Service

FERC Docket: RP03-479-000

First Revised Sheet No. 1C First Revised Sheet No. 1C : Superseded Superseding: Original Sheet No. 1C

TABLE OF CONTENTS

ORIGINAL VOLUME NO. 2

Rate

Schedule Sheet No.

E-17 3365 National Helium Corporation

FERC Docket: RP99-506-000

First Revised Sheet No. 2 First Revised Sheet No. 2: Superseded

Superseding: Original Sheet No. 2

PRELIMINARY STATEMENT

Panhandle Eastern Pipe Line Company is a Natural Gas company engaged in the business of transporting and storing Natural Gas. Panhandle provides service to a variety of Shippers in the United States and Canada, principally in the territory traversed by its pipeline system in the States of Texas, Kansas, Missouri, Illinois, Indiana, Ohio, and Michigan. Its main transmission line extends in a northeasterly direction from its principal sources of supply in the States of Texas, Oklahoma, and Kansas into the State of Michigan up to the international boundary between the United States and Canada. Panhandle also leases extensive underground Natural Gas Storage facilities.

The Transportation and Storage of Natural Gas is undertaken by the Company only under written contracts acceptable to the Company after consideration of its commitments to others, delivery capacity, and other pertinent factors.

This FERC Gas Tariff is filed in compliance with Part 154, Subpart A, and Part 284, Subpart A, Chapter 1, Title 18 of the Code of Federal Regulations.

FERC Docket: RP03-382-000

Fifth Revised Sheet No. 3 Fifth Revised Sheet No. 3 : Superseded Superseding: Fourth Revised Sheet No. 3

SYSTEM MAP APPEARS HERE

Information has been removed for Non-Internet Public treatment.

FERC Docket: RP03-382-000

Fourth Revised Sheet No. 3A Fourth Revised Sheet No. 3A : Superseded Superseding: Third Revised Sheet No. 3A

FIELD ZONE MAP APPEARS HERE

Information has been removed for Non-Internet Public treatment.

FERC Docket: RP03-382-000

Sixth Revised Sheet No. 3B sixth Revised Sheet No. 3B : Superseded Superseding: Fifth Revised Sheet No. 3B

MARKET ZONE MAP APPEARS HERE

Information has been removed for Non-Internet Public treatment.

FERC Docket: RP03-325-000

Seventieth Revised Sheet No. 4 : Superseded

Superseding: Sixty-Ninth Revised Sheet No. 4

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE FT
FIRM TRANSPORTATION SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

			Base Rate	Section 25.1	Section 18	Maximum Rate Per Dt. Excl. of	Overrun Charge Per	Minimum Rate	
Fuel	- '			- 11	a 1 1/	an = .0./	D: 2/		
Reimbur	Descrip rsement	tion	Per Dt.	Adjustment	Surch. 1/	GRI 2/	Dt. 3/	Per Dt.	
0.70 %	Gathering Reservati Commodity		\$ 8.67 0.16		\$ (0.01) 0.18 >	\$ 8.65 0.34 >	28.50 >	- 0.16 >	-
1.19 %	Transmissio Field Zon Reservati Commodity	e on Rate	\$ 4.73 1.95		\$ (0.01) 0.18 >		15.55 >	- 1.95 >	-
	Market Zone Access Cha Commodity	rge	\$ 2.80		\$ (0.01) 0.18 >	\$ 2.78 0.21 >	9.21 >	- 0.03 >	- -
0.43 %		arge (per 100 Mile Reservation Rate Commodity Rate	\$ 0.60	>		\$ 0.60 0.25 >	1.97 >	- 0.25 >	-
	101 - 200	Reservation Rate Commodity Rate		>		\$ 1.20 0.50 >	3.94	- 0.50	-
0.86	201 - 300	Reservation Rate Commodity Rate		>		\$ 1.80 0.75 >	5.91 -	- 0.75	-
1.29	301 - 400	Reservation Rate Commodity Rate		>		\$ 2.40 1.00 >	7.88 -	- 1.00	-
1.72	401 - 500	Reservation Rate Commodity Rate		>		\$ 3.00 1.25 >	9.85 -	- 1.25	-
2.15	501 - 600	Reservation Rate Commodity Rate		>		\$ 3.60 1.50 >	11.82	- 1.50	-
2.58	601 - 700	Reservation Rate Commodity Rate		>		\$ 4.20 1.75 >	13.79	- 1.75	-
3.01	701 - 800	Reservation Rate Commodity Rate		>		\$ 4.80 2.00 >	15.76	- 2.00	-
3.44	801 - 900	Reservation Rate Commodity Rate		>		\$ 5.40 2.25 >	17.73	- 2.25	-
3.87	901 - 1000	Reservation Rate Commodity Rate		>		\$ 6.00 2.50 >	19.70	- 2.50	-
4.30 Sur	rcharges	4							

the provisions of the General Terms and Conditions under which they are authorized.

```
2/ GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions.

Minimum Reservation and Commodity Charge $ 0.00

Reservation Charge

High Load Factor Customer (>than 50%) $ 0.05

Low Load Factor Customer (< than 50%) 0.031

Commodity Charge 0.40 >
```

3/ Maximum firm volumetric rate applicable for capacity release, exclusive of surcharges.

FERC Docket: RP03-325-000

Seventieth Revised Sheet No. 5 Seventieth Revised Sheet No. 5 : Superseded

Superseding: Sixty-Ninth Revised Sheet No. 5

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE EFT
ENHANCED FIRM TRANSPORTATION SERVICE

		Base Rate	Section 25.1	Section 18	Maximum Rate Per Dt. Excl. of		Minimum Rate	
Fuel Descripti		Da D+	Adjustment	O	GRI 2/	D+ 2/	D D+	
Reimbursement	.On	Per Dt.	Adjustment	Surch. 1/	GRI 2/	Ы. 3/	rer Dt.	
Gathering Reservation Commodity R		\$ 8.67 0.16		\$ (0.01) 0.18 >	\$ 8.65 0.34 >	28.50 >	0.16 >	-
Transmission Field Zone Reservatio Commodity 1.19 %	n Rate	\$ 4.73 1.95		\$ (0.01) 0.18 >	\$ 4.71 2.13 >	15.55 >	- 1.95 >	-
Market Zone Access Cha Commodity	rge	\$ 3.30 0.06		\$ (0.01) 0.18 >		10.85 >	- 0.06 >	- -
	arge (per 100 Mile Reservation Rate Commodity Rate	\$ 0.60	>		\$ 0.60 0.25 >	1.97 >	- 0.25 >	-
101 - 200	Reservation Rate Commodity Rate		>		\$ 1.20 0.50 >	3.94	- 0.50	-
0.86 201 - 300	Reservation Rate Commodity Rate		>		\$ 1.80 0.75 >	5.91	- 0.75	-
	Reservation Rate Commodity Rate		>		\$ 2.40 1.00 >	7.88 -	- 1.00	-
401 - 500	Reservation Rate Commodity Rate		>		\$ 3.00 1.25 >	9.85 -	- 1.25	-
	Reservation Rate Commodity Rate		>		\$ 3.60 1.50 >	11.82	- 1.50	-
	Reservation Rate Commodity Rate		>		\$ 4.20 1.75 >	13.79	- 1.75	-
	Reservation Rate Commodity Rate		>		\$ 4.80 2.00 >	15.76 -	- 2.00	-
	Reservation Rate Commodity Rate		>		\$ 5.40 2.25 >	17.73	- 2.25	-
	Reservation Rate Commodity Rate		>		\$ 6.00 2.50 >	19.70	- 2.50	-
4.30 Surcharges	-							

^{1/} Surcharges pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently

effective Sheet No. 19. Reservation and Commodity surcharges will be billed in accordance with the provisions of the General Terms and Conditions under which they are authorized.

```
2/ GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions.

Minimum Reservation and Commodity Charge $ 0.00
Reservation Charge
High Load Factor Customer (>than 50%) $ 0.05
Low Load Factor Customer (< than 50%) 0.031
Commodity Charge 0.40 >
```

3/ Maximum firm volumetric rate applicable for capacity release, exclusive of surcharges.

FERC Docket: RP03-325-000

Seventieth Revised Sheet No. 6 Seventieth Revised Sheet No. 6: Superseded

Superseding: Sixty-Ninth Revised Sheet No. 6

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE SCT
SMALL CUSTOMER TRANSPORTATION SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

					Maximum Rate Per		
Fuel		Base Rate	Section 25.1	Section 18	Dt. Excl. of	Overrun Charge	Minimum Rate
	escription ent	Per Dt.	Adjustment	Surch. 1/	GRI 2/	Per Dt.	Per Dt.
Gather 0.70 %	ring Charge	54.45 >	(0.06) >	0.12 >	54.51 >	28.66 >	0.16 >
	mission Charge eld Zone	31.57 >	(0.06) >	0.12 >	31.63 >	17.50 >	1.95 >
I	rket Zone Access Charge Mileage Charge (per 100 Miles)	20.73 >	(0.06) >	0.12 >	20.79 >	10.91 >	0.06 > -
0.43 %	1 - 100 Miles)	4.01 >			4.01 >	2.22 >	0.25 >
0.86	101 - 200	8.02			8.02	4.44	0.50
1.29	201 - 300	12.03			12.03	6.66	0.75
1.72	301 - 400	16.04			16.04	8.88	1.00
2.15	401 - 500	20.05			20.05	11.10	1.25
2.58	501 - 600	24.06			24.06	13.32	1.50
3.01	601 - 700	28.07			28.07	15.54	1.75
3.44	701 - 800	32.08			32.08	17.76	2.00
3.87	801 - 900	36.09			36.09	19.98	2.25
4.30	901 - 1000	40.10			40.10	22.20	2.50

Surcharges

 $^{\ 1/\} Surcharges$ pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently

effective Sheet No. 19. Commodity surcharges will be billed in accordance with the provisions of the General Terms and Conditions under which they are authorized.

^{2/} GRI Funding Unit pursuant to Section 18.1 and of the General Terms and Conditions. Maximum Commodity Charge 0.60 \rightarrow Minimum Commodity Charge 0.00 \rightarrow

FERC Docket: RP03-325-000

Seventy-Third Revised Sheet No. 7 : Superseded

Superseding: Seventy-Second Revised Sheet No. 7

CURRENTLY EFFECTIVE RATES

RATE SCHEDULE IT

INTERRUPTIBLE TRANSPORTATION SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

	Base	Section 25.1		Maximum Rate Per Dt. Excl.		E. a l
Description Reimbursement	Rate Per Dt.	25.1 Adjustment	18 Surch. 1/	of GRI 2/	Rate Per Dt.	Fuel
Gathering Charge	28.66 >	(0.03) >	0.18 >	28.81 >	0.16 >	0.70 %
Transmission Charge Field Zone	17.50 >	(0.03) >	0.18 >	17.65 >	1.95 >	1.19 %
Market Zone Access Charge Mileage Charge (per 100 Miles)	9.24 >	(0.03) >	0.18 >	9.39 >	0.03 >	-
1 - 100 Miles)	2.22 >			2.22 >	0.25 >	0.43 %
101 - 200	4.44			4.44	0.50	0.86
201 - 300	6.66			6.66	0.75	1.29
301 - 400	8.88			8.88	1.00	1.72
401 - 500	11.10			11.10	1.25	2.15
501 - 600	13.32			13.32	1.50	2.58
601 - 700	15.54			15.54	1.75	3.01
701 - 800	17.76			17.76	2.00	3.44
801 - 900	19.98			19.98	2.25	3.87
901 - 1000	22.20			22.20	2.50	4.30

Surcharges

^{1/} Surcharges pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently effective Sheet No. 19. Commodity surcharges will be billed in accordance with the provisions of the General Terms and Conditions under which they are authorized.

^{2/} GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions. Maximum Commodity Charge 0.40 \rightarrow Minimum Commodity Charge 0.00 \rightarrow

FERC Docket: RP03-325-000

Seventy-Third Revised Sheet No. 8 Seventy-Third Revised Sheet No. 8: Superseded

Superseding: Seventy-Second Revised Sheet No. 8

CURRENTLY EFFECTIVE RATES RATE SCHEDULE EIT ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

	Base	Section	Section	Maximum Rate Per Dt. Excl.	Minimum	
Description	Rate Per Dt.	25.1 Adjustment	18 Surch. 1/	of GRI 2/	Rate Per Dt.	Fuel Reimbursement
Gathering Charge	28.66 >	3	0.18 >		0.16 >	0.70 %
Transmission Charge						
Field Zone	17.50 >	(0.03) >	0.18 >	17.65 >	1.95 >	1.19 %
Market Zone Access Charge Mileage Charge	10.91 >	(0.03) >	0.18 >	11.06 >	0.06 >	-
(per 100 Miles) 1 - 100	2.22 >			2.22 >	0.25 >	0.43 %
101 - 200	4.44			4.44	0.50	0.86
201 - 300	6.66			6.66	0.75	1.29
301 - 400	8.88			8.88	1.00	1.72
401 - 500	11.10			11.10	1.25	2.15
501 - 600	13.32			13.32	1.50	2.58
601 - 700	15.54			15.54	1.75	3.01
701 - 800	17.76			17.76	2.00	3.44
801 - 900	19.98			19.98	2.25	3.87
901 - 1000	22.20			22.20	2.50	4.30

Surcharges

^{1/} Surcharges pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently effective Sheet No. 19. Commodity surcharges will be billed in accordance with the provisions of the General Terms and Conditions under which they are authorized.

^{2/} GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions. Maximum Commodity Charge 0.40 \rightarrow Minimum Commodity Charge 0.00 \rightarrow

FERC Docket: RP03- 4-000

Thirteenth Revised Sheet No. 9 Thirteenth Revised Sheet No. 9: Pending

Superseding: Twelfth Revised Sheet No. 9

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE IOS
IN/OUT STORAGE

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Deliverability Charge	\$ 3.35 1/	-	-
Capacity Charge	40.28 > 1/	-	-
Injection Charge	0.33	0.33>	1.56%
Withdrawal Charge	0.33	0.33	0.53
Overrun Charge	14.85	_	-

^{1/} Maximum rate applicable for capacity release.

FERC Docket: RP03- 4-000

Ninth Revised Sheet No. 10 Ninth Revised Sheet No. 10 : Pending

Superseding: Eighth Revised Sheet No. 10

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE IIOS
INTERRUPTIBLE IN/OUT STORAGE

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Monthly Inventory Charge	14.85>	-	-
Injection Charge	0.33	0.33>	1.56%
Withdrawal Charge	0.33	0.33	0.53

FERC Docket: RP03- 4-000

Fifteenth Revised Sheet No. 11 Fifteenth Revised Sheet No. 11: Pending

Superseding: Fourteenth Revised Sheet No. 11

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE WS
WINTER STORAGE SERVICE

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Market Area			
Deliverability Charge	\$ 2.97 1/	-	-
Capacity Charge	42.46 > 1/	-	-
Injection Charge	3.85	3.85>	1.67%
Withdrawal Charge	3.85	3.85	0.89
Overrun Charge	13.50		-
Field Area			
Deliverability Charge	\$ 3.35 1/	-	-
Capacity Charge	40.28 1/	-	-
Injection Charge	0.33	0.33>	1.56%
Withdrawal Charge	0.33	0.33	0.53
Overrun Charge	14.85	-	-
Surcharges Dakota Gasification 2/			
Deliverability - Market Area	\$0	.00	

^{1/} Maximum rate applicable for capacity release.

^{2/} Dakota Gasification Transition Cost Surcharge pursuant to Section 18.11 of the General Terms and Conditions.

FERC Docket: RP03- 4-000

Fifteenth Revised Sheet No. 12 Fifteenth Revised Sheet No. 12: Pending

Superseding: Fourteenth Revised Sheet No. 12

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE PS
PEAKING STORAGE SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Deliverability Charge	\$ 2.97 1/	-	-
Capacity Charge	42.46 > 1/	-	-
Injection Charge	3.85	3.85>	1.67%
Withdrawal Charge	3.85	3.85	0.89
Overrun Charge	13.50	-	-

Surcharges

Dakota Gasification 2/ Deliverability - Market Area

\$0.00

^{1/} Maximum rate applicable for capacity release.

^{2/} Dakota Gasification Transition Cost Surcharge pursuant to Section 18.11 of the General Terms and Conditions.

FERC Docket: RP03- 4-000

Eleventh Revised Sheet No. 13 Eleventh Revised Sheet No. 13: Pending

Superseding: Tenth Revised Sheet No. 13

CURRENTLY EFFECTIVE RATES

RATE SCHEDULE IWS

INTERRUPTIBLE WINTER STORAGE SERVICE

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Market Area			
Monthly Inventory Charge	13.50>	-	-
Injection Charge	3.85	3.85>	1.67%
Withdrawal Charge	3.85	3.85	0.89
Field Area			
Monthly Inventory Charge	14.85>	-	-
Injection Charge	0.33	0.33>	1.56%
Withdrawal Charge	0.33	0.33	0.53
Surcharges			
Dakota Gasification 1/ Monthly Inventory - Market Area	0.0	0 >	

^{1/} Dakota Gasification Transition Cost Surcharge pursuant to Section 18.11 of the General Terms and Conditions.

FERC Docket: RP03- 4-000

Fourteenth Revised Sheet No. 14 Fourteenth Revised Sheet No. 14: Pending

Superseding: Thirteenth Revised Sheet No. 14

CURRENTLY EFFECTIVE RATES
RATE SCHEDULE FS
FLEXIBLE STORAGE SERVICE

Description	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement	
Market Area				
Deliverability Charge	\$ 2.97 1/	-	-	
Capacity Charge	42.46> 1/	-	-	
Injection Charge	3.85	3.85>	1.67%	
Withdrawal Charge	3.85	3.85	0.89	
Overrun Charge	13.50	-	-	
Field Area				
Deliverability Charge	\$ 3.35 1/	-	-	
Capacity Charge	40.28 > 1/	-	-	
Injection Charge	0.33	0.33>	1.56%	
Withdrawal Charge	0.33	0.33	0.53	
Overrun Charge	14.85	-	-	
Surcharges Dakota Gasification 2/				
Deliverability - Market Area	\$0.00			

^{1/} Maximum rate applicable for capacity release.

^{2/} Dakota Gasification Transition Cost Surcharge pursuant to Section 18.11 of the General Terms and Conditions.

FERC Docket: RP03-325-000

Forty-Ninth Revised Sheet No. 15 Forty-Ninth Revised Sheet No. 15: Superseded

Superseding: Forty-Eighth Revised Sheet No. 15

CURRENTLY EFFECTIVE RATES RATE SCHEDULE LFT LIMITED FIRM TRANSPORTATION SERVICE

n .1			Base Rate	Section 25.1	Section 18	Maximum Rate Per Dt. Excl. of		Minimum Rate	
Fuel I Reimburser	Descripti ment	on	Per Dt.	Adjustment	Surch. 1/	GRI 2/	Dt. 3/	Per Dt.	
Res	ering servation nmodity R		\$ 5.82 0.16 >			\$ 5.80 0.34	28.50 >	- 0.16 >	-
Fie Res	smission eld Zone servation ommodity	Rate	\$ 3.17 1.95 >		\$ (0.01) 0.18 >		15.55 >	- 1.95 >	-
Ac	rket Zone ccess Cha ommodity	rge	\$ 2.22 0.06 >			\$ 2.20 0.24 >	10.85 >	- 0.06 >	-
M±		arge (per 100 Mile Reservation Rate Commodity Rate	\$ 0.40	•		\$ 0.40 0.25 >	1.97 >	- 0.25 >	-
	01 - 200	Reservation Rate Commodity Rate		•		\$ 0.80 0.50 >	3.94	- 0.50	-
	01 - 300	Reservation Rate Commodity Rate		•		\$ 1.20 0.75 >	5.91 -	- 0.75	-
	01 - 400	Reservation Rate Commodity Rate		•		\$ 1.60 1.00 >	7.88	- 1.00	-
	01 - 500	Reservation Rate Commodity Rate		•		\$ 2.00 1.25 >	9.85 -	- 1.25	-
5(01 - 600	Reservation Rate Commodity Rate		•		\$ 2.40 1.50 >	11.82	- 1.50	-
	01 - 700	Reservation Rate Commodity Rate				\$ 2.80 1.75 >	13.79	- 1.75	-
	01 - 800	Reservation Rate Commodity Rate		•		\$ 3.20 2.00 >	15.76 -	- 2.00	-
	01 - 900	Reservation Rate Commodity Rate		,		\$ 3.60 2.25 >	17.73 -	- 2.25	-
	01 - 1000	Reservation Rate Commodity Rate		•		\$ 4.00 2.50 >	19.70	- 2.50	-
4.30 Surch	narges								

Surcharges

^{1/} Surcharges pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently

effective Sheet No. 19. Reservation and Commodity surcharges will be billed in accordance with the provisions of the General Terms and Conditions under which they are authorized.

^{2/} GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions.

```
Minimum Reservation and Commodity Charge $ 0.00
Reservation Charge
High Load Factor Customer (>than 50%) $ 0.05
Low Load Factor Customer (< than 50%) 0.031
Commodity Charge 0.40 >
```

3/ Maximum firm volumetric rate applicable for capacity release, exclusive of surcharges.

FERC Docket: RP99-290-000

Second Revised Sheet No. 16 Second Revised Sheet No. 16: Superseded Superseding: First Revised Sheet No. 16

CURRENTLY EFFECTIVE RATES RATE SCHEDULE GPS GAS PARKING SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

	Maximum Rate	Minimum Rate
	Per Dt. Per Day	Per Dt. Per Day
Description		
Market Zone Daily Parking Rate	12.24 >	0.00 >
Field Zone Daily Parking Rate	11.80 >	0.00 >

Effective Date: 05/01/2003 Status: Effective

FERC Docket: RP03-325-000

Fourteenth Revised Sheet No. 17 Fourteenth Revised Sheet No. 17 : Superseded

Superseding: Thirteenth Revised Sheet No. 17

CURRENTLY EFFECTIVE RATES RATE SCHEDULE HFT HOURLY FIRM TRANSPORTATION SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

Fuel			Base Rate	Section 25.1	Section 18	Maximum Rate Per Dt. Excl. of	Overrun Charge Per	Minimum Rate	
	Description ment	nc	Per Dt.	Adjustment	Surch. 1/	GRI 2/	Dt. 3/	Per Dt.	
Rea	ering servation mmodity Ra		\$ 8.67 0.16			\$ 8.65 0.34 >	28.50 >	- 0.16 >	-
Fi Re	smission (eld Zone eservation ommodity I	n Rate	\$ 4.73 1.95		\$ (0.01) 0.18 >		15.55 >	- 1.95 >	-
A	rket Zone ccess Char ommodity I		\$ 3.30			\$ 3.28 0.24 >	10.85	- 0.06 >	- -
		arge (per 100 Mile Reservation Rate Commodity Rate		>		\$ 0.60 0.25 >	1.97 >	- 0.25 >	-
	01 - 200	Reservation Rate Commodity Rate		>		\$ 1.20 0.50 >	3.94	- 0.50	-
0.86 2	01 - 300	Reservation Rate Commodity Rate		>		\$ 1.80 0.75 >	5.91 -	- 0.75	-
	01 - 400	Reservation Rate Commodity Rate		>		\$ 2.40 1.00 >	7.88 -	- 1.00	-
4	01 - 500	Reservation Rate Commodity Rate		>		\$ 3.00 1.25 >	9.85 -	- 1.25	-
	01 - 600	Reservation Rate Commodity Rate		>		\$ 3.60 1.50 >	11.82	- 1.50	-
	01 - 700	Reservation Rate Commodity Rate	\$ 4.20 1.75	>		\$ 4.20 1.75 >	13.79	- 1.75	-
	01 - 800	Reservation Rate Commodity Rate	\$ 4.80	>		\$ 4.80 2.00 >	15.76 -	- 2.00	-
	01 - 900	Reservation Rate Commodity Rate	\$ 5.40 2.25	>		\$ 5.40 2.25 >	17.73 -	- 2.25	-
3.87 9.	01 - 1000	Reservation Rate Commodity Rate		>		\$ 6.00 2.50 >	19.70	- 2.50	-
4.00									

Surcharges

 $^{\ 1/\}$ Surcharges pursuant to Sections 18.2 and 18.10 of the General Terms and Conditions. See currently

effective Sheet No. 19. Reservation and Commodity surcharges will be billed in accordance with

the provisions of the General Terms and Conditions under which they are authorized.

```
2/ GRI Funding Unit pursuant to Section 18.1 of the General Terms and Conditions.

Minimum Reservation and Commodity Charge $ 0.00

Reservation Charge

High Load Factor Customer (>than 50%) $ 0.05

Low Load Factor Customer (< than 50%) 0.031

Commodity Charge 0.40 >
```

3/ Maximum firm volumetric rate applicable for capacity release, exclusive of surcharges.

Effective Date: 11/01/2002 Status: Effective

FERC Docket: RP03- 4-000

Second Revised Sheet No. 18 Second Revised Sheet No. 18: Pending Superseding: First Revised Sheet No. 18

CURRENTLY EFFECTIVE RATES RATE SCHEDULE DVS DELIVERY VARIANCE SERVICE

Each rate set forth on this tariff sheet is the currently effective rate pertaining to the particular rate schedule to which it is referenced, but each such rate is separate and independent and the change in any such rate shall not thereby effect a change in any other rate schedule.

	Maximum Rate Per Dt.	Minimum Rate Per Dt.	Fuel Reimbursement
Description			
Reservation Rate	\$ 6.71	-	
Commodity Rate	0.0066	\$ 0.0066	2.09 %

FERC Docket: RP01-556-000

Twenty-Seventh Revised Sheet No. 19 Twenty-Seventh Revised Sheet No. 19: Superseded Superseding: Twenty-Sixth Revised Sheet No. 19

CURRENTLY EFFECTIVE RATES

STATEMENT OF SURCHARGES

2 /	General Terms & Conditions Section	Description	Firm Transp Reservation		Interruptible Transportation Commodity 2/	SCT Commodity	
3/	18.2	ACA	-	0.21 >	0.21 >	0.21 >	
	18.10	Trunkline/TLC Settlement	\$ (0.01)	(0.03)	(0.03)	(0.09)	
			\$ (0.01) =====	0.18 >	0.18 >	0.12 >	

 $^{1/\,\,}$ Applicable to Transportation provided pursuant to Rate Schedules FT, EFT, LFT and HFT.

 $^{2/\,}$ Applicable to Transportation provided pursuant to Rate Schedules IT and EIT.

^{3/} Applicable to Transportation provided pursuant to Rate Schedule SCT.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

First Revised Sheet No. 20 First Revised Sheet No. 20: Superseded Superseding: Original Sheet No. 20

Notice of Cancellation of Sheet Nos. 20-25R

Notice is hereby given that effective June 15, 2003, Original Sheet Nos. 20, 21, 22, 23, 24A, 25, 25A, 25B, 25C, 25D, 25E, 25F, 25G, 25H, 25I, 25J, 25K, 25L, 25M, 25N, 25O, 25P, 25Q and 25R; and Sub First Revised Sheet No. 24 are cancelled and reserved for future use.

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Original Sheet No. 21 Original Sheet No. 21: Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Take-or-Pay Settlement Cost Surcharge Pursuant to Section 18.3

	Total Principal Amount (1)	Monthly Amount (2)
	(1)	(2)
Small General Service - Central Zone SG-2 and SSS-2		
Associated Natural Gas Company	\$ 20,128.20	\$ 559.12
Auburn, City of	-	_
Bowling Green Gas Company	.	. .
Bushnell, City of	29,073.10	807.59
Clarence, City of	-	_
Divernon, Village of	-	_
Edinburg, Village of	-	_
Franklin, Village of	.	-
Fulton, City of	•	2,314.11
Hermann, City of	46,723.39	
Macon, City of	36,267.40	
Madison, City of	-	_
Monroe City, City of	26 , 278.68	729.96
Montgomery City, City of	-	_
Morton, City of	_	_
Paris, City of	10,392.78	288.69
Pawnee, Village of	-	_
Perry, City of	-	_
Pittsfield, City of	-	_
Pleasant Hill, Village of	-	_
Riverton, Village of	-	_
Roodhouse, City of	-	_
Rossville, Village of	-	_
Shelbina, City of	12,193.43	338.71
Stonington, Village of	4,011.41	111.43
United Cities Gas Co Town	-	_
United Cities Gas Co Rurals	-	
Waverly, City of	-	-
Westville, Village of	-	-
White Hall, City of	-	-
Winchester, City of	-	-

FERC Docket: RS92- 22-005

Original Sheet No. 22 Original Sheet No. 22: Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Take-or-Pay Settlement Cost Surcharge Pursuant to Section 18.3

		Monthly Amount (2)
	(1)	(2)
Small General Service - Western Zone SG-3 and SSS-3		
Gruver, City of Hardesty, Town of	\$ -	\$ -
Hazelton, City of	_	_
LaCygne, Village of	_	_
Louisburg, City of	8,991.19	249.76
Miami Pipe Line Company, The	-	_
Morse Utility Company	-	-
Seiling Public Works Authority	8,413.59	233.71
Stinnett, City of	-	-
Sunray Utilities, City of		-
Taloga, Town of	2,943.71	81.77
Vici Public Works Authority	8,221.84	228.38
Limited Service - Eastern Zone LS-1		
Columbia Gas Transmission Corp.	-	_
East Ohio Gas Company, The	-	-
Limited Service - Central Zone LS-2		
Northern Illinois Gas Company	-	-
Seasonal Service - Eastern Zone SS-1		
Michigan Consolidated Gas Co.	19,872.51	552.01
Combined Service - Eastern Zone CS-1		
Michigan Gas Storage Company	-	-

⁽¹⁾ Individual customer amounts attributable to Transportation contracts resulting from conversions of sales service to Transportation service under Rate Schedules PT-Firm and SCT have been combined with sales service rate schedules.

⁽²⁾ Based upon 36-Month Amortization Period.

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Original Sheet No. 23 Original Sheet No. 23: Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Settlement Direct Bill Surcharge Pursuant to Section 18.5

Customer	Settlement Direct Bill Surcharge Amount	Monthly Principal Surcharge Amount 1/
	(1)	(2)
Associated Natural Gas Company Battle Creek Gas Company Central Illinois Light Company Central Illinois Public Service Co. Citizens Gas and Coke Utility Citizens Gas Fuel Company Columbia Gas Transmission Corp. Dayton Power and Light Co. Franklin, Village of Great River Gas Co.(United Cities) Hazelton, City of Illinois Power Company Indiana Gas Company, Inc. Kansas Power and Light Company Kokomo Gas and Fuel Company KPL Company (Gas Service Co.) LaCygne, Village of Lapel, Town of Louisburg, City of	\$ 207,653 266,435 4,004,033 3,835,036 3,739,481 612,049 17,379 71,983 23,200 620,731 2,655 4,317,193 8,000,000 593,314 1,660,535 356,906 14,233 59,600	\$ 3,461 4,441 66,734 63,917 62,325 10,201 290 1,200 387 10,346 44 71,953 133,333 9,889 27,676 5,948 237 993
Miami Pipe Line Company Missouri Public Service Company	- 798 , 056	- 13,301
Morse Utility Company Northern Illinois Gas Company Northern Indiana Fuel & Light Co. Northern Indiana Public Service Co. Ohio Gas Company Ohio Valley Gas Corporation Pittsboro, Town of Pittsfield, City of Richmond Gas Corporation Southeastern Michigan Gas Co. Sunray, City of Union Electric Company United Cities Gas Co.(Rurals & Town)	8 57,432 727,986 4,210,970 1,844,173 552,437 16,804 223,221 - 835,371 574 3,772,410 120,630	957 12,133 70,183 30,736 9,207 280 3,720 - 13,923 10 62,874 2,011

^{1/} Based on 60-month Amortization Period.

FERC Docket: RP93-172-007

Sub First Revised Sheet No. 24 Sub First Revised Sheet No. 24: Superseded

Superseding: Sub Original Sheet No. 24

PANHANDLE EASTERN PIPE LINE COMPANY

Account 191 Direct Bill Amounts Pursuant to Section 18.8

Customer	Revised April 30, 1993 Balance To Be Direct Billed 1/	Principal	Nine-Month Adjustment Period Amount To Be Direct Billed 3/
	(1)	(2)	(3)
Associated Natural Gas Company Central Illinois Light Company Central Illinois Public Service Co. Citizens Gas and Coke Utility Columbia Gas Transmission Corp. Dayton Power and Light Co. Illinois Power Company Kansas Power and Light Company (G-3) Kansas Power and Light Company (G-2) Kokomo Gas and Fuel Company Louisburg, City of Miami Pipe Line Company Missouri Public Service Company Morse Utility Company Northern Indiana Public Service Co. Ohio Gas Company Sunray, City of Union Electric Company United Cities Gas Co. (G-2)	1,614,142 19,862 109,213 3,262,420 268,302	\$ 36,973 76,816 227,316 134,512 1,655 9,101 271,868 22,359 39,671 56,906 12 (2) 6,644 94 291,951 40,569 129 272,803 23,160	\$ 49,979 46,975 238,244 115,803 2,237 12,303 311,212 25,269 48,025 72,947 (28) (55) 8,744 125 401,374 44,833 174 315,824 25,158
United Cities Gas Co. (SG-2 Rural) United Cities Gas Co. (SG-2 Town)	94,992 48,049	7,916 4,004	10,982 2,413

- 1/ Interest on these balances will be calculated from May 1, 1993 through the date of payment in the same manner as provided in Section 154.67 of the Commission's Regulations which is applicable to pipeline refunds.
- 2/ Customer may elect to make payments of the Total Principal Amount based on a 12-month amortization period commencing with the initial billing plus carrying charges calculated in the same manner as provided in Section 154.67 of the Commission's Regulations which is applicable to pipeline refunds.
- 3/ Interest on these balances will be calculated from February 1, 1994 through the date of payment in the same manner as provided in Section 154.67 of the Commission's Regulations which is applicable to pipeline refunds.

FERC Docket: RP94-393-000

Original Sheet No. 25 original Sheet No. 25 : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	S	Settlement Direct Bill Amount (3)
Battle Creek Gas Company Citizens Gas Fuel Company Citizens Gas & Coke Company	G-1 G-1 G-1		\$	7,055 3,251 33,439
Indiana Gas Company	G-1			51,928
Kokomo Gas & Fuel Company	G-1			9,077
Michigan Gas Utilities	G-1			2,215
Northern Indiana Fuel & Light Company, Inc.	G-1			3,598
Northern Indiana Public Service Company	G-1			21,254
Ohio Valley Gas Corporation Ohio Gas Company	G-1 G-1			1,450 7,664
Richmond Gas Corporation d/b/a Indiana Gas Company	G-1 G-1			4,511
Southeastern Michigan Gas Company	G-1			9,384
Central Illinois Light Company	G-2			27,883
Central Illinois Public Service Company	G-2			20,764
Kansas Power & Light Co. d/b/a Western Resources, Inc.	G-2			3,145
Great River Gas Co. d/b/a United Cities Gas Company	G-2			2,705
Illinois Power Company	G-2			24,247
Missouri Public Service Company	G-2			3,231
Union Electric Company	G-2			17,649
Kansas Power & Light Co. d/b/a Western Resources, Inc.	G-3			1,545
Bainbridge, Town of Columbia Gas Transmission Corporation	SG-1 SG-1			57 64
Dayton Power & Light Company	SG-1			204
Lapel, Town of	SG-1			187
Montezuma, Town of	SG-1			176
Pittsboro, Town of	SG-1			103
Roachdale, Town of	SG-1			135
Westfield Gas Corporation	SG-1			221
Associated Natural Gas Company	SG-2			820
Auburn, City of	SG-2			336
Bowling Green Gas Company	SG-2 SG-2			278 494
Bushnell, City of Clarence, City of	SG-2 SG-2			494 81
Divernon, City of	SG-2 SG-2			116
Edinburg, Village of	SG-2			114
Franklin, Village of	SG-2			67
Fulton, City of	SG-2			729
Hermann, City of	SG-2			454
Macon, City of	SG-2			483
Madison, City of	SG-2			50
Monroe, City of	SG-2			232
Montgomery, City of	SG-2 SG-2			151 962
Morton, City of Paris, City of	SG-2 SG-2			144
Pawnee, Village of	SG-2 SG-2			219
Perry, City of	SG-2			58
Pittsfield, City of	SG-2			547
Pleasant Hill, Village of	SG-2			96
Riverton, Village of	SG-2			243

FERC Docket: RP94-393-000

Original Sheet No. 25A original Sheet No. 25A : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)		Settlement Direct Bill Amount (3)
Roodhouse, City of Rossville, Village of Shelbina, City of Stonington, Village of United Cities Gas Co Town United Cities Gas Co Rural Waverly, City of Westville, Village of White Hall, City of Winchester, City of Gruver, City of Hardesty, Town of Hazelton, City of Lacygne, Village of Louisburg, City of The Miami Pipeline Company Morse Utility Company Seiling Public Works Authority Stinnett, City of Taloga, Town of Taloga, Town of Taloga, Town of Taloga Gas Storage Co. Columbia Gas Transmission Corporation Northern Illinois Gas Company Michigan Gas Storage Company (CD) Michigan Gas Storage Company (ACV) Ohio Gas Company Vinited Cities Gas Company Mid-Louisiana Gas Co. United Gas Pipe Line Co. Mississippi River Transmission Corporation Columbia Gas Transmission Corporation Columbia Gas Pipe Line Co. Southern Natural Gas Co. Transcontinental Gas Pipe Line Co. Southern Natural Gas Pipe Line Co. Southern Natural Gas Pipe Line Co. Diamond Shamrock Corp.			φ	
KN Energy, Inc. Seward County Gas Co. Kansas Power & Light Co. d/b/a Western Resources, Inc.	T-41 T-52 T-53			3 21 -

FERC Docket: RP94-393-000

Original Sheet No. 25B original Sheet No. 25B : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Kansas Power & Light Co. d/b/a Western Resources, Inc. KN Energy, Inc. Consolidated Gas Trans. Corp. Dekalb Swine Breeders, Inc. Trunkline Gas Company Columbia Gas Transmission Corporation MIGC, Inc. Colorado Interstate Gas Company KN Energy, Inc. Citizens Gas Fuel Company Ohio Gas Company Southeastern Michigan Gas Company Northern Indiana Fuel & Light Company, Inc. National Steel Corporation General Motors Corporation General Motors Corporation Ohio Gas Company Northern Indiana Fuel & Light Company, Inc. Southeastern Michigan Gas Company Indiana Gas Company Citizens Gas Fuel Company Michigan Gas Utilities Columbia Gas Othio East Ohio Gas Company East Ohio Gas Company East Ohio Gas Company Michigan Gas Storage Co. Battle Creek Gas Company Michigan Gas Storage Co. Battle Creek Gas Company Citizens Gas Fuel Company Michigan Gas Storage Co. Battle Creek Gas Company Citizens Gas Func Indiand Cogeneration Venture, L.P. Midland Cogeneration Venture, L.P. O-I Brockway Glass, Inc. Indiana Gas Company National Steel Corporation Central Illinois Light Company Great River Gas Co. d/b/a United Cities Gas Company General Motors Corporation Great River Gas Co. d/b/a United Cities Gas Company Central Illinois Light Company			
KPL Gas Service Co. d/b/a Western Resources, Inc. Illinois Power Company Kraft, Inc. d/b/a Kraft General Foods, Inc. Central Illinois Light Company Central Illinois Light Company Keystone Steel & Wire Company Teepak, Inc. Kansas Pipeline Partnership Kansas Pipeline Partnership	PT-Firm	2437 2470 2485 2487 2539 2593 2594 2459 2486	1,091 1,108 335 403 1,674 119 91 813 788

FERC Docket: RP94-393-000

Original Sheet No. 25C original Sheet No. 25C : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	:	Settlement Direct Bill Amount (3)
Mueller Brass Co.	PT-Int	483	\$	144
Dunn/Seco Partners	PT-Int	719		599
Michigan Sugar Company	PT-Int	726		452
Michigan Gas Utilities	PT-Int	789		845
Michigan Gas Utilities	PT-Int	791		75
Citizens Gas & Coke Utility	PT-Int	792		26
Citizens Gas Fuel Company	PT-Int	811		175
Kokomo Gas & Fuel Company	PT-Int	825		1,290
Kokomo Gas & Fuel Company	PT-Int	826		62
East Ohio Gas Company	PT-Int	831		10,632
Delhi Gas Pipeline Corporation	PT-Int	833		564
Kokomo Gas & Fuel Company	PT-Int	834		25
Kokomo Gas & Fuel Company	PT-Int	835		22
Kokomo Gas & Fuel Company	PT-Int	836		1
Battle Creek Gas Company	PT-Int	843		514
Michigan Gas Utilities	PT-Int	866		58
Baltimore Gas & Electric Company	PT-Int	877		216
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	884		259
Northern Indiana Fuel & Light Company, Inc.	PT-Int	887		203
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	899		134
Indiana Gas Company	PT-Int	903		405
Indiana Gas Company	PT-Int	904		697
Indiana Gas Company	PT-Int	905		931
Indiana Gas Company	PT-Int	908		318
Battle Creek Gas Company	PT-Int	909		291
Battle Creek Gas Company	PT-Int	910		975
Indiana Gas Company	PT-Int	911		1,753
Southeastern Michigan Gas Company	PT-Int	912		177
Indiana Gas Company	PT-Int	913		493
Indiana Gas Company	PT-Int	914		716
Indiana Gas Company	PT-Int	933		132
Kokomo Gas & Fuel Company	PT-Int	935		684
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	936		552
Michigan Gas Utilities	PT-Int	938		275
Indiana Gas Company	PT-Int	962		111
Ohio Gas Company	PT-Int	971		150
Indiana Gas Company	PT-Int	975		650
Asphalt Products Corp.	PT-Int	976		22
Ace Asphalt & Paving Co., Inc.	PT-Int	977		41
Slater Steels Corp.	PT-Int	979		174
St. Joseph Mercy Hospital	PT-Int	988		14
Industrial Steel Treating Co.	PT-Int	989		16
Indiana Gas Company	PT-Int	995		268
William Beaumont Hospital	PT-Int	997		12
East Ohio Gas Company	PT-Int	1003		90
Consumers Power Company	PT-Int	1008		6,661
Hadson Gas Systems, Inc.	PT-Int	1009		648
Valley Asphalt Company	PT-Int	1012		8
Indiana Gas Company	PT-Int	1029		27

FERC Docket: RP94-393-000

Original Sheet No. 25D original Sheet No. 25D : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Quivira Gas Company	PT-Int	1037	\$ 1,344
Indiana Gas Company	PT-Int	1038	111
Indiana Gas Company	PT-Int	1039	244
Kokomo Gas & Fuel Company	PT-Int	1040	1
Wayne County Dep. of Public Services	PT-Int	1044	25
Indiana Gas Company	PT-Int	1049	103
Centana Energy Corporation	PT-Int	1052	902
Indiana Gas Company	PT-Int	1059	343
Access Energy Pipeline Corporation	PT-Int	1064	203
Ohio Gas Company	PT-Int	1066	23
Battle Creek Gas Company	PT-Int	1067	212
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1068	51
Columbia Gas of Ohio	PT-Int	1069	2,387
Columbia Gas of Virginia	PT-Int	1070	300
Columbia Gas of New York	PT-Int	1071	423
Columbia Gas of Kentucky	PT-Int	1072	373
Columbia Gas of Maryland	PT-Int	1073	306
Columbia Gas of Pennsylvania	PT-Int PT-Int	1074 1083	1,011 6
Engineered Heat Treat, Inc.	PT-Int	1088	7
Hitachi Magnetics Corporation Indiana Gas Company	PT-Int	1102	102
Indiana Gas Company Indiana Gas Company	PT-Int	1102	311
James River - KVP Group	PT-Int	1105	1
Ohio Gas Company	PT-Int	1106	49
Indiana Gas Company	PT-Int	1107	805
Michigan Sugar Company	PT-Int	1118	15
Indiana Gas Company	PT-Int	1119	284
Indiana Gas Company	PT-Int	1120	99
Indiana Gas Company	PT-Int	1121	77
Indiana Gas Company	PT-Int	1122	276
Indiana Gas Company	PT-Int	1123	16
Indiana Gas Company	PT-Int	1125	51
Indiana Gas Company	PT-Int	1126	399
Indiana Gas Company	PT-Int	1127	175
Ohio Gas Company	PT-Int	1187	7
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	1188	17
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	1189	186
Indiana Gas Company	PT-Int PT-Int	1190 1191	117
Indiana Gas Company Citizens Gas & Coke Utility	PT-Int	1191	459 358
Ohio Gas Company	PT-Int	1196	5
Indiana Gas Company	PT-Int	1198	276
Howard Plating Industries	PT-Int	1202	1
Enamalum Corporation	PT-Int	1221	1
Vassar Manufacturing	PT-Int	1224	15
Ohio Gas Company	PT-Int	1226	3
Battle Creek Gas Company	PT-Int	1227	94
Central Soya Co., Inc.	PT-Int	1232	109
Indiana Gas Company	PT-Int	1235	60

FERC Docket: RP94-393-000

Original Sheet No. 25E original Sheet No. 25E : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1239	\$ 87
Indiana Gas Company	PT-Int	1240	129
Indiana Gas Company	PT-Int	1246	247
Indiana Gas Company	PT-Int	1247	72
Indiana Gas Company	PT-Int	1248	80
Indiana Gas Company	PT-Int	1249	69
Indiana Gas Company	PT-Int	1250	262
Indiana Gas Company	PT-Int	1251	186
Hoover Group	PT-Int	1255	_
Indiana Gas Company	PT-Int	1258	370
Indiana Gas Company	PT-Int	1259	783
Indiana Gas Company	PT-Int	1261	341
Oppenheimer Livonia Associates	PT-Int	1263	4
Thermo Processing Co., Inc.	PT-Int	1270	13
Thermo Electron Corp.	PT-Int	1274	2
Indiana Gas Company	PT-Int	1279	177
Access Energy Pipeline Corporation	PT-Int	1288	214
Essex Group, Inc.	PT-Int	1292	4
State of Michigan	PT-Int	1296	1,176
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1302	40
Ohio Gas Company	PT-Int	1303	247
MichCon Trading Co. d/b/a Coenergy Trading Company	PT-Int	1317	1,072
Consumers Power Company	PT-Int	1332	2
Green Bay Food Company	PT-Int	1334	40
Clawson School District	PT-Int	1340	2
Indiana Gas Company	PT-Int	1341	94
Indiana Gas Company	PT-Int	1342	158
Indiana Gas Company	PT-Int	1343	91
Indiana Gas Company	PT-Int	1344	66
Indiana Gas Company	PT-Int	1345	75
Citizens Gas & Coke Utility	PT-Int	1353	60
Citizens Gas & Coke Utility	PT-Int	1354	209
Indiana Gas Company	PT-Int	1356	190
Citizens Gas Supply Corporation	PT-Int	1362	5,052
Ohio Gas Company	PT-Int	1364	4
Ohio Gas Company	PT-Int	1365	39
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1397	183
Indiana Gas Company	PT-Int	1401	76
Indiana Gas Company	PT-Int	1402	132
Indiana Gas Company	PT-Int	1403	89
Foodland Distributors	PT-Int	1412	1
Citizens Gas & Coke Utility	PT-Int	1422	28,894
Citizens Gas & Coke Utility	PT-Int	1426	79
Battle Creek Gas Company	PT-Int	1444	1,588
Indiana Gas Company	PT-Int	1446	25,782
East Ohio Gas Company	PT-Int	1447	411
Central Soya Co., Inc.	PT-Int	1451	66
Metallurgical Processing, Inc.	PT-Int	1452	136
Ohio Gas Company	PT-Int	1461	4,080

FERC Docket: RP94-393-000

Original Sheet No. 25F original Sheet No. 25F : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
General Electric Company	PT-Int	1463	\$ 266
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1464	290
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	1465	91
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	1466	1 , 657
Kokomo Gas & Fuel Company	PT-Int	1470	5,911
Ohio Valley Gas Corporation	PT-Int	1472	700
Indiana Gas Company	PT-Int	1473	12
Elias Brothers Restaurants Inc.	PT-Int	1474	7
Madonna College	PT-Int	1481	-
National Steel Corporation	PT-Int	1492	205
Indiana Gas Company	PT-Int	1496	56
SEMCO Energy Services, Inc.	PT-Int	1506	227
E.I. du Pont de Nemours	PT-Int	1514	55
Southeastern Michigan Gas Company	PT-Int	1524	465
Battle Creek Gas Company	PT-Int	1530	101
Citizens Gas Fuel Company	PT-Int	1534	827
SEMCO Energy Services, Inc.	PT-Int	1549	23
MichCon Trading Co. d/b/a Coenergy Trading Company	PT-Int	1551	318 90
Indiana Gas Company	PT-Int	1552	90 72
Indiana Gas Company	PT-Int PT-Int	1558 1569	· -
Ford Motor Company Ford Motor & Rouge Steel	PT-Int	1570	2,122 5,948
Citizens Gas Fuel Company	PT-Int	1571	1,015
Ohio Gas Company	PT-Int	1572	950
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	1575	53
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	1578	5,942
Unicorp Energy d/b/a Westcoast Gas Services (U.S.A.), Inc.	PT-Int	1580	908
Llano, Inc.	PT-Int	1583	488
Southeastern Michigan Gas Company	PT-Int	1586	82
Continental Natural Gas, Inc.	PT-Int	1603	141
Eastex Gas Transmission Company	PT-Int	1605	16
Eastex Gas Transmission Company	PT-Int	1606	40
Woodward Pipeline Inc.	PT-Int	1608	4
Natural Gas Clearinghouse	PT-Int	1610	136
SEMCO Energy Services, Inc.	PT-Int	1611	5
Access Energy Pipeline Corporation	PT-Int	1613	196
Indiana Gas Company	PT-Int	1614	60
Central Soya Co., Inc.	PT-Int	1632	65
Nagasco Marketing, Inc.	PT-Int	1633	28
Consolidated Fuel Supply, Inc.	PT-Int	1634	484
Standard Gas Marketing d/b/a Transok Gas Company	PT-Int	1637	13
Ohio Gas Marketing Corp.	PT-Int	1641	5
PSI, Inc. d/b/a Aquila Energy Marketing Corporation	PT-Int	1682	304
Boyd, Rosene & Associates Inc.	PT-Int	1689	165
Xebec Gas Co. d/b/a Enmark Gas Corp.	PT-Int PT-Int	1698 1700	55 23
Dart Container Corporation	PT-Int PT-Int	1700	23 85
Seagull Marketing Services, Inc.	PT-Int PT-Int	1702	85 5
Unicorp Energy d/b/a Westcoast Gas Services (U.S.A.), Inc. Boyd, Rosene & Associates Inc.	PT-Int	1709	1,300
Doya, Roberta & Roboctated The.	T T T11C	1,00	1,500

FERC Docket: RP94-393-000

Original Sheet No. 25G original Sheet No. 25G : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	:	Settlement Direct Bill Amount (3)
Llano, Inc.	PT-Int	1736	\$	851
Citizens Gas & Coke Utility	PT-Int	1737		451
LL & E Gas Marketing, Inc.	PT-Int	1739		182
Michigan Gas Utilities	PT-Int	1761		9
Victoria Gas Corp.	PT-Int	1763		36
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	1764		126
Xebec Gas Co. d/b/a Enmark Gas Corp.	PT-Int	1771		8
Olympic Pipeline Co. d/b/a Olympic Fuels Company	PT-Int	1775		2,192
V.H.C. Gas Systems, L.P. d/b/a Valero Gas Marketing, L.P.	PT-Int	1776		241
Citizens Gas & Coke Utility	PT-Int	1779		24
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	1781		91
Panhandle Trading Co. d/b/a 1 Source Energy Services Company	PT-Int	1784		488
Ohio Gas Company	PT-Int	1801		323
Union Gas Limited	PT-Int	1819		3,801
Michigan Gas Utilities	PT-Int	1824		82
Consumers Power Company	PT-Int	1829		5
SEMCO Energy Services, Inc.	PT-Int	1839		67
Battle Creek Gas Company	PT-Int	1843		4
Battle Creek Gas Company	PT-Int	1853		5
Michigan Gas Utilities	PT-Int	1854		19
Bundy Tubing Corporation	PT-Int	1860		16
American Central Gas Marketing Company	PT-Int	1863		4,116
Access Energy Corporation	PT-Int	1890		17
Consumers Power Company	PT-Int	1895		25
Natural Gas Clearinghouse	PT-Int	1896		366
Columbia Gas of Ohio	PT-Int	1898		7,118
Northern Indiana Fuel & Light Company, Inc.	PT-Int	1902		79
Indiana Gas Company	PT-Int	1906		10
Michigan Gas Utilities	PT-Int	1914		21
Xebec Gas Co. d/b/a Enmark Gas Corp.	PT-Int	1919		56
Battle Creek Gas Company	PT-Int	1924		9
Battle Creek Gas Company	PT-Int	1925		7
Krupp & Associates	PT-Int	1926		129
Ohio Gas Company	PT-Int	1933		2
Kerr-McGee Corporation	PT-Int	1938		106
Michigan Gas Utilities	PT-Int	1939		344
Union Gas Limited	PT-Int	1960		1,151
Boyd, Rosene & Associates Inc.	PT-Int	2035		25
Battle Creek Gas Company	PT-Int	2040		28
Mountain Iron & Supply Company	PT-Int	2043		1
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	2059		40
Unicorp Energy d/b/a Westcoast Gas Services (U.S.A.), Inc.	PT-Int	2075		86
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	2084		66
Northern Indiana Fuel & Light Company, Inc.	PT-Int	2085		34
Gulf Ohio Pipeline Corporation	PT-Int	2086		378
GasTrak Corporation	PT-Int	2093		1,473
Mobil Oil Corp. d/b/a Mobil Natural Gas Inc.	PT-Int	2102		70
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	2127		81
Louisiana State Gas Corporation	PT-Int	2130		4

FERC Docket: RP94-393-000

Original Sheet No. 25H original Sheet No. 25H : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)		Settlement Direct Bill Amount (3)
Consolidated Fuel Corporation	PT-Int	2134	\$	82
TranState Gas Service Co. d/b/a Truimph Natural Gas, Inc.	PT-Int	2135	~	351
Battle Creek Gas Company	PT-Int	2139		900
Michigan Gas Utilities	PT-Int	2142		1,105
Mobil Oil Corp. d/b/a Mobil Natural Gas Inc.	PT-Int	2149		986
Golden Gas Energies, Inc.	PT-Int	2151		67
Central Soya Co., Inc.	PT-Int	2165		13
National Steel Corporation	PT-Int	2178		59
Reliance Gas Marketing Co. d/b/a Anadarko Trading Company	PT-Int	2182		102
Consolidated Fuel Corporation	PT-Int	2191		67
Unicorp Energy d/b/a Westcoast Gas Services (U.S.A.), Inc.	PT-Int	2195		34
Consolidated Fuel Corporation	PT-Int	2197		11
Gulf Energy Marketing Company	PT-Int	2201		1
Consolidated Fuel Corporation	PT-Int	2205		42
Consolidated Fuel Corporation	PT-Int	2206		15
Consolidated Fuel Corporation	PT-Int	2207		15
Gulf Ohio Pipeline Corporation	PT-Int	2215		57
Bethlehem Steel Corp.	PT-Int	2220		59
Consolidated Fuel Corporation	PT-Int	2227		49
Consolidated Fuel Corporation	PT-Int	2232		1 , 679
Panhandle Trading Co. d/b/a 1 Source Energy Services Company	PT-Int	2244		38
Consolidated Fuel Corporation	PT-Int	2247		1
GasTrak Corporation	PT-Int	2259		76
Michigan Gas Utilities	PT-Int	2271		24
James River - KVP Group	PT-Int	2274		12
BP Chemicals America, Inc.	PT-Int	2277		1
Indiana Gas Company	PT-Int PT-Int	2279 2316		55 33
Loutex Energy d/b/a LEDCO, Inc.	PT-Int PT-Int	2316		33 6
Ohio Gas Company	PT-Int	2344		3
Indiana Gas Company Citizens Gas Supply Corporation	PT-Int	2376		1,301
Williams Gas Marketing Company	PT-Int	2405		105
Consolidated Fuel Corporation	PT-Int	2427		42
Conoco, Inc.	PT-Int	2428		33
Ohio Gas Company	PT-Int	2444		1
National Steel Corporation	PT-Int	2494		3,185
Richmond Gas Corporation d/b/a Indiana Gas Company	PT-Int	2497		15
Indiana Gas Company	PT-Int	2502		65
Indiana Gas Company	PT-Int	2503		125
Indiana Gas Company	PT-Int	2504		28
Unicorp Energy d/b/a Westcoast Gas Services (U.S.A.), Inc.	PT-Int	2508		18
Indiana Gas Company	PT-Int	2541		132
Ohio Gas Company	PT-Int	2547		8
Citizens Gas Fuel Company	PT-Int	2618		244
Indiana Gas Company	PT-Int	2627		29
Northern Indiana Fuel & Light Company, Inc.	PT-Int	2638		87
Kokomo Gas & Fuel Company	PT-Int	2641		452

FERC Docket: RP94-393-000

Original Sheet No. 251 Original Sheet No. 251 : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Midwest Solvents Co.	PT-Int	482	\$ 117
University of Illinois & Ill. Power Company	PT-Int	526	691
Keystone Steel & Wire Company	PT-Int	794	826
A.E. Staley Manufacturing Co.	PT-Int	798	1,569
Union Electric Company	PT-Int	799	256
Union Electric Company	PT-Int	800	184
Macon Municipal Utilities	PT-Int	823	54
Macon Municipal Utilities	PT-Int	824	46
HUMKO Products, Division of Kraft	PT-Int	829	843
Archer Daniels Midland Company	PT-Int	830	899
Pekin Energy Company	PT-Int	832	81
Fulton, City of	PT-Int	837	123
Great River Gas Co. d/b/a United Cities Gas Company	PT-Int	838	93
Great River Gas Co. d/b/a United Cities Gas Company	PT-Int	839	27
Briggs Manufacturing Co.	PT-Int	841	119
Caterpillar, Inc.	PT-Int	844	1,545
Sherex Chemical Company, Inc.	PT-Int	845	633
Monroe, City of	PT-Int	861	46
Mueller Company	PT-Int	864	16
Springfield Public S.D. #186	PT-Int	865	101
Monroe, City of	PT-Int	873	96
Teepak, Inc.	PT-Int	875	665
Borden Chemicals & Plastics	PT-Int	879	982
Carnation Co. d/b/a Nestle Food Company	PT-Int	880	139
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	881	318
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	882	60
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	883	237
Morton, Village of	PT-Int	885	46
Union Electric Company	PT-Int	892	41
Union Electric Company	PT-Int	893	16
Union Electric Company	PT-Int	894	177
NTN-Bower Corp.	PT-Int	901	111
Georgia-Pacific Corp.	PT-Int	916	374
Quincy Soybean Company	PT-Int	917	218
Midwest Grain Products of Illinois	PT-Int	918	230
Sisters of Third Order St. Francis	PT-Int	919	238
Can-Am Industries, Inc. d/b/a Titan Wheel International	PT-Int	924	209
McGraw Edison Power System	PT-Int	930	24
St. Mary Hospital	PT-Int	931 932	39 42
Proctor Community Hospital	PT-Int	932	
Lauhoff Grain Company	PT-Int	943 957	510
VHA-Illinois Blessing	PT-Int PT-Int	957 958	21 19
VHA-Illinois McDonough National Standard Co FPD	PT-Int	960	5
Industrial Machinery Division - GD	PT-INC PT-Int	961	39
B. F. Goodrich Chemical Group d/b/a B. F. Goodrich Company	PT-INC PT-Int	961	159
Union Electric Company	PT-INC PT-Int	978	23
Illinois Power Company	PT-Int	981	182
Kansas Power & Light Co. d/b/a Western Resources, Inc.	PT-INC PT-Int	982	763
named fower a figure co. a/b/a western resources, file.	1 1 111 C	202	100

FERC Docket: RP94-393-000

Original Sheet No. 25J <code>Original Sheet No. 25J : Superseded</code>

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Archer Daniels Midland Company	PT-Int	985	\$ 1,779
Union Electric Company	PT-Int	987	71
Ball-InCon Packaging d/b/a Ball Glass Container Corp.	PT-Int	991	439
Bradley University	PT-Int	992	125
Methodist Medical Center of Illinois	PT-Int	993	106
Central Illinois Public Service Company	PT-Int	1001	111
Wayne Pet Foods	PT-Int	1010	35
Lear Gas Marketing Co. d/b/a Transok Gas Company	PT-Int	1017	72
Haeger Potteries of Macomb, Inc.	PT-Int	1019	45
National Starch & Chemical Corporation	PT-Int	1022	139
National By-Products, Inc.	PT-Int	1024	89
Eaton Corporation	PT-Int	1027	32
Quaker Oats Company	PT-Int	1030	100
Quaker Oats Company	PT-Int	1032	112
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	1036	7
Centana Energy Corporation	PT-Int	1051	816
Centana Energy Corporation	PT-Int	1053	219
Swift-Eckrich	PT-Int	1054	51
Central Illinois Public Service Company	PT-Int	1056	365
Prince Manufacturing Company	PT-Int	1058	28
Lakeview Med. Center d/b/a United Samaritans Medical Center	PT-Int	1063	46
Access Energy Pipeline Corporation	PT-Int	1065	987
Gates Rubber Company	PT-Int	1075	25
Peoria Blacktop, Inc.	PT-Int	1077	51
Pekin Memorial Hospital	PT-Int	1078	38
Decatur Memorial Hospital	PT-Int	1100	46
Morton, Village of	PT-Int	1108	2
Fulton, City of	PT-Int	1109	24
Fulton, City of	PT-Int	1110	13
Fulton, City of	PT-Int	1111	24
Fulton, City of	PT-Int	1112	32
Fulton, City of	PT-Int	1113	57
Fulton, City of	PT-Int	1114	14
Fulton, City of	PT-Int	1115	35
Fulton, City of	PT-Int	1116	62
Centana Energy Corporation	PT-Int	1130	402
Roanoke Farmers Association	PT-Int	1183	11
Illinois Central College	PT-Int	1184	45
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	1194	127
Pillsbury Company	PT-Int	1206	97
Sarah Bush Lincoln Health Center	PT-Int	1208	63
Illinois Roses Limited	PT-Int	1213	69
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	1231	34
Great River Gas Co. d/b/a United Cities Gas Company	PT-Int	1243	56
Peoria Housing Authority	PT-Int	1244	132
Shelbina, City of	PT-Int	1252	34
Lear Gas Marketing Co. d/b/a Transok Gas Company	PT-Int	1257	34
Union Electric Company	PT-Int	1281	56
PMP Fermentation Products, Inc.	PT-Int	1282	100

FERC Docket: RP94-393-000

Original Sheet No. 25K original Sheet No. 25K : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

	38 138 165
Peoria, County of PT-Int 1283 \$	
Archer Daniels Midland Company PT-Int 1286	165
Cimarron River Pipeline System PT-Int 1289	T 62
Royal Haeger Lamp Company PT-Int 1290	28
Trailmobile, Inc. PT-Int 1291	40
Union Electric Company PT-Int 1294	24
Kansas Pipeline Partnership PT-Int 1298	315
Kansas Pipeline Partnership PT-Int 1301	24
Interstate Brands Corporation PT-Int 1305	6
Williamsville C.U.S.D. #15 PT-Int 1311	7
Rochester C.U.S.D. #3A PT-Int 1312	9
Lincoln Land Community College PT-Int 1313	30
Sangamon State University PT-Int 1314	22
Ball-Chatham C.U.S.D. #5 PT-Int 1315	11
Peoria Public Schools District #150 PT-Int 1318	104
Springfield, City of PT-Int 1320	42
Memorial Medical Center PT-Int 1321	128
W.R. Grace & Co. d/b/a The Henry Fertilizer Company PT-Int 1322	71
Eureka Hospital PT-Int 1347	1
	25
	5 7
±	
	18
Abraham Lincoln Memorial Hospital PT-Int 1363 Speedrack, Inc. PT-Int 1366	25 2
	9
Lincoln Elementary S.D. #27 PT-Int 1407	
Riverton, Village of PT-Int 1419	9
3 1 1	,264
Bartlow Brothers, Inc. PT-Int 1441	17
	,722
Lucky Lady Laundries PT-Int 1462	6
R.R. Donnelley & Sons Company PT-Int 1475	98
Community U.S.D. #7 PT-Int 1476	4
Lincoln Community High School PT-Int 1491	9
Forman C.U.S.D. #124 PT-Int 1501	8
Quincy College Corp. d/b/a Quincy University Corporation PT-Int 1505	10
	,429
Union Electric Company PT-Int 1521	46
Union Electric Company PT-Int 1522	57
Union Electric Company PT-Int 1523	10
Golden Gas Energies, Inc. PT-Int 1527	47
Union Electric Company PT-Int 1538	51
Victoria Gas Corp. PT-Int 1553	2
Caterpillar, Inc. PT-Int 1560	47
San Jose C.U.S.D. #122 PT-Int 1561	5
Pillsbury Company PT-Int 1567	37
Central Illinois Public Service Company PT-Int 1577	1
Access Energy Pipeline Corporation PT-Int 1594	53
Eastex Gas Transmission Company PT-Int 1595	17

FERC Docket: RP94-393-000

Original Sheet No. 25L original Sheet No. 25L : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Eastex Gas Transmission Company	PT-Int	1596	\$ 51
Clinton Gas Transmission, Inc.	PT-Int	1619	396
Union Electric Company	PT-Int	1629	34
Utilicorp United, Inc. d/b/a Missouri Public Service	PT-Int	1650	991
Macon Municipal Utilities	PT-Int	1661	94
Quincy Public Schools	PT-Int	1665	4
Illini Bluffs U.S.D. #327	PT-Int	1669	
Quincy Pepsi-Cola Bottling Co.	PT-Int	1672	1
Central Illinois Public Service Company	PT-Int	1673	1
Sunset Home of the United Methodist Church	PT-Int	1674	1
United States Gypsum Co.	PT-Int	1687	234
Associated Intrastate Pipeline Company	PT-Int	1692	796
Associated Intrastate Pipeline Company Associated Intrastate Pipeline Company	PT-Int	1703	1,883
Woodward Pipeline Inc.	PT-Int	1710	411
Burtonview Grain, Coal & Lumber Company	PT-Int	1716	411
Delavan Co-Operative Elevator Company	PT-Int	1717	
	PT-Int	1719	1
Farmers Cooperative Association	PT-Int	1720	1
Emden Farmers Grain Company	PT-Int	1721	1
Farmers Grain Company of Latham			2
East Lincoln Farmers' Grain Co.	PT-Int	1722	_
Elmwood Elevator Company	PT-Int	1723	1
Putnam Grain Company	PT-Int	1725	-
Funk's Grove Grain Company	PT-Int	1726	2
Farmers Grain Company	PT-Int	1727	5
Tremont Co-Op Grain Company	PT-Int	1728	6
Colt Industries, Inc.	PT-Int	1731	1
Continental Bondware	PT-Int	1732	6
Farmers Grain & Coal Company	PT-Int	1734	4
Assumption Cooperative Grain Company	PT-Int	1749	3
Ashland Farmers Elevator Company	PT-Int	1751	2
Schuyler-Brown F.S., Inc.	PT-Int	1752	2
Culver-Fancy Prairie Co-Op	PT-Int	1753	2
Gill Grain Company	PT-Int	1756	1
Rumbold Grain Co., Inc.	PT-Int	1758	5
Minier Co-Op Grain Company	PT-Int	1759	5
Arthur Community Unit Schools	PT-Int	1762	2
Mt. Pulaski Community Unit S.D d/b/a Transok Inc.	PT-Int	1768	7
Atwood Grain & Supply Company	PT-Int	1772	1
Stanford Grain Company	PT-Int	1787	1
Lehn & Finks Products Group	PT-Int	1798	42
Vesta Energy Company	PT-Int	1799	65
Mountain Industrial Gas Co. d/b/a Coastal Gas Marketing Co.	PT-Int	1803	113
GasTrak Corporation	PT-Int	1806	90
Access Energy Pipeline Corporation	PT-Int	1810	2
Union Electric Company	PT-Int	1813	6,472
Natural Gas Clearinghouse	PT-Int	1814	133
Union Electric Company	PT-Int	1816	163
Union Electric Company	PT-Int	1817	25
Eastex Gas Transmission Company	PT-Int	1820	8

FERC Docket: RP94-393-000

Original Sheet No. 25M original Sheet No. 25M : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Eastex Gas Transmission Company	PT-Int	1821	\$ 82
Hopedale Medical Complex	PT-Int	1845	9
Central Illinois Public Service Company	PT-Int	1846	9
Eureka College	PT-Int	1852	8
Access Energy Pipeline Corporation	PT-Int	1855	-
Archer Daniels Midland Company	PT-Int	1864	123
Amgas, Inc.	PT-Int	1866	945
Bishop Pipeline Corporation d/b/a MG Natural Gas Corp.	PT-Int	1868	464
St. John's Hospital	PT-Int	1870	195
Stonington C.U.S.D. #7	PT-Int	1872	7
Frericks Gardens, Inc.	PT-Int	1874	2
Great River Gas Co. d/b/a United Cities Gas Company	PT-Int	1876	300
Ramada Renaissance Hotel	PT-Int	1877	17
Woodward Pipeline Inc.	PT-Int	1878	255
International Paper Company	PT-Int	1879	4
Green Valley Community H.S.	PT-Int	1880	2
Sunrise Resources Inc. d/b/a Energy Dynamics, Inc.	PT-Int	1882	10
Morton, Village of	PT-Int	1887	2
Enron Gas Marketing, Inc.	PT-Int	1903	41
Union Electric Company	PT-Int	1907	20
Shelbina, City of	PT-Int	1909	10
Rossville Grain Co.	PT-Int	1910	2
Mountain Iron & Supply Company	PT-Int	1911	122
Union Electric Company	PT-Int	1917	22
Union Electric Company	PT-Int	1918	8
Associated Natural Gas, Inc.	PT-Int	1921	94
Eastex Gas Transmission Company	PT-Int	1928	11
Rossville-Alvin U.S.D. #7	PT-Int	1931	5
Quivira Gas Company	PT-Int	1943	2
Ladd Gas Marketing, Inc. d/b/a Amax Gas Marketing, Inc.	PT-Int	1959	45
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	1965	91
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	1966	84
Eastex Gas Transmission Company	PT-Int	1967	40
Slater Steels Corp.	PT-Int	1969	90
Beason C.U.S.D. #17	PT-Int	1970	4
Douglas County Board	PT-Int	1971	25
Peoria Journal Star, Inc.	PT-Int	1972	5
Broadwell C.C.S.D. #68	PT-Int	1973	1
Sparland C.U.S.D. #3	PT-Int	1974	4 8
Mid County Community Unit #4	PT-Int	1975 1976	5
Henry-Senachwine Consolidated #20 Chester E. Lincoln C.C.S.D #61	PT-Int PT-Int	1979	4
Tazewell-Mason Counties	PT-Int	1980	_ 4
Henry Comm. Cons. G.S. #35	PT-Int	1981	7
Hartsburg-Emden C. U. D #21	PT-Int	1982	5
N. Pekin-Marquette Heights S. D. #102	PT-Int	1983	3
Shiloh School District #2	PT-Int	1984	4
Prairie Farms Dairy, Inc.	PT-Int	1985	18
Apostolic Christian Home	PT-Int	1987	
npostorie onristian nome	1 1 111C	100	

FERC Docket: RP94-393-000

Original Sheet No. 25N original Sheet No. 25N : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
South Pekin Grade School #137	PT-Int	1988	\$ 2
Villa Grove C.U.S.D. #302	PT-Int	1989	4
Roanoke Benson S.D. #60	PT-Int	1990	2
Oak Grove Schools #68	PT-Int	1992	4
Eureka C.U.S.D. #140	PT-Int	1993	3
Peoria Heights C.U.S.D. #325	PT-Int	1994	8
St. Patrick Roman Catholic Congregation	PT-Int	1995	2
Newman C.U.S.D. #303	PT-Int	1996	3
Washington School District #52	PT-Int	1997	4
Limestone Walter S.D. #316	PT-Int	1999	_
Federal Warehouse Company	PT-Int	2000	7
Peoria Park District	PT-Int	2001	10
Hotel Pere Marquette	PT-Int	2002	26
Catholic Diocese of Peoria	PT-Int	2003	52
Monroe School District #70	PT-Int	2004	2
Fleming-Potter Company, Inc.	PT-Int	2005	29
Logan County Courthouse	PT-Int	2006	1
Illinois Valley Central U.D. #321	PT-Int	2007	10
Limestone Community H.S. #310	PT-Int	2008	7
Olympia C.U.S.D. #16	PT-Int	2009	11
Farmington Central C.U.S.D. #265	PT-Int	2010	10
Illini Bluffs U.S.D. #327	PT-Int	2011	2
Pleasant Valley S.D. #62	PT-Int	2013	2
East Peoria Elementary School District #86	PT-Int	2014	12
Delavan C.U.S.D. #703	PT-Int	2015	4
Washington District 50 Schools	PT-Int	2016	2
Atwood-Hammond Community Unit #39	PT-Int	2017	3
Tuscola C.S.D. #301	PT-Int	2018	7
Dunlap School District #323	PT-Int	2019	13
Washington Community H.S. #308	PT-Int	2020	5
Pekin High School #303	PT-Int	2021	15
Creve Coeur S.D. #76	PT-Int	2022	2
Pekin Public Schools District #108	PT-Int	2023	14
Bellevue Public School #152	PT-Int	2024	1
Metamora Township H.S.	PT-Int	2025	4
Tremont C.U.D. #702	PT-Int	2026	5
East Peoria Community High School	PT-Int	2027	7
Norwood School District #63	PT-Int	2028	2
Bartonville Grade School #66	PT-Int	2029	1
Robein School District #85	PT-Int	2030	1
Rankin Community District #98	PT-Int	2031	1
Edgar County C.U.S.D. #6	PT-Int	2032	4
Washington CentColumbia S.D.	PT-Int	2033	3
Lonza, Inc.	PT-Int	2038	8
Kraft, IncInd. Foods d/b/a Kraft General Foods, Inc.	PT-Int	2045	116
H.M. Buckley & Sons, Inc.	PT-Int	2046	40
Access Energy Pipeline Corporation	PT-Int	2047	22
Ohio Gas Marketing Corp. d/b/a AM International/Bruning Div.	PT-Int	2051	24
Seagull Marketing Services, Inc.	PT-Int	2053	251

FERC Docket: RP94-393-000

Original Sheet No. 250 original Sheet No. 250 : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)		Settlement Direct Bill Amount (3)
Consumers Gas Marketing, Inc.	PT-Int	2056	\$	19
Colony Natural Gas Corporation	PT-Int	2066	Ċ	30
Cabot Energy Marketing Corp. d/b/a Anthem Energy Co., L.P.	PT-Int	2068		60
Union Electric Company	PT-Int	2069		54
Eastex Gas Transmission Company	PT-Int	2073		17
Franklin C.U.S.D. #1	PT-Int	2076		1
Macon Municipal Utilities	PT-Int	2077		2
Amgas, Inc.	PT-Int	2092		11
PSI, Inc. d/b/a Aquila Energy Marketing Corporation	PT-Int	2094		149
MidCon Marketing Corp. d/b/a Midcon Gas Services Corp.	PT-Int	2099		93
Amgas, Inc.	PT-Int	2104		2
Sears, Roebuck & Company	PT-Int	2105		14
Hagerty Brothers Company	PT-Int	2108		4
Amgas, Inc.	PT-Int	2113		2
Lowpoint-Washburn C.U.S.D #21	PT-Int	2114		2
Amgas, Inc.	PT-Int	2115		2
Springfield Hilton	PT-Int	2116		17
Amgas, Inc.	PT-Int	2117		3
Amgas, Inc.	PT-Int	2118		12
Amgas, Inc.	PT-Int	2119		2 2
Amgas, Inc.	PT-Int PT-Int	2120 2121		4
Edinburg C.U.S.D. #4 Jeckel Pork Farms, Inc.	PT-Int	2123		2
Associated Intrastate Pipeline Company	PT-Int	2125		2,855
Bishop Pipeline Corporation d/b/a MG Natural Gas Corp.	PT-Int	2128		2 , 033
Tazewell County Board	PT-Int	2131		4
Thermic Refractories, Inc.	PT-Int	2133		47
Lincoln College	PT-Int	2145		7
Continental Regency	PT-Int	2146		5
Mobil Oil Corp. d/b/a Mobil Natural Gas Inc.	PT-Int	2148		508
Wyoming Comm. H.S. Dist. No. 71	PT-Int	2150		2
Southern Illinois University	PT-Int	2160		24
GasTrak Corporation	PT-Int	2161		51
Elmwood C.U.S.D. #322	PT-Int	2162		2
Morton, Village of	PT-Int	2172		160
Quantum Chemical Corporation	PT-Int	2175		386
Ivex Corporation	PT-Int	2177		146
Archer Daniels Midland Company	PT-Int	2179		20
W.A. Sadler Resources, Inc.	PT-Int	2189		47
Jumer Hotels, Ltd.	PT-Int	2190		9
Union Electric Company	PT-Int	2193		5
Sunrise Resources d/b/a Energy Dynamics, Inc.	PT-Int	2196		14
Access Energy Corporation	PT-Int	2200		22
Mountain Iron & Supply Company	PT-Int	2210		2
American Cyanamid Company	PT-Int	2211		43
Bishop Pipeline Corporation d/b/a MG Natural Gas Corp.	PT-Int	2213		82
Ohio Gas Marketing Corp. d/b/a Gulf Ohio Corporation	PT-Int	2226		6
Consolidated Fuel Supply, Inc.	PT-Int	2228		59 135
Consolidated Fuel Supply, Inc.	PT-Int	2229		135

FERC Docket: RP94-393-000

Original Sheet No. 25P original Sheet No. 25P : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
GasTrak Corporation	PT-Int	2240	\$ 248
Access Energy Corporation	PT-Int	2242	238
Union Electric Company	PT-Int	2248	51
Mountain Iron & Supply Company	PT-Int	2252	2
Mountain Iron & Supply Company	PT-Int	2253	1
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	2269	45
Merrill's Contractors, Inc.	PT-Int	2272	12
Shelbina, City of	PT-Int	2281	9
Aratex Services, Inc.	PT-Int	2282	22
Town House Condo. Association	PT-Int	2289	4
San Jose Cooperative Company	PT-Int	2291	1
Sankey Construction, Inc.	PT-Int	2292	12
Ohio Gas Marketing Corp. d/b/a Gulf Ohio Corporation	PT-Int	2331	2
Quantum Chemical Corporation	PT-Int	2336	152
Mountain Iron & Supply Company	PT-Int	2377	4
Mountain Iron & Supply Company	PT-Int	2397	1
Amgas, Inc.	PT-Int	2402	4
Amgas, Inc.	PT-Int	2410	14
Amgas, Inc.	PT-Int	2411	12
Mountain Iron & Supply Company	PT-Int	2413	5
Archer Daniels Midland Company	PT-Int	2432	14
Amgas, Inc.	PT-Int	2436	2
Raynor Motel Company d/b/a Holiday Inn East	PT-Int	2438	13
Mountain Iron & Supply Company	PT-Int	2441	5
Amgas, Inc.	PT-Int	2463	1
Amgas, Inc.	PT-Int	2464	1
Mountain Iron & Supply Company	PT-Int	2475	7
GasTrak Corporation	PT-Int	2478	25
Amgas, Inc.	PT-Int	2500	7
Mountain Iron & Supply Company	PT-Int	2505	6
GasTrak Corporation	PT-Int	2507	15
Amgas, Inc.	PT-Int	2510	5
Amgas, Inc.	PT-Int	2511	-
Amgas, Inc.	PT-Int	2512	1
Mobil Natural Gas, Inc.	PT-Int	2514	86
Union Electric Company	PT-Int	2519	5
Phillips Natural Gas Company	PT-Int	2523	1
Dominican Sisters of Springfield, Illinois	PT-Int	2526	5
Amgas, Inc.	PT-Int	2529	3
Amgas, Inc.	PT-Int	2530	2
Amgas, Inc.	PT-Int	2531	2
Amgas, Inc.	PT-Int	2542	-
Amgas, Inc.	PT-Int	2543	5
Amgas, Inc.	PT-Int	2544	5 1
Amgas, Inc.	PT-Int PT-Int	2551 2564	5
Union Electric Company	PT-Int PT-Int	2600	3
Amgas, Inc.			10
Amgas, Inc.	PT-Int PT-Int	2603 2610	2
Union Electric Company	LI-TIIC	7010	۷

FERC Docket: RP94-393-000

Original Sheet No. 25Q original Sheet No. 25Q : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

BuyerShipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Amgas, Inc.	PT-Int	2614	\$ 1
Amgas, Inc.	PT-Int	2615	1
Kansas Pipeline Partnership	PT-Int	874	844
GasTrak Corporation	PT-Int	1789	52
Williams Gas Marketing Company	PT-Int	1884	49
Williams Gas Marketing Company	PT-Int	1885	68
Enron Gas Marketing, Inc.	PT-Int	1904	5
Mountain Iron & Supply Company	PT-Int	1912	16
Associated Natural Gas, Inc.	PT-Int	1922	38
Seagull Marketing Services, Inc.	PT-Int	2054	106
Mountain Iron & Supply Company	PT-Int	2091	67
Reed Minerals Div., Harsco Corp.	PT-Int	2214	15
Reliance Gas Marketing Co. d/b/a Anadarko Trading Company	PT-Int	2491	73
Mobil Natural Gas, Inc.	PT-Int	2515	62
Phillips Natural Gas Company	PT-Int	2522	16
Cimarron River Pipeline System	PT-Int	842	11
Public Service Company of Colorado	PT-Int	889	105
Getty Gas Gathering, Inc.	PT-Int	994	56
Getty Gas Gathering, Inc.	PT-Int	1041	623
Energy Pipeline Company	PT-Int	1124	111
Delhi Gas Pipeline Corporation	PT-Int	1182	45
Gulf Energy Marketing Company	PT-Int	1285	442
Colorado Springs, City of	PT-Int	1361	186
American Central Gas Marketing Company	PT-Int	1443	138
Natural Gas Clearinghouse	PT-Int	1459	187
Hadson Gas Systems, Inc.	PT-Int	1498	451
Consolidated Fuel Supply, Inc.	PT-Int	1519	625
Golden Gas Energies, Inc.	PT-Int	1526	31
Access Energy Pipeline Corporation	PT-Int	1546	1
V.H.C. Gas Systems, L.P. d/b/a Valero Gas Marketing, L.P.	PT-Int	1548	133
Public Service Company of Colorado	PT-Int	1554	71
Continental Natural Gas, Inc.	PT-Int	1555	26
Peoples Natural Gas Company	PT-Int	1573	527
Associated Intrastate Pipeline Company	PT-Int	1579	93
PSI, Inc. d/b/a Aquila Energy Marketing Corporation	PT-Int	1581	70
Panda Resources, Inc.	PT-Int	1582	1
Clinton Gas Transmission, Inc.	PT-Int	1588	- 25
Bishop Pipeline Corporation d/b/a MG Natural Gas Corp.	PT-Int	1598	35
Continental Natural Gas, Inc.	PT-Int PT-Int	1600 1609	69 252
Associated Intrastate Pipeline Company	PT-INC PT-Int	1612	190
Panda Resources, Inc.	PT-Int	1616	241
Ladd Petroleum Corp. d/b/a Amax Oil & Gas, Inc.	PT-INC PT-Int	1618	18
Mountain Iron & Supply Company	PT-Int	1704	99
Associated Intrastate Pipeline Company PSI, Inc. d/b/a Aquila Energy Marketing Corporation	PT-INC PT-Int	1741	1
Coastal States Gas Transmission Company	PT-INC PT-Int	1770	11
Panda Resources, Inc.	PT-INC PT-Int	1777	4
Enron Gas Marketing, Inc.	PT-Int	1786	21
Cabot Pipeline Corp.	PT-INC PT-Int	1827	333
capot riperine corp.	1 1 111 C	102/	223

FERC Docket: RP94-393-000

Original Sheet No. 25R original Sheet No. 25R : Superseded

PANHANDLE EASTERN PIPE LINE COMPANY

Buyer/Shipper	Rate Schedule (1)	Contract No. (2)	Settlement Direct Bill Amount (3)
Panda Resources, Inc.	PT-Int	1828	\$ 6
Mountain Industrial Gas Co. d/b/a Coastal Gas Marketing Co.	PT-Int	1844	17
Williams Gas Marketing Company	PT-Int	1899	167
Enron Gas Marketing, Inc.	PT-Int	1905	1
Mountain Iron & Supply Company	PT-Int	1913	61
Kansas Pipeline Partnership	PT-Int	1920	1
Associated Natural Gas, Inc.	PT-Int	1923	9
American Central Gas Marketing Company	PT-Int	1944	7
Cities Service Helex, Inc. d/b/a Oxy Helex, Inc.	PT-Int	1962	5
Macey & Mershon Oil Inc.	PT-Int	2036	42
Seagull Marketing Services, Inc.	PT-Int	2055	92
Panhandle Trading Co. d/b/a 1 Source Energy Services Company		2061	695
Delta Gas Resources	PT-Int	2097	1
Amarillo Natural Gas, Inc.	PT-Int	2194	88
Consolidated Fuel Corporation	PT-Int	2219	29
Citizens Gas Supply Corporation	PT-Int	2250	363
Delhi Gas Pipeline Corporation	PT-Int	2261	24
Amarillo Natural Gas, Inc.	PT-Int	2319	108
Natural Gas Clearinghouse	PT-Int	2334	675
Entrade Corp. d/b/a Tenneco Gas Marketing Company	PT-Int	2339	62
Mobil Natural Gas, Inc.	PT-Int	2409	45
PMC, A General Partnership	PT-Int	2451	92
Eastex Gas Transmission Company	PT-Int	2476	5
Eastex Gas Transmission Company	PT-Int	2477	5
Getty Gas Gathering, Inc.	PT-Int	2480	25
Eastex Gas Transmission Company	PT-Int	2483	6
Amoco Production Company	PT-Int	2496	1,617
Marathon Oil Company	PT-Int	2598	360
Amoco Production Company	PT-Int	2602	1,793
Union Texas Product Corp. d/b/a Western Gas Resources, Inc.	PT-Int	2605	13
Union Pacific Resources Co. d/b/a Union Pacific Fuels, Inc.	PT-Int	2661	10
Marathon Oil Company	PT-Int	2672	9
Total			\$713,144

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

First Revised Sheet No. 26 First Revised Sheet No. 26: Superseded Superseding: Original Sheet No. 26

CUSTOMIZED RESERVATION PATTERN (CRP)

MAXIMUM RESERVATION RATES

EFFECTIVE NOVEMBER 1, _____ TO OCTOBER 31, _____

	Data	November	December	January	February	March	
Shipper	Rate Schedule	₹/Dt.	₹/Dt.	\$/Dt.	\$/Dt.	\$/Dt.	
		-	-	-	-	-	
	Date	April	May J	June Jul	y August	t September	October
Shipper	Rate Schedule	\$/Dt.	\$/Dt. \$	3/Dt. \$/D	₹. <u>\$/Dt</u>	\$\overline{7Dt.}	\$\overline{7Dt.}

FERC Docket: RP97-185-000

Original Sheet No. 27 Original Sheet No. 27 : Superseded

NEGOTIATED RATES

	Rate	Negotiated		Point(s) of	Point(s) of
Shipper	Schedule	Rate	Quantity	Receipt	Delivery

These negotiated rate agreements do not deviate in any material respect from the form of service agreement in the tariff.

Effective Date: 01/09/1997 Status: Effective FERC Docket: RP97-185-000

Sheet Nos. 28-29 Sheet Nos. 28-29 : Superseded

RESERVED FOR FUTURE USE.

FERC Docket: RS92- 22-005

Original Sheet No. 30 Original Sheet No. 30: Superseded

RATE SCHEDULE FT FIRM TRANSPORTATION SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- (c) which receives such Gas from Panhandle at a physical Point(s) of Delivery equipped with a Flow Control Device or at a Pool Point(s).

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the firm Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement on a uniform hourly basis up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement.

Service at the primary and secondary Point(s) of Receipt and Point(s) of Delivery shall be provided on a firm basis subject to the scheduling, curtailment and interruption provisions of Sections 8 and 9 of the General Terms and Conditions.

FERC Docket: RP96-224-000

First Revised Sheet No. 31 First Revised Sheet No. 31: Superseded

Superseding: Original Sheet No. 31

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for service under this Rate Schedule FT shall be as follows:

3.1 Transmission Charge

If Shipper designates a Point(s) of Receipt or a Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Monthly Transmission Charge which consists of a reservation component and a commodity component.

- (a) The reservation component of the Transmission Charge shall be the product of the MDCQ times the applicable Transmission reservation rates per Dt. as set forth on the effective Tariff Sheet No. 4. The reservation component shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available. Shipper may elect to participate in the Customized Reservation Patternb program pursuant to Section 3.10 herein.
- (b) The commodity component of the Transmission Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 4.

FERC Docket: RS92- 22-005

Original Sheet No. 32 Original Sheet No. 32: Superseded

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

(c) If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 4 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on effective Tariff Sheet No. 4 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates primary Point(s) of Receipt or Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for firm service hereunder, Shipper shall pay a Monthly Gathering Charge which shall consist of a reservation component and a commodity component.

- (a) The reservation component of the Gathering Charge shall be the product of the MDCQ times the applicable reservation rate per Dt. as set forth on the effective Tariff Sheet No. 4. The reservation component shall be pro rated for the first and last contract Months to adjust for the number of Days during the Months service was available.
- (b) The commodity component of the Gathering Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable commodity rate per Dt. as set forth on the effective Tariff Sheet No. 4.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 33 Substitute Original Sheet No. 33 : Superseded

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

If Shipper designates a Point(s) of Receipt on the Gathering portion of Panhandle's system as a secondary firm Point(s) of Receipt pursuant to Section 10.3(c) of the General Terms and Conditions, the otherwise applicable charges shall also include the Gathering commodity rate and overrun charge per Dt. as set forth on effective Tariff Sheet No. 4.

3.3 Special Rate for Limited Class of Backhaul Service

For firm service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 4. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for firm Transportation under Rate Schedule FT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include the Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which the Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 34 Fourth Revised Sheet No. 34 : Superseded

Superseding: Third Revised Sheet No. 34

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

- 3.6 Overrun Charges and Unauthorized Overrun Penalties
 - (a) Overrun Charge for Takes in Excess of MDCQ

If during one or more Days in the billing Month Shipper takes Quantities of Gas in excess of the applicable Maximum Daily Contract Quantity as stated in the Service Agreement (hereinafter called overrun quantities) for firm service hereunder, Shipper shall be subject to an Overrun Charge in addition to the applicable reservation and commodity charges and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The applicable Overrun Charge per Dt. for such overrun quantities shall be as set forth on the effective Tariff Sheet No. 4. In addition, subject to the provisions of Section 3.6(b) of this Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty as set forth in Section 12.16 of the General Terms and Conditions.

(b) Unauthorized Overrun Penalty for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If the OFO requires the elimination of unauthorized overruns, the applicable Unauthorized Overrun Penalty shall be as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

FERC Docket: RP99-482-000

Second Revised Sheet No. 35 second Revised Sheet No. 35 : Superseded

Superseding: First Revised Sheet No. 35
RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 4 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 4 for service hereunder.

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

3.10 Customized Reservation Patternb (CRPb)

(a) The CRPp election provides an alternative method of paying the uniform monthly billing of the Reservation Charge set forth on the effective Tariff Sheet No. 4. Billing and payment of CRPb Reservation Charges shall be in accordance with the reservation pattern elected by Shipper. Rates reflecting such Shipper CRPp election shall be posted on the Messengerb system and will be filed with the Commission on or before the first day of the Month of effectiveness of such election. The total annual Reservation Charges resulting from CRPb will equal the total Reservation Charges Shipper would pay without the CRPb election for the same annual period. An adjustment, if necessary, will be included on the invoice for the last month of the CRPb period to ensure that the total Reservation Charges due Panhandle under Panhandle's uniform currently effective rates equals the Reservation Charges received by Panhandle under the CRPb rates.

FERC Docket: RP96-224-001

Sub Original Sheet No. 35A Sub Original Sheet No. 35A: Superseded

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

All rights and obligations of Section 17 of the General Terms and Conditions shall apply to the Reservation Charge for each billing Month as elected by Shipper. In the event any CRPp invoice shall be based on rates in effect subject to refund and refunds shall be required, refunds shall be calculated as if the CRPp election had not been made.

(b) Shipper may elect flexible maximum reservation charges under CRPp that are derived from the maximum uniform charges for service on the effective Tariff Sheet No. 4; provided, however, CRPp election is not available to a Shipper with an effective rate agreement pursuant to Section 3.7 herein nor a Replacement Shipper pursuant to Section 15 of the General Terms and Conditions. Shipper shall submit to Panhandle an executed Exhibit D to its Service Agreement in order to participate in the CRPp program.

In addition to the otherwise applicable charges set forth for monthly service on the effective Tariff Sheet No. 4 for November through March, a Shipper electing CRPp will be allowed to pay up to 80% of the Reservation Charges due for the period April through October in the preceding November through March period. CRPb elections which specify the portion of the April through October Reservation Charges to be paid during the preceding November through March period will be due by October 1 of each year to be effective November 1. Revised CRPb elections will be allowed up to the fifteenth Business Day preceding the Month of Gas flow, for prospective application only, so long as the Shipper's total annual obligations for Reservation Charges during the 12 month CRPb Period are not affected by such revisions and provided that such revisions do not result in the Reservation Charge for any month during the April through October period exceeding the charge which would have been applicable had the CRPb election not been made.

(c) Maximum CRPp reservation rates for each Shipper shall be reflected on effective Tariff Sheet No. 26.

FERC Docket: RP97-291-004

Third Revised Sheet No. 35B Third Revised Sheet No. 35B: Superseded

Superseding: Second Revised Sheet No. 35B
RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE

(d) The usage rates set forth on effective Tariff Sheet No. 4 shall not be affected by the provisions of this Section 3.10.

3.11 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2 and 3.3 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit C of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit C of the executed Service Agreement. Nothing in this Section 3.11 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges, plus all applicable surcharges.

FERC Docket: RP97-185-002

First Revised Sheet No. 36 First Revised Sheet No. 36: Superseded Superseding: Original Sheet No. 36

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 37 Original Sheet No. 37: Superseded

RATE SCHEDULE EFT ENHANCED FIRM TRANSPORTATION SERVICE

1. AVATLABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- (c) which receives such Gas from Panhandle at a physical Point(s) of Delivery or at a Pool Point(s).

This service is available for firm service at a Point(s) of Delivery without regard to the presence of a Flow Control Device.

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the firm Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement on a basis consistent with Section 5 hereof.

Service at the primary and secondary Point(s) of Receipt and Point(s) of Delivery shall be provided on a firm basis subject to the scheduling, curtailment and interruption provisions of Sections 8 and 9 of the General Terms and Conditions.

FERC Docket: RP96-224-000

First Revised Sheet No. 38 First Revised Sheet No. 38: Superseded

Superseding: Original Sheet No. 38

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for service under this Rate Schedule ${\tt EFT}$ shall be as follows:

3.1 Transmission Charge

If Shipper designates a Point(s) of Receipt or a Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Monthly Transmission Charge which consists of a reservation component and a commodity component.

- (a) The reservation component of the Transmission Charge shall be the product of the MDCQ times the applicable Transmission reservation rates per Dt. as set forth on the effective Tariff Sheet No. 5. The reservation component shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available. Shipper may elect to participate in the Customized Reservation Patternb program pursuant to Section 3.10 herein.
- (b) The commodity component of the Transmission Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 5.
- (c) If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 5 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 39 Substitute Original Sheet No. 39: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 5 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates primary Point(s) of Receipt or Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for firm service hereunder, Shipper shall pay a Monthly Gathering Charge which shall consist of a reservation component and a commodity component.

- (a) The reservation component of the Gathering Charge shall be the product of the MDCQ times the applicable reservation rate per Dt. as set forth on the effective Tariff Sheet No. 5. The reservation component shall be pro rated for the first and last contract Months to adjust for the number of Days during the Months service was available.
- (b) The commodity component of the Gathering Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable commodity rate per Dt. as set forth on the effective Tariff Sheet No. 5.

If Shipper designates a Point(s) of Receipt on the Gathering portion of Panhandle's system as a secondary firm Point(s) of Receipt pursuant to Section 10.3(c) of the General Terms and Conditions, the otherwise applicable charges shall also include the Gathering commodity rate and overrun charge per Dt. as set forth on the effective Tariff Sheet No. 5.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 40 Substitute Original Sheet No. 40: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

3.3 Special Rate for Limited Class of Backhaul Service

For firm service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 5. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for firm Transportation under Rate Schedule EFT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include the Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which the Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Second Revised Sheet No. 41 Second Revised Sheet No. 41: Superseded

Superseding: First Revised Sheet No. 41

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

3.6 Overrun Charges and Unauthorized Overrun Penalties

(a) Overrun Charge for Takes in Excess of MDCQ

If during one or more Days in the billing Month Shipper takes Quantities of Gas in excess of the applicable Maximum Daily Contract Quantity as stated in the Service Agreement (hereinafter called overrun quantities) for firm service hereunder, Shipper shall be subject to an Overrun Charge in addition to the applicable reservation and commodity charges and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The applicable Overrun Charge per Dt. for such overrun quantities shall be as set forth on the effective Tariff Sheet No. 5. In addition, subject to the provisions of Section 3.6(c) of this Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty as set forth in Section 12.16 of the General Terms and Conditions.

(b) Takes in Excess of Permissible Hourly Deliveries

If, pursuant to Section 5 hereof, Shipper has been notified that it must limit takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery, Shipper must use its best efforts to limit its takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery as soon as possible, but in no event later than two hours following notification. If, after such two hours, the Shipper's takes exceed such hourly limitation for any hour during the next twenty-four hour period, Shipper shall pay the overrun charges as set forth on the effective Tariff Sheet No. 5, in addition to the applicable commodity rates.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 42 Fourth Revised Sheet No. 42: Superseded

Superseding: Third Revised Sheet No. 42

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

(c) Unauthorized Overrun Penalty for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If the OFO requires the elimination of unauthorized overruns, the applicable Unauthorized Overrun Penalty shall be as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 5 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 5 for service hereunder.

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

FERC Docket: RP99-482-000

Second Revised Sheet No. 42A Second Revised Sheet No. 42A: Superseded

Superseding: First Revised Sheet No. 42A

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

3.10 Customized Reservation Patternp (CRPp)

- (a) The CRPp election provides an alternative method of paying the uniform monthly billing of the Reservation Charge set forth on the effective Tariff Sheet No. 5. Billing and payment of CRPb Reservation Charges shall be in accordance with the reservation pattern elected by Shipper. Rates reflecting such Shipper CRPb election shall be posted on the Messengerb system and will be filed with the Commission on or before the first day of the Month of effectiveness of such election. The total annual Reservation Charges resulting from CRPb will equal the total Reservation Charges Shipper would pay without the CRPb election for the same annual period. An adjustment, if necessary, will be included on the invoice for the last month of the CRPb period to ensure that the total Reservation Charges due Panhandle under Panhandle's uniform currently effective rates equals the Reservation Charges received by Panhandle under the CRPb rates. All rights and obligations of Section 17 of the General Terms and Conditions shall apply to the Reservation Charge for each billing Month as elected by Shipper. In the event any CRPp invoice shall be based on rates in effect subject to refund and refunds shall be required, refunds shall be calculated as if the CRPb election had not been made.
- (b) Shipper may elect flexible maximum reservation charges under CRPp that are derived from the maximum uniform charges for service on the effective Tariff Sheet No. 5; provided, however, CRPp election is not available to a Shipper with an effective rate agreement pursuant to Section 3.7 herein nor a Replacement Shipper pursuant to Section 15 of the General Terms and Conditions. Shipper shall submit to Panhandle an executed Exhibit B to its Service Agreement in order to participate in the CRPp program.

FERC Docket: RP97-291-004

Third Revised Sheet No. 42B Third Revised Sheet No. 42B: Superseded

Superseding: Second Revised Sheet No. 42B
RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

In addition to the otherwise applicable charges set forth for monthly service on the effective Tariff Sheet No. 5 for November through March, a Shipper electing CRPb will be allowed to pay up to 80% of the Reservation Charges due for the period April through October in the preceding November through March period. CRPb elections which specify the portion of the April through October Reservation Charges to be paid during the preceding November through March period will be due by October 1 of each year to be effective November 1. Revised CRPp elections will be allowed up to the fifteenth Business Day preceding the Month of Gas flow, for prospective application only, so long as the Shipper's total annual obligations for Reservation Charges during the 12 month CRPb Period are not affected by such revisions and provided that such revisions do not result in the Reservation Charge for any month during the April through October period exceeding the charge which would have been applicable had the CRPp election not been made.

- (c) Maximum CRPb reservation rates for each Shipper shall be reflected on effective Tariff Sheet No. 26.
- (d) The usage rates set forth on effective Tariff Sheet No. 5 shall not be affected by the provisions of this Section 3.10.

3.11Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2 and 3.3 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit C of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

FERC Docket: RP99-482-000

Fourth Revised Sheet No. 43 Fourth Revised Sheet No. 43: Superseded

Superseding: Third Revised Sheet No. 43
RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit C of the executed Service Agreement. Nothing in this Section 3.11 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges, plus all applicable surcharges.

5. HOURLY DELIVERIES

Shipper shall deliver or cause to be delivered Natural Gas for the account of Shipper on a uniform hourly basis as nearly as practicable. At each Point(s) of Delivery Shipper may receive from Panhandle in any hour such Quantities of Gas as it requires or as are consistent with available pipeline pressures at such point(s); provided that Panhandle shall be entitled to require Shipper to restrict takes at any point during any hour to one-sixteenth of the Natural Gas nominated for Transportation to that Point(s) of Delivery on that Day by notifying Shipper through the Messengerb system, the Web Site and by telephone or facsimile communication.

6. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

7. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 44 Original Sheet No. 44: Superseded

RATE SCHEDULE SCT SMALL CUSTOMER TRANSPORTATION SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which is a municipality or a small distribution company engaged in the resale of Natural Gas, or a Farm Tap or Irrigation Tap customer;
- (b) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule for a Maximum Daily Contract Quantity (MDCQ) not to exceed 10,000 Dt. per Day;
- (c) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper;
- (d) which receives such Gas from Panhandle at a physical Point(s) of Delivery; and
- (e) which transports its total system requirements under this Rate Schedule, provided, however, that on any Day Shipper has taken its Maximum Daily Contract Quantity, Shipper may receive additional Quantities under Rate Schedule IT, EIT or pursuant to a Release Agreement under Section 15 of the General Terms and Conditions.

This service is available for firm service at a Point(s) of Delivery without regard to the presence of a Flow Control Device.

A Shipper with an executed Service Agreement under this Rate Schedule shall have its service and Service Agreement automatically converted to service and form of agreement under Rate Schedule EFT effective as of the date any Gas service is received through such Shipper's facilities other than under this Rate Schedule, except as provided in (e) above.

FERC Docket: RS92- 22-005

Original Sheet No. 45 Original Sheet No. 45: Superseded

RATE SCHEDULE SCT (Continued) SMALL CUSTOMER TRANSPORTATION SERVICE

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the firm Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at the Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at the Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement (not to exceed 10,000 Dt. per Day) on a basis consistent with Section 5 hereof.

Service provided at the primary and secondary Point(s) of Receipt and Point(s) of Delivery shall be provided on a firm basis subject to the scheduling, curtailment and interruption provisions of Sections 8 and 9 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3.RATES

The rates and charges for service under this Rate Schedule SCT shall be as follows:

FERC Docket: RP99-192-000

First Revised Sheet No. 46 First Revised Sheet No. 46 : Superseded

Superseding: Substitute Original Sheet No. 46
RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER TRANSPORTATION SERVICE

3.1 Transmission Charge

If Shipper designates a Point(s) of Receipt or a Point(s) of Delivery on the transmission portion of Panhandle's system, the applicable Transmission Charge shall be the product of the actual Quantity of Gas delivered during the Month up to Shipper's MDCQ times the applicable Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 6. If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 6 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 6 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates a primary Point(s) of Receipt or a Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for firm service hereunder, Shipper shall pay a Gathering charge which shall be the product of the actual Quantity of Gas delivered during the Month times the applicable commodity rate per Dt. as set forth on the effective Tariff Sheet No. 6.

If Shipper designates a Point(s) of Receipt on the Gathering portion of Panhandle's system as a secondary Point(s) of Receipt pursuant to Section 10.3(c) of the General Terms and Conditions, the otherwise applicable charges shall also include the Gathering commodity rate per Dt. as set forth on the effective Tariff Sheet No. 6.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Second Revised Sheet No. 47 Second Revised Sheet No. 47: Superseded

Superseding: First Revised Sheet No. 47

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER TRANSPORTATION SERVICE

3.3 Special Rate for Limited Class of Backhaul Service

For firm service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 6. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for firm Transportation under Rate Schedule SCT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that if the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone Charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

- 3.6 Overrun Charges and Unauthorized Overrun Penalties
 - (a) Overrun Charge for Takes in Excess of MDCQ

If during one or more Days in the billing Month Shipper takes Quantities of Gas in excess of the applicable Maximum Daily Contract Quantity as stated in the Service Agreement (hereinafter called overrun Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 48 Fourth Revised Sheet No. 48: Superseded

Superseding: Third Revised Sheet No. 48

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER TRANSPORTATION SERVICE

quantities) for firm service hereunder, Shipper shall be subject to an Overrun Charge in addition to the applicable commodity charge and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The applicable charge per Dt. for such overrun quantities shall be as set forth on the effective Tariff Sheet No. 6. In addition, subject to the provisions of Section 3.6(c) of this Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty as set forth in Section 12.16 of the General Terms and Conditions.

(b) Takes in Excess of Permissible Hourly Deliveries

If, pursuant to Section 5 hereof, Shipper has been notified that it must limit takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery, Shipper must use its best efforts to limit its takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery as soon as possible, but in no event later than two hours following notification. If, after such two hours, the Shipper's takes exceed such hourly limitation for any hour during the next twenty-four hour period, Shipper shall pay the overrun charges as set forth on the effective Tariff Sheet No. 6, in addition to the applicable commodity rates.

(c) Unauthorized Overrun Penalty for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If the OFO requires the elimination of unauthorized overruns, the applicable Unauthorized Overrun Penalty shall be as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

FERC Docket: RP96-348-000

First Revised Sheet No. 49 First Revised Sheet No. 49: Superseded

Superseding: Original Sheet No. 49

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER TRANSPORTATION SERVICE

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 6 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 6 for service hereunder.

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges, plus all applicable surcharges.

FERC Docket: RP99-482-000

Second Revised Sheet No. 50 Second Revised Sheet No. 50: Superseded

Superseding: First Revised Sheet No. 50
RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER TRANSPORTATION SERVICE

5. HOURLY DELIVERIES

Shipper shall deliver or cause to be delivered Natural Gas for the account of Shipper on a uniform hourly basis as nearly as practicable. At each Point(s) of Delivery Shipper may receive from Panhandle in any hour such Quantities of Gas as it desires or as are consistent with available pipeline pressures at such point(s); provided that Panhandle shall be entitled to require Shipper to restrict takes at any point during any hour to one-sixteenth of the Natural Gas nominated for Transportation to that Point(s) of Delivery on that Day by notifying Shipper through the Messengerb system, the Web Site and by telephone or facsimile communication.

6. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

7. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 51 Original Sheet No. 51: Superseded

RATE SCHEDULE IT INTERRUPTIBLE TRANSPORTATION SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- (c) which receives such Gas from Panhandle at a physical Point(s) of Delivery equipped with a Flow Control Device or at a Pool Point(s).

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the interruptible Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at a Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement on a uniform hourly basis up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement, subject to the availability of capacity.

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

FERC Docket: RS92- 22-005

Original Sheet No. 52 Original Sheet No. 52: Superseded

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE

3.RATES

The rates and charges for interruptible service under this Rate Schedule shall be as follows:

3.1 Transmission Charge

If Shipper designates interruptible Point(s) of Receipt or interruptible Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Transmission Charge which shall be the product of the actual Quantity of Gas delivered during the Month times the Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 7. If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rate shall be as set forth on the effective Tariff Sheet No. 7 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 7 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates Point(s) of Receipt or Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for interruptible service hereunder, Shipper shall pay a Gathering Charge which shall be the product of the actual Quantity of Gas delivered during the Month times the applicable rate per Dt. as set forth on the effective Tariff Sheet No. 7.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 53 Substitute Original Sheet No. 53: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE

3.3 Special Rate for Limited Class of Backhaul Service

For interruptible service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 7. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for interruptible Transportation under Rate Schedule IT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

FERC Docket: RS92- 22-005

Original Sheet No. 54 Original Sheet No. 54: Superseded

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE

3.6 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 7 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.7 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 7 for service hereunder.

3.8 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

4.MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable balancing charges plus all applicable surcharges.

5.RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 55} \; \texttt{Original Sheet No. 55} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE

6.GENERAL TERMS AND CONDITIONS

FERC Docket: RS92- 22-005

Original Sheet No. 56 Original Sheet No. 56: Superseded

RATE SCHEDULE EIT
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- (c) which receives such Gas from Panhandle at a Point(s) of Delivery on a basis consistent with Section 5, below.

This service is available for interruptible service at a Point(s) of Delivery without regard to the presence of a Flow Control Device.

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the interruptible Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at the Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at a Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement, subject to the availability of capacity.

FERC Docket: RS92- 22-005

Original Sheet No. 57 Original Sheet No. 57: Superseded

RATE SCHEDULE EIT (Continued) ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3.RATES

The rates and charges for interruptible service under this Rate Schedule shall be as follows:

3.1 Transmission Charge

If Shipper designates interruptible Point(s) of Receipt or interruptible Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Transmission Charge which shall be the product of the actual Quantity of Gas delivered during the Month times the Transmission commodity rate per Dt. as set forth on the effective Tariff Sheet No. 8. If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rate shall be as set forth on the effective Tariff Sheet No. 8 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 8 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates Point(s) of Receipt or Point(s) of Delivery in the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for interruptible service hereunder, Shipper shall pay a Gathering Charge which shall be the product of the actual Quantity of Gas delivered during the Month times the applicable rate per Dt. as set forth on the effective Tariff Sheet No. 8.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 58 Substitute Original Sheet No. 58: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

3.3 Special Rate for Limited Class of Backhaul Service

For interruptible service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission Charge per Dt. as set forth in the effective Tariff Sheet No. 8. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for interruptible Transportation under Rate Schedule EIT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Third Revised Sheet No. 59 Third Revised Sheet No. 59: Superseded

Superseding: Second Revised Sheet No. 59

RATE SCHEDULE EIT (Continued)

ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

3.6 Overrun Charge for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If, during an OFO Shipper has been notified, pursuant to Section 5 hereof, that it must limit takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery, Shipper must use its best efforts to limit its takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery as soon as possible, but in no event later than two hours following notification. If, after such two hours, the Shipper's takes exceed such hourly limitation for any hour during the next twenty-four hour period, Shipper shall pay an Overrun Penalty as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 8 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 8 for service hereunder.

FERC Docket: RP99-482-000

Second Revised Sheet No. 60 Second Revised Sheet No. 60: Superseded

Superseding: First Revised Sheet No. 60

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges plus all applicable surcharges.

5. HOURLY DELIVERIES

Shipper shall deliver Natural Gas or cause Natural Gas to be delivered for the account of Shipper on a uniform hourly basis as nearly as practicable. At each Point(s) of Delivery Shipper may receive from Panhandle in any hour such Quantities of Gas as it desires or as are consistent with available pipeline pressures at such point(s); provided that Panhandle shall be entitled to require Shipper to restrict takes at any point during any hour to one-sixteenth of the Natural Gas nominated for Transportation to that Point(s) of Delivery on that Day by notifying Shipper through the Messengerb system, the Web Site and by telephone or facsimile communication.

6. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

7. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 61 original Sheet No. 61: Superseded

RATE SCHEDULE IOS IN/OUT STORAGE SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule.

2.APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the IOS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection by Panhandle of Shipper's Gas into Storage, Storage in Panhandle's Field Area Storage Facilities and the subsequent withdrawal from Storage of Shipper's Gas, all on a firm basis. Injections may be made at any time, subject to daily nomination by Shipper and available Maximum Stored Quantity. The Maximum Daily Injection Quantity (MDIQ) shall be 1/200th of the Maximum Stored Quantity. Withdrawals may be made at any time, subject to daily nomination by Shipper and available Stored Volume. The Maximum Daily Withdrawal Quantity (MDWQ) shall be 1/100th of the Maximum Stored Quantity.

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into IOS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of IOS Storage in accordance with Section 14.2 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

FERC Docket: RS92- 22-005

Original Sheet No. 62 Original Sheet No. 62: Superseded

RATE SCHEDULE IOS (Continued) IN/OUT STORAGE SERVICE

3.RATES

3.1 Capacity Charge

Shipper shall pay a Monthly Capacity Charge which shall be the product of 1/12th the Maximum Stored Quantity times the capacity charge per Dt. as set forth on the effective Tariff Sheet No. 9.

3.2 Deliverability Charge

Shipper shall pay a Monthly Deliverability Charge which shall be the product of the Maximum Daily Withdrawal Quantity times the deliverability charge per Dt. as set forth on the effective Tariff Sheet No. 9.

3.3 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually delivered at the IOS Point and injected into Storage during the Month times the injection charge per Dt. as set forth on the effective Tariff Sheet No. 9.

3.4 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the IOS Point during the Month times the withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 9.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 63 Fifth Revised Sheet No. 63: Superseded

Superseding: Fourth Revised Sheet No. 63

RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE

3.6 Overrun Charge and Unauthorized Overrun Penalty

Shipper shall pay the overrun charge and, if applicable, the Unauthorized Overrun Penalty per Dt. of Gas actually delivered at the IOS Point and injected into Storage or withdrawn from Storage and delivered at the IOS Point during any Day in excess of the MDIQ and MDWQ set out in the Service Agreement, as applicable. The Overrun Charge per Dt. shall be as set forth on the effective Tariff Sheet No. 9 and the Unauthorized Overrun Penalty shall be as set forth in Sections 12.16 and 12.17 of the General Terms and Conditions.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 9 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 9 for service hereunder.

3.9 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2, 3.3 and 3.4 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit A of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems

FERC Docket: RP97-291-004

Third Revised Sheet No. 64 Third Revised Sheet No. 64: Superseded

Superseding: Second Revised Sheet No. 64
RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE

such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit A of the executed Service Agreement. Nothing in this Section 3.9 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the capacity charge, the deliverability charge, the injection charge, the withdrawal charge, any overrun charges plus all applicable surcharges.

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 65 Original Sheet No. 65: Superseded

RATE SCHEDULE IIOS INTERRUPTIBLE IN/OUT STORAGE SERVICE

1.AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the IOS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection by Panhandle of Shipper's Gas into Storage in Panhandle's Field Area Storage Facilities, Storage and the subsequent withdrawal from Storage of Shipper's Gas, all on an interruptible basis. Injections may be made at any time, subject to daily nomination by Shipper and available IIOS Maximum Stored Quantity. The Maximum Daily Injection Quantity shall be 1/200th of the Maximum Stored Quantity. Withdrawals may be made at any time, subject to daily nomination by Shipper, available Stored Volume and available capacity. The Maximum Daily Withdrawal Quantity shall be 1/100th of the Maximum Stored Quantity. Shipper may be required to withdraw all, or any part, of its Stored Volume under this IIOS Rate Schedule within 30 Days following notice from Panhandle. Any Stored Volume not withdrawn within 30 Days of Panhandle giving Shipper notice shall become the property of Panhandle at no cost to Panhandle, free and clear of any adverse claims.

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into IIOS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of IIOS Storage in accordance with Section 14.2 of the General Terms and Conditions.

FERC Docket: RS92- 22-005

Original Sheet No. 66 Original Sheet No. 66: Superseded

RATE SCHEDULE IIOS (Continued)
INTERRUPTIBLE IN/OUT STORAGE SERVICE

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3.RATES

The rates and charges for Storage service under this Rate Schedule shall be as follows:

3.1 Monthly Inventory Charge

Shipper shall pay a Monthly Inventory Charge which shall be the product of the inventory charge per Dt. as set forth on the effective Tariff Sheet No. 10 times the average daily IIOS Stored Volume for the Month.

3.2 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually delivered at the IOS Point and injected into Storage during the Month times the injection charge per Dt. as set forth on the effective Tariff Sheet No. 10.

3.3 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the IOS Point during the Month times the withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 10.

3.4 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

FERC Docket: RS92- 22-005

Original Sheet No. 67 Original Sheet No. 67: Superseded

RATE SCHEDULE IIOS (Continued)
INTERRUPTIBLE IN/OUT STORAGE SERVICE

3.5 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 10 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.6 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 10 for service hereunder.

4.MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Monthly inventory charge, the injection charge, the withdrawal charge, plus all applicable surcharges.

5.RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6.GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 68 Substitute Original Sheet No. 68: Superseded

RATE SCHEDULE WS WINTER STORAGE SERVICE

1. AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule for Storage service from either Panhandle's Market Area Storage Facilities or Field Area Storage Facilities.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the applicable WS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement. In addition, where Shipper is utilizing a third party to make Gas available for Storage in Panhandle's Market Area Storage Facilities under this Rate Schedule, the Shipper may designate as the WS Point an off-system Market Area Storage Facility that Panhandle has leased from a third party, subject to operational feasibility, the terms and conditions applicable to Panhandle's agreement with the third party providing the off-system storage, the General Terms and Conditions herein, and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection of Shipper's Gas into storage at the applicable WS Point during the Summer Period, Storage in Panhandle's Market Area Storage Facilities or Field Area Storage Facilities, as elected by the Shipper in the Service Agreement, and the subsequent withdrawal from Storage of Shipper's Gas at the applicable WS Point during the Winter Period, all on a firm basis. Winter Period injections and Summer Period withdrawals will be permitted on an interruptible basis, subject to Panhandle's management of Storage to preserve deliverability. Maximum Daily Withdrawal Quantities (MDWQ) and Maximum Daily Injection Quantities (MDIQ) shall be in accordance with the WS Storage Agreement but shall not exceed 1/100th of Maximum Stored Quantity and 1/200th of Maximum Stored Quantity, respectively. Available withdrawal quantities will vary according to the percentage of Shipper's Stored Volume to the Maximum Stored Quantity as follows:

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 69 Substitute Original Sheet No. 69: Superseded

RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE

Percentage of Stored Volume

to the

Maximum Stored Quantity Available Withdrawal Quantities

100% - 30% 100% of MDWQ <30% - 10% 75% of MDWQ <10% - 0% 60% of MDWQ

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into WS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of WS Storage in accordance with Section 14.2 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for Storage service under this Rate Schedule from Panhandle's Market Area Storage Facilities and Field Area Storage Facilities shall be as follows:

3.1 Capacity Charge

Shipper shall pay a Monthly Capacity Charge which shall be the product of 1/12th the Maximum Stored Quantity times the applicable capacity charge per Dt. as set forth on the effective Tariff Sheet No. 11.

3.2 Deliverability Charge

Shipper shall pay a Monthly Deliverability Charge which shall be the product of the Maximum Daily Withdrawal Quantity times the applicable deliverability charge per Dt. as set forth on the effective Tariff Sheet No. 11.

3.3 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually received from

FERC Docket: RP00-395-003

Third Revised Sheet No. 70 Third Revised Sheet No. 70: Superseded

Superseding: Second Revised Sheet No. 70

RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE

Shipper at the WS Point and injected into Storage during the Month times the applicable injection charge per Dt. as set forth on the effective Tariff Sheet No. 11.

3.4 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the WS Point during the Month times the applicable withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 11.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

3.6 Overrun Charge and Unauthorized Overrun Penalty

Shipper shall pay the applicable Overrun Charge and, if applicable, the Unauthorized Overrun Penalty per Dt. of Gas actually delivered at the applicable WS Point and injected into Storage or withdrawn from Storage and received for transportation at the applicable WS Point during any Day in excess of the MDIQ and MDWQ set out in the Service Agreement, as applicable. The Overrun Charge per Dt. shall be as set forth on the effective Tariff Sheet No. 11 and the Unauthorized Overrun Penalty shall be as set forth in Sections 12.16 and 12.17 of the General Terms and Conditions. For purposes of computing Overrun Charges and Unauthorized Overrun Penalties applicable to withdrawals in excess of MDWQ when Stored Volume falls below 30% of MSQ, the MDWQ as stated in the WS Storage Agreement shall be deemed to be reduced in accordance with the ratchets set forth in Section 2 of this Rate Schedule. The Overrun Charge and Unauthorized Overrun Penalty will not apply to Winter Period injections and Summer Period withdrawals if the Quantities injected or withdrawn are within the MDIQ or MDWQ, as applicable.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 11 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

FERC Docket: RP97-291-004

Third Revised Sheet No. 71 Third Revised Sheet No. 71: Superseded

Superseding: Second Revised Sheet No. 71
RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 11 for service hereunder. If the Stored Volume of the Shipper is not reduced to 20% of Maximum Stored Quantity or less prior to April 1 of any Contract Year, then Panhandle shall retain .25% of the excess Stored Volume over 20% of the Maximum Stored Quantity to compensate for additional compressor fuel required.

3.9 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2, 3.3 and 3.4 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit A of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit A of the executed Service Agreement. Nothing in this Section 3.9 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

FERC Docket: RP97-185-002

Sub Original Sheet No. 71A Sub Original Sheet No. 71A: Superseded

RATE SCHEDULE WS (Continued) WINTER STORAGE SERVICE

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the capacity charge, the deliverability charge, the injection charge, the withdrawal charge, any applicable overrun charges plus all applicable surcharges.

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part bereof

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 72 Substitute Original Sheet No. 72: Superseded

RATE SCHEDULE PS PEAKING STORAGE SERVICE

1. AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Storage Agreement in the form prescribed under this Tariff for service under this Rate Schedule.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the PS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement. In addition, where Shipper is utilizing a third party to make Gas available for Storage under this Rate Schedule, the Shipper may designate as the PS Point an off-system Market Area Storage Facility that Panhandle has leased from a third party, subject to operational feasibility, the terms and conditions applicable to Panhandle's agreement with the third party providing the off-system storage, the General Terms and Conditions herein, and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection of Shipper's Gas into Storage at the PS Point during the Summer Period, Storage in Panhandle's Market Area Storage Facilities and the subsequent withdrawal from Storage at the PS Point, during the Winter Period, all on a firm basis. Injections must be scheduled in accordance with the schedule set out in the Service Agreement, which schedule shall be subject to annual revision by Panhandle to reflect the cycling requirements of Storage available to Panhandle. Injections during the Month of April may be suspended by Shipper if the PS Stored Volume does not fall below 17% of the Maximum Stored Quantity. On any Day during the Summer Period, if the PS Stored Volume is at or above the monthly scheduled amount, injections may be suspended by Shipper. The monthly injection schedule will be supplied to each Shipper prior to the Summer Period injection cycle. Winter Period injections will be permitted on an interruptible basis, subject to Panhandle's management of Storage to preserve deliverability. Maximum Daily Withdrawal Quantities shall be in accordance with the Service Agreement but, shall not exceed 1/71st of the Maximum Stored Quantity. Available withdrawal quantities will vary according to the percentage of Stored Volume to the Maximum Stored Quantity as follows:

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 73 Substitute Original Sheet No. 73: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE

Percentage of Stored Volume

to the

Maximum Stored Quantity Available Withdrawal Quantities

100% - 30% 100% of MDWQ <30% - 10% 44% of MDWQ <10% - 0% 11% of MDWQ

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into PS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of PS Storage in accordance with Section 14.2 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. SCHEDULING OF RECEIPTS AND DELIVERIES

Storage service hereunder shall comply with the scheduling provisions of Section 8 of the General Terms and Conditions. In accordance with the schedule set out in the Service Agreement, Shipper must provide and nominate the Monthly Stored Volume during each Month of the Summer Period. If Shipper does not provide and nominate sufficient Quantities for storage injections during any Month of the Summer Period, Panhandle shall have the right to inject Gas into Storage on Shipper's behalf equal to the shortfall and Shipper shall pay Panhandle a rate equal to the rate for service under Rate Schedule EIT for receipts at the IOS point and deliveries to the PS Point near the Michigan and Ohio state line and Shipper shall purchase the injected Gas from Panhandle in accordance with Section 12.11(a)(2) of the General Terms and Conditions at the >5%-10% percentage imbalance level.

FERC Docket: RS92- 22-005

Original Sheet No. 74 Original Sheet No. 74: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE

4.RATES

The rates and charges for service under this Rate Schedule shall be as follows:

4.1 Capacity Charge

Shipper shall pay a Monthly Capacity Charge which shall be the product of 1/12th the Maximum Stored Quantity times the capacity charge per Dt. as set forth on the effective Tariff Sheet No. 12.

4.2 Deliverability Charge

Shipper shall pay a Monthly Deliverability Charge which shall be the product of the Maximum Daily Withdrawal Quantity times the deliverability charge per Dt. as set forth on the effective Tariff Sheet No. 12.

4.3 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually received from Shipper at the PS Point and injected into Storage during the Month times the injection charge per Dt. as set forth on the effective Tariff Sheet No. 12.

4.4 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the PS Point during the Month times the withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 12.

4.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

FERC Docket: RP00-395-003

Third Revised Sheet No. 75 Third Revised Sheet No. 75: Superseded

Superseding: Second Revised Sheet No. 75

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE

4.6 Overrun Charge and Unauthorized Overrun Penalty

Shipper shall pay the Overrun Charge and, if applicable, the Unauthorized Overrun Penalty per Dt. of Gas actually delivered at the PS Point and injected into Storage or withdrawn from Storage and received for transportation at the PS Point during any Day in excess of the MDIQ and MDWQ set out in the Storage Agreement and in this Rate Schedule, as applicable. The Overrun Charge per Dt. shall be as set forth on the effective Tariff Sheet No. 12 and the Unauthorized Overrun Penalty shall be as set forth in Sections 12.16 and 12.17 of the General Terms and Conditions. For purposes of computing Overrun Charges and Unauthorized Overrun Penalties applicable to withdrawals in excess of MDWQ when Stored Volume falls below 30% of MSQ, the MDWQ as stated in the PS Storage Agreement shall be deemed to be reduced in accordance with the ratchets set forth in Section 2 of this Rate Schedule. The Overrun Charge and Unauthorized Overrun Penalty shall not apply to Winter Period injections if the Quantities injected are within the MDIQ.

4.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 12 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

4.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 12 for service hereunder. If the Stored Volume of any Shipper is not reduced to 20% of Maximum Stored Quantity or less prior to April 1 of any Contract Year, then Panhandle shall retain .25% of the excess Stored Volume over 20% of the Maximum Stored Quantity to compensate for additional compressor fuel required.

FERC Docket: RP97-291-004

Third Revised Sheet No. 76 : Superseded

Superseding: Second Revised Sheet No. 76
RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE

4.9 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 4.1, 4.2, 4.3 and 4.4 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit A of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit A of the executed Service Agreement. Nothing in this Section 4.9 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

5. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the capacity charge, the deliverability charge, the injection charge, the withdrawal charge, any applicable overrun charges plus all applicable surcharges.

FERC Docket: RP97-185-002

Original Sheet No. 76A original Sheet No. 76A: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE

6. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

7. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 77 Substitute Original Sheet No. 77: Superseded

RATE SCHEDULE IWS
INTERRUPTIBLE WINTER STORAGE SERVICE

1. AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule for Storage service from either Panhandle's Market Area Storage Facilities or Field Area Storage Facilities.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the applicable WS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement. In addition, where Shipper is utilizing a third party to make Gas available for Storage in Panhandle's Market Area Storage Facilities under this Rate Schedule, the Shipper may designate as the WS Point an off-system Market Area Storage Facility that Panhandle has leased from a third party, subject to operational feasibility, the terms and conditions applicable to Panhandle's agreement with the third party providing the off-system storage, the General Terms and Conditions herein, and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection of Shipper's Gas into Storage at the applicable WS Point during the Summer Period, Storage in Panhandle's Market Area Storage facilities or Field Area Storage Facilities, as elected by the Shipper in the Service Agreement, and the subsequent withdrawal from Storage at the applicable WS Point during the Winter Period, all on an interruptible basis. Maximum Daily Withdrawal Quantities (MDWQ) and Maximum Daily Injection Quantities (MDIQ) shall be in accordance with the IWS Storage Agreement but shall not exceed 1/100th of Maximum Stored Quantity and 1/200th of Maximum Stored Quantity, respectively. Available withdrawal quantities shall vary according to the percentage of Shipper's Stored Volume to the Maximum Stored Quantity as follows:

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 78 Substitute Original Sheet No. 78: Superseded

RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE

Percentage of Stored Volume

to the

Maximum Stored Quantity Available Withdrawal Quantities

100%	-	30%	100%	of	MDWQ
<30%	-	10%	75%	of	MDWQ
<10%	-	0%	60%	of	MDWQ

Daily withdrawal and injection Quantities of Gas shall be subject to available capacity and IWS Stored Volume attributable to a Shipper. All services performed under this Rate Schedule shall be performed on a capacity available basis. Shipper may be required to withdraw all, or any part, of its Stored Volume under this IWS Rate Schedule within 30 Days following notification from Panhandle. Any Stored Volume not withdrawn within 30 Days of Panhandle giving Shipper notice shall become the property of Panhandle at no cost to Panhandle, free and clear of any adverse claims.

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into IWS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of IWS Storage in accordance with Section 14.2 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for Storage service under this Rate Schedule from Panhandle's Market Area Storage Facilities and Field Area Storage Facilities shall be as follows:

3.1 Monthly Inventory Charge

Shipper shall pay a Monthly Inventory Charge which shall be the product of the applicable inventory charge per Dt. as set forth on the effective Tariff Sheet No. 13 and the average daily IWS Stored Volume for the Month.

FERC Docket: RS92- 22-005

Original Sheet No. 79 Original Sheet No. 79: Superseded

RATE SCHEDULE IWS (Continued) INTERRUPTIBLE WINTER STORAGE SERVICE

3.2 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually received from Shipper at the WS Point injected into Storage during the Month times the applicable injection charge per Dt. as set forth on the effective Tariff Sheet No. 13.

3.3 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the WS Point during the Month times the applicable withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 13.

3.4 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

3.5 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 13 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.6 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 13 for service hereunder.

FERC Docket: RS92- 22-005

Original Sheet No. 80 Original Sheet No. 80: Superseded

RATE SCHEDULE IWS (Continued) INTERRUPTIBLE WINTER STORAGE SERVICE

4.MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Monthly inventory charge, the injection charge, the withdrawal charge, plus all applicable surcharges.

5.RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6.GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part bereof

FERC Docket: RP94-160-000

First Revised Sheet No. 81 First Revised Sheet No. 81: Superseded

Superseding: Original Sheet No. 81
RATE SCHEDULE GDS
GENERAL DELIVERY SERVICE

1. AVAILABILITY

This Rate Schedule is available for combined firm Natural Gas Transportation, Gathering and Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a General Delivery Service ("GDS") Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule; and
- (b) which has executed a Transportation Agreement subject to Panhandle's Rate Schedule EFT or Rate Schedule SCT and a Storage Agreement with Panhandle subject to Panhandle's Rate Schedule IOS with reserved capacities consistent with Section 2 below.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the service management assistance Panhandle will provide in accordance with Section 7 hereof, the General Terms and Conditions herein and the further provisions of the Service Agreement.

Under this Rate Schedule Panhandle will manage Transportation and Storage services designated by the Shipper to relieve the Shipper of submitting nominations at the Point of Delivery as required by Section 8 of the General Terms and Conditions so that the Shipper will not be subject to Daily Scheduling Charges set forth in Section 12.11(h) of the General Terms and Conditions. The Shipper must permit Panhandle to manage its Transportation service under Rate Schedule EFT or Rate Schedule SCT and Storage service under Rate Schedule IOS. The Shipper may also permit Panhandle to manage Storage service under Rate Schedule WS, PS or FS. The service management assistance provided hereunder is available only at a single Point of Delivery as specified in the Service Agreement hereunder. However, in accordance with Section 11.4 of the General Terms and Conditions, multiple measuring stations belonging to a single operator may constitute a single Point of Delivery.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 82 Fourth Revised Sheet No. 82: Superseded

Superseding: Third Revised Sheet No. 82

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE

Service under this Rate Schedule shall be provided to the extent of the Transportation and Storage service the Shipper specifies in the Service Agreement hereunder and the Quantities of Gas Shipper makes available to Panhandle for Transportation and Storage service; provided, the Maximum Daily Withdrawal Quantity under Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS, may not be less than twenty percent (20%) of the Maximum Daily Contract Quantity specified in the Service Agreements under Rate Schedule EFT or Rate Schedule SCT that Shipper permits Panhandle to manage pursuant to this Rate Schedule.

If, after any six consecutive Months, Shipper has experienced excessive Transportation imbalances or overruns or both, and if Shipper does not agree to increase the Maximum Stored Quantity under Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS, specified in the Service Agreement hereunder, Panhandle shall be permitted to terminate the service management assistance provided hereunder at the beginning of any subsequent Month except to the extent required to provide "No Notice Service" as described in Section 3 hereof. In the event service management assistance provided hereunder is terminated in whole or in part, Transportation and Storage services shall continue in accordance with the applicable Rate Schedules and Service Agreements.

3. NO NOTICE SERVICE

Under Rate Schedule GDS, Panhandle shall provide no notice service at the Point of Delivery on any Day up to the Maximum Daily Withdrawal Quantity of Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS and, the Maximum Daily Withdrawal Quantity of Rate Schedules WS, PS and other FS specified in the Service Agreement hereunder subject to the prior receipt by Panhandle of sufficient Stored Quantities under Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS and, if specified in the Service Agreement hereunder, Rate Schedules WS, PS and other

4. STORAGE AND TRANSPORTATION OVERRUN CHARGES AND UNAUTHORIZED OVERRUN PENALTIES

Service under this Rate Schedule GDS is subject to the following overrun charges:

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 83 Fourth Revised Sheet No. 83: Superseded

Superseding: Third Revised Sheet No. 83

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE

(a) If on any Day the Quantities of Gas delivered at the Point of Delivery vary from the Quantities nominated by Shipper pursuant to Section 7 hereof for receipt under the Transportation Agreement, adjusted for Fuel Reimbursement, and if such variances exceed the Maximum Daily Withdrawal Quantity or Maximum Daily Injection Quantity specified in the Service Agreement hereunder, the Shipper shall be subject to the Overrun Charge and, if applicable, the Unauthorized Overrun Penalty in addition to the applicable reservation and commodity charges. The Overrun Charges shall be the rate as set forth on the effective Tariff Sheet No. 9 multiplied by the Quantities of Gas that exceed the MDWQ or the MDIQ, as applicable. The Unauthorized Overrun Penalty shall be as set forth in Section 12.16 of the General Terms and Conditions. Shipper shall also be subject to monthly balancing charges and the Unauthorized Overrun Penalty when Panhandle issues an OFO as set forth in Sections 12.11 and 12.17 of the General Terms and Conditions.

(b) Nothing in this Rate Schedule GDS shall permit Shipper to take at its Point of Delivery, on any Day, Quantities of Gas in excess of the MDCQ of its Transportation service included in the GDS service hereunder or, in any hour, in excess of any hourly swing limitation invoked by Panhandle. Section 3.6 of Rate Schedule EFT and Rate Schedule SCT regarding overruns of Transportation service remain in effect for the Transportation service included in GDS service.

5. RATES

The rates and charges for service under this Rate Schedule shall include all of the applicable charges set forth in Rate Schedule EFT, or Rate Schedule SCT, as applicable, and Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS. If Shipper also specifies service under Panhandle's Rate Schedules WS, PS or other FS for inclusion in the service provided hereunder, the rates and charges for firm Storage service under these Rate Schedules shall also be included under this Rate Schedule.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 84 Fourth Revised Sheet No. 84: Superseded

Superseding: Third Revised Sheet No. 84

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE

6. MONTHLY BILL

The Monthly Bill for the service provided hereunder shall be equal to all of the charges set out in the Monthly Bill of Rate Schedule EFT or Rate Schedule SCT, as applicable, and Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS. If Shipper also specifies Rate Schedules WS, PS or other FS for service hereunder, then the Monthly Bill for the service provided hereunder shall also include all of the charges set out in the Monthly Bill of Rate Schedules WS, PS and other FS. In addition, the Monthly Bill for service provided hereunder shall also include Storage and Transportation Overrun Charges and Unauthorized Overrun Penalties specified in Section 4.

7. OPERATING PLAN

For each Month during the term of the Service Agreement under this Rate Schedule, Panhandle and Shipper shall develop a Monthly operating plan. Under the operating plan, Shipper shall be responsible for nominating and scheduling Quantities of Gas under Rate Schedule EFT or Rate Schedule SCT at the Point(s) of Receipt and providing a Storage use priority plan for service under Rate Schedule ToS, service under Rate Schedule WS, service under Rate Schedule PS or service under Rate Schedule FS.

If Shipper and Panhandle fail to agree on a Storage use priority plan prior to five Days before the beginning of a Month Panhandle shall use the following priority schedule for allocation of actual volumes to the underlying services:

```
Winter Period -

1st - EFT or SCT, up to nominations at the Point(s) of Receipt, less Fuel Reimbursement then,

2nd - WS withdrawals (if applicable) then,

3rd - PS withdrawals (if applicable) then,

4th - FS withdrawals (if applicable) then,

5th - IOS or eligible FS withdrawals/injections

Summer Period -

1st - PS injection (if applicable) then,

2nd - FS injection (if applicable) then,

3rd - WS injection (if applicable) then,

4th - EFT or SCT deliveries then,

5th - IOS or eligible FS injections/withdrawals
```

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 85 Fourth Revised Sheet No. 85: Superseded

Superseding: Third Revised Sheet No. 85

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE

The operating plan shall authorize Panhandle to schedule the injection of Gas into IOS Storage or an eligible Storage Agreement under Rate Schedule FS, (if Shipper has delivered sufficient Gas to Panhandle) and/or the withdrawal of Gas from IOS Storage or an eligible Storage Agreement under Rate Schedule FS, (up to Shipper's Stored Volume) for Shipper's account, in accordance with the operating plan, without further nominations from Shipper. In the event that imbalances or overruns occur due to insufficient actual receipts, insufficient Stored Volume, insufficient Maximum Stored Quantity or any other reason, the Monthly balancing charges and Overrun Charges set out in the underlying agreements and the General Terms and Conditions, as well as any Storage Overrun Charges and Unauthorized Overrun Penalties, as defined in Section 4(a) hereof, shall be applicable. Daily Scheduling Charges which would otherwise apply under Rate Schedules EFT and SCT shall not apply under this Rate Schedule GDS. As a part of the operating plan hereunder, Shipper may provide Panhandle with written authorization to nominate receipts of Gas on Shipper's behalf at any Point(s) of Receipt on Panhandle's system at which Panhandle agrees that Shipper has a contract for Gas purchases and at which receipts can be verified on a "real time" basis. If so authorized, Panhandle will permit the operating plan to include a ranking for "intra-day" nomination of receipts at such point(s), in accordance with Section 8.2 of the General Terms and Conditions. GDS service based on such receipts is conditioned upon verification of such receipts and Shipper will remain liable for storage Overrun Charges and Unauthorized Overrun Penalties in the event its supplier does not perform in accordance with the nominations so tendered by Panhandle on Shipper's behalf.

8. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

9. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RP00-395-003

Second Revised Sheet No. 86 Second Revised Sheet No. 86 : Superseded

Superseding: Substitute First Revised Sheet No. 86
RATE SCHEDULE TBS
TRANSPORTATION BALANCING SERVICE

1. AVAILABILITY

This Rate Schedule is available for combined firm Natural Gas Transportation and Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Transportation Balancing Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which has executed a Transportation Agreement under Rate Schedule FT, EFT, SCT, IT, EIT, LFT or HFT or is the agent of such a party; and
- (c) which has executed a Storage Agreement subject to Panhandle's Rate Schedule IOS or an eligible Storage Agreement under Rate Schedule FS consistent with Section 2 below.

2. APPLICABILITY AND CHARACTER OF SERVICE

Panhandle shall provide Transportation Balancing Service with respect to the Transportation of Natural Gas up to the Maximum Daily Contract Quantity (MDCQ) set out in the designated Transportation Agreements under Rate Schedule FT, EFT, SCT, IT, EIT, LFT, or HFT by coordinating injections and withdrawals from firm Storage service under Rate Schedule IOS or a qualifying Storage Agreement under Rate Schedule FS up to the Maximum Stored Quantity (MSQ), Maximum Daily Injection Quantity (MDIQ) and Maximum Daily Withdrawal Quantity (MDWQ) set out in Shipper's Storage Agreement, subject to the availability of capacity, the General Terms and Conditions herein and the further provisions of the Transportation Balancing Service Agreement. Storage service under Rate Schedule FS shall qualify to be utilized for service under Rate Schedule TBS if the FS Storage Agreement provides for both injections and withdrawals at any time during the year.

In no event shall Panhandle be obligated to provide any Transportation Balancing Service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

The quantity of Transportation Balancing Service ("TBS") available to Shipper shall be defined by the capacity of the Storage Service Agreement specified in the TBS Agreement.

FERC Docket: RS92- 22-005

Original Sheet No. 87 Original Sheet No. 87: Superseded

RATE SCHEDULE TBS (Continued) TRANSPORTATION BALANCING SERVICE

Five Days prior to the beginning of each Month, Shipper must provide to Panhandle a schedule of the designated Transportation Agreements which shall utilize the Transportation Balancing Service during the next Month. When multiple Transportation Agreements are to be balanced, Shipper must provide the allocation scheme for applying the Transportation Balancing Service. The allocation method for the designated Transportation Agreements must be either (1) prorata based upon delivery nominations or (2) a fixed percentage, unless otherwise agreed to by Panhandle.

Throughout the applicable Month, when Shipper's receipts and deliveries under its Transportation Agreements are not in balance, Panhandle shall be authorized to schedule the injection of Gas into IOS Storage and/or the withdrawal of Gas from IOS Storage for Shipper's account, without further nominations from Shipper. Such injections or withdrawals shall be allocated to out-of-balance designated Transportation Agreements in order to achieve a balance, to the extent permitted by Shipper's MSQ, MDIQ, MDWQ and Stored Volume. If imbalances or overruns occur due to insufficient confirmed receipts, insufficient IOS Stored Volume, insufficient Maximum Stored Quantity, insufficient MDIQ or MDWQ or for any other reason, the balancing and overrun charges set out in the designated Service Agreements and the General Terms and Conditions shall be applicable to Shipper.

If, as of the first Day of any Month, any of the designated Transportation Agreements has a Monthly imbalance in excess of a contract's MAIQ or MMIQ, then Panhandle shall be authorized to schedule the injection of Gas into IOS Storage and/or the withdrawal of Gas from IOS Storage for Shipper's account, without further nominations from Shipper. Such injections or withdrawals shall be allocated to the out-of-balance designated Transportation Agreements in order to achieve a balance within MAIQ or eliminate the MMIQ, to the extent permitted by Shipper's MSQ, MDIQ, MDWQ and Stored Volume. If imbalances in excess of MAIQ or MMIQ persist due to insufficient confirmed receipts, insufficient IOS Stored Volume, insufficient Maximum Stored Quantity or for any other reason, the Monthly imbalance provisions set out in the designated Service Agreements and the General Terms and Conditions shall be applicable to Shipper.

FERC Docket: RP00-395-003

First Revised Sheet No. 88 First Revised Sheet No. 88: Superseded

Superseding: Original Sheet No. 88

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE

The Daily Scheduling Charge tolerance level as specified in Section 12.11(h) of the General Terms and Conditions shall be expanded for the designated Service Agreements based upon the level of each Shipper's MDIQ, MDWQ and available Stored Volume applicable to each designated Service Agreement.

RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

4. MONTHLY BILL

The Monthly Bill for Transportation Balancing Service provided hereunder shall be equal to all of the charges set out in the Monthly Bill of Rate Schedule IOS or FS, whichever is specified in Shipper's Transportation Balancing Service Agreement.

5. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RP94-160-000

Original Sheet No. 89 Original Sheet No. 89: Superseded

RATE SCHEDULE FS FLEXIBLE STORAGE SERVICE

1. AVAILABILITY

This Rate Schedule is available for Natural Gas Storage service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule for Storage service from either Panhandle's Market Area Storage Facilities or Field Area Storage Facilities.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle under firm, interruptible or third party transportation agreements at the applicable WS Point and nominated for Storage under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement. In addition, where Shipper is utilizing a third party to make Gas available for Storage in Panhandle's Market Area Storage Facilities under this Rate Schedule, the Shipper may designate as the WS Point an off-system Market Area Storage Facility that Panhandle has leased from a third party, subject to operational feasibility, the terms and conditions applicable to Panhandle's agreement with the third party providing the off-system storage, the General Terms and Conditions herein, and the further provisions of the Service Agreement.

The service provided hereunder consists of the injection of Shipper's Gas into Storage at the applicable WS Point during the Summer Period, Storage in Panhandle's Market Area Storage Facilities or Field Area Storage Facilities, as elected by the Shipper in the Service Agreement, and the subsequent withdrawal from Storage of Shipper's Gas at the applicable WS Point during the Winter Period, all on a firm basis. Winter Period injections and Summer Period withdrawals will be permitted on an interruptible basis, subject to Panhandle's management of Storage to preserve deliverability. Maximum Daily Withdrawal Quantities (MDWQ) and Maximum Daily Injection Quantities (MDIQ) shall be in accordance with the FS Storage Agreement. Available withdrawal quantities will vary according to the percentage of Shipper's Storage Agreement.

FERC Docket: RP94-160-000

Original Sheet No. 90 Original Sheet No. 90: Superseded

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE

The available Stored Volume shall be increased, up to the Maximum Stored Quantity, by the Quantity of Gas Shipper transfers into FS Storage in accordance with Section 14.2 of the General Terms and Conditions. The available Stored Volume shall be reduced by the Quantity of Gas Shipper transfers out of FS Storage in accordance with Section 14.2 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Storage service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for Storage service under this Rate Schedule from Panhandle's Market Area Storage Facilities and Field Area Storage Facilities shall be as follows:

3.1 Capacity Charge

Shipper shall pay a Monthly Capacity Charge which shall be the product of 1/12th the Maximum Stored Quantity times the applicable capacity charge per Dt. as set forth on the effective Tariff Sheet No. 14.

3.2 Deliverability Charge

Shipper shall pay a Monthly Deliverability Charge which shall be the product of the Maximum Daily Withdrawal Quantity set out in the FS Storage Agreement times the applicable deliverability charge per Dt. as set forth on the effective Tariff Sheet No. 14.

3.3 Injection Charge

Shipper shall pay an Injection Charge which shall be the product of the Quantity of Gas actually received from Shipper at the WS Point and injected into Storage during the Month times the applicable injection charge per Dt. as set forth on the effective Tariff Sheet No. 14.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 91 Fourth Revised Sheet No. 91: Superseded

Superseding: Third Revised Sheet No. 91

RATE SCHEDULE FS (Continued)
FLEXIBLE STORAGE SERVICE

3.4 Withdrawal Charge

Shipper shall pay a Withdrawal Charge which shall be the product of the Quantity of Gas actually withdrawn from Storage and delivered to Shipper at the WS Point during the Month times the applicable withdrawal charge per Dt. as set forth on the effective Tariff Sheet No. 14.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

3.6 Overrun Charge and Unauthorized Overrun Penalty

Shipper shall pay the applicable Overrun Charge and, if applicable, the Unauthorized Overrun Penalty per Dt. of Gas actually delivered at the applicable WS Point and injected into Storage or withdrawn from Storage and received for transportation at the applicable WS Point during any Day in excess of the MDIQ and MDWQ set out in the Service Agreement, as applicable. The Overrun Charge per Dt. shall be as set forth on the effective Tariff Sheet No. 14 and the Unauthorized Overrun Penalty shall be as set forth in Sections 12.16 and 12.17 of the General Terms and Conditions. For purposes of computing Overrun Charges and Unauthorized Overrun Penalties applicable to withdrawals in excess of MDWQ when available withdrawal quantities are reduced as a result of Stored Volume being below the levels specified in the FS Storage Agreement, the MDWQ as stated in the FS Storage $\,$ Agreement shall be deemed to be reduced in accordance with the ratchets, if any, set forth in the FS Storage Agreement. The Overrun Charge will not apply to injections during the Withdrawal Period or withdrawals during the Injection Period as set forth in the FS Storage Agreement, if the Quantities injected or withdrawn are within the MDIQ or MDWQ, as applicable.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 14 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

FERC Docket: RP99-192-000

Fourth Revised Sheet No. 92 Fourth Revised Sheet No. 92: Superseded

Superseding: Third Revised Sheet No. 92

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 14 for service hereunder. If the Stored Volume of the Shipper is not reduced to 20% of Maximum Stored Quantity or less prior to the end of the withdrawal period of any Contract Year as specified in the Service Agreement, then Panhandle shall retain .25% of the excess Stored Volume over 20% of the Maximum Stored Quantity to compensate for additional compressor fuel required.

3.9 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2, 3.3 and 3.4 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit A of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit A of the executed Service Agreement. Nothing in this Section 3.9 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

FERC Docket: RP97-185-002

Sub Original Sheet No. 92A Sub Original Sheet No. 92A: Superseded

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the capacity charge, the deliverability charge, the injection charge, the withdrawal charge, any applicable overrun charges plus all applicable surcharges.

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RP95-397-003

Substitute Original Sheet No. 93 Substitute Original Sheet No. 93 : Superseded

RATE SCHEDULE LFT LIMITED FIRM TRANSPORTATION SERVICE

1. AVATLABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- (c) which receives such Gas from Panhandle at a physical Point(s) of Delivery or at a Pool Point(s).

This service is available for firm service at a Point(s) of Delivery without regard to the presence of a Flow Control Device.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the firm Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement on a basis consistent with Section 5 hereof.

Service at the primary and secondary Point(s) of Receipt and Point(s) of Delivery shall be provided on a firm basis subject to Panhandle's right to not schedule service in whole or in part on any Day, but not more than ten (10) Days in each Month. Service that is scheduled under Rate Schedule LFT shall be considered firm service and thus shall be subject to the Nominations and Scheduling of Service, and Curtailment and Interruption procedures included in Sections 8 and 9 of the General Terms and Conditions. Shippers under Rate Schedule LFT may release all or a portion of their MDCQ in accordance with the procedures established for the Assignment of Service Rights under Section 15 of the General Terms and Conditions.

FERC Docket: RP00-162-000

First Revised Sheet No. 94 First Revised Sheet No. 94 : Superseded

Superseding: Substitute Original Sheet No. 94
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

If Panhandle receives an acceptable request for firm Transportation Service pursuant to Rate Schedule FT, EFT or HFT that can only be provided by reducing the MDCQ of services under this Rate Schedule, Panhandle shall notify the Shipper(s) that has the lowest priority of service under this Rate Schedule of the reductions in its or their MDCQ under this Rate Schedule which are necessary to furnish Transportation Service under Rate Schedule FT, EFT or HFT. Priority of service for purposes of the preceding sentence shall be determined as provided in Section 7.1 of the General Terms and Conditions. Such reduction(s) shall be made on the date(s) specified in such notification, which date(s) shall be no less than thirty (30) Days after the date of such notification. Such notified Shipper(s) under this Rate Schedule may, within said thirty (30) Days, convert that portion of service provided under this Rate Schedule which is equal to such reduction, to service under Rate Schedule FT, EFT or HFT, by executing a Service Agreement under Rate Schedule FT, EFT or HFT, which conversion shall have priority over the pending request for service under Rate Schedule FT, EFT or HFT which caused such notification, assuming such Rate Schedule LFT Shipper agrees to a service under Rate Schedule FT, EFT or HFT that will have a net present value that equals or exceeds the net present value of the Rate Schedule FT, EFT or HFT pending request, or such notified Shipper(s) may convert that portion of service provided under this Rate Schedule which is equal to such reduction to service under Rate Schedules IT or EIT by executing a Service Agreement for such service, or such notified Shipper(s) may terminate any remaining portion of service provided under this Rate Schedule. Conversions to Rate Schedules IT or EIT service shall maintain the original priority date under this Rate Schedule, for purposes of priority of service pursuant to Sections 8 and 9 of the General Terms and Conditions.

In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

FERC Docket: RP95-397-003

Substitute Original Sheet No. 95 Substitute Original Sheet No. 95: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

3. RATES

The rates and charges for service under this Rate Schedule LFT shall be as follows:

3.1 Transmission Charge

If Shipper designates a Point(s) of Receipt or a Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Monthly Transmission Charge which consists of a reservation component and a commodity component.

- (a) The reservation component of the Transmission Charge shall be the product of the MDCQ times the applicable Transmission reservation rates per Dt. as set forth on the effective Tariff Sheet No. 15. The reservation component shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available.
- (b) The commodity component of the Transmission Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 15.
- (c) If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 15 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 15 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

FERC Docket: RP95-397-003

Substitute Original Sheet No. 96 Substitute Original Sheet No. 96: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

3.2 Gathering Charge

If Shipper designates primary Point(s) of Receipt or Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for firm service hereunder, Shipper shall pay a Monthly Gathering Charge which shall consist of a reservation component and a commodity component.

- (a) The reservation component of the Gathering Charge shall be the product of the MDCQ times the applicable reservation rate per Dt. as set forth on the effective Tariff Sheet No. 15. The reservation component shall be pro rated for the first and last contract Months to adjust for the number of Days during the Months service was available.
- (b) The commodity component of the Gathering Charge shall be the product of the actual Quantity of Gas delivered during the Month times the applicable commodity rate per Dt. as set forth on the effective Tariff Sheet No. 15.

If Shipper designates a Point(s) of Receipt on the Gathering portion of Panhandle's system as a secondary firm Point(s) of Receipt pursuant to Section 10.3(c) of the General Terms and Conditions, the otherwise applicable charges shall also include the Gathering commodity rate and overrun charge per Dt. as set forth on the effective Tariff Sheet No. 15.

3.3 Special Rate for Limited Class of Backhaul Service

For firm service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 15. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

FERC Docket: RP00-395-003

Second Revised Sheet No. 97 Second Revised Sheet No. 97: Superseded

Superseding: First Revised Sheet No. 97

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

3.4 Transportation Involving Pooling

The rates for firm Transportation under Rate Schedule LFT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include the Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which the Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

- 3.6 Overrun Charges and Unauthorized Overrun Penalties
 - (a) Overrun Charge for Takes in Excess of MDCQ

If during one or more Days in the billing Month Shipper takes Quantities of Gas in excess of the applicable Maximum Daily Contract Quantity as stated in the Service Agreement (hereinafter called overrun quantities) for firm service hereunder, Shipper shall

FERC Docket: RP00-395-003

Second Revised Sheet No. 98 second Revised Sheet No. 98 : Superseded

Superseding: First Revised Sheet No. 98

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

be subject to an Overrun Charge in addition to the applicable reservation and commodity charges and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The applicable Overrun Charge per Dt. for such overrun quantities shall be as set forth on the effective Tariff Sheet No. 15. In addition, subject to the provisions of Section 3.6(d) of this Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty as set forth in Section 12.16 of the General Terms and Conditions.

(b) Overrun Charges for Takes in Excess of Monthly Allowable Quantities

For Monthly takes in excess of the Monthly Allowable Quantity of Gas, Shipper shall be subject to an Overrun Charge in addition to the applicable reservation and commodity charges, any other applicable Overrun Charges and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The Monthly Allowable Quantity of Gas shall be determined by multiplying the Maximum Daily Contract Quantity set forth in the Service Agreement under this Rate Schedule times the number of days of firm service allowed for such Month. The number of days of firm service allowed for each Month shall be the result of the number of days in the Month minus ten. The applicable charge per Dt. for any Quantity of Gas taken during the Month in excess of the Monthly Allowable Quantity shall be the Overrun Charge Per Dt. as set forth on the effective Tariff Sheet No. 15.

(c) Takes in Excess of Permissible Hourly Deliveries

If, pursuant to Section 5 hereof, Shipper has been notified that it must limit takes to one-sixteenth of the Quantities nominated for delivery at the Point(s) of Delivery, Shipper must use its best efforts to limit its takes to one-sixteenth of the Quantities

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 99 Fourth Revised Sheet No. 99: Superseded

Superseding: Third Revised Sheet No. 99

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

nominated for delivery at the Point(s) of Delivery as soon as possible, but in no event later than two hours following notification. If, after such two hours, the Shipper's takes exceed such hourly limitation for any hour during the next twenty-four hour period, Shipper shall pay the Overrun Charges as set forth on the effective Tariff Sheet No. 15, in addition to the applicable commodity rates.

(d) Unauthorized Overrun Penalty for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If the OFO requires the elimination of unauthorized overruns, the applicable Unauthorized Overrun Penalty shall be as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

(e) Overrun Charge for Takes During a Period in Which Service is not Scheduled

If, pursuant to Section 2 hereof, Shipper has been notified that its service is not being scheduled, either completely or partially, Shipper must limit its takes to the level permitted in the notification for scheduled service. If Shipper's takes exceed the level permitted during such period service is not scheduled, the applicable charge per Dt. for volumes in excess of those permitted shall be \$10.00 or two times the Mid-Continent Spot Price calculated in accordance with Section 12.11(c) of the General Terms and Conditions, whichever is greater.

Effective Date: 10/18/1999 Status: Effective

FERC Docket: RP99-482-000

Second Revised Sheet No. 100 Second Revised Sheet No. 100: Superseded

Superseding: First Revised Sheet No. 100
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 15 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 15 for service hereunder.

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

3.10 Customized Reservation Patternb (CRPb)

(a) The CRPb election provides an alternative method of paying the uniform monthly billing of the Reservation Charge set forth on the effective Tariff Sheet No. 15. Billing and payment of CRPb Reservation Charges shall be in accordance with the reservation pattern elected by Shipper. Rates reflecting such Shipper CRPb election shall be posted on the Messengerb system and will be filed with the Commission on or before the first day of the Month of effectiveness of such election. The total annual Reservation Charges resulting from CRPb will equal the total Reservation Charges Shipper would pay without the CRPb election for the same annual period. An adjustment, if necessary, will be included on the invoice for the last month of the CRPb period to ensure that the total Reservation Charges due Panhandle under Panhandle's uniform currently effective rates equals the Reservation Charges received by Panhandle under the CRPb rates. All rights and obligations of Section 17 of the General Terms and Conditions shall apply to the Reservation Charge for each billing Month as elected by Shipper. In the event any CRPp invoice shall be based on rates in effect subject to refund and refunds shall be required, refunds shall be calculated as if the CRPp election had not been made.

Effective Date: 01/09/1997 Status: Effective

FERC Docket: RP97-185-000

First Revised Sheet No. 100A First Revised Sheet No. 100A: Superseded

Superseding: Sub Original Sheet No. 100A
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

(b) Shipper may elect flexible maximum reservation charges under CRPb that are derived from the maximum uniform charges for service on the effective Tariff Sheet No. 15; provided, however, CRPb election is not available to a Shipper with an effective rate agreement pursuant to Section 3.7 herein nor a Replacement Shipper pursuant to Section 15 of the General Terms and Conditions. Shipper shall submit to Panhandle an executed Exhibit D to its Service Agreement in order to participate in the CRPb program.

In addition to the otherwise applicable charges set forth for monthly service on the effective Tariff Sheet No. 15 for November through March, a Shipper electing CRPb will be allowed to pay up to 80% of the Reservation Charges due for the period April through October in the preceding November through March period. CRPb elections which specify the portion of the April through October Reservation Charges to be paid during the preceding November through March period will be due by October 1 of each year to be effective November 1. Revised CRPb elections will be allowed up to the fifteenth Business Day preceding the Month of Gas flow, for prospective application only, so long as the Shipper's total annual obligations for Reservation Charges during the 12 month CRPb Period are not affected by such revisions and provided that such revisions do not result in the Reservation Charge for any month during the April through October period exceeding the charge which would have been applicable had the CRPb election not been made.

- (c) Maximum CRPb reservation rates for each Shipper shall be reflected on effective Tariff Sheet No. 26.
- (d) The usage rates set forth on effective Tariff Sheet No. 15 shall not be affected by the provisions of this Section 3.10.

3.11 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2 and 3.3 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit E of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Effective Date: 10/18/1999 Status: Effective

FERC Docket: RP99-482-000

Sixth Revised Sheet No. 101 Sixth Revised Sheet No. 101: Superseded

Superseding: Fifth Revised Sheet No. 101
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit C of the executed Service Agreement. Nothing in this Section 3.11 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges, plus all applicable surcharges.

5. HOURLY DELIVERIES

Shipper shall deliver or cause to be delivered Natural Gas for the account of Shipper on a uniform hourly basis as nearly as practicable. At each Point(s) of Delivery Shipper may receive from Panhandle in any hour such Quantities of Gas as it requires or as are consistent with available pipeline pressures at such point(s); provided that Panhandle shall be entitled to require Shipper to restrict takes at any point during any hour to one-sixteenth of the Natural Gas nominated for Transportation to that Point(s) of Delivery on that Day by notifying Shipper through the Messengerp system, the Web Site and by telephone or facsimile communication.

Effective Date: 01/09/1997 Status: Effective

FERC Docket: RP97-185-002

Original Sheet No. 101A Original Sheet No. 101A: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE

6. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

7. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

Effective Date: 03/01/1998 Status: Effective

FERC Docket: RP98-121-001

Sub Second Revised Sheet No. 102 Sub Second Revised Sheet No. 102: Superseded

Superseding: First Revised Sheet No. 102

RATE SCHEDULE GPS GAS PARKING SERVICE

1. AVATLABILITY

This Rate Schedule is available for Gas Parking service provided by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called Shipper):

- (a) which has executed a Service Agreement in the form prescribed under this Tariff for service under this Rate Schedule; and
- (b) which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for subsequent delivery by Panhandle for the account of Shipper.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to Gas made available by Shipper to Panhandle in connection with firm or interruptible transportation agreements and nominated for Parking service under this Rate Schedule, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.

The service provided hereunder is an interruptible service and shall be provided only after all other services offered by Panhandle are scheduled and to the extent permitted by Panhandle's system. Providing Parking service shall not impede Panhandle's ability to meet its firm and interruptible service obligations, including Panhandle's system requirements and Panhandle's otherwise applicable firm obligations under a GDS Service Agreement to provide no-notice service up to the MDCQ of the underlying firm Transportation Agreements. The service shall consist of: (1) the receipt/delivery of a Quantity of Gas for Parking for Shipper's account at the points designated by Shipper, up to the maximum parked quantity as specified in the Service Agreement; (2) the Parking of such Gas for the account of Shipper at the Parking point designated by Shipper in the Service Agreement; and (3) the subsequent redelivery/receipt of such Gas to/from Shipper.

Panhandle shall receive from Shipper such daily quantities of Gas as are nominated and scheduled for Parking and park such quantity for a minimum of one (1) day. Gas may be parked for a period of up to thirty one (31) days and such period may be extended by Panhandle as permitted by system operating conditions.

Shipper's account may hold a negative parked quantity, which shall mean the Quantity of Gas that Shipper has received from parking in excess of the sum of the parked quantity at any point in time. Nominations to receive such excess Quantity of Gas shall be confirmed at Panhandle's reasonable discretion.

FERC Docket: RP00-395-003

First Revised Sheet No. 102A First Revised Sheet No. 102A: Superseded

Superseding: Original Sheet No. 102A

RATE SCHEDULE GPS (Continued)
GAS PARKING SERVICE

Panhandle shall accept Shipper's request to schedule a negative parked Quantity when:

- (a) The nomination does not interfere with Panhandle's ability to provide services under all other Rate Schedules;
- (b) Shipper's Transportation Service Agreement has sufficient available capacity to receive or deliver Gas at the Parking point;
- (c) Line pack is sufficient to satisfy the Shipper's request;
- (d) The nomination of such excess Quantity of Gas would have no adverse effect on system operations; and
- (e) The nomination does not result in an imbalance due Panhandle on Shipper's Transportation Service Agreement.

A Pooling Shipper may elect to have Panhandle utilize the Pooling Shipper's Gas Parking account to balance nominated receipts and deliveries under Shipper's associated Pooling Transportation Agreement at a Pool Point on a daily basis. Pooling Shipper shall make such election and specify the Pooling Transportation Agreement and Pool Point at the time the Parking Service Agreement is executed.

Effective Date: 10/18/1999 Status: Effective

FERC Docket: RP99-482-000

Second Revised Sheet No. 103 Second Revised Sheet No. 103 : Superseded

Superseding: Sub First Revised Sheet No. 103
RATE SCHEDULE GPS (Continued)
GAS PARKING SERVICE

The Shipper may be required, upon notification from Panhandle by telephone as well as through the Messengerp system or by facsimile, to cease or reduce deliveries to or receipts from its Parking service within the Gas Day or remove all, or any part, of its parked quantity under this Rate Schedule as rapidly as is consistent with Panhandle's operating capabilities. Any parked quantity not removed in accordance with Panhandle's notice within five (5) days of such notice shall become the property of Panhandle at no cost to Panhandle, free and clear of any adverse claims; provided, however, that Panhandle shall extend the time available for Shipper to remove its parked quantity by one Day for every Day that Shipper has been unable to remove gas due to operational conditions on Panhandle's system.

The Shipper may be required, upon notification from Panhandle by telephone as well as through the Messengerb system or by facsimile, to deliver all, or any part, of its negative parked quantity as rapidly as is consistent with Panhandle's operating capabilities. Any Quantity of Gas not delivered into Parking in accordance with Panhandle's notice within three (3) days of Panhandle giving Shipper such notice shall be subject to a daily penalty of \$10.00 per Dt. or two times the Mid-Continent Spot Price calculated in accordance with Section 12.11(c) of the General Terms and Conditions of Panhandle's FERC Gas Tariff, whichever is greater; provided, however, that Panhandle shall extend the time available for Shipper to return gas to Parking by one Day for every Day that Shipper has been unable to return gas into Parking due to operational conditions on Panhandle's system.

In no event shall Panhandle be obligated to provide any Parking service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for Parking service under this Rate Schedule shall be as follows:

3.1 Daily Parking Charge

The Daily Parking Charge shall be the absolute value of the product of the daily Parked Quantity for each day of the Month and the daily parking rate per Dt. as set forth on the effective Tariff Sheet No. 16.

3.2 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 16 shall be the applicable Maximum Rate set forth thereon. If an amount less than the applicable Maximum Rate and not less than the applicable Minimum Rate is agreed upon, such amount shall be applicable prospectively.

Effective Date: 03/01/1998 Status: Effective

FERC Docket: RP98-121-001

Sub First Revised Sheet No. 104 Sub First Revised Sheet No. 104: Superseded

Superseding: First Revised Original Sheet No. 104

RATE SCHEDULE GPS (Continued)
GAS PARKING SERVICE

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the sum of the Daily Parking Charges incurred during the applicable Month.

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 105 1st Rev Original Sheet No. 105 : Superseded

RATE SCHEDULE HFT
HOURLY FIRM TRANSPORTATION SERVICE

1. AVAILABILITY

This Rate Schedule is available for Natural Gas Transportation and Gathering service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) when Panhandle has determined that it has sufficient available and uncommitted capacity to perform service requested by any party (hereinafter called Shipper):

- 1.1 which has executed a Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule;
- 1.2 which delivers Natural Gas or causes Natural Gas to be delivered to Panhandle for delivery by Panhandle for the account of Shipper; and
- 1.3 which receives Gas from Panhandle at a physical Point(s) of Delivery, which is directly connected to Panhandle's system, equipped with a Flow Control Device and EGM equipment capable of verifying changes in gas flow on a real-time basis.

2. APPLICABILITY AND CHARACTER OF SERVICE

- 2.1 This Rate Schedule shall apply to the firm Transportation and Gathering of Natural Gas on Panhandle's system, subject to the General Terms and Conditions herein and the further provisions of the Service Agreement.
- 2.2 The service provided by Panhandle under this Rate Schedule consists of the receipt of Natural Gas by Panhandle at Point(s) of Receipt located on Panhandle's system and specified in the executed Service Agreement and the delivery of such Gas, after appropriate reductions, at Point(s) of Delivery located on Panhandle's system and specified in the executed Service Agreement up to the Maximum Daily Contract Quantity (MDCQ) set out in the executed Service Agreement.
- 2.3 Shipper shall elect to receive the Hourly Delivery Quantity of its MDCQ during a specified hourly period within the Day as set forth on the executed Service Agreement. Shipper's Hourly Delivery Quantity shall be the MDCQ divided by the specified Hourly Delivery Period set forth in the Service Agreement. The Hourly Delivery Period shall be the minimum amount of time within which Panhandle shall be required to deliver Shipper's MDCQ. Shipper shall take the gas uniformly during the selected Hourly Delivery Period.

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 106 1st Rev Original Sheet No. 106: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

- 2.4 Service at a designated Point of Delivery under this Rate Schedule is limited to only one Service Agreement.
- 2.5 Service at the primary and secondary Point(s) of Receipt and eligible Point(s) of Delivery shall be provided on a firm basis subject to the scheduling, curtailment and interruption provisions of this Rate Schedule and Sections 8 and 9 of the General Terms and Conditions.
- 2.6 In addition to the nomination timeline provisions of Section 8 of the General Terms and Conditions, Shipper must notify Panhandle Gas Control by telephone at least three (3) hours prior to actual gas flow at the Point(s) of Delivery designated for service under this Rate Schedule.
- 2.7 Shipper may release all or a portion of its MDCQ pursuant to Section 15 of the GT&C provided, however, any such released capacity shall be subject to the terms and conditions of this Rate Schedule HFT. Capacity may be released for a minimum of one Day.
- 2.8 In no event shall Panhandle be obligated to provide any Transportation or Gathering service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for service under this Rate Schedule HFT shall be as follows:

3.1 Transmission Charge

If Shipper designates a Point(s) of Receipt or a Point(s) of Delivery on the transmission portion of Panhandle's system, Shipper shall pay a Monthly Transmission Charge which consists of a reservation component and a commodity component.

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 107 1st Rev Original Sheet No. 107: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

(a) The reservation component of the Transmission Charge shall be the product of the Billing MDCQ and the applicable reservation rates per Dt. as set forth on the effective Tariff Sheet No. 17. The Billing MDCQ shall be determined by multiplying the MDCQ by 16 hours divided by the Hourly Delivery Period as set forth in the Service Agreement.

The reservation component shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available. Shipper may elect to participate in the Customized Reservation Patternp program pursuant to Section 3.10 herein.

- (b) The commodity component of the Transmission Charge shall be the product of the actual Quantity of Gas delivered during the Month and the applicable Transmission commodity rates per Dt. as set forth on the effective Tariff Sheet No. 17.
- (c) If both the Point(s) of Receipt and the Point(s) of Delivery are located within the Field Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 17 for service within the Field Zone only; if both the Point(s) of Receipt and Point(s) of Delivery are located within the Market Zone, the applicable rates shall be as set forth on the effective Tariff Sheet No. 17 for service within the Market Zone only; if the Point(s) of Receipt and the Point(s) of Delivery are located in different zones, the applicable rates shall be derived by adding the effective rates for service in the Field Zone and the Market Zone.

3.2 Gathering Charge

If Shipper designates primary Point(s) of Receipt or Point(s) of Delivery on the Gathering portion of Panhandle's system or utilizes the Gathering portion of Panhandle's system for firm service hereunder, Shipper shall pay a Monthly Gathering Charge which shall consist of a reservation component and a commodity component.

(a) The reservation component of the Gathering Charge shall be the product of the Billing MDCQ and the applicable reservation rate per Dt. as set forth on the effective Tariff Sheet No. 17. The Billing MDCQ shall be determined by multiplying the MDCQ by 16 hours divided by the Hourly Delivery Period as set forth in the Service Agreement. The reservation component shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available.

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 108 1st Rev Original Sheet No. 108: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

(b) The commodity component of the Gathering Charge shall be the product of the actual Quantity of Gas delivered during the Month and the applicable commodity rate per Dt. as set forth on the effective Tariff Sheet No. 17.

If Shipper designates a Point(s) of Receipt on the Gathering portion of Panhandle's system as a secondary firm Point(s) of Receipt pursuant to Section 10.3(c) of the General Terms and Conditions, the otherwise applicable charges shall also include the Gathering commodity rate and overrun charge per Dt. as set forth on the effective Tariff Sheet No. 17.

3.3 Special Rate for Limited Class of Backhaul Service

For firm service hereunder for all volumes received at or East of Tuscola and delivered on Panhandle's Transmission facilities at or West of the Haven, Kansas Compressor Station, the rate charged shall be one-half the applicable Transmission charge per Dt. as set forth on the effective Tariff Sheet No. 17. Backhaul services provided which do not satisfy the foregoing specified criteria shall be charged the generally applicable rate.

3.4 Transportation Involving Pooling

The rates for firm Transportation under Rate Schedule HFT involving pooling, as set forth in Section 14 of the General Terms and Conditions, shall be as specified in this Section 3; provided that the rates charged for service under a Pooling Transportation Agreement shall not include the Field Zone Transmission Charges, so long as title to such Gas passes to another Shipper and the Quantities of Gas are delivered for immediate receipt and subsequent Transportation on Panhandle's system under a Service Agreement to which the Field Zone Transmission Charges apply. If the Pooling Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Field Zone charges, such charges shall not apply to the Quantities transported under the Corresponding Transportation Agreement but shall apply to the Pooling Transportation Agreement for such Month. In addition, if the Corresponding Shipper notifies Panhandle in writing at the time its initial nomination for the Month is submitted that it has agreed to pay the Gathering charges, such charges shall not apply to the Quantities transported under the Pooling Transportation Agreement but shall apply to the Corresponding Transportation Agreement for such Month.

FERC Docket: RP00-395-003

First Revised Sheet No. 109 First Revised Sheet No. 109: Superseded

Superseding: Sub Original Sheet No. 109

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

3.5 Surcharges

Shipper shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

- 3.6 Overrun Charges and Unauthorized Overrun Penalties
 - (a) If during one or more Days in the billing Month Shipper takes Quantities of Gas in excess of the applicable Maximum Daily Contract Quantity as stated in the Service Agreement (hereinafter called overrun quantities) for firm service hereunder, Shipper shall be subject to an Overrun Charge in addition to the applicable reservation and commodity charges and any balancing charges that may be applicable pursuant to Section 12 of the General Terms and Conditions. The applicable Overrun Charge per Dt. for such overrun quantities shall be as set forth on the effective Tariff Sheet No. 17. In addition, subject to the provisions of Section 3.6(b) of this Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty as set forth in Section 12.16 of the General Terms and Conditions.
 - (b) Unauthorized Overrun Penalty for Takes During an OFO

Panhandle may issue an OFO and, if it does so, Panhandle shall notify Shippers through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Once issued the OFO shall continue until Panhandle notifies Shippers to the contrary using the above methods. If the OFO requires the elimination of unauthorized overruns, the applicable Unauthorized Overrun Penalty shall be as set forth in Section 12.17(f)(ii) of the General Terms and Conditions.

(c) Takes in Excess of Shipper's Contracted Hourly Delivery Quantity

If Shipper's takes exceed the Hourly Delivery Quantity as stated in Shipper's Service Agreement for any hour during the Day, the applicable charge per Dt. for such excess quantities shall be \$10 per Dt.

FERC Docket: RP00-162-000

Original Sheet No. 110 Original Sheet No. 110: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

by 3.413 (industry standard conversion factor to convert \$/MWh to \$/MMBtu). If the reported price referenced above is not published for the Day required, Panhandle shall utilize another publication that publishes a comparable location and price.

3.7 Range of Rates

Unless otherwise agreed to in writing by Shipper and Panhandle, any rate applicable to a Shipper for service hereunder, by reference to the effective Tariff Sheet No. 17 shall be the applicable Maximum Rate(s) set forth thereon. If an amount less than the applicable Maximum Rate(s) and not less than the applicable Minimum Rate(s) is agreed upon, such amount shall be applicable prospectively.

3.8 Fuel Reimbursement

Shipper shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 17 for service hereunder.

3.9 Balancing Charges

If balancing charges are incurred in accordance with Section 12 of the General Terms and Conditions, then such balancing charges shall also be applicable.

FERC Docket: RP00-162-000

Original Sheet No. 111 Original Sheet No. 111: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

3.10 Customized Reservation Patternp (CRPp)

- (a) The CRPp election provides an alternative method of paying the uniform monthly billing of the Reservation Charge set forth on the effective Tariff Sheet No. 17. Billing and payment of CRPb Reservation Charges shall be in accordance with the reservation pattern elected by Shipper. Rates reflecting such Shipper CRPb election shall be posted on the Messengerb system and will be filed with the Commission on or before the first day of the Month of effectiveness of such election. The total annual Reservation Charges resulting from CRPb will equal the total Reservation Charges Shipper would pay without the CRPb election for the same annual period. An adjustment, if necessary, will be included on the invoice for the last month of the CRPb period to ensure that the total Reservation Charges due Panhandle under Panhandle's uniform currently effective rates equals the Reservation Charges received by Panhandle under the CRPb rates. All rights and obligations of Section 17 of the General Terms and Conditions shall apply to the Reservation Charge for each billing Month as elected by Shipper. In the event any CRPp invoice shall be based on rates in effect subject to refund and refunds shall be required, refunds shall be calculated as if the CRPb election had not been made.
- (b) Shipper may elect flexible maximum reservation charges under CRPb that are derived from the maximum uniform charges for service on the effective Tariff Sheet No. 17; provided, however, CRPb election is not available to a Shipper with an effective rate agreement pursuant to Section 3.7 herein nor a Replacement Shipper pursuant to Section 15 of the General Terms and Conditions. Shipper shall submit to Panhandle an executed Exhibit B to its Service Agreement in order to participate in the CRPb program.

FERC Docket: RP00-162-000

Original Sheet No. 112 Original Sheet No. 112: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

In addition to the otherwise applicable charges set forth for monthly service on the effective Tariff Sheet No. 17 for November through March, a Shipper electing CRPb will be allowed to pay up to 80% of the Reservation Charges due for the period April through October in the preceding November through March period. CRPb elections which specify the portion of the April through October Reservation Charges to be paid during the preceding November through March period will be due by October 1 of each year to be effective November 1. Revised CRPb elections will be allowed up to the fifteenth Business Day preceding the Month of Gas flow, for prospective application only, so long as the Shipper's total annual obligations for Reservation Charges during the 12 month CRPb Period are not affected by such revisions and provided that such revisions do not result in the Reservation Charge for any month during the April through October period exceeding the charge which would have been applicable had the CRPb election not been made.

- (c) Maximum CRPp reservation rates for each Shipper shall be reflected on effective Tariff Sheet No. 26.
- (d) The usage rates set forth on effective Tariff Sheet No. 17 shall not be affected by the provisions of this Section 3.10.

3.11 Negotiated Rates

Shipper and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to the charges identified in Sections 3.1, 3.2 and 3.3 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than the Minimum Rate; may be based on a rate design other than straight fixed variable; and may include a minimum quantity. Such Negotiated Rate shall be set forth on Exhibit C of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any Shipper that does not choose a Negotiated Rate.

Shippers paying a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of scheduling, curtailment and interruption, and calculating the economic value of a request for unsubscribed firm capacity. In addition, a Shipper that does not agree to pay any surcharges will not be granted a scheduling preference that deems such surcharges have been paid. Replacement Shippers may not bid or pay a rate greater than the Maximum Rate and are not eligible for Negotiated Rates.

FERC Docket: RP00-162-000

Original Sheet No. 113 Original Sheet No. 113: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE

In the event that capacity subject to a Negotiated Rate which is based on a rate design other than straight fixed variable is released, Shipper and Panhandle may agree on billing adjustments to the Releasing Shipper that may vary from or are in addition to those set forth in Section 15.6 of the General Terms and Conditions in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. Such payment obligation and crediting mechanism for capacity release shall be set forth on Exhibit C of the executed Service Agreement. Nothing in this Section 3.10 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Transmission Charge, the Gathering Charge, any applicable overrun and balancing charges including charges for takes in excess of Shipper's contracted hourly deliveries, plus all applicable surcharges.

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

FERC Docket: RP00-395-003

Original Sheet No. 114 Original Sheet No. 114: Superseded

RATE SCHEDULE DVS DELIVERY VARIANCE SERVICE

1. AVAILABILITY

This Rate Schedule is available for Delivery Variance Service performed by Panhandle Eastern Pipe Line Company (hereinafter called Panhandle) for any party (hereinafter called DVS Party) which has executed a Delivery Variance Service Agreement in the form prescribed under Panhandle's FERC Gas Tariff for service under this Rate Schedule and

- (a) which has executed a Transportation Agreement; or
- (b) which operates a Point of Delivery.

2. APPLICABILITY AND CHARACTER OF SERVICE

This Rate Schedule shall apply to the firm Delivery Variance Service provided by Panhandle subject to the General Terms and Conditions herein and the further provisions of the Service Agreement. The service provided hereunder is available only to the DVS Party at a single Point of Delivery as designated in the Service Agreement.

Under this Rate Schedule, Panhandle shall manage delivery point daily scheduling variances up to the Maximum Daily Variance Quantity (MDVQ) as set forth in the Service Agreement; provided, however, the MDVQ shall not entitle the Shipper to take Quantities of Gas in excess of the MDCQ of the underlying Transportation Agreement(s). The maximum MDVQ under a DVS Service Agreement shall be determined on a case-by-case basis after evaluating Panhandle's existing firm obligations and system requirements.

If the DVS Party is a Shipper and the daily scheduling variance at the Point of Delivery exceeds the tolerance level, the MDVQ shall be added to the absolute value of the Shipper's calculated tolerance Quantity for the underlying Transportation Agreement as provided under Section 12.11(h) of the General Terms and Conditions. Any variance then remaining shall be subject to the daily scheduling charge.

FERC Docket: RP00-395-003

Original Sheet No. 115 Original Sheet No. 115: Superseded

RATE SCHEDULE DVS (Continued) DELIVERY VARIANCE SERVICE

If the DVS Party is a delivery point operator, the MDVQ shall be added to the absolute value of the calculated tolerance Quantity for the delivery meter as provided in Section 12.11(h) of the General Terms and Conditions. Any variance then remaining shall be allocated back to the applicable Transportation Agreement(s) based on the predetermined allocation methodology for the designated Point of Delivery and shall be subject to the daily scheduling charge.

The service provided hereunder shall not affect the scheduling priorities of the underlying Transportation Agreements, as set forth in Section 8 of the General Terms and Conditions. Also, the service provided hereunder shall not affect the calculation of any overrun charge or overrun penalty applicable to the underlying Transportation Agreement(s).

In no event shall Panhandle be obligated to provide any Delivery Variance Service for which capacity is not available or which would require the construction or acquisition of new facilities or the modification or expansion of existing facilities.

3. RATES

The rates and charges for service under Rate Schedule DVS shall be:

3.1 Reservation Charge

The reservation charge applicable to this Rate Schedule shall be the product of the MDVQ times the reservation rate per Dt. as set forth on the effective Tariff Sheet No. 18. The service charge shall be pro-rated for the first and last contract Months to adjust for the number of Days during those Months service was available.

3.2 Commodity Charge

The commodity charge applicable to this Rate Schedule shall be the product of the actual Quantity of Gas applicable to the Delivery Variance Service each Day times the commodity rate per Dt. as set forth on the effective Tariff Sheet No. 18.

FERC Docket: RP00-395-003

Original Sheet No. 116 Original Sheet No. 116: Superseded

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE

3.3 Surcharges

DVS Party shall pay all surcharges specified in the General Terms and Conditions or which otherwise may be applicable to service under this Rate Schedule from time to time.

3.4 Range of Rates

Unless otherwise agreed to in writing by the DVS Party and Panhandle, any rate applicable to a DVS Party for service hereunder, by reference to the effective Tariff Sheet No. 18 shall be the applicable Maximum Rate set forth thereon. If an amount less than the applicable Maximum Rate is agreed upon, such amount shall be applicable prospectively.

3.5 Fuel Reimbursement

DVS Party shall reimburse Panhandle in kind for fuel usage and lost or unaccounted for Gas pursuant to the terms and conditions of the Service Agreement and as stated on the effective Tariff Sheet No. 18 for service hereunder.

3.6 Negotiated Rates

The DVS Party and Panhandle may agree, on a prospective basis, to a Negotiated Rate with respect to charges identified in Section 3.1 and 3.2 herein which may be less than, equal to or greater than the Maximum Rate; shall not be less than zero; and may be based on a rate design other than straight fixed variable. Such Negotiated Rate shall be set forth on Exhibit A of the executed Service Agreement and on the currently effective Tariff Sheet No. 27. The Maximum Rate shall be available to any DVS Party that does not choose a Negotiated Rate. Nothing in this Section 3.6 shall authorize Panhandle or Shipper to negotiate terms and conditions of service.

4. MONTHLY BILL

The Monthly Bill for service under this Rate Schedule shall be equal to the Reservation Charge, Commodity Charge and applicable surcharges.

FERC Docket: RP00-395-003

Original Sheet No. 117 Original Sheet No. 117: Superseded

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE

5. RESERVATIONS

Panhandle reserves the right from time to time to unilaterally make any changes to, or to supersede, the rates and charges and other terms in this Rate Schedule and the other provisions of Panhandle's FERC Gas Tariff, and the applicability thereof, subject to the provisions of the Natural Gas Act and the Commission's Regulations thereunder.

6. GENERAL TERMS AND CONDITIONS

The General Terms and Conditions of Panhandle's FERC Gas Tariff are applicable to this Rate Schedule and are hereby made a part hereof.

Effective Date: 02/01/2002 Status: Effective FERC Docket: RP00-395-003

Sheet Nos. 118-199 Sheet Nos. 118-199 : Superseded

SHEET NOS. 118 THROUGH 199 RESERVED FOR FUTURE USE

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Original Sheet No. 200 Original Sheet No. 200: Superseded

PANHANDLE EASTERN PIPE LINE COMPANY GENERAL TERMS AND CONDITIONS

FERC Docket: RP02-466-000

First Revised Sheet No. 201 First Revised Sheet No. 201: Superseded Superseding: Original Sheet No. 201

GENERAL TERMS AND CONDITIONS

INDEX

Section	Title	Sheet	No.
1.	DEFINITIONS	203	
2.	ELECTRONIC COMMUNICATION SYSTEM	210	
3.	QUALITY OF GAS	213	
4.	MEASUREMENT	215	
5.	RESPONSIBILITY DURING TRANSPORTATION OR STORAGE	218	
6.	PROCEDURES FOR REQUESTING SERVICE RIGHTS	219	
7.	CONTRACTING FOR SERVICE RIGHTS	232	
8.	NOMINATION AND SCHEDULING OF SERVICE	238	
9.	CURTAILMENT AND INTERRUPTION	245	
10.	POINTS OF RECEIPT	249	
11.	POINTS OF DELIVERY	253	
12.	CONDITIONS OF RECEIPT AND DELIVERY	256	
13.	POLICY FOR CONSTRUCTION OF NEW RECEIPT OR DELIVERY FACILITIES	273	
14.	POOLING AND IN-FIELD TRANSFERS OF STORAGE INVENTORY	274	
15.	PROCEDURES FOR ASSIGNMENT OF SERVICE RIGHTS	276	
16.	ASSIGNMENT OF UPSTREAM SERVICE RIGHTS	284	
17.	STATEMENTS AND PAYMENTS	285	
18.	PERIODIC RATE ADJUSTMENTS AND SURCHARGES	288	

FERC Docket: RP03-479-000

Third Revised Sheet No. 202 Third Revised Sheet No. 202: Superseded Superseding: Second Revised Sheet No. 202

GENERAL TERMS AND CONDITIONS

INDEX (Continued)

Section	Title	Sheet No.
19.	WARRANTY AND ASSIGNMENTS	322
20.	FORCE MAJEURE	323
21.	TERMINATION FOR DEFAULT	324
22.	REGULATIONS	325
23.	OPERATION OF RATE SCHEDULES IN CONJUNCTION WITH MARKETING AND GATHERING AFFILIATES	326
24.	FUEL REIMBURSEMENT ADJUSTMENT	327
25.	FLOW THROUGH OF CASH-OUT REVENUES AND PENALTIES IN EXCESS OF COSTS	329
26.	RESERVED FOR FUTURE USE	331
27.	MISCELLANEOUS	337

Effective Date: 03/01/1997 Status: Suspended

FERC Docket: RP96-348-000

Third Revised Sheet No. 203 Third Revised Sheet No. 203: Suspended

Superseding: Substitute Second Revised Sheet No. 203

GENERAL TERMS AND CONDITIONS

The following General Terms and Conditions are applicable to any service provided hereunder:

1. DEFINITIONS

- 1.1 British Thermal Unit (Btu) shall mean the amount of heat required to raise the temperature of one (1) avoirdupois pound of pure water from 58.5øF to 59.5øF under standard conditions as described in Section 4.2.
- 1.2 Business Day shall mean from 8:00 A.M. to 5:00 P.M. Central Time on any weekday excluding Panhandle's company holidays.
- 1.3 Commission or FERC shall mean the Federal Energy Regulatory Commission or any successor regulatory agency.
- 1.4 Contract Year as used in Rate Schedules FT, EFT, SCT, IT, EIT, IOS, IIOS, GDS, TBS and LFT shall mean an annual period beginning the first Day of the Month following the effectiveness of the Service Agreement and ending twelve Months later; and as used in Rate Schedules WS, PS, IWS and FS shall mean an annual period beginning April 1 and ending the following March 31.
- 1.5 Corresponding Transportation Agreement shall mean a Transportation Agreement that identifies a Pool Point as a Point of Receipt and provides for further delivery downstream of the Pool Point.
- 1.6 Corresponding Shipper shall mean a Shipper that nominates receipts of Gas at a Pool Point for subsequent delivery under a Corresponding Transportation Agreement.
- 1.7 Daily Scheduling Charge shall mean the charge that applies to a Transportation Agreement when the difference between the sum of the actual Quantities of Gas delivered at any Point of Delivery and the sum of Quantities of Gas nominated at such point exceeds the tolerance level. The Daily Scheduling Charge under normal conditions is set forth in Section 12.11(h) of the General Terms and Conditions and the Daily Scheduling Charge that applies when Panhandle declares an extreme conditions situation is set forth in Section 12.17 of the General Terms and Conditions.

FERC Docket: RP00-395-003

Sixth Revised Sheet No. 203 Sixth Revised Sheet No. 203: Superseded

Superseding: Fifth Revised Sheet No. 203

GENERAL TERMS AND CONDITIONS

The following General Terms and Conditions are applicable to any service provided hereunder:

1. DEFINITIONS

- 1.1 British Thermal Unit (Btu) shall mean the amount of heat required to raise the temperature of one (1) avoirdupois pound of pure water from 58.5 øF to 59.5 øF under standard conditions as described in Section 4.2.
- 1.2 Business Day shall mean Monday through Friday from 8:00 A.M. to 5:00 P.M. Central Time excluding Federal Banking Holidays.
- 1.3 Central Time shall mean Central Clock Time whether Central Standard Time or Central Daylight Savings Time.
- 1.4 Commission or FERC shall mean the Federal Energy Regulatory Commission or any successor regulatory agency.
- 1.5 Contract Year as used in Rate Schedules FT, EFT, SCT, IT, EIT, IOS, IIOS, GDS, TBS, LFT, HFT and DVS shall mean an annual period beginning the first Day of the Month following the effectiveness of the Service Agreement and ending twelve Months later; and as used in Rate Schedules WS, PS, IWS and FS shall mean an annual period beginning April 1 and ending the following March 31.
- 1.6 Corresponding Transportation Agreement shall mean a Transportation Agreement that identifies a Pool Point as a Point of Receipt and provides for further delivery downstream of the Pool Point.
- 1.7 Corresponding Shipper shall mean a Shipper that nominates receipts of Gas at a Pool Point for subsequent delivery under a Corresponding Transportation Agreement.
- 1.8 Daily Scheduling Charge shall mean the charge that applies to a Transportation Agreement when the difference between the sum of the actual Quantities of Gas delivered at any Point of Delivery and the sum of Quantities of Gas nominated at such point exceeds the tolerance level. The Daily Scheduling Charge under normal conditions is set forth in Section 12.11(h) of the General Terms and Conditions and the Daily Scheduling Charge that applies when Panhandle issues an OFO is set forth in Section 12.17 of the General Terms and Conditions.

FERC Docket: RP00-395-003

Third Revised Sheet No. 204 Third Revised Sheet No. 204: Superseded

Superseding: Second Revised Sheet No. 204

GENERAL TERMS AND CONDITIONS (Continued)

- 1.9 Day shall mean a period of twenty-four (24) consecutive hours beginning and ending at 9:00 A.M. Central Time. The reference date for any Day shall be the date of the beginning of such Day.
- 1.10 Dekatherm (Dt.) shall mean one thousand cubic feet of Gas (Mcf) with a heat content of 1,000 Btu per cubic foot on a measurement basis specified under Section 4 herein and based on the same pressure base as the volumes.
- 1.11 DVS Party shall mean the entity that has executed a Service Agreement under Rate Schedule DVS with Panhandle.
- 1.12 Electronic Gas Measurement (EGM) shall mean that form of measurement equipment which may consist of a computerized remote terminal unit, transducers and other associated power, radio and sensing and other electronic devices to accomplish Gas measurement and transfer of data, without the use of charts.
- 1.13 Electronic Funds Transfer shall mean payments made or effected by wire transfer (Fedwire, CHIPS, or Bank Book Entry) or Automated Clearing House (ACH) or any other recognized electronic or automated payment mechanism that is agreed upon by Shipper and Panhandle in the future.
- 1.14 Field Area Storage Facilities shall mean the storage facilities Panhandle owns and leases located upstream of the suction side of Panhandle's Haven, Kansas Compressor Station.
- 1.15 Field Zone shall mean the portion of Panhandle's system located on the suction side of Panhandle's Haven, Kansas Compressor Station and extending westward through Kansas, Oklahoma and Texas.
- 1.16 Flow Control Device shall mean a physical valve at a Point of Receipt or Point of Delivery on Panhandle's pipeline which provides Panhandle with control over flowing volumes at that point.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 205 Fourth Revised Sheet No. 205: Superseded

Superseding: Third Revised Sheet No. 205

GENERAL TERMS AND CONDITIONS (Continued)

- 1.17 Former Tariff shall mean Panhandle's FERC GAS TARIFF,
 Original Volume No. 1 as it was in effect on the Day prior
 to the effectiveness hereof.
- 1.18 Fuel Reimbursement shall mean the reimbursement of Panhandle by Shipper in kind for fuel usage and lost or unaccounted for Gas.
- 1.19 Gathering shall mean the receipt by Panhandle of Natural Gas from a wellhead or other facility connected to Panhandle's Gathering facilities and its Transportation to the outlet side of Panhandle's facilities at the terminus of any portion of Panhandle's Gathering system or such other mutually agreeable point(s) on Panhandle's Gathering system as set forth in the Transportation Agreement.
- 1.20 IOS Point shall mean the point near Panhandle's Haven, Kansas Compressor Station at which Gas is received for injection into Storage and Gas withdrawn from Storage is delivered to Panhandle under Rate Schedules IOS and IIOS.
- 1.21 Long-Term Agreement shall mean a Transportation or Storage Agreement with a primary term of one year or more from the effective date of the Agreement.
- 1.22 Market Area Storage Facilities shall mean the storage facilities Panhandle owns and leases located downstream of the discharge side of Panhandle's Haven, Kansas Compressor Station.
- 1.23 Market Zone shall mean the portion of Panhandle's system located on the discharge side of Panhandle's Haven, Kansas Compressor Station and extending eastward through Kansas, Missouri, Illinois, Indiana, Ohio and Michigan.
- 1.24 Maximum Daily Contract Quantity (MDCQ) shall mean the maximum Quantity of Gas which Panhandle is obligated to deliver on any Day at a Point of Delivery to Shipper as specified in an executed Transportation Agreement.

FERC Docket: RP00-395-003

Seventh Revised Sheet No. 206 Seventh Revised Sheet No. 206: Superseded

Superseding: Sixth Revised Sheet No. 206

GENERAL TERMS AND CONDITIONS (Continued)

1.25 Maximum Daily Injection Quantity (MDIQ) shall mean the maximum Quantity of Gas which Panhandle is obligated to inject into Storage for the account of Shipper on any Day as specified in an executed Storage Agreement.

- 1.26 Maximum Daily Variance Quantity (MDVQ) shall mean the maximum Quantity of Gas that Panhandle is obligated to manage under Rate Schedule DVS.
- 1.27 Maximum Daily Withdrawal Quantity (MDWQ) shall mean the maximum Quantity of Gas which Panhandle is obligated to withdraw from Storage for the account of Shipper on any Day as specified in an executed Storage Agreement.
- 1.28 Maximum Rate shall mean the applicable maximum rate as set forth on the currently effective Tariff Sheet Nos. 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15, 16, 17, and 18 plus all surcharges specified in the General Terms and Conditions as may be applicable from time to time.
- 1.29 Maximum Stored Quantity (MSQ) shall mean the maximum Quantity of Gas which Shipper is permitted to have in Storage at any time during the Contract Year as specified in an executed Storage Agreement.
- 1.30 Messenger shall mean the electronic communication system offered by Panhandle on a nondiscriminatory basis to any Shipper that has compatible equipment for electronic transmission of data, provided that such Shipper has been assigned a USERID and password and agrees to comply with the procedures for access to Messenger and with the procedures for use of the Messenger system.
- 1.31 Minimum Rate shall mean the applicable minimum rate as set forth on the currently effective Tariff Sheet Nos. 4, 5, 6, 7, 8, 9, 10, 11, 12, 13, 14, 15, 16, 17 and 18.
- 1.32 Month shall mean the period beginning at 9:00 a.m. Central Time on the first Day of the calendar Month and ending at 9:00 a.m. Central Time on the first Day of the next succeeding calendar Month.
 - 1.33 Mcf shall mean 1,000 cubic feet of Gas.
 - 1.34 Mmcf shall mean 1,000,000 cubic feet of Gas.
 - 1.35 Natural Gas or Gas shall mean either Natural Gas unmixed, or a mixture of natural and artificial Gas.

FERC Docket: RP00-395-003

Seventh Revised Sheet No. 207 Seventh Revised Sheet No. 207 : Superseded

Superseding: Sixth Revised Sheet No. 207

GENERAL TERMS AND CONDITIONS (Continued)

- 1.36 Negotiated Rate shall mean the rate agreed to by Shipper and Panhandle which may be less than, equal to or greater than the Maximum Rate but shall not be less than the Minimum Rate. The Negotiated Rate may be based on a rate design other than straight fixed variable and may include a minimum quantity.
- 1.37 Operational Flow Order (OFO) shall mean an order issued by Panhandle in the circumstances specified in Section 12.17 requiring a Shipper(s) to take a certain action(s) or to refrain from taking a certain action(s).
- 1.38 Overrun Charge shall mean the charge that applies when a Shipper takes Quantities of Gas in excess of the MDCQ as stated in Shipper's Service Agreement under Rate Schedules FT, EFT, SCT, LFT and HFT. Overrun Charge shall also mean the charge that applies when Shipper exceeds the MDIQ or MDWQ under Shipper's Service Agreement as reflected in Section 3.6 of Rate Schedules IOS, WS and FS and Section 4.6 of Rate Schedule PS.
- 1.39 Panhandle shall mean Panhandle Eastern Pipe Line Company.
- 1.40 Parking shall mean the receipt by Panhandle of Gas from or for the account of Shipper, the holding of such Gas for a period of time and the subsequent redelivery to Shipper.
- 1.41 Parking Agreement shall mean an agreement for Parking service provided under Rate Schedule GPS.
- 1.42 Pool Point shall mean a point on Panhandle's system located at or west of Panhandle's Haven, Kansas Compressor Station which has been identified as a Point of Receipt in a Corresponding Transportation Agreement, subject to Section 14 herein. A Pool Point may not be located at a point where Gas physically enters or leaves Panhandle's system.
- 1.43 Pooling Shipper shall mean a Shipper under a Pooling Transportation Agreement.
- 1.44 Pooling Transportation Agreement shall mean a Transportation Agreement under Rate Schedules FT, EFT, IT, EIT or LFT that designates a Pool Point as the sole Point of Delivery.

FERC Docket: RP00-395-003

Sixth Revised Sheet No. 208 Sixth Revised Sheet No. 208: Superseded

Superseding: Fifth Revised Sheet No. 208

GENERAL TERMS AND CONDITIONS (Continued)

- 1.45 Point of Receipt shall mean a point at the inlet side of Panhandle's facilities at a point of interconnection between the facilities of Panhandle and facilities of Shipper or Shipper's designee, a Pool Point or such other mutually agreeable point as set forth in the Service Agreement. Point(s) of Receipt located on the Gathering and transmission portions of Panhandle's system shall be identified in the Service Agreement.
- 1.46 Point of Delivery shall mean a point at the outlet side of Panhandle's facilities at a point of interconnection between the facilities of Panhandle and the facilities of Shipper or Shipper's designee, a Pool Point or such other mutually agreeable point as set forth in the Service Agreement. Point(s) of Delivery located on the Gathering and transmission portions of Panhandle's system shall be identified in the Service Agreement.
- 1.47 Primary Path shall mean the portion of capacity physically located between the designated Primary Points of Receipt and Delivery.
- 1.48 PS Point shall mean the point near the Michigan and Ohio state line traversed by Panhandle's pipeline facilities at which Gas is received for injection into Panhandle's Market Area Storage Facilities and Gas withdrawn from such Storage facilities is delivered to Panhandle under Rate Schedule PS.
- 1.50 Seasonal Storage shall mean Storage service provided under Rate Schedules WS, PS, IWS, or FS.
- 1.51 Service Agreement shall mean a Transportation Agreement, Storage Agreement, Parking Agreement, or Delivery Variance Service Agreement.
- 1.52 Storage shall mean the receipt by Panhandle of Gas from or for the account of Shipper, the retention by Panhandle of that Gas for a period of time in an underground facility and the redelivery to Shipper.
- 1.53 Storage Agreement shall mean an agreement for Storage service provided under Rate Schedules IOS, IIOS, WS, PS, IWS or FS.
- 1.54 Stored Volume shall mean the Quantity of Gas held by Panhandle in Storage for Shipper's account at any time.

FERC Docket: RP00-395-003

Eighth Revised Sheet No. 209 Eighth Revised Sheet No. 209: Superseded

Superseding: Seventh Revised Sheet No. 209

GENERAL TERMS AND CONDITIONS (Continued)

1.55 Summer Period shall mean the period from April 1 through October 31.

- 1.56 Thermal or Thermally Equivalent shall mean an equal number of Btu's.
- 1.57 Transportation shall mean forward haul, backhaul or any other method of transport of Gas as defined in the FERC's Regulations, except Storage.
- 1.58 Transportation Agreement shall mean an agreement for Transportation service provided under Rate Schedule FT, EFT, SCT, IT, EIT, LFT, GPS or HFT.
- 1.59 Unauthorized Gas shall mean any Quantity of Gas, either received or delivered by Panhandle which cannot be identified as being received or delivered pursuant to a Transportation or Storage withdrawal nomination on behalf of an existing Shipper.
- 1.60 Unauthorized Overrun Penalty shall mean the penalty that applies when, on any Day, Shipper takes Quantities of Gas in excess of the MDCQ as stated in the Transportation Agreement or Quantities of Gas in excess of the MDIQ or MDWQ as stated in the Storage Agreement and such excess Quantities have not been scheduled by Panhandle.
- 1.61 Web Site shall mean Panhandle's HTML page accessible via the Internet's World Wide Web located at http://www.cmsenergy.com.
- 1.62 Winter Period shall mean the period November 1 through March 31 of the next year.
- 1.63 WS Point shall mean the point at which Gas is received for injection into Storage and Gas withdrawn from Storage is delivered to Panhandle under Rate Schedules WS, IWS and FS. If Shipper's Service Agreement provides for Storage service under Rate Schedule WS, IWS, or FS from Panhandle's Market Area Storage Facilities, the WS Point shall be at a point near the Michigan and Ohio state line traversed by Panhandle's pipeline facilities; if Shipper's Service Agreement provides for Storage service under Rate Schedule WS, IWS, or FS from Panhandle's Field Area Storage Facilities, the WS Point shall be at a point near Panhandle's Haven, Kansas Compressor Station.

FERC Docket: RP02-466-000

Second Revised Sheet No. 210 second Revised Sheet No. 210 : Superseded

Superseding: First Revised Sheet No. 210
GENERAL TERMS AND CONDITIONS
(Continued)

2. ELECTRONIC COMMUNICATION SYSTEM

- 2.1 Panhandle has established an interactive electronic communication system, the Messenger system, for use by any party, including Shippers and potential Shippers. The Messenger system will be available on a nondiscriminatory basis to any party (hereinafter the "User") that has compatible equipment for electronic transmission of data, provided that such party has been assigned a user identification (USERID) and password, executes a Messenger Agreement in the form prescribed in this tariff and agrees to comply with the procedures for use of the Messenger system. Panhandle will provide timely and equal access to any and all information available on the Messenger system.
- 2.2 The Messenger system will provide on-line help, an interactive search function that permits Users to locate all information concerning a specific transaction, and menus that permit Users to access separately each record in the Transportation log, notices of available capacity, and standards of conduct information. The Messenger system will permit Users to electronically download a file(s) from the Messenger system and to split files into smaller files prior to such download. Panhandle will maintain and retain daily backup records of the information displayed on the Messenger system for three years and permit Users to review those records, which will be archived and reasonably accessible. Information on the most recent entries will appear ahead of older information. Panhandle will periodically purge transactions from current Messenger system files when transactions have been completed. Information that has been purged from the Messenger system will be archived and will be available in hard copy for public inspection by appointment during regular business hours at Panhandle's principal place of business for at least three years thereafter and will be made available in electronic format.

FERC Docket: RP99-482-000

Second Revised Sheet No. 211 second Revised Sheet No. 211 : Superseded

Superseding: First Revised Sheet No. 211
GENERAL TERMS AND CONDITIONS
(Continued)

2.3 The Messengerb system may be used to: (a) interactively request service, execute, or amend Service Agreements, and provide nominations pursuant to the applicable rate schedule and these General Terms and Conditions; (b) exercise a Right of First Refusal as an existing Shipper pursuant to Section 7.2 of the General Terms and Conditions or submit a bid as a Potential Shipper under such section; and (c) assign Service Rights pursuant to Section 15 of the General Terms and Conditions.

- 2.4 Through the Messengerb system the User may obtain:
 - (a) Information concerning the availability of capacity for firm and interruptible Transportation and Storage services (i) at receipt points (ii) on the mainline, (iii) at delivery points, and (iv) in Storage fields, and whether the capacity is available from Panhandle directly, through exercise of a Shipper's Right of First Refusal, or through assignment from a Shipper.
 - (b) A listing of Point(s) of Receipt on Panhandle's system including the following information:
 - (i) Designation of Point of Receipt;
 - (ii) Location of Point of Receipt;
 - (iii) Operator name;
 - (iv) Availability of EGM;

or a third party.

- (v) Whether the Point of Receipt is located on the transmission or Gathering portions of Panhandle's system;
- (vi)Total firm capacity subscribed at the point; (vii)Whether the point is measured by Panhandle

FERC Docket: RP99-482-000

Third Revised Sheet No. 212 Third Revised Sheet No. 212: Superseded

Superseding: Second Revised Sheet No. 212

GENERAL TERMS AND CONDITIONS

(Continued)

- (c) The minimum price, quantity, terms, special conditions, and releasing party names for all release notices submitted pursuant to Section 15 of the General Terms and Conditions.
- (d) The price, term, and quantity offered for assignment of service rights under Section 15 of the General Terms and Conditions.
- (e) The price, term, quantity, and names of parties for all completed assignments of service rights under Section 15 of the General Terms and Conditions.
- (f) The log of data used to allocate capacity.
- (g) Panhandle's currently effective FERC Gas Tariff's General Terms and Conditions.
- (h) On a best efforts basis, information concerning operational pressures at major pipeline interconnection points.
- 2.5 Electronic data interchange will be available on a nondiscriminatory basis to any party. A mutually acceptable agreement between Panhandle and such party is required for electronic transmission of data.
- 2.6 Any provisions of this FERC Gas Tariff requiring that matters be written or in writing are satisfied by either Shipper or Panhandle utilizing electronic transmission through the Messengerb system in accordance with the procedures for utilization of the Messengerb system or electronic data interchange as provided for in the Commission approved or permitted data sets, provided, however, that notification utilizing electronic data interchange or the Messengerb system which require actions by Shipper within 48 hours or less shall also be communicated by telephone or in writing. All other provisions requiring items or information to be written or in writing remain unchanged.
- 2.7 A Messengerb system User shall utilize the Messengerb system only in a manner permitted by, and subject to the conditions of, its Messengerb Agreement. Panhandle's liability for any damages which result from any use of the Messengerb system shall be limited to the direct results of Panhandle's negligence, bad faith, fraud or willful misconduct and shall not extend to consequential, indirect or punitive damages.

FERC Docket: RS92- 22-005

Original Sheet No. 213 Original Sheet No. 213: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

3. QUALITY OF GAS

- Gas received and delivered shall be merchantable Natural Gas; shall be free of water and liquid hydrocarbons; shall contain not more than 7 pounds of water vapor per MMcf, 1.0 grain of hydrogen sulphide and 20 grains of total sulphur (reduced by sulphur caused by odorization equipment) per 100 cubic feet, 2% of carbon dioxide (by volume), and 50 parts per million of oxygen; shall not contain any active bacteria or bacterial agent, including but not limited to sulphate reducing bacteria and acid producing bacteria; shall not contain any hazardous or toxic substances; and shall not exceed 120ø Fahrenheit in temperature. The Gas shall have a total or gross heating value of not less than nine hundred fifty (950) Btu and not more than one thousand two hundred (1,200) Btu per cubic foot at the Point of Receipt. Panhandle shall not be obligated to accept Gas for Transportation or Storage which does not meet these quality provisions. Shipper shall reimburse Panhandle for any loss, cost, damage, or expense incurred by Panhandle as a direct or indirect result of Shipper's failure to comply with the provisions of this Section 3.1.
- 3.2 If the Gas offered for Transportation by Shipper fails at any time to conform to any of the specifications set forth herein, then Panhandle shall have the right, either by written, oral or telephonic notice to Shipper, to refuse to accept all or any portion of such quality deficient Gas. If Panhandle refuses to accept Gas tendered by Shipper because such Gas does not conform to the specifications set forth herein, Shipper shall not be relieved of its obligation to pay any Access, Reservation, Transmission or Gathering charges or any other applicable charge in Panhandle's tariff, Rate Schedules or Service Agreements.

FERC Docket: RS92- 22-005

Original Sheet No. 214 Original Sheet No. 214: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

3.3 Panhandle shall have the unqualified right to commingle Gas transported and/or stored hereunder with Gas from other sources. It is recognized that Gas delivered at a Point of Delivery may not be the same molecules as those received at a Point of Receipt. The Quantities of Gas delivered pursuant to a Transportation Agreement at a Point of Delivery shall be Thermally Equivalent to the Quantities of Gas received at the Point of Receipt, adjusted for Fuel Reimbursement.

FERC Docket: RS92- 22-005

Original Sheet No. 215 Original Sheet No. 215: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

4. MEASUREMENT

4.1 Quantities of Gas received or delivered hereunder at each Point of Receipt and Point of Delivery shall be determined by either of Panhandle or Shipper responsible for measurement at such point, as listed on Exhibit A of the Service Agreement.

The heating value, specific gravity and supercompressibility (Mol percent of nitrogen and carbon
 dioxide) values of the Gas shall be determined at the
 beginning of receipt and delivery and annually thereafter,
 or at such frequency as may be found acceptable in
 practice. Water content shall be determined at intervals
 as found necessary in practice. Shipper may be required
 to test, at the receipt meter, for bacteria or bacterial
 agents if Panhandle so requires. Tests for bacteria or
 bacterial agents shall be conducted on samples taken from
 the meter run or other appurtenant piping using American
 Petroleum Institute (API) test method API-RP38 or any
 other test method acceptable to Panhandle which is
 currently available or may become available at any time
 during the term of the Transportation Agreement.

4.2 Such measurement, both volumetric and thermal, shall be at a temperature of 600 Fahrenheit and at a pressure of 14.73 psia dry, and on the basis of the methods prescribed and published by the American Gas Association in conjunction with the ANSI/API 2530 Report as currently available and in accordance with any subsequent amendments thereof accepted and agreed upon between the parties if orifice meters are used. If positive displacement or turbine meters are used, measurement shall be performed in accordance with AGA-7 as currently available and with any subsequent amendments thereof accepted and agreed upon between the parties. Btu content shall be determined by taking the arithmetic average of the heating value as recorded by recording calorimeter or other method of determination generally accepted in the industry. Flowing temperature shall be determined by a recording thermometer or EGM.

FERC Docket: RS92- 22-005

Original Sheet No. 216 Original Sheet No. 216: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

4.3 All measuring stations shall be acceptable to Panhandle and Shipper and installed and operated to conform to recognized industry standards, using flange connections and, where necessary, pulsation dampening equipment. Auxiliary equipment shall be installed so as not to interfere with the measurement of either party.

- 4.4 Either Panhandle or Shipper may witness the installation or maintenance of the other's equipment. Neither Panhandle nor Shipper shall be required to verify accuracy more often than once during any 30 Day period. If, upon test, accuracy is found to be within 2% by volume (chromatograph/calorimeter within 0.5%), measurements taken by such equipment shall be assumed to have been accurate. Any error exceeding these percentages shall require correction to zero error of previous recordings back to the date error began. If the beginning date of error is unknown, correction shall extend to one-half the time since the last test, not to exceed 16 Days.
- 4.5 Where wellhead compressors have been installed, the maximum permissible Square Root Error Indicator (SREI) shall be no greater than 1% for volumes less than 2 MMcf per Day and .5% for volumes greater than 2 MMcf per Day. The SREI shall be determined by tests at the orifice taps and shall be performed on the full operating range of the compressor.

FERC Docket: RS92- 22-005

Original Sheet No. 217 Original Sheet No. 217: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

When the information necessary for billing purposes is in the control of the operator of the measuring facility and not in Panhandle's control, then the operator of that measuring facility shall furnish to Panhandle daily operational volumes of the previous twenty-four hours within a twenty-four hour period upon completion of the gas Day. In addition, the operator of the measuring facility shall furnish actual volume information to Panhandle on or before the fifth Day following each Month. If volumes are not available, an estimate shall be provided by the operator of the measuring facility to Panhandle by no later than 3:00 p.m. Central time on the fifth Day of each Month. If an estimate is not made by the operator of the measuring facility, then Panhandle shall estimate a Quantity of Gas based on the best information available. If an estimate is used for billing, Panhandle shall provide in the succeeding Month's billing, an adjustment based on any difference between actual Quantities of Gas and estimated Quantities of Gas.

FERC Docket: RP96- 51-000

First Revised Sheet No. 218 First Revised Sheet No. 218: Superseded

Superseding: Original Sheet No. 218

GENERAL TERMS AND CONDITIONS (Continued)

5. RESPONSIBILITY DURING TRANSPORTATION, STORAGE OR PARKING

As between Panhandle and Shipper, Panhandle shall be deemed to be in control and possession of the Gas transported and/or stored and/or parked hereunder only after the Gas is received at the Point of Receipt, and before it is delivered to or for the account of Shipper at the Point of Delivery. Shipper shall be deemed to be in control and possession at all other times. Whichever of Panhandle or Shipper is deemed to be in control and possession of the Gas shall be responsible for and shall indemnify the other party with respect to any losses, injuries, claims, liabilities or damages caused thereby and occurring while the Gas is in its possession.

FERC Docket: RP99-506-000

Fifth Revised Sheet No. 219 Fifth Revised Sheet No. 219: Superseded

Superseding: Fourth Revised Sheet No. 219
GENERAL TERMS AND CONDITIONS
(Continued)

- 6. PROCEDURES FOR REQUESTING SERVICE RIGHTS
 - 6.1 Persons desiring Transportation, Storage or Parking service must deliver a valid executed request for service rights through the Messengerp system or in writing in the form specified by Panhandle's current service rights request form, as such may be revised from time to time. Written requests must be submitted to the following address:

Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-4272 or 1-800-275-7375 Facsimile: (713) 989-1178

- 6.2 Acceptance of a request for service rights and the continuation of service is contingent upon the Shipper satisfying, on an on-going basis, a credit appraisal by Panhandle. Panhandle shall apply consistent evaluation practices to determine the acceptability of Shipper's overall financial condition. Such credit appraisal shall be based upon the following information and criteria:
 - (a) Shipper shall provide current financial statements, annual reports, 10-K reports or other filings with regulatory agencies which discuss the Shipper's financial status, a list of all corporate affiliates, parent companies and subsidiaries, and any reports from credit reporting and bond rating agencies which are available. Panhandle shall determine the acceptability of Shipper's overall financial condition, working capital, and profitability trends.
 - (b) Shipper shall provide a bank reference and at least two trade references. The results of reference checks and any credit reports submitted in 6.2(a) must show that Shipper's obligations are being paid on a reasonably prompt basis.

FERC Docket: RS92- 22-005

Original Sheet No. 220 Original Sheet No. 220: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(c) Shipper shall confirm in writing that it is not operating under any chapter of the bankruptcy laws and is not subject to litigation or debt reduction procedures under state laws, such as an assignment for the benefit of creditors, or any informal creditors' committee agreement. An exception can be made for a Shipper who is a debtor in possession operating under Chapter XI of the Federal Bankruptcy Act but only with adequate assurance that the service billing shall be paid promptly as a cost of administration under the Federal Court's jurisdiction.

(d) Shipper shall confirm in writing that it is not subject to any uncertainty in any pending liquidation or regulatory proceedings in state or Federal courts or is not aware of any change in business conditions which could cause a substantial deterioration in its financial condition, a condition of insolvency or the inability to exist as an on-going business entity.

(e) If Shipper has an on-going business relationship with Panhandle, no delinquent balances should be outstanding for services provided previously by Panhandle and Shipper must have paid its account balances according to the established terms and not made deductions or withheld payment for claims not authorized by contract.

- (f) Shipper shall confirm in writing that no significant collection lawsuits or judgments are outstanding which would seriously reflect upon the business entity's ability to remain solvent.
- 6.3 Upon notification by Panhandle that Shipper has failed to satisfy the credit criteria, or subsequently during the term of the Service Agreement no longer satisfies the credit criteria such Shipper may still obtain credit approval by Panhandle if it elects to provide one of the following: (a) an advance deposit; (b) a Standby

FERC Docket: RP99-506-000

First Revised Sheet No. 221 First Revised Sheet No. 221: Superseded

Superseding: Original Sheet No. 221
GENERAL TERMS AND CONDITIONS
(Continued)

Irrevocable Letter of Credit; (c) security interest in collateral found to be satisfactory to Panhandle; (d) a guaranty, acceptable to Panhandle, by another person or entity which satisfies credit appraisal. An advance deposit, Standby Irrevocable Letter of Credit, or security interest should at all times equal the three highest Months of estimated usage during the term of the Service Agreement.

- 6.4 If Shipper's credit standing ceases to meet Panhandle's credit requirements during the period of service, then Panhandle has the right to require security or prepayment as specified in 6.3 above. If security or prepayment is not tendered in a time period as reasonably determined by Panhandle, then Panhandle is not required to continue service.
- 6.5 If Shipper is unable to maintain credit approval, the executed Service Agreement shall terminate as of the first Day of the Month following written notice to Shipper.
- 6.6 In the event a Shipper or potential Shipper may have a complaint, the Shipper or potential Shipper shall:
 - (a) Provide Panhandle a written or oral description of the complaint, including the identification of the request for service rights, if applicable, by contacting Panhandle at the following:

Panhandle Eastern Pipe Line Company Attn: Customer Services P. O. Box 4967 Houston, Texas 77210-4967 Phone: (713) 627-4272 or 1-800-275-7375 Facsimile: (713) 989-1178

(b) Panhandle shall respond initially within two Business Days and in writing within 30 Days advising Shipper or potential Shipper of the disposition of the complaint.

FERC Docket: RP96-214-000

Third Revised Sheet No. 222 Third Revised Sheet No. 222: Superseded

Superseding: Second Revised Sheet No. 222

GENERAL TERMS AND CONDITIONS

(Continued)

6.7 A request for service rights shall not be valid and no request date shall be assigned until Panhandle has received in writing all of the information required or requested under Sections 6.1 and 6.2 herein. If Panhandle requests additional information or assurances in accordance with this Section, and such information is provided within seven Days of such request, Shipper's request for service rights shall be deemed to have been received when the information set forth in Sections 6.1 and 6.2 was received. Otherwise, Shipper's request for service rights shall become valid and a request date assigned when such additional information or assurances are actually received in writing by Panhandle.

- 6.8 Panhandle may require such other information as is required to comply with regulatory reporting or filing requirements.
- 6.9 To be considered valid, a request for firm service rights must be accompanied by payment of the lesser of (1) all applicable reservation charges, capacity charges and deliverability charges for one Month or (2) \$10,000. Such payment must be made by Shipper to Panhandle. When service is commenced, this payment, with carrying charges computed in a manner consistent with Section 154.501(d) of the Commission's Regulations, shall be used to offset Shipper's Monthly bill. If the request is (1) withdrawn prior to the date Panhandle tenders a Service Agreement, (2) becomes void in accordance with Section 6.11, or

FERC Docket: MT94- 19-000

First Revised Sheet No. 223 First Revised Sheet No. 223 : Superseded

Superseding: Original Sheet No. 223
GENERAL TERMS AND CONDITIONS
(Continued)

(3) Panhandle declines to tender a Service Agreement, this payment shall be refunded with carrying charges. If Shipper's request expires, as set forth in Section 7, Shipper's payment shall be non-refundable.

- 6.10 If any information provided by Shipper pursuant to this Section 6 changes, Shipper shall be obligated to provide prompt written notice of such changes to Panhandle.
- 6.11 If Panhandle tenders a Service Agreement within ninety (90) Days of the date Shipper submits a valid request for service, Shipper shall have thirty (30) Days from the date the Service Agreement is sent to execute the Service Agreement, as set forth in Section 7.1 of these General Terms and Conditions. If Panhandle does not tender a Service Agreement within ninety (90) Days, Shipper's request shall be deemed to be null and void and of no further effect without prejudice to the Shipper's rights to submit a new request for service rights, and Shipper's payment pursuant to Section 6.9 shall be refunded with carrying charges.
- 6.12 Requests for Transportation or Storage Service rights must be submitted in the form specified in this Section 6.12.

First Revised Sheet No. 224 First Revised Sheet No. 224: Superseded Superseding: Original Sheet No. 224 GENERAL TERMS AND CONDITIONS (Continued) SERVICE RIGHTS REQUEST FORM FOR PANHANDLE EASTERN PIPE LINE COMPANY (PANHANDLE) TRANSPORTATION AND STORAGE SERVICE Dat.e Customer (Shipper) Information 1. Shipper Name: State of Incorporation (if applicable): Or other legal description (e.g., partnership): 2.Shipper is a(n): (check one) T₁DC Intrastate Marketer Interstate End User Producer Other Describe 3. Representative(s) authorized to sign affidavit under Section 9.4 of the General Terms and Conditions: Name Title __ Pipeline and Service Information 4. Type of Service: (check one) Transportation: FT EFT SCT
Storage: IOS IIOS WS
Other: GDS TBS SCT IT EIT PS For FT, EFT or SCT, the service is Storage Related in accordance with Section 12.15: Yes ____ No _ If Yes: Market Area Storage Related ____ Field Area Storage Related For WS, IWS or FS, service to be provided from: Market Area Storage Facilities ___ Field Area Storage Facilities

Effective Date: 04/01/1994 Status: Effective

FERC Docket: RP94-160-000

Effective Date: 10/01/1994 Status: Effective FERC Docket: MT94- 19-000 Second Revised Sheet No. 225 Second Revised Sheet No. 225 : Superseded Superseding: First Revised Sheet No. 225 GENERAL TERMS AND CONDITIONS (Continued) 5. Regulation: 284B (Section 311, on behalf of LDC or Intrastate) 284G (Open Access pursuant to Panhandle's Certificate issued in Docket No. CP86-585) 6. Requested Term of Service: _ Initial Delivery Date: Requested Termination Date: ___ Requestor Information 7. Name Title Telephone : All information provided in this Service Rights Request Form is true, accurate and complete to the best of $\ensuremath{\mathsf{my}}$ knowledge. Signature* : Title * See Remarks following No. 21. Contract Volumes 8. Maximum Daily Contract Quantity (MDCQ) Dt./Day (For FT, EFT, SCT, IT or EIT) * Maximum Daily Contract Quantity (MDCQ) for Dt./Day Summer Period * Maximum Daily Contract Quantity (MDCQ) for Dt./Day Winter Period ** Maximum Daily Injection Quantity (MDIQ) Dt./Day ** Maximum Daily Withdrawal Quantity (MDWQ) Dt./Day

- * Available only for FT, EFT or SCT Transportation used to deliver Gas to Storage or receive Gas from Storage in accordance with Section 12.15
- ** For IOS, IIOS, WS, PS, IWS or FS

** Maximum Stored Quantity (MSQ)

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Original Sheet No. 226 <code>Original Sheet No. 226 : Superseded</code>

GENERAL TERMS AND CONDITIONS (Continued)

Rat	e Informat	ion						
9.	Does Requ		ee to			able maximum	m rate(s) for
		equested,		Reques		agree to po Less than t		
	Field Zon	e Charge			Delivera	ability Cha	rge	
	Market Zo: Access				Capacity	/ Charge		
	Mileage (per	Charge 100 miles)			Injectio	on Charge		 -
	Gathering	Charge			Withdrav	wal Charge		
On I	Behalf Of	Party:						
10.	If Shipper requests service under Part 284, Subpart B (NGPA Section 311 service), the following information concerning the party on whose behalf the Transportation shall be performed (hereinafter called the On Behalf Of party) must be provided with the request:							
	(a) The	exact leg	gal na	ame of	the On Be	ehalf Of pa	rty:	
	(b) The	On Behali	Of]	party's	address	(if other	than Sh	ipper):
	(c) Is	the On Beh A Local I An Intras)istr	ibution	Company			

FERC Docket: RS92- 22-005

Original Sheet No. 227 Original Sheet No. 227: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(d) Certification by the On Behalf Of party that the Transportation to be provided shall be on its behalf and specifications that the local distribution company or intrastate pipeline company shall (i) at some point have physical custody and transport the Gas to be transported hereunder or (ii) hold title to the Gas for a purpose related to the On Behalf Of party's status and function as an intrastate pipeline or local distribution company.

Agency Information

- 11. If Shipper is acting as agent for another entity, the following information concerning the principal must be provided with the request:
 - (a) The exact legal name of the principal:
 - (b) The principal's address:

Address Information

12. Address For Notices (Please include street address for express mail service.):

Company :
Party Code: (If known) Address Seq No. (If known)
Contact :
Title :
Address :
Address :
City : State: Zip Code
Telephone : () Facsimile: ()

Effective Date: 10/01/1994 Status: Effective FERC Docket: MT94- 19-000 First Revised Sheet No. 228 First Revised Sheet No. 228 : Superseded Superseding: Original Sheet No. 228 GENERAL TERMS AND CONDITIONS (Continued) 13. Address For Nominations (Person(s) designated by Shipper to provide scheduling information and volume nominations.): Company (If known) Address Seq No. (If known) Party Code: Contact : Title Address Address City : State: Zip Code Telephone :() Facsimile: () Outside Normal Business Hours Title : Address Address State: Zip Code Facsimile: () City Telephone : () 14. Address For Billing: Company : Party Code: (If known) Address Seq No. (If known) Contact : Title Address Address : City : State: Zip Code Telephone :() Facsimile: ()

FERC Docket: N	d Sheet No. 229 Fi Original Sheet No. GENERAL TE	rst Revised		: Superseded
DELIVERY POI	NT INFORMATION			
expedite pro	The completeness cessing of your runder Rate Schedu Device.)	equest. (Pri	mary and Secon	dary Point(s)
15.(a). Pr	imary Point(s) of C (If applicable)	Delivery:		or
	Meter Name State		Maximum	Daily
			(Firm	Only)
	ndary Point(s) of (If applicable):_			_ or
	Meter Name		County	
RECEIPT POIN	T INFORMATION (Ch	eck all that	apply)	
	The completeness cessing of your re		of this inform	ation will
	Secondary Points Currently effecti secondary points) Requesting Primar (answer #17 also. Requesting propose connection). (an	<pre>ve Master Rec vy Point(s) of) ed Point(s) c</pre>	Receipt.	

FERC Docket: MT94- 19-000

First Revised Sheet No. 230 First Revised Sheet No. 230 : Superseded Superseding: Original Sheet No. 230

GENERAL TERMS AND CONDITIONS (Continued)

17.	(a) Primary	y Point	(s) of	Receipt	:		
No.	Meter No.	Meter Name		State	Cou	inty	Maximum Daily Contract Quantity (Firm Only)
1. 2. 3. 4. 5.							
	(b) Second	dary Po	int(s)	of Rece	ipt:		
No.	Meter No.	Meter Name		State	Cou	inty	
1. 2. 3. 4. 5.							
18.	Proposed Po Proposed Po State: Onshore/Of:	oint Nar	me: Cou	nty:			Range:
	Ti			()			
	Billing Cor	ntact:	Title: Addres Phone:	s:	Sta	ıte:	Zip:
FERC		Origin a	and Con				ates in which Gas ly consumed (C).)
State	е		gin or umption		State		igin or nsumption

FERC Docket: MT94- 19-000

First Revised Sheet No. 231 First Revised Sheet No. 231 : Superseded

Superseding: Original Sheet No. 231

GENERAL TERMS AND CONDITIONS (Continued)

20. Name(s) of Upstream (U) and/or Downstream (D) Pipeline(s):

Party Code	Upstre	eam/ Contr	act
(if known)	Pipeline	Downstream	Date*
	-		
			
			

^{*}Indicate the date of contract with any interstate pipeline company when requesting Transportation under Section 284G.

21. Please send Transportation/Storage Request Form-New Service to:

Panhandle Eastern Pipe Line Company Attention: Marketing Services P.O. Box 1642 Houston, Texas 77251-1642

OR Facsimile: (713) 627-4829

Remarks - If Service Agreement is executed on behalf of the Shipper by a person other than an Officer of the Company, documentation confirming that person's authorization to execute contracts must be provided at the time the Shipper returns Service Agreement.

If Shipper has not previously completed a Credit Application, a Credit Application must be submitted.

FERC Docket: RP97-185-000

Second Revised Sheet No. 232 Second Revised Sheet No. 232 : Superseded

Superseding: First Revised Sheet No. 232
GENERAL TERMS AND CONDITIONS
(Continued)

7. CONTRACTING FOR SERVICE RIGHTS

This Section 7 governs the order in which requests for Transportation or Storage service rights shall be accommodated where capacity is not available for all valid requests. It does not govern scheduling, which is governed by Section 8 herein.

7.1 Order For Accommodating Firm Service Rights

- Subject to the provisions of Section 7.2 hereof, as firm capacity becomes available for (1) Transportation, (2) Storage from Field Area Storage Facilities under Rate Schedule IOS, WS or FS, or (3) Storage from Market Area Storage Facilities under Rate Schedule WS, PS or FS, requests for firm services shall be fulfilled first, commencing with the requested service offering the greatest economic value for the capacity available. If the economic values, as determined below, of separate service requests are equal, then service shall be offered in sequence starting with the request for such service with the earliest date. If separate service requests have equal economic values and the same date of request, service shall be offered to such requests on a pro-rata basis.
- (b) The economic value of request(s) for service rights shall be the net present value of the reservation charges, capacity charges and deliverability charges that Shipper would pay at the rates Shipper has requested, which shall not be less than the Minimum Rate nor greater than the Maximum Rates as stated on the currently effective Tariff Sheet governing such service, over the term of service specified in the request(s), utilizing as the annual discount factor the overall rate of return underlying Panhandle's currently effective rates. For purposes of determining the economic value of request(s) for service rights under Rate Schedule SCT, the net present value will be determined using the applicable reservation rate per Dt. under Rate Schedule EFT at a load factor of 100%. Shippers requesting service at a Negotiated Rate which exceeds the Maximum Rate will be considered to be paying the Maximum Rate for purposes of determining the economic value of request(s) for service rights.

FERC Docket: RS92- 22-005

Original Sheet No. 233 Original Sheet No. 233: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(c)Within thirty (30) Days of the date firm capacity becomes available on a particular segment of Panhandle's system, Panhandle shall tender Service Agreements to potential Shipper(s) whose request(s) for firm service can be accommodated through the capacity that becomes available in accordance with the order established under Section 7.1(a).

If Panhandle tenders a Service Agreement, but does not receive the executed Service Agreement within 30 Days from the date the Service Agreement is sent, the request for service rights shall expire without prejudice to the prospective Shipper's right to submit a new request for service rights. In the event the prospective Shipper returns a Service Agreement with a rate less than that used to determine the economic value of the request or with a term less than that used to determine the economic value of the request or both, Panhandle shall redetermine the economic value of the Shipper's request in accordance with Section 7.1(b) hereof. If the request continues to qualify for service, and Panhandle agrees to the rate, Panhandle shall execute the Service Agreement. If the request does not continue to qualify for service, the Service Agreement shall be deemed null and void, and the prospective Shipper's request shall be modified in accordance with the terms proposed by the prospective Shipper.

- 7.2 Continuation of Long-Term Agreements for Firm Service; Right of First Refusal
 - (a) Panhandle shall continue to provide service pursuant to a Long-Term Agreement beyond the term specified in such Agreement if:
 - (i) The Long-Term Agreement is extended according to its terms, or

Effective Date: 08/01/2000 Status: Effective

FERC Docket: RP00-375-000

Second Revised Sheet No. 234 Second Revised Sheet No. 234: Superseded

Superseding: First Revised Sheet No. 234

GENERAL TERMS AND CONDITIONS

(Continued)

If the Long-Term Agreement meets the (ii) eligibility standards below and Shipper agrees to match the bid having the greatest economic value, as defined in Section 7.1(b) hereof, for the firm service Shipper receives, or any portion thereof, which Panhandle has been offered in accordance with Section 7.2(b) and executes a new Service Agreement matching those terms. A Long-Term Agreement for firm service is eligible for the right of first refusal provisions in this Section 7.2 if the Long-Term Agreement (1) is in effect prior to August 1, 2000 or (2) commences on or after August 1, 2000 and the Long-Term Agreement provides for twelve (12) or more Months of consecutive service at the Maximum Rate applicable to the service or, if the service is not available for twelve (12) consecutive Months, the Long-Term Agreement provides for service for more than one year at the Maximum Rate applicable to the service.

- (b) The procedures specified in this Section 7.2(b) shall be followed to determine whether a Long-Term Agreement will be continued in accordance with Section 7.2(a)(ii).
 - (i) Shipper must give timely notice that it wants to continue service beyond the term of the Agreement. For the notice to be timely, Shipper must notify Panhandle within the following periods:

2 years or longer Not earlier than 18 Months and

not later than 12 Months

Less than 2 years $$\operatorname{\textsc{Not}}$$ earlier than 9 Months and

 $\hbox{not later than 6 Months}\\$

Panhandle shall notify each Shipper not earlier than 45 Days and not later than 30 Days prior to the last Day Shipper can give timely notice hereunder.

Effective Date: 08/01/2000 Status: Effective

FERC Docket: RP00-375-000

Original Sheet No. 234A <code>Original Sheet No. 234A</code> : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(ii) Upon receipt of the Shipper's notice, Panhandle shall post on the Messenger system for a period of 45 Days ("Posting Period") the Maximum Daily Contract Quantity under the Shipper's Long-Term Agreement and the primary Point(s) of Receipt and Point(s) of Delivery thereunder.

FERC Docket: RP97-481-000

Fifth Revised Sheet No. 235 Fifth Revised Sheet No. 235: Superseded

Superseding: Fourth Revised Sheet No. 235
GENERAL TERMS AND CONDITIONS
(Continued)

(iii) During the Posting Period Panhandle shall accept requests for all or a portion of the Shipper's service rights under the Long-Term Agreement from any prospective Shipper that has submitted a valid request for service rights in accordance with the provisions of Section 6.1 hereof.

If, during the Posting Period, Panhandle (iv) receives an acceptable offer for all or a portion of the service rights under Shipper's Long-Term Agreement, Panhandle shall notify Shipper in writing of the offer having the greatest economic value, as defined in Section 7.1(b) hereof; provided, that for purposes of value comparisons under this section the rate utilized shall be limited to the Maximum Rate that can be charged to the existing Shipper and the longest term utilized shall be five (5) years. If Shipper elects to match the offer, Shipper shall notify Panhandle of such election in writing within 30 Days after receiving notice from Panhandle and shall execute a new Service Agreement matching the offer within 30 days after Panhandle has tendered the Service Agreement. If Shipper elects not to match the offer or does not execute the Service Agreement within 30 days, Panhandle will tender a Service Agreement to the prospective Shipper submitting the offer having the greatest economic value. If the Service Agreement is not executed within 30 days, the request for service rights shall expire without prejudice to the prospective Shipper's right to submit a new request for service rights. Panhandle shall then notify the Shipper in writing of the acceptable offer, if any, having the next greatest economic value in accordance with the provisions of this Section 7.2(b)(iv). there is no other acceptable offer, the Shipper may continue service in accordance with Section 7.2(b)(v).

FERC Docket: RP99-482-000

Fourth Revised Sheet No. 236 Fourth Revised Sheet No. 236: Superseded

Superseding: Third Revised Sheet No. 236
GENERAL TERMS AND CONDITIONS
(Continued)

- (V) If no acceptable offers are received, Panhandle shall so notify Shipper within 15 days after the close of the Posting Period. In such event, Shipper may continue to receive service under a new Long-Term Agreement with any term Shipper chooses, at the applicable Maximum Rate or at a rate agreed to by Panhandle and Shipper. Shipper must notify Panhandle of its intent and indicate the term of the new Long-Term Agreement within 15 days of having been notified in writing by Panhandle that no acceptable offer was received. Panhandle will then be obligated to tender the new Long-Term Agreement to Shipper within 15days of Shipper's notification. If Shipper (1) fails to provide Panhandle the term of the requested new Long-Term Agreement within the required 15 day period, or (2) does not return an executed Service Agreement reflecting such term to Panhandle within 30 days of the date such contract is tendered, then Shipper shall be deemed to have elected not to continue service under a Long-Term Agreement pursuant to the provisions of Section 7.2 herein.
- (vi) Panhandle shall post the winning bid and bidder on the Messengerp system for ninety (90) Days whether or not the bidder executes a Service Agreement.
- (c) If a Long-Term Agreement is not continued in accordance with Section 7.2(a) above, such Agreement shall terminate as specified therein.
- (d) The provisions of this Section 7.2 shall not apply to firm Transportation service initiated pursuant to the exercise of conversion rights under 18 CFR, Section 284.10, and which took effect between February 13, 1991 and May 18, 1992.
- 7.3 Panhandle shall not be obligated to tender, execute or continue a Service Agreement for service at any rate less than the Maximum Rate set forth in the effective tariff sheets applicable to the service requested. It shall be Panhandle's sole discretion to tender, execute or continue a Service Agreement at any rate less than the applicable Maximum Rate for the service requested. Panhandle and Shipper must agree to any rate requested at less than the applicable Maximum Rate before Panhandle becomes obligated to tender, execute or continue a Service Agreement for firm service at any rate less than the applicable Maximum Rate.

FERC Docket: RP96- 51-000

Third Revised Sheet No. 237 Third Revised Sheet No. 237: Superseded

Superseding: Second Revised Sheet No. 237

GENERAL TERMS AND CONDITIONS (Continued)

7.4 Once a Service Agreement for firm Transportation or Storage is executed, the economic value of the Service Agreement shall be determined in accordance with Section 7.1(b). During the term of the Service Agreement any change Shipper may make in the terms of the Service Agreement shall not reduce such economic value of the Service Agreement to Panhandle without Panhandle's written consent.

7.5 Elimination of Inactive Interruptible Service Agreements

Panhandle may terminate Shippers' Service Agreements under Rate Schedules IT, EIT, IIOS, IWS and GPS, upon thirty (30) days prior written notice to Shippers who have not submitted nominations to either transport, store Gas, or park Gas under such Service Agreements within the immediately preceding twelve (12) Months.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 238 Second Revised Sheet No. 238: Superseded

Superseding: First Revised Sheet No. 238

GENERAL TERMS AND CONDITIONS

(Continued)

3. NOMINATION AND SCHEDULING OF SERVICE

- 8.1(a) If Shipper desires the Transportation or Storage of Gas on any Day under Panhandle's firm or interruptible Rate Schedules, Shipper shall furnish to Panhandle written nominations indicating Quantities of Gas to be received and delivered or stored by Panhandle, by individual Point of Receipt and individual Point of Delivery, as required below. Such nominations shall reflect the Quantities of Gas for each Point of Receipt and each Point of Delivery. Such nominations also shall be in balance, except to the extent they reflect any previous imbalance make-up Quantities of Gas and shall reflect scheduled daily variations. Nominations for Point(s) of Receipt shall include applicable Fuel Reimbursement Quantities. Overrun Quantities may be requested either in Shipper's standard nomination or in a separate nomination.
 - (b) If, at any time Shipper's receipt nominations and receipt actuals or delivery nominations and delivery actuals for any contract are out of balance by the greater of 1,000 Dt. or 5% of the nomination or if the difference between actual receipts and actual deliveries is greater than 5% of the delivery nominations, Panhandle shall be entitled, upon notice to Shipper, to adjust Shipper's nominations to bring them into balance. If actual receipts or deliveries are reduced by a meter operator, Panhandle shall be entitled, upon notice to Shipper, to adjust Shipper's nominations for the affected contracts to bring them into balance. Panhandle's entitlement to adjust Shipper's nominations shall not relieve Shipper of the obligation to pay any balancing, scheduling or any other penalties provided for in these General Terms and Conditions.

Effective Date: 10/01/2002 Status: Effective

FERC Docket: RP02-466-000

Fifth Revised Sheet No. 239 Fifth Revised Sheet No. 239: Superseded

Superseding: Sub Fourth Revised Sheet No. 239

GENERAL TERMS AND CONDITIONS (Continued)

8.2 Panhandle will accept nominations for service as follows:

(a) Next Day Service

A new nomination for service commencing the next Day will be accepted if the nomination is submitted in writing on the Day prior to the proposed effective date.

The nomination timeline in Central Time on the Day prior to Gas flow shall be the following:

The Timely Nomination Cycle

11:30 a.m. Nomination leaves control of the Shipper

11:45 a.m. Receipt of nomination by Panhandle (including from title transfer tracking service providers)

12:00 p.m. Panhandle provides quick response for validity of data elements

3:30 p.m. Completed confirmations from upstream and downstream connected parties provided to Panhandle

4:30 p.m. Scheduled Quantities made available by Panhandle for receipt by Shipper and point operator

In addition, at the end of each Day Panhandle shall make available to Shippers information containing scheduled Quantities, including scheduled intra-day nominations and any other scheduling changes.

Shipper shall specify a begin date, time and end date for each standard nomination. Such begin date and end date shall be within the term of Shipper's Service Agreement. To the extent that Shipper desires to change its nomination for any Day, Shipper must submit a new nomination for such Day. When a nomination for a date range is received, each day within that range is considered an original nomination. When a subsequent nomination is received for one or more days within that range, the previous nomination

Effective Date: 10/01/2002 Status: Effective

FERC Docket: RP02-466-000

Third Revised Sheet No. 239A Third Revised Sheet No. 239A: Superseded

Superseding: Sub Second Revised Sheet No. 239A

GENERAL TERMS AND CONDITIONS

(Continued)

is superseded by the subsequent nomination only to the extent of the days specified. The days of the previous nomination outside the range of the subsequent nomination are unaffected. Nominations have a prospective effect only. If a Shipper fails to provide a nomination on or before the nomination deadline, Panhandle shall deem the Shipper's nomination to be zero.

Panhandle shall have the right to refuse to receive or deliver any Gas not timely and properly nominated and confirmed. Shipper shall indemnify and hold Panhandle harmless from and Panhandle shall not be liable to Shipper or any other person as a direct or indirect consequence of such refusal.

(b) Intra-day Nominations

Any nomination submitted after the deadline set forth in Section 8.2(a) above shall be an intraday nomination. An intra-day nomination shall be effective for one Day only. An intra-day nomination shall specify an effective date, time and the Daily Quantity. The interconnected parties shall agree on the hourly flows for such Day. Nominations for intra-day changes in service will be accepted subject to operator confirmation, verification, and operating conditions. Quantities shall be deemed to be flowing if the operator at the Point of Receipt has confirmed delivery to Panhandle.

The intra-day nomination timeline in Central Time on the Day prior to Gas flow shall be the following:

The Evening Nomination Cycle

6:00 p.m.	Nomination leaves control of the Shipper
6:15 p.m.	Receipt of nomination by Panhandle (including from title transfer tracking service providers)
6:30 p.m.	Panhandle provides quick response for validity of data elements
9:00 p.m.	Completed confirmations from upstream and downstream connected parties provided to Panhandle

Effective Date: 10/01/2002 Status: Effective

FERC Docket: RP02-466-000

First Revised Sheet No. 239B First Revised Sheet No. 239B : Superseded

Superseding: Sub Original Sheet No. 239B GENERAL TERMS AND CONDITIONS

(Continued)

10:00 p.m. Scheduled Quantities provided by Panhandle to affected Shippers, point operators and bumped parties (notice to bumped parties)

Scheduled Quantities resulting from an Evening Nomination that does not cause another Shipper to receive notice that it is being bumped should be effective at 9:00 a.m. on the gas Day; and when an Evening Nomination causes another Shipper to receive notice that it is being bumped, the scheduled Quantities should be effective at 9:00 a.m. on the gas Day.

The intra-day nomination timelines in Central Time on the Day of Gas flow shall be the following:

(1) The Intra-day 1 Nomination Cycle

10:00	a.m.	Nomination	leaves	control	of	the
		Shipper				

- 10:15 a.m. Receipt of nomination by Panhandle (including from title transfer tracking service providers)
- 10:30 a.m. Panhandle provides quick response for validity of data elements
- 1:00 p.m. Completed confirmations from upstream and downstream connected parties
- 2:00 p.m. Scheduled Quantities provided by Panhandle to affected Shippers, point operators and bumped parties (notice to bumped parties)
- 5:00 p.m. Intra-day 1 nomination effective
- (2) The Intra-day 2 Nomination Cycle
- 5:00 p.m. Nomination leaves control of the Shipper
- Receipt of nomination by Panhandle 5:15 p.m. (including from title transfer tracking service providers)
- 5:30 p.m. Panhandle provides quick response for validity of data elements
- 8:00 p.m. Completed confirmations from upstream and downstream connected parties

FERC Docket: RP99-482-000

First Revised Sheet No. 239C First Revised Sheet No. 239C : Superseded Superseding: Sub Original Sheet No. 239C

Superseding: Sub Original Sheet No. 239C
GENERAL TERMS AND CONDITIONS
(Continued)

9:00 p.m.Scheduled Quantities provided by Panhandle to Shipper and point operator

9:00 p.m.Intra-day 2 nomination effective

Bumping is not allowed during the Intra-day 2 Nomination Cycle.

In the event that all or part of an interruptible Shipper's scheduled Quantity is reduced as the result of a nomination by a firm Shipper in the Evening Nomination Cycle or the Intra-day 1 Nomination Cycle, such interruptible Shipper shall be notified of the bump through the Messengerp system, the Web Site and by telephone and facsimile communication.

FERC Docket: RP99-378-000

Second Revised Sheet No. 240 second Revised Sheet No. 240 : Superseded

Superseding: Sub First Revised Sheet No. 240
GENERAL TERMS AND CONDITIONS
(Continued)

- (c) Panhandle may waive any part of the notice requirements of Section 8.2(a) or Section 8.2(b), upon request, if operating conditions permit such waiver. Shipper shall notify Panhandle immediately of any unexpected changes in volumes tendered for receipt or delivery, whether or not such notice conforms to the times set out herein.
- 8.3 Shippers shall cause the Operator of each Point of Receipt and each Point of Delivery designated in any nomination under Section 8.2 to confirm all such nominations in writing prior to implementation by Panhandle.
- 8.4 At least ten Days prior to the first Day of January,
 April, July and October, Shipper shall furnish Panhandle a
 written schedule of the estimated daily Quantities of Gas
 to be scheduled for Transportation or Storage injection or
 withdrawal during the quarter beginning on such date.
- 8.5(a) Shipper shall cause Operators at their Points of Receipt and Points of Delivery to provide in writing to Panhandle a predetermined daily allocation methodology for measured Quantities of Gas (based on scheduled nominations) before the start of the Day. Parties may agree upon one of the following allocation methodologies: ranked, pro rata, percentage, swing and operator provided value. In the event the parties cannot agree on an allocation methodology, pro rata based on confirmed nominations will be used as the default method. The party responsible for custody transfer (the party performing the measurement function) must provide the allocation. Panhandle will accept such allocation if operationally or administratively feasible.

FERC Docket: RP97- 4-002

Third Revised Sheet No. 241 Third Revised Sheet No. 241: Superseded

Superseding: Second Revised Sheet No. 241

GENERAL TERMS AND CONDITIONS (Continued)

- (b) Changes to the daily allocation methodology may be made during the Month. Such changes must be submitted before the start of the Day. No retroactive reallocations of any transactions shall be permitted.
- (c) If the daily allocation methodology at a Point of Delivery identifies a Service Agreement, other than the meter holder's Service Agreement to be allocated the last Quantities through the meter, the Shipper whose Service Agreement has been identified must agree to the allocation methodology before it may become effective. Quantities delivered under Rate Schedule GDS must be allocated as the last Quantities through the meter.
- 8.6 Shipper may change the person(s) designated in the service rights request form to provide the nomination information set forth in this section to Panhandle. Such change shall be effective immediately upon receipt by Panhandle of Shipper's written notification. If Shipper designates another person to provide this information, Panhandle shall be entitled to rely on the nominations provided by Shipper's prior designee until such nominations are changed in accordance with this Section 8.
- 8.7 Nominations shall also specify such additional information which may be needed for Panhandle to accept Commission approved or permitted standard data elements necessary to perform service.
- 8.8 Quantities of Gas nominated for Transportation or Storage by Shippers pursuant to Section 8.3 shall be scheduled by Panhandle for receipt and delivery in the following order:

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 242 Fifth Revised Sheet No. 242: Superseded

Superseding: Fourth Revised Sheet No. 242

GENERAL TERMS AND CONDITIONS (Continued)

(a) Scheduling of Receipts

- (1) Firm service from primary Point(s) of Receipt, adjusted for Fuel Reimbursement.
- (2) Firm service from secondary Point(s) of Receipt within the Primary Path, adjusted for Fuel Reimbursement, beginning with service charged the applicable Maximum Rate followed by firm service from secondary Point(s) of Receipt within the Primary Path charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Within each category, Quantities of Gas will be scheduled pro-rata.
- (3) Firm service from secondary Point(s) of Receipt outside the Primary Path, adjusted for Fuel Reimbursement, beginning with service charged the applicable Maximum Rate followed by firm service from secondary Point(s) of Receipt outside the Primary Path charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Within each category, Quantities of Gas will be scheduled pro-rata.
- (4) Interruptible service from Point(s) of Receipt, beginning with interruptible service charged the applicable Maximum Rate, followed by interruptible service from Point(s) of Receipt charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Interruptible service provided under Rate Schedule EIT will be scheduled before interruptible service provided under Rate Schedule IT charged a rate equally proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Authorized Quantities in excess of the MDCQ shall be scheduled as interruptible. Within each category of interruptible service, Transportation shall be scheduled in sequence, starting with the service with the earliest request date.
- (5) Gas Parking Service.

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 243 Fifth Revised Sheet No. 243: Superseded

Superseding: Fourth Revised Sheet No. 243

GENERAL TERMS AND CONDITIONS (Continued)

(6) Within each service category provided in (1) through (4) above, a Shipper may, through electronic data interchange, provide a ranking of individual Points of Receipt within a Shipper's Service Agreement in the event the entirety of the Shipper nomination cannot be

(b) Scheduling of Deliveries

scheduled.

- (1) Firm service to primary Point(s) of Delivery.
- (2) Firm service to secondary Point(s) of Delivery within the Primary Path beginning with service charged the applicable Maximum Rate followed by firm service to secondary Point(s) of Delivery within the Primary Path charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Within each category, Quantities of Gas will be scheduled pro-rata.
- (3) Firm Service to secondary Point(s) of Delivery outside the Primary Path beginning with service charged the applicable Maximum Rate followed by firm service to secondary Point(s) of Delivery outside the Primary Path charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Within each category, Quantities of Gas will be scheduled pro-rata.
- (4) Interruptible service to Point(s) of Delivery beginning with interruptible service charged the applicable Maximum Rate, followed by interruptible service charged a rate less than the applicable Maximum Rate in sequence starting with the rate most proximate to the Maximum Rate (expressed as a percentage of the Maximum Rate). Interruptible service provided under Rate Schedule EIT shall be scheduled before interruptible service provided under Rate Schedule IT charged a rate equally proximate to the maximum rate (expressed as a percentage of the maximum rate). Authorized Quantities in excess of the MDCQ shall be scheduled as interruptible. Within each category of interruptible service, service shall be scheduled in sequence starting with the service with the earliest request date.

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 244 Fifth Revised Sheet No. 244: Superseded

Superseding: Sub Fourth Revised Sheet No. 244

GENERAL TERMS AND CONDITIONS
(Continued)

(5) Gas Parking Service.

- (6) Within each service category provided in (1) through (4) above, a Shipper may, through electronic data interchange, provide a ranking of individual Points of Delivery within a Shipper's Service Agreement in the event the entirety of the Shipper nomination cannot be scheduled.
- (c) If Panhandle is unable to schedule all Quantities nominated because of a point of constraint, other than at a Point of Receipt or Delivery, Panhandle will schedule Gas through such point of constraint in accordance with the location of the point of constraint. If the point of constraint is located at or west of Panhandle's Haven, Kansas Compressor Station, Panhandle will schedule firm service from primary Point(s) of Receipt and from secondary Point(s) of Receipt equally if the point of constraint is within the Primary Path of the Transportation service using the secondary Point(s) of Receipt, and service from other Point(s) of Receipt in the order specified in Section 8.8(a)(3) through (5). If the point of constraint is located east of Panhandle's Haven, Kansas Compressor Station, Panhandle will schedule firm service from primary Point(s) of Delivery and from secondary Point(s) of Delivery equally if the point of constraint is within the Primary Path of the Transportation service using the secondary Point(s) of Delivery, and service to other Point(s) of Delivery in the order specified in Section 8.8(b)(3) through (5).
- (d) Subject to Section 8.2(b), scheduled service only can be displaced by a shipper moving to its primary point(s) of receipt or delivery or by an event of Force Majeure.
- 8.9 Section 9.7 describes scheduling provisions during curtailment.

FERC Docket: RP96-348-003

Sub Second Revised Sheet No. 245 Sub Second Revised Sheet No. 245: Superseded

Superseding: Substitute First Revised Sheet No. 245 GENERAL TERMS AND CONDITIONS

(Continued)

9. CURTAILMENT AND INTERRUPTION

- 9.1 Panhandle shall have the unqualified right to interrupt Transportation Services or Storage Services or Gas Parking Service or all such services at any time under Panhandle's interruptible Rate Schedules to provide service under Panhandle's firm Rate Schedules. Gas Parking Service will be interrupted first and then interruptible services shall be interrupted in sequence, starting with interruptible service charged the rate least proximate to the applicable maximum rate. Within each rate category of interruptible service, service shall be interrupted in sequence, starting with the service most recently requested and, if necessary, pro-rata by request date.
- 9.2 Panhandle shall have the right to curtail or discontinue Transportation Services or Storage Services or Gas Parking Service or all such services, in whole or in part on all or a portion of its system at any time for reasons of Force Majeure or when in Panhandle's sole judgment, capacity or operating conditions so require or it is desirable or necessary to make modifications, repairs or $% \left(\left(1\right) \right) =\left(1\right) \left(\left(1\right) \right) \left(1\right) \left(1\right$ operating changes to its system. Panhandle shall provide Shipper such notice of the curtailment as is reasonable under the circumstances.
- 9.3 In the event of curtailment pursuant to Section 9.2 above, Gas Parking Service shall be curtailed first and then interruptible services shall be curtailed in sequence, starting with interruptible service charged the rate least proximate to the applicable Maximum Rate. Within each rate category of interruptible service, service shall be curtailed in sequence, starting with the service most recently requested and, if necessary, pro-rata by request date. For any condition requiring curtailment at or west of Panhandle's Haven, Kansas Compressor Station, Panhandle shall curtail firm services in the following order (1) service from secondary Point(s) of Receipt outside the Primary Path, (2) service from secondary Point(s) of Receipt within the Primary Path and (3) service from Primary Point(s) of Receipt. For any condition requiring curtailment east of Panhandle's Haven, Kansas Compressor Station, Panhandle shall curtail firm services in the following order (1) service from secondary Point(s) of Delivery outside the Primary Path, (2) service from secondary Point(s) of Delivery within the Primary Path and (3) service from Primary Point(s) of Delivery. Within each category of firm service curtailment shall be on a pro-rata basis based on scheduled Quantities of Gas regardless of rates or request date.

FERC Docket: RS92- 22-005

Original Sheet No. 246 Original Sheet No. 246: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

- (b) For any condition requiring curtailment or interruption which affects portions of Panhandle's system downstream of the Haven, Kansas compressor station, firm service shall be curtailed starting with (i) firm service utilizing a secondary Point of Delivery then (ii) firm service utilizing a primary Point of Delivery. Within each of these categories, service shall be curtailed pro-rata.
- 9.4 When in Panhandle's judgment Panhandle will be unable to satisfy all scheduled deliveries on any portion of its system due to the failure of Shippers to deliver or cause to be delivered scheduled receipts, Panhandle may reduce, in whole or in part, deliveries to Shippers with receipts scheduled on any portion of Panhandle's system showing deficient receipts, in the order specified in Section 9.3 hereof, subject to the following conditions:
 - (a) Notice, pursuant to Section 9.6 hereof, shall be given to the affected Shippers that deliveries must be reduced within a time that is reasonable under the then existing conditions.
 - (b) Shippers who are able to confirm their receipts to Panhandle within the time set out in such notice shall be permitted to maintain their service. Confirmation shall mean that Panhandle is able, in good faith, to determine that the Shipper's Natural Gas is being received, or Shipper has provided a notarized affidavit by an officer or representative empowered to bind Shipper that, based on personal knowledge formed after diligent investigation, the scheduled quantities (or stated lesser quantities) are being delivered to Panhandle at the Point(s) of Receipt nominated by Shipper for the applicable Service Agreement(s).
 - (c) If Shipper is unable to provide confirmation within the time set out in the notice, deliveries to Shipper shall be resumed when such confirmation is provided or when the curtailment ceases.

Effective Date: 03/01/1997 Status: Suspended

FERC Docket: RP96-348-000

First Revised Sheet No. 246 First Revised Sheet No. 246: Suspended

Superseding: Original Sheet No. 246
GENERAL TERMS AND CONDITIONS
(Continued)

When in Panhandle's judgment Panhandle will be unable to satisfy all scheduled deliveries on any portion of its system due to the failure of Shippers to deliver or cause to be delivered scheduled receipts, Panhandle may reduce, in whole or in part, deliveries to Shippers with receipts scheduled on any portion of Panhandle's system showing deficient receipts, in the order specified in Section 9.3 hereof, subject to the following conditions:

- (a) Notice, pursuant to Section 9.6 hereof, shall be given to the affected Shippers that deliveries must be reduced within a time that is reasonable under the then existing conditions.
- (b) Shippers who are able to confirm their receipts to Panhandle within the time set out in such notice shall be permitted to maintain their service. Confirmation shall mean that Panhandle is able, in good faith, to determine that the Shipper's Natural Gas is being received, or Shipper has provided a notarized affidavit by an officer or representative empowered to bind Shipper that, based on personal knowledge formed after diligent investigation, the scheduled quantities (or stated lesser quantities) are being delivered to Panhandle at the Point(s) of Receipt nominated by Shipper for the applicable Service Agreement(s).
- (c) If Shipper is unable to provide confirmation within the time set out in the notice, deliveries to Shipper shall be resumed when such confirmation is provided or when the curtailment ceases.

FERC Docket: RS92- 22-005

Original Sheet No. 247 Original Sheet No. 247: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(d) If Shipper's affidavit is in fact false, then in addition to any remedies available to Panhandle at law, Shipper shall be deemed in violation of Panhandle's curtailment order and the provisions of Section 9.5 shall apply.

9.5 Curtailment Compliance

- (a) Without regard to any other remedy provided by law or by the provisions hereof, Panhandle shall be entitled to seek an order from the Commission or any other appropriate tribunal requiring compliance with curtailment or interruption ordered by Panhandle in compliance with this Section 9 or any directive from any governmental authority having jurisdiction in the premises.
 - (b) All volumes received and/or taken in violation of Panhandle's curtailment or interruption orders shall constitute unauthorized receipts or deliveries of Gas for which a charge of \$10.00 per Dt. or two times the Mid-Continent Spot Price calculated in accordance with Section 12.11(c), whichever is greater, shall be assessed in addition to any other applicable rate, charge or penalty. Such charge shall be applicable to all such unauthorized receipts and deliveries following notification of curtailment or interruption pursuant to Section 9.6, below, which may be given by telephone, confirmed by facsimile transmission or any other reasonable means.
- 9.6 Situation Reports, Notices and Indemnification
 - (a) Panhandle shall provide Shipper with notice of curtailment or interruption at a time and in a manner that is reasonable under then existing conditions, and shall in any event confirm in writing or by facsimile transmission the notice given if originally provided telephonically.

FERC Docket: RP97- 4-002

First Revised Sheet No. 248 First Revised Sheet No. 248: Superseded

Superseding: Original Sheet No. 248
GENERAL TERMS AND CONDITIONS
(Continued)

(b) Shipper shall have the responsibility to inform its end-users, suppliers, other transporters and all others involved in the transaction, as to any curtailment or interruption.

(c) Shipper shall indemnify Panhandle against and hold Panhandle harmless from any and all damages, claims, suits, actions or proceedings whatsoever threatened or initiated as a result of any curtailment or interruption invoked by Panhandle; which shall include any curtailment or interruptions described in any part of this Section 9. Shipper shall not be required to indemnify Panhandle as stated above to the extent that the curtailment or interruption is a result of Panhandle's negligence, bad faith, fault or willful misconduct.

9.7 Scheduling During Curtailment

Mid-Month nominations may be accepted at any time to allow for the reduction of Quantities of Gas flowing within an area affected by curtailment. Mid-Month nominations that would disadvantage existing scheduled firm Quantities of Gas through a constrained area will not be accepted for the remainder of the Month while curtailment remains in effect.

When curtailment continues into the following Month, the scheduling provisions in Section 8.8(a) and (b) herein shall be performed to determine the Quantities of Gas that shall flow in the constrained area for the new Month.

FERC Docket: RP00-395-003

First Revised Sheet No. 249 First Revised Sheet No. 249: Superseded

Superseding: Second Sub Original Sheet No. 249

GENERAL TERMS AND CONDITIONS (Continued)

10. POINTS OF RECEIPT

- 10.1 A Shipper may designate any Point of Receipt for Gas to be transported or stored under a firm Rate Schedule. Each Point of Receipt shall be identified in the Service Agreement as either a primary or secondary firm Point of Receipt. The sum of the Quantities of Gas from all primary firm Points of Receipt, adjusted for Fuel Reimbursement, specified in the Service Agreement may not exceed the firm Maximum Daily Contract Quantity stated in the Service Agreement. Unless the Shipper specifies otherwise, the Quantity of Gas assigned to each secondary firm Point of Receipt the Shipper designates in a rate zone or mileage segment of the Market Zone shall equal the sum of the Quantities of Gas assigned to the primary firm Points of Receipt in the same rate zone or, in the Market Zone, the same mileage segment of the Market Zone and upstream thereof. In addition, a Shipper may segment, to the extent operationally feasible, its Primary Path into two (2) or more discrete segments for its own use or in connection with an assignment pursuant to Section 15 of these
- 10.2 A Shipper may nominate, subject to the provisions of Sections 4, 8, 9 and 12 herein, Quantities of Gas to be transported under a firm Rate Schedule from any Point of Receipt identified in the Service Agreement up to the Quantity of Gas assigned to such Point of Receipt in the Service Agreement.

General Terms and Conditions.

- (a) If the Shipper's Service Agreement has not been assigned pursuant to Section 15 of these General Terms and Conditions, the sum of the Quantities of Gas nominated by the Shipper from any secondary firm Point(s) of Receipt, together with the sum of the Quantities of Gas nominated from a primary firm Point(s) of Receipt, adjusted for Fuel Reimbursement, shall not exceed the Maximum Daily Contract Quantity specified in the Shipper's Service Agreement; provided, however, if the Shipper utilizes two (2) or more discrete pipeline segments, the sum of the Quantities of Gas nominated by the Shipper from all Point(s) of Receipt may exceed the Maximum Daily Contract Quantity specified in the Service Agreement so long as the Ouantities nominated for transportation in a pipeline segment do not exceed the Maximum Daily Contract Quantity applicable to such segment.
- (b) If the Shipper's Service Agreement has been assigned in part to a Replacement Shipper pursuant to Section 15 of these General Terms and Conditions:

FERC Docket: RP00-395-003

First Revised Sheet No. 250 First Revised Sheet No. 250 : Superseded

Superseding: Second Sub Original Sheet No. 250 GENERAL TERMS AND CONDITIONS

(Continued)

- (i) the sum of the Quantities of Gas nominated by the Replacement Shipper from any secondary firm Point(s) of Receipt, together with the sum of the Quantities of Gas nominated from primary firm Point(s) of Receipt, adjusted for Fuel Reimbursement, shall not exceed the Maximum Daily Contract Quantity in the Replacement Shipper's Service Agreement; provided, however, if the Replacement Shipper utilizes two (2) or more discrete pipeline segments, the sum of the Quantities of Gas nominated by the Replacement Shipper from all Point(s) of Receipt may exceed the Maximum Daily Contract Quantity specified in the Service Agreement so long as the Quantities nominated for transportation in a pipeline segment do not exceed the Maximum Daily Contract Quantity applicable to such segment; and
- (ii) the sum of the Quantities of Gas nominated by the Releasing Shipper from any secondary firm Point(s) of Receipt, together with the sum of the Quantities of Gas nominated from primary firm Point(s) of Receipt, adjusted for Fuel Reimbursement, shall not exceed the unassigned Maximum Daily Contract Quantity in the Releasing Shipper's Service Agreement; provided, however, that where a Releasing Shipper releases a segment or segments of its service rights, the Releasing Shipper may nominate for Transportation on any Day, up to the unassigned Maximum Daily Contract Quantity in its Service Agreement applicable to any segment wherein it has retained service rights.
- (iii) If the Releasing Shipper and the Replacement Shipper nominate Quantities of Gas in segments that overlap, the Quantities shall be scheduled in accordance with Section 8.8 of these General Terms and Conditions. If the Quantities have equal priority and the sum of the Quantities cannot be scheduled, the Quantities shall be scheduled pro-rata unless the Releasing Shipper specifies otherwise in its release notice.
- (c) If the Shipper's Service Agreement has been assigned entirely to a Replacement Shipper pursuant to Section 15 of these General Terms and Conditions, the Replacement Shipper may nominate Quantities at Point(s) of Receipt to the same extent the Releasing Shipper could have if the Service Agreement had not been assigned.
- (d) If a Shipper nominates Quantities of Gas for transportation from a secondary firm Point(s) of Receipt, Shipper shall retain its priority at the primary firm Point(s) of Receipt designated in the

FERC Docket: RP00-162-000

Third Revised Sheet No. 251 Third Revised Sheet No. 251: Superseded

Superseding: Substitute First Revised Sheet No. 251 $_{\mbox{\scriptsize GENERAL}}$ TERMS AND CONDITIONS

(Continued)

Service Agreement and may nominate from the primary firm Point(s) of Receipt at any time, subject to the provisions of Sections 8 and 9 of these General Terms and Conditions.

- 10.3(a) The reservation components under Rate Schedules FT, EFT, LFT and HFT shall be computed using the Quantities of Gas assigned to the primary firm Point(s) of Receipt in the Service Agreement except as provided in Sections 10.3(b) or 10.3(c). Subject to the provisions of Section 10.5 hereof, the reservation components shall be pro-rated to adjust for changes during the Month for the Point(s) of Receipt designated in the Service Agreement or the Quantities assigned thereto.
 - (b) The Quantities of Gas assigned to a secondary firm Point(s) of Receipt shall serve to increase the reservation components to reflect the additional rate zones or mileage segments associated with such Quantities if: (1) the Quantity assigned to a secondary firm Point(s) of Receipt located on Panhandle's Gathering facilities exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Receipt on such Gathering facilities; or (2) the Quantity assigned to a secondary firm Point(s) of Receipt located in the Field Zone exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Receipt located in the Field Zone and upstream thereof; or (3) the Quantity assigned to a secondary firm Point(s) of Receipt in the Market Zone exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Receipt located in the same mileage segment of the Market Zone and upstream thereof.
 - (c) A Shipper with a primary firm Point of Receipt located in the Field Zone may assign Quantities of Gas to be transported under a firm Rate Schedule from a secondary firm Point(s) of Receipt located on Panhandle's Gathering facilities up to the Maximum Daily Contract Quantity assigned to the Field Zone without affecting the computation of the Gathering reservation component, notwithstanding the provisions of Section 10.3(b).

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 252 Fifth Revised Sheet No. 252: Superseded

Superseding: Fourth Revised Sheet No. 252

GENERAL TERMS AND CONDITIONS (Continued)

10.4 If a Shipper has executed a Transportation Agreement for firm service designating a primary firm Point of Receipt in the Field Zone and Point of Delivery in the Market Zone, the Shipper may designate a Pool Point as an alternative firm Point of Receipt for Quantities not in excess of the Maximum Daily Contract Quantity assigned to Point(s) of Receipt located in the Field Zone. If the Shipper designates a Pool Point as an alternative firm Point of Receipt, the Shipper is not required to reduce the Maximum Daily Contract Quantity applicable to Point(s) of Receipt located in the Field Zone, but the Shipper may not nominate or have transported Quantities of Gas in excess of the Maximum Daily Contract Quantity at such Point(s) of Receipt. Shipper's nomination of Quantities at the Pool Point does not relieve the Shipper of its obligation to pay the reservation charges applicable to service from Point(s) of Receipt located in the Field

- 10.5(a) Subject to Section 7.4 and the availability of capacity, Shipper may submit a request to change any firm primary Point of Receipt or to modify the Quantity of Gas assigned to any firm primary Point of Receipt upon prior written notice to Panhandle. Panhandle will respond to the notice as soon as possible but in no event later than two (2) Business Days following receipt of the notice. Such request may be submitted twice in any thirty (30) Day period.
 - (b) A Shipper may submit a request to change any secondary firm Point of Receipt or to modify the Quantity of Gas assigned to any secondary firm Point of Receipt at any time. Panhandle shall tender a revised contract within one (1) Business Day of receipt of such request.
- 10.6 Upon agreement by operator and Panhandle, multiple measuring stations belonging to a single operator may constitute a single Point of Receipt.
- 10.7 Shippers under Rate Schedule FT, EFT, SCT, IT, EIT, LFT, HFT or GPS which designate and utilize a Pool Point as a Point of Receipt shall be subject to the terms and conditions of Section 14 herein.

FERC Docket: RP00-395-003

Second Revised Sheet No. 253 Second Revised Sheet No. 253: Superseded

Superseding: First Revised Sheet No. 253

GENERAL TERMS AND CONDITIONS (Continued)

11. POINTS OF DELIVERY

- 11.1 A Point of Delivery shall be equipped with a Flow Control Device if service at the Point of Delivery is provided under Rate Schedule FT, HFT or IT.
- 11.2 A Shipper may designate any Point of Delivery for Gas to be delivered under a particular Transportation Rate Schedule. Each Point of Delivery for firm service shall be identified in the Service Agreement as either primary firm Point(s) of Delivery or secondary firm Point(s) of Delivery. The sum of the Quantities of Gas of all primary firm Point(s) of Delivery specified in the Transportation Agreement may not exceed the firm Maximum Daily Contract Quantity stated in the Transportation Agreement. Unless the Shipper specifies otherwise, the Quantity of Gas assigned to each secondary firm Point of Delivery the Shipper designates in a rate zone or mileage segment of the Market Zone shall equal the sum of the Quantities of Gas assigned to the primary firm Points of Delivery in the same rate zone or, in the Market Zone, the same mileage segment of the Market Zone or upstream portion thereof. In addition, a Shipper may segment, to the extent operationally feasible, its Primary Path into two (2) or more discrete segments for its own use or in connection with an assignment pursuant to Section 15of these General Terms and Conditions.
- 11.3 Subject to the conditions of the Rate Schedule under which Shipper is receiving service, Shipper may have multiple primary and secondary Points of Delivery.
- 11.4 Upon agreement by operator and Panhandle, multiple measuring stations belonging to a single operator may constitute a single Point of Delivery.
- 11.5 Subject to the conditions of the Rate Schedule under which Shipper is receiving service and the provisions of Sections 4, 8, 9 and 10 herein, Shipper may nominate Quantities of Gas to be delivered to any Point of Delivery identified in the Transportation Agreement, up to the Quantity of Gas assigned to such Point of Delivery in the Transportation Agreement.
 - (a) If the Shipper's Service Agreement has not been assigned in whole or in part pursuant to Section 15 of these General Terms and Conditions, the sum of the

FERC Docket: RS92- 22-012

Second Sub Original Sheet No. 254 Second Sub Original Sheet No. 254: Superseded

Superseding: Sub Original Sheet No. 254
GENERAL TERMS AND CONDITIONS
(Continued)

Quantities of Gas nominated by the Shipper to any secondary firm Point(s) of Delivery, together with the sum of the Quantities of Gas nominated to a primary firm Point(s) of Delivery shall not exceed the Maximum Daily Contract Quantity specified in the Shipper's Service Agreement for service under any Rate Schedule; provided, however, if the Shipper utilizes two (2) or more discrete pipeline segments, the sum of the Quantities of Gas nominated by the Shipper to all Point(s) of Delivery may exceed the Maximum Daily Contract Quantity specified in the Service Agreement so long as the Quantities nominated for transportation in a pipeline segment do not exceed the Maximum Daily Contract Quantity applicable to such segment.

- (b) If the Shipper's Service Agreement has been assigned in part pursuant to Section 15 of these General Terms and Conditions:
 - the sum of the Quantities of Gas nominated by the Replacement Shipper to any secondary firm Point(s) of Delivery, together with the sum of the Quantities of Gas nominated to primary firm Point(s) of Delivery, shall not exceed the Maximum Daily Contract Quantity in the Replacement Shipper's Service Agreement; provided, however, if the Replacement Shipper utilizes two (2) or more discrete pipeline segments, the sum of the Quantities of Gas nominated by the Replacement Shipper to all Point(s) of Delivery may exceed the Maximum Daily Contract Quantity specified in the Service Agreement so long as the Quantities nominated for transportation in a pipeline segment do not exceed the Maximum Daily Contract Quantity applicable to such segment; and

FERC Docket: RP00-395-003

First Revised Sheet No. 254A First Revised Sheet No. 254A: Superseded

Superseding: Original Sheet No. 254A $$\operatorname{\mathtt{GENERAL}}$ TERMS AND CONDITIONS

(Continued)

(ii) the sum of the Quantities of Gas nominated by the Releasing Shipper to any secondary firm Point(s) of Delivery, together with the sum of the Quantities of Gas nominated to primary firm Point(s) of Delivery, shall not exceed the unassigned Maximum Daily Contract Quantity in the Releasing Shipper's Service Agreement; provided, however, that where a Releasing Shipper releases a segment or segments of its service rights, the Releasing Shipper may nominate for Transportation on any Day, up to the unassigned Maximum Daily Contract Quantity in its Service Agreement applicable to any segment wherein it has retained service rights.

- (iii) If the Releasing Shipper and the Replacement Shipper nominate Quantities of Gas in segments that overlap, the Quantities shall be scheduled in accordance with Section 8.8 of these General Terms and Conditions. If the Quantities have equal priority and the sum of the Quantities cannot be scheduled, the Quantities shall be scheduled pro-rata unless the Releasing Shipper specifies otherwise in its release notice.
- (c) If the Shipper's Service Agreement has been assigned entirely to a Replacement Shipper pursuant to Section 15 of these General Terms and Conditions, the Replacement Shipper may nominate Quantities at Point(s) of Delivery to the same extent the Releasing Shipper could have if the Service Agreement had not been assigned.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 255 Fourth Revised Sheet No. 255: Superseded Superseding: Third Revised Sheet No. 255

GENERAL TERMS AND CONDITIONS

ENERAL TERMS AND CONDITION (Continued)

- (d) If a Shipper nominates Quantities of Gas for Transportation to a secondary firm Point(s) of Delivery, Shipper shall retain its priority at the primary firm Point(s) of Delivery designated in the Transportation Agreement and may nominate from the primary firm Point(s) of Delivery, subject to the provisions of Sections 8 and 9 of these General Terms and Conditions.
- 11.6(a) The reservation components applicable to Shippers under Rate Schedules FT, EFT, LFT and HFT shall be computed using the Quantities of Gas assigned to the primary firm Point(s) of Delivery in the Service Agreement except as provided in Section 11.6(b). Subject to the provisions of Section 11.9 hereof, the reservation components shall be pro-rated to adjust for changes during the Month in the Point(s) of Delivery designated in the Service Agreement or the Quantities assigned thereto.
 - (b) Quantities of Gas assigned to a secondary firm Point(s) of Delivery shall serve to increase the reservation components to reflect the additional rate zones or mileage segments associated with such Quantities if: (1) the Quantity assigned to a secondary firm Point(s) of Delivery located on Panhandle's Gathering Facilities exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Delivery on such Gathering facilities and downstream thereof; or (2) the Quantity assigned to a secondary firm Point(s) of Delivery located in the Field Zone exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Delivery located in the Field Zone and downstream thereof; or (3) the Quantity assigned to a secondary firm Point(s) of Delivery in the Market Zone exceeds the Maximum Daily Contract Quantity assigned to all primary firm Points of Delivery located in the same mileage segment of the Market Zone and downstream thereof.
- 11.7 A secondary firm Point of Delivery is not eligible for service under Rate Schedule GDS.
- 11.8 Shippers under Rate Schedule FT, EFT, IT, EIT, LFT or GPS which designate and utilize a Pool Point as a Point of Delivery shall be subject to the terms and conditions of Section 14 herein.

FERC Docket: RP00-395-003

Third Revised Sheet No. 255A Third Revised Sheet No. 255A: Superseded

Superseding: Second Revised Sheet No. 255A

GENERAL TERMS AND CONDITIONS

(Continued)

11.9(a) Subject to the availability of capacity and the provisions of Section 7.4, Shipper may submit a request to change a primary firm Point of Delivery or to modify the Quantity of Gas assigned to any firm primary Point of Delivery upon prior written notice to Panhandle. Panhandle will respond to the notice as soon as possible but in no event later than two (2) Business Days following receipt of the notice. Such a request may be submitted twice in any thirty (30) Day period. A change in the primary firm Point(s) of Delivery or modification in the Quantity of Gas at the primary firm Point of Delivery pursuant to this Section shall not be made so as to reduce the reservation charges applicable under the Service Agreement. A Shipper may add a primary Point(s) of Delivery that is not between the Point(s) of Receipt and Point(s) of Delivery specified in the Service Agreement and may increase the Maximum Daily Contract Quantity at any primary Point of Delivery over and above the prevailing Quantities under the Service Agreement pursuant to Section 7.

- (b) A Shipper may submit a request to change any secondary firm Point(s) of Delivery or to modify the Quantity of Gas assigned to any secondary firm Point(s) of Delivery at any time. Panhandle shall tender a revised contract within one (1) Business Day of receipt of such request.
- (c) A change to a primary or secondary firm Point of Delivery located at a Pool Point shall not be permitted if it results in firm delivery capacity in excess of firm receipt capacity at any such Pool Point.

FERC Docket: RS92- 22-005

Original Sheet No. 256 Original Sheet No. 256: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

12. CONDITIONS OF RECEIPT AND DELIVERY

Deliveries of Gas by Panhandle to, or for the account of, any Shipper are absolutely dependent upon confirmed receipts of Gas, scheduled for delivery or provided for under Rate Schedule GDS. Panhandle shall, in no circumstances, be obligated to deliver any Quantities of Gas for which the Thermal Equivalent, adjusted for Fuel Reimbursement, has not been received by Panhandle for service under any Service Agreement.

- 12.1 Panhandle and Shipper, by mutual agreement, shall establish the date of commencement of receipt and delivery of Gas hereunder.
- 12.2 Receipts and deliveries of Gas hereunder shall be at uniform hourly rates of flow unless otherwise provided pursuant to the Rate Schedule under which Shipper is receiving service. If the Quantities of Gas received and delivered are not in balance on any particular Day such imbalance shall be corrected by Shipper as promptly as is reasonably consistent with operating conditions.
- 12.3 Unless otherwise agreed to in writing by Panhandle and Shipper, deliveries of Gas at a Point of Delivery shall be at such pressure as may exist in Panhandle's pipeline at such point from time to time.
- 12.4 Deliveries of Gas at the Point of Receipt shall be at a pressure sufficient to enter Panhandle's pipeline system at such point.
- 12.5 Shipper shall be responsible for upstream and downstream Transportation and Storage arrangements. To the extent Shipper's upstream or downstream transporters do not deliver or receive Gas for any reason, Panhandle shall not be obligated to receive or deliver Gas.
- 12.6 If a Shipper is unable to accept Quantities of Gas
 tendered at the Point(s) of Delivery on any Day, then
 Panhandle shall not be obligated to receive Gas at the
 Point(s) of Receipt on such Day.

FERC Docket: RS92- 22-005

Original Sheet No. 257 Original Sheet No. 257: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

- 12.7 Should Shipper be unable to provide Quantities of Gas scheduled at the Point(s) of Receipt on any Day, then Panhandle shall not be obligated to deliver Gas at the Point(s) of Delivery on such Day.
- 12.8 Panhandle shall have the unqualified right to commingle Gas transported hereunder with Gas from other sources, and to treat and handle all such Gas as its own. Unless Shipper or Shipper's designee elects to exercise its rights to process Gas for removal of moisture, helium, natural gasoline, butane, propane, or other liquefiables or inerts and has made arrangements for such processing at an existing point on Panhandle's system, Panhandle will have the unqualified right to process such gas for the purpose of removing such moisture, helium, natural gasoline, butane, propane, or other liquefiables or inerts, and the ownership of such moisture, helium, natural gasoline, butane, propane, or other liquefiables or inerts, shall be vested in Panhandle. Panhandle must be advised prior to January 1 and July 1 of each year by Shipper or Shipper's designee for the following six (6) months of Shipper's or Shipper's designee's exercise of their right to process Gas for removal of such moisture, helium, natural gasoline, butane, propane, or other liquefiables or inerts and of the arrangements made therefor. It is recognized that Gas delivered at a Point of Delivery may not be the same molecules as those received at any Point of Receipt, and that Panhandle's sole obligation is to deliver the Thermal Equivalent of Quantities of Gas received, adjusted for Fuel Reimbursement.
- 12.9 Shipper shall have the responsibility to maintain a concurrent balance between Quantities of Gas received, adjusted for appropriate Fuel Reimbursement, and Quantities of Gas delivered, based on the best information available to Shipper. Panhandle shall make available to Shipper information regarding daily receipts and daily deliveries based on the best operating information (hereinafter called Operating Data) available

FERC Docket: RP00-162-000

Sixth Revised Sheet No. 258 Sixth Revised Sheet No. 258: Superseded

Superseding: Fifth Revised Sheet No. 258
GENERAL TERMS AND CONDITIONS
(Continued)

to Panhandle and shall make such information available to all parties to the transaction. Panhandle shall make Operating Data available in the Messengerb system, within one Business Day following the conclusion of the production Day for meters equipped with EGM or the availability of chart measurement for meters not so equipped.

12.10Allocation of Deliveries

Quantities of Gas delivered at any Point of Delivery on any Day shall be allocated first to nominated and scheduled WS, IWS, PS and FS Storage. After all nominated Storage has been accounted for, all Transportation Agreements shall be allocated pursuant to the allocation methodology as described in Section 8.5(a).

12.11Balancing Charges

If Shipper does not maintain a concurrent balance between Quantities of Gas received, adjusted for Fuel Reimbursement, and Quantities of Gas delivered, the following charges shall apply to imbalances in such Quantities.

(a) Firm Service Monthly Contract Balancing

For the purposes of this Section 12.11(a), Maximum Accumulated Imbalance Quantity (MAIQ) shall be established as the product of 1.5 times the Maximum Daily Contract Quantity (MDCQ) under a firm Transportation Agreement or 1,000 Dt., whichever is greater.

Each Month, Panhandle shall determine for each Shipper receiving service under Rate Schedule FT, EFT, SCT, GDS, LFT or HFT the Shipper's imbalance based on the Quantities of Gas received and delivered in the prior Month after minimizing the imbalances in accordance with Section 12.11(d) of these General Terms and Conditions. The resulting excess receipts and deliveries shall be handled as follows:

FERC Docket: RS92- 22-005

Original Sheet No. 259 Original Sheet No. 259: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(1) Excess Receipts

When actual receipts, adjusted for Fuel Reimbursement, exceed actual deliveries, the excess shall be applied to any prior accumulated imbalance due Panhandle first. If receipts still exceed deliveries, then an imbalance due Shipper shall be carried without charge up to the MAIQ. Any imbalance in excess of the MAIQ shall be sold by Shipper to Panhandle according to the following schedule:

Percentage Imbalance Level

ın	Exces	SS OI	MAIQ			Sales Pi	rice	
0%	-	5%		. 9	Х	Mid-Continent	Spot	Price
> 5%	-	10%		. 8	Х	Mid-Continent	Spot	Price
>10%	-	15%		. 7	Х	Mid-Continent	Spot	Price
>15%	-	20%		. 6	Х	Mid-Continent	Spot	Price
>20%	5			.5	Х	Mid-Continent	Spot	Price

The percentage imbalance level shall be calculated by dividing the imbalance in excess of MAIQ by the actual Quantity of Gas received during the Month unless the percentage imbalance level would be lower if calculated using operating data provided pursuant to Section 12.9 of these General Terms and Conditions, in which case the percentage imbalance level shall be calculated using such operating data.

(2) Excess Deliveries

When actual deliveries exceed actual receipts, adjusted for Fuel Reimbursement, the excess shall be applied to any prior accumulated imbalance due Shipper first. If deliveries still exceed receipts, then an imbalance due Panhandle shall be carried without charge up to the MAIQ. Any imbalance in excess of the MAIQ shall be purchased by Shipper from Panhandle according to the following schedule:

FERC Docket: RS92- 22-005

Original Sheet No. 260 Original Sheet No. 260: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

Percentage
Imbalance Level
in Excess of MAIQ
0% - 5%
1.1 x Mid-Continent Spot Price
> 5% - 10%
- 15%
1.2 x Mid-Continent Spot Price
> 1.3 x Mid-Continent Spot Price
> 1.4 x Mid-Continent Spot Price
> 20%
1.5 x Mid-Continent Spot Price

The percentage imbalance level shall be calculated by dividing the imbalance in excess of MAIQ by the actual Quantity of Gas delivered during the Month unless the percentage imbalance level would be lower if calculated using operating data provided pursuant to Section 12.9 of these General Terms and Conditions, in which case the percentage imbalance level shall be calculated using such operating data.

(b) Interruptible Service Monthly Contract Balancing

For the purposes of this Section 12.11(b), the Maximum Monthly Imbalance Quantity (MMIQ) shall be established as the product of .05 times the actual volumes delivered under an interruptible Transportation Agreement for the Month, or 1,000 Dt., whichever is greater.

Each Month, Panhandle shall determine for each Shipper receiving service under Rate Schedules IT or EIT the Shipper's imbalance based on the Quantities of Gas received and delivered in the prior Month after minimizing the imbalances in accordance with Section 12.11(d) of these General Terms and Conditions. The resulting excess receipts and deliveries shall be handled as follows:

FERC Docket: RS92- 22-005

Original Sheet No. 261 Original Sheet No. 261: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(1) Excess Receipts

When actual receipts, adjusted for Fuel Reimbursement exceed actual deliveries, an imbalance due Shipper shall be carried without charge up to the MMIQ. Any imbalance in excess of the MMIQ shall be sold by Shipper to Panhandle according to the following schedule:

The percentage imbalance level shall be calculated by dividing the imbalance in excess of MMIQ by the actual quantity of gas received unless the percentage imbalance level would be lower if calculated using operating data provided pursuant to Section 12.9 of these General Terms and Conditions, in which case the percentage imbalance level shall be calculated using such operating data. Any IT or EIT excess receipts up to the MMIQ must be nominated and scheduled to be transported during the next Month. Any portion of the MMIQ existing at the end of a Month which has not been nominated and scheduled to be made up during the next Month shall be sold by Shipper to Panhandle at the 0-5% Percentage Imbalance Level in Excess of MMIQ.

FERC Docket: RS92- 22-005

Original Sheet No. 262 Original Sheet No. 262: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(2) Excess Deliveries

When actual deliveries exceed actual receipts, adjusted for Fuel Reimbursement, an imbalance due Panhandle shall be carried without charge up to the MMIQ. Any imbalance in excess of the MMIQ shall be purchased by Shipper from Panhandle according to the following schedule:

Percentage Imbalance Level in Excess of MMIQ 0% - 5% 1.1 x Mid-Continent Spot Price > 5% - 10% 1.2 x Mid-Continent Spot Price >1.3 x Mid-Continent Spot Price >1.5% - 20% 1.4 x Mid-Continent Spot Price >20% 1.5 x Mid-Continent Spot Price

The percentage imbalance level shall be calculated by dividing the imbalance in excess of MMIQ by the actual quantity of gas delivered unless the percentage imbalance level would be lower if calculated using operating data provided pursuant to Section 12.9 of these General Terms and Conditions, in which case the percentage imbalance level shall be calculated using such operating data. Any IT or EIT excess deliveries up to the MMIQ must be nominated and scheduled to be received by Panhandle from Shipper during the next Month. Any portion of the $\widehat{\text{MMIQ}}$ existing at the end of a Month which has not been nominated and scheduled to be made up during the next Month shall be purchased by Shipper from Panhandle at the 0-5% Percentage Imbalance Level in Excess of MMTO.

FERC Docket: RP00-613-000

Second Revised Sheet No. 263 Second Revised Sheet No. 263: Superseded

Superseding: First Revised Sheet No. 263

GENERAL TERMS AND CONDITIONS

(Continued)

(c) Mid-Continent Spot Price Calculation

For sales of excess receipts under Sections 12.11(a)(1) and 12.11(b)(1) herein and purchases of excess deliveries under Sections 12.11(a)(2) and 12.11(b)(2), the Mid-Continent Spot Price shall be the average of the Kansas/Oklahoma Field Zone Spot Price for Gas delivered to Panhandle from the table "Spot Prices on Interstate Pipeline Systems" for deliveries during the applicable Month in which the imbalance occurred contained in the first issue of NATURAL GAS WEEK published the following month. If the reported price referenced above is not published for the Month required, Panhandle shall determine the Mid-Continent Spot Price using another publication that publishes the spot price for Gas.

- (d) Minimization of Imbalances for Transportation Agreements
 - (1) Contract Imbalance Netting

In order to minimize the quantity of excess $% \left(1\right) =\left(1\right) ^{2}$ receipts pursuant to Sections 12.11(a)(1) and 12.11(b)(1), above and excess deliveries pursuant to Sections 12.11(a)(2) and 12.11(b)(2), above, all of a Shipper's firm and interruptible Transportation Agreements shall be matched by Point(s) of Receipt and Point(s) of Delivery. The total quantity of Gas received and delivered under the Shipper's Transportation Agreements within the same Operational Impact Area, as defined in Section 12.11(d)(3) below, shall be netted and excess receipts or excess deliveries shall be determined only after such netting. Such netting of contract imbalances does not relieve the Shipper of the obligation to pay all applicable transportation charges for the Quantity of Gas actually delivered to Shipper during the Month.

FERC Docket: RP00-613-000

Original Sheet No. 263A Original Sheet No. 263A: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(2) Contract Imbalance Trading

- (i) Shipper may authorize contract imbalances under Shipper's Transportation Service Agreements within the same Operational Impact Area, as defined in Section 12.11(d)(3), to be posted for trading after the Shipper has minimized excess receipts and deliveries pursuant to Section 12.11(d)(1) above.
- (ii) Authorizations to post imbalances that are received by Panhandle by 11:45 a.m. shall be effective by 8:00 a.m. the next Business Day (Central Clock Time). Imbalances authorized for posting shall be posted on or before the ninth (9th) Business Day of the Month.
- (iii) Panhandle shall provide the ability to post and trade imbalances until at least the close of the seventeenth (17th) Business Day of the Month.
- (iv) Shippers may trade contract imbalances with other Shippers having Transportation Agreements within the same Operational Impact Area, as defined in Section 12.11(d)(3) below. Contract imbalances must be traded with contract imbalances in the opposite direction and must move Shipper's imbalance closer to zero. When trading imbalances, a Quantity must be specified.
- (v) Panhandle shall enable the imbalance trading process by:
 - . Receiving the request for imbalance trade;
 - . Receiving the imbalance trade confirmation;
 - . Sending the imbalance trade notification;
 - . Reflecting the trade prior to or on the next monthly Shipper imbalance or cashout.
- (vi) Imbalance trades can only be withdrawn by the initiating Shipper and only prior to the confirming Shipper's confirmation of the trade. Imbalance trades are considered final when confirmed by the confirming Shipper and effectuated by Panhandle. Imbalance trades

FERC Docket: RP03-479-000

First Revised Sheet No. 263B First Revised Sheet No. 263B : Superseded

Superseding: Original Sheet No. 263B

GENERAL TERMS AND CONDITIONS
(Continued)

shall be deemed to be effectuated by Panhandle when Panhandle sends the imbalance trade notification.

- (vii) After receipt of an imbalance trade confirmation, Panhandle shall send the imbalance trade notification to the initiating Shipper and the confirming Shipper no later than noon (Central Time) on the next Business Day.
- (3) Operational Impact Area

For purposes of this Section 12.11(d), two or more Transportation Agreements shall be deemed to be within the same Operational Impact Area only if the Transportation Agreements provide for the Transportation of Natural Gas solely within the Field Zone; or if the Transportation Agreements provide for either the receipt or delivery of Natural Gas within the Market Zone.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 264 Substitute Original Sheet No. 264: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(e) All sales and purchases of excess receipts and excess deliveries pursuant to this Section 12 shall be deemed to take place on the first Day of the Month following the accumulation of the applicable imbalance. Settlement of accounts for such sales and purchases shall be made with the Monthly bill, for the Month in which the applicable imbalance accumulated. Settlements of sales and purchases shall be in the form of a credit to Shipper's bill in the case of sales to Panhandle, and as an addition to Shipper's bill, in the case of sales to Shipper.

(f) Prior Month Adjustments

In the event there are adjustments to Shipper's receipts or deliveries or both applicable to a prior Month ("Prior Month Adjustment" or "PMA") such adjustments shall be used to mitigate any amount paid by Shipper or amount credited by Panhandle pursuant to Section 12.11(a) or Section 12.11(b), if applicable to the prior Month. Otherwise, the PMA will be accounted for as current Month volumes. The Shipper will not be penalized in the current Month for PMA volumes that worsen the Shipper's current accumulated imbalance.

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 265 Fifth Revised Sheet No. 265: Superseded

Superseding: Sub Fourth Revised Sheet No. 265
GENERAL TERMS AND CONDITIONS
(Continued)

(g) Hourly Delivery Notification

If a Shipper under Rate Schedule EFT, EIT, SCT, GDS or LFT, is duly notified by Panhandle to limit hourly deliveries to one-sixteenth of the Quantity scheduled for delivery on that Day and continues to receive more Gas at the Point(s) of Delivery than the amount allowed, Shipper shall pay \$10 per Dt. or two times the Mid-Continent Spot Price calculated in accordance with Section 12.11(c), whichever is greater, for the excess deliveries in addition to all other applicable charges.

- (h) Daily Scheduling Charge
 - (1) If the sum of actual deliveries at any Point of Delivery under Transportation Agreements varies from the Sum of Quantities of Gas scheduled for delivery at such Point of Delivery by 10% or 100 Dt., whichever is greater (hereinafter called the tolerance level) plus the available MDVQ if the delivery point operator is a DVS Party, a daily Scheduling Charge shall be assessed; provided that, no Scheduling Charge shall be assessed on an interruptible Transportation Agreement for the Day on which such Shipper's scheduled Quantity has been reduced as a result of a nomination under a Transportation Agreement for firm service during the Evening Nomination Cycle or the Intra-day 1 Nomination Cycle in Section 8.2(b) herein. The charge shall be applied to each Transportation Agreement with respect to which actual deliveries, as allocated in accordance with Section 12.10, above, vary from the scheduled Quantity in excess of the tolerance level; provided that, if the Transportation Agreement is a designated Transportation Agreement under a TBS Agreement, the tolerance level shall be expanded by the MDIQ, MDWQ, and available Stored Volume applicable to the designated Transportation Agreement under the TBS Agreement and if the Transportation Agreement is an underlying Transportation Agreement under Rate Schedule DVS, the tolerance level shall be increased by the MDVQ. For service under Rate Schedule FT and IT, the Scheduling Charge shall be equal to the sum of the IT Commodity rates per Dt. applicable to the Transmission Access Charge and the Transmission Mileage Charge for 401-500 miles set forth on effective Tariff Sheet No. 7;

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 266 Fifth Revised Sheet No. 266: Superseded

Superseding: Fourth Revised Sheet No. 266

GENERAL TERMS AND CONDITIONS

(Continued)

for service provided under Rate Schedules EFT, SCT, EIT, LFT and HFT the Scheduling Charge shall be equal to the sum of the EIT Commodity rates per Dt. applicable to the Transmission Access Charge and the Transmission Mileage Charge for 401-500 miles set forth on effective Tariff Sheet No. 8.

(2) If Panhandle issues an OFO as to any portion of its system, requiring Shipper(s) to reduce scheduling variances, the tolerance level, as defined above, for that portion of its system shall be reduced from 10% to 5% and the charge for variances beyond the reduced tolerance level shall be as stated in Section 12.17 herein in lieu of the Scheduling Charge described in this Section 12.11(h)(1). For Shipper(s) under Rate Schedule SCT, the tolerance level shall be reduced from 10% to 5%, or 500 Dt., whichever is greater.

FERC Docket: RP96-348-000

First Revised Sheet No. 267 First Revised Sheet No. 267: Superseded

Superseding: Original Sheet No. 267

GENERAL TERMS AND CONDITIONS (Continued)

(i) Farm Tap

A Point of Delivery on Panhandle's pipeline system which is required under the terms of a right-of-way agreement between the landowner and Panhandle and at which Gas is delivered under Rate Schedules SCT or EFT for domestic and/or agricultural use shall be defined as a Farm Tap. A Shipper's deliveries at a Farm Tap shall not be subject to the terms set out in (g) and (h) above but shall be subject to the terms of (a) and (c) above. Excess receipts and excess deliveries shall be determined on a basis consistent with the measurement frequency for the Farm Tap delivery meters.

(j) Irrigation Tap

A Point of Delivery on Panhandle's pipeline system which is required under the terms of a right-of-way agreement between the landowner and Panhandle and at which Gas is delivered under Rate Schedules EFT or EIT for purposes of irrigation shall be defined as an Irrigation Tap. A Shipper's deliveries at an Irrigation Tap shall not be subject to the terms set out in (g) and (h) above but shall be subject to the terms of (a), (b) and (c) above. Excess receipts and excess deliveries shall be determined on a quarterly, rather than on a Monthly basis.

FERC Docket: RP95-383-002

Substitute First Revised Sheet No. 268 Substitute First Revised Sheet No. 268: Superseded

Superseding: Original Sheet No. 268

GENERAL TERMS AND CONDITIONS (Continued)

- 12.12(a) Upon termination of firm Transportation Service, Shipper shall reduce to zero any remaining imbalance within 60 Days of the date that Panhandle notifies Shipper of such imbalance. Such reduction may be accomplished by assigning the imbalance within MAIQ to another firm Service Agreement as long as such assignment will not result in an imbalance in excess of MAIQ on the Service Agreement to which the imbalance is assigned. If Shipper does not reduce the remaining imbalance to zero within such period, Panhandle shall retain those quantities it owes Shipper free and clear of any adverse claims and shall charge Shipper and Shipper shall be obligated to pay for those quantities Shipper owes Panhandle at a rate equal to 1.5 times the Mid-Continent Spot Price as defined in Section 12.11(c) for the Month in which the Service Agreement terminated.
- 12.12(b) Upon termination of firm Storage Service, Shipper shall withdraw, or otherwise dispose of, all of its Stored Volume by the date the firm Storage Agreement terminates. If Shipper fails to withdraw, or otherwise dispose of, all of its Stored Volume prior to the termination of the firm Storage Agreement, Panhandle may retain any remaining quantities of Stored Volume free and clear of any adverse claims; provided, however, that Panhandle will notify Shipper in writing 60 Days prior to termination of its firm Storage Agreement of the quantity of Stored Volume being held by Panhandle for Shipper's account. If Shipper in good faith disagrees with the Quantity of Stored Volume set forth in the notification required above, then within 15 days of such notification, Shipper shall notify Panhandle in writing of such dispute and request a reconciliation. If, as of the date of termination of Shipper's firm Storage Agreement, the Quantity of Stored Volume contained in Panhandle's notice has not been finally reconciled and such failure to reconcile results in a retention hereunder by Panhandle, then title to the difference between the Quantity subject to retention according to Panhandle's records and that reflected on Shipper's

FERC Docket: RP95-383-002

Original Sheet No. 268A Original Sheet No. 268A: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

records shall not finally pass to Panhandle until such reconciliation has been completed. Panhandle shall extend the time available for Shipper to remove its Gas from storage by one Day for every Day that Shipper has been unable to withdraw properly nominated gas due to operational or force majeure conditions on Panhandle's system.

12.13 Unauthorized Gas

Panhandle shall have the right to dispose of Unauthorized \mbox{Gas} as follows:

(a) Unauthorized Receipts

When Unauthorized Gas is delivered to Panhandle, Panhandle shall purchase the Unauthorized Gas from the operator of the Point(s) of Receipt at which the Unauthorized Gas is received at 50% of the Mid-Continent Spot Price, as defined in Section 12.11(c), for the Month in which the Unauthorized Gas is delivered to Panhandle.

FERC Docket: RP94-160-000

First Revised Sheet No. 269 First Revised Sheet No. 269: Superseded

Superseding: Sub Original Sheet No. 269
GENERAL TERMS AND CONDITIONS
(Continued)

agreements. If there are no Transportation Agreements between Panhandle and the operator of the Point(s) of Delivery at which the Unauthorized Gas is taken, Panhandle shall sell the Unauthorized Gas to the operator at 150 percent of the Mid-Continent Spot Market Price, as defined in Section 12.11(c) for the Month in which the Unauthorized Gas is taken.

12.14 Unauthorized Overrun Receipts

Panhandle shall notify Shipper of excess receipts in order to permit Shipper to reduce receipt quantities. If the excess receipts continue and such receipts jeopardize the safety of the pipeline or Panhandle's ability to serve other Shippers, Panhandle shall be entitled to refuse to accept such excess receipts. If, despite such refusal the excess receipts continue all further excess receipts as to which notice has been given shall be treated as excess receipts under Section 12.11(a) or (b) with a percentage imbalance level greater than 20%.

12.15 Storage Related Transportation

In order to accommodate the transportation to and from Storage used to meet Winter Period requirements, the Maximum Daily Contract Quantity in the Transportation Agreement shall be reduced during the Summer Period in accordance with this Section 12.15.

- (a) For Transportation to and from Storage provided by Panhandle under Rate Schedule WS, PS or FS, the Maximum Daily Contract Quantity in the Summer Period shall be reduced if the Transportation Agreement provides that:
 - (i) During the Summer Period the sole primary Point of Delivery is the WS Point or PS Point; and the Maximum Daily Contract Quantity at the

FERC Docket: RP94-160-000

First Revised Sheet No. 270 First Revised Sheet No. 270: Superseded

Superseding: Original Sheet No. 270
GENERAL TERMS AND CONDITIONS
(Continued)

Point(s) of Receipt is equal to the Maximum Daily Injection Quantity, adjusted for Fuel Reimbursement, under Rate Schedule WS, PS or FS; and

- (ii) During the Winter Period the sole primary Point of Receipt is the WS Point or the PS Point; and the sole primary Point of Delivery is the outlet side of Panhandle's pipeline facilities that interconnect Panhandle's system with the facilities of Shipper or Shipper's designee; and the Maximum Daily Contract Quantity at the primary Point of Receipt is equal to the Maximum Daily Withdrawal Quantity, adjusted for Fuel Reimbursement, under Rate Schedule WS, PS or FS.
- (b) For transportation to and from storage provided by a person other than Panhandle, the Maximum Daily Contract Quantity in the Summer Period shall be reduced if the Transportation Agreement provides that:
 - (i) During the Summer Period the sole primary Point of Delivery is the outlet side of Panhandle's pipeline facilities that interconnect Panhandle's system with the facilities of another person; and the Maximum Daily Contract Quantity at the Point(s) of Receipt is equal to the Maximum Daily Injection Quantity, adjusted for Fuel Reimbursement, under the Shipper's firm Storage service agreement;
 - (ii) During the Winter Period the sole primary Point of Receipt shall be the inlet side of Panhandle's pipeline facilities that interconnect Panhandle's system with the facilities of another person; the sole primary Point of Delivery is the outlet side of Panhandle's pipeline facilities that interconnect Panhandle's system

FERC Docket: RS92- 22-005

Original Sheet No. 271 Original Sheet No. 271: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

with the facilities of Shipper or Shipper's designee; and the Maximum Daily Contract Quantity at the Point of Receipt is equal to the Maximum Daily Withdrawal Quantity, adjusted for Fuel Reimbursement, under the Shipper's firm Storage service agreement;

(iii) The primary Point of Delivery in the Summer Period and the primary Point of Receipt in the Winter Period is an interconnection in the Market Zone between Panhandle's facilities and the facilities of another person that can both receive Gas from, and deliver Gas to, Panhandle and that delivers to Panhandle for the account of the Shipper, under the affected Transportation Agreement only Quantities that have been withdrawn from firm Storage.

Panhandle shall not execute a Service Agreement for storage related transportation to or from an interconnection with another person until the Shipper provides Panhandle with written verification that it is entitled to receive storage service, with an MDIQ and MDWQ, adjusted as appropriate for Fuel Reimbursement, matching the MDCQ the Shipper has requested at the interconnection during the Summer and Winter Periods, and that the Shipper is entitled to require that such Quantities be received from and delivered to Panhandle during the Summer and Winter Periods. Storage related transportation may be suspended if and to the extent that the Shipper does not subsequently verify (1) that it continues to be entitled to receive a firm storage service, (2) the Shipper's MDIQ, MDWQ, and MSQ (3) Quantities the Shipper is entitled to require to be received from, and delivered to, Panhandle at the interconnection, and (4) the Shipper's Stored Volumes at the beginning of the Winter Period.

FERC Docket: RP00-395-003

Second Revised Sheet No. 272 Second Revised Sheet No. 272: Superseded

Superseding: Sub First Revised Sheet No. 272

GENERAL TERMS AND CONDITIONS

(Continued)

(c) A Shipper with a Storage Related Transportation Agreement whose Maximum Daily Contract Quantity has been reduced in accordance with this Section 12.15 may identify secondary Point(s) of Receipt and Point(s) of Delivery in the Transportation Agreement and may nominate Quantities of Gas at such Point(s) in accordance with Sections 10 and 11 of these General Terms and Conditions. In addition, the Shipper may change the primary and secondary Point(s) of Receipt and Delivery in accordance with Section 10.5 and Section 11.9 of these General Terms and Conditions. If the Shipper changes the primary Point(s) of Receipt or Point(s) of Delivery, and the Transportation Agreement no longer qualifies under the terms of this Section 12.15 for the reduced Maximum Daily Contract Quantity during the Summer period, the Maximum Daily Contract Quantity during the Winter Period shall apply throughout the year as of the Day the change in the primary point(s) becomes effective.

12.16 Unauthorized Overrun Penalty

- (a) If on any Day during the month, Shipper takes Quantities of Gas in excess of the MDCQ as stated in the Transportation Agreement or Quantities of Gas in excess of the MDIQ or MDWQ as stated in the Storage Agreement and such Quantity has not been scheduled by Panhandle, then, in addition to the Overrun Charge set forth in the applicable Rate Schedule, Shipper shall be subject to an Unauthorized Overrun Penalty of two times the greater of the highest daily price published in Gas Daily, Daily Price Survey, Citygates - Chicago LDCs or Citygates Mich.- Mich Con for the day (Saturday and Sunday shall be the preceding Friday price) that Unauthorized Overrun Penalties are incurred for each Dt. taken in excess of the greater of the MDCQ, MDIQ, or MDWQ, as applicable, or the scheduled Quantity of Gas. For purposes of determining the Quantities of Gas subject to the Unauthorized Overrun Penalty provided for in this Section 12.16, Panhandle will allow a daily tolerance of four (4) percent of the MDCQ, MDIQ, or MDWQ, as applicable, or fifty (50) Dt. whichever is greater, and on a monthly basis there will be a two (2) percent tolerance. The Quantities of Gas subject to an Unauthorized Overrun Penalty each month shall be the greater of (1) the sum of the daily Unauthorized Overrun Penalty volumes after the four (4) percent daily tolerance; or (2) the monthly Unauthorized Overrun Penalty volumes determined after the two (2) percent monthly tolerance.
- (b) Aggregation by Shipper

For purposes of calculating the daily tolerance and the Quantity of $\ensuremath{\mathsf{Gas}}$ subject to the Unauthorized Overrun Penalty under this Section 12.16, a Shipper shall be permitted to aggregate the MDCQs of its firm Transportation Agreements at a designated Point of

FERC Docket: RP00-395-003

Third Revised Sheet No. 272A Third Revised Sheet No. 272A: Superseded

Superseding: Second Revised Sheet No. 272A

GENERAL TERMS AND CONDITIONS

(Continued)

Delivery. For purposes of aggregation under this Section 12.16(b), Shipper shall have scheduled quantities under the firm Transportation Agreements to be aggregated to no other Point of Delivery.

- (c) Panhandle will use the quality specifications, including Btu measurements, shown at the meter with no retroactive adjustment applied in determining the Unauthorized Overrun Penalty provided for in this Section 12.16.
- (d) If a Shipper takes Quantities of Gas in excess of MDCQ which cause a Shipper to incur an Unauthorized Overrun Penalty under this Section 12.16, such Quantities of Gas will not be considered in any calculation of an otherwise applicable Daily Scheduling Charge pursuant to Section 12.11(h)(1) of the General Terms and Conditions. However, if Panhandle has issued an OFO, the penalty provided for in Section 12.17(f)(ii) below shall apply in lieu of the Unauthorized Overrun Penalty in this Section 12.16.

12.17 Operational Flow Orders

(a) Conditions for the Issuance of an Operational Flow Order

Panhandle will have the right to issue an Operational Flow Order (OFO) when, in Panhandle's reasonable judgment, such OFO is required to alleviate conditions which threaten or could threaten system integrity, safety or reliability of service or to ensure Shipper(s) compliance with the provisions contained in this Tariff. An OFO may be issued to a particular Shipper(s) creating the need for the OFO or to Shippers on all or part of the system, when in the absence of such Shipper(s) action(s), an OFO $\,$ would not be required. When an OFO is issued, Panhandle will endeavor to minimize the Shippers and Quantities affected. In this regard, Panhandle will not require Shipper(s) under Rate Schedules EFT and SCT in the affected segment to restrict takes at a Point(s) of Delivery during any hour to one-sixteenth of the nominated and scheduled Quantity to the Point(s) for the Day unless deliveries to other Shipper(s) in the affected segment have been ordered to be within the tolerance levels set forth in this Section 12.17.

(b) Circumstances Under Which an OFO May Be Issued

Panhandle may issue an OFO in any circumstance which would, in Panhandle's reasonable judgment, impair Panhandle's ability to receive or deliver Quantities of Gas in accordance with its service obligations including, but not limited to, when:

FERC Docket: RP00-395-003

First Revised Sheet No. 272B First Revised Sheet No. 272B : Superseded

Superseding: Sub Original Sheet No. 272B
GENERAL TERMS AND

GENERAL TERMS AND CONDITIONS (Continued)

- (i) operating pressures on the affected portion of the system are significantly less than or greater than normal system operating pressures despite Panhandle's efforts to maintain normal pressures and a further decline or increase in operating pressures would impair Panhandle's ability to receive or deliver scheduled Quantities of Gas;
- (ii) a Shipper fails to maintain receipts or deliveries as required in this tariff;
- (iii) unscheduled pipeline maintenance and repairs
 affect capacity;
- (iv) wells or pipelines or other essential equipment freeze to the extent that such freezing damages or destroys or otherwise impairs Panhandle's essential facilities or ability to monitor and control essential facilities or results in the loss of supplies of natural gas from other pipelines, Panhandle's storage facilities or producers of Gas;
- (v) a loss of Gas supply from producers, storage providers or other suppliers reduces pressure or supplies available for delivery;
- (vi) Federal or state rules, regulations, orders, such as safety inspection orders and environmental safety orders, require Panhandle to reduce operating pressures in, or remove from service, a portion of Panhandle's system facilities;
- (vii) a mechanical or physical failure affects Panhandle's ability to deliver or receive gas or operate storage, including, but not limited to, pipeline failure, compressor failure, regulator failure, or other similar mechanical or physical failure;
- - (ix) external operations, including but not limited to highway construction, require Panhandle to reduce operating pressures in, or remove from service, a portion of Panhandle's system facilities; or
 - (x) a Shipper's failure to comply with the provisions of this Tariff that adversely affects the operations of Panhandle's

FERC Docket: RP00-395-003

Original Sheet No. 272C Original Sheet No. 272C : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

pipeline system including, but not limited to, a Shipper's failure to adhere to the quality of gas specifications set forth in Section 3 of these General Terms and Conditions.

(c) Remedial Actions to Be Taken Before Issuing an OFO

Prior to issuing an OFO Panhandle will take all reasonable actions to minimize the issuance and adverse consequences of the OFO. These actions may include, but are not limited to, requiring firm shippers to utilize primary Points of Receipt and Delivery and compliance with the provisions of this Tariff and the Service Agreement.

(d) Notice

- Panhandle will provide as much advance notice of an OFO as is possible in the circumstances. Ordinarily, the notice will be issued by 10:30 a.m. CT on the Day before the OFO is to be effective. The OFO will ordinarily become effective at the commencement of the Day. When operating conditions so require, Panhandle may issue the notice of the OFO after 10:30 a.m. and may provide that the OFO will become effective before the commencement of the next Day. When an OFO becomes effective at any time other than the commencement of any Day, Shippers will use their best efforts to comply with the OFO as soon as possible but in no event later than two hours, after which the applicable Scheduling Charges or Unauthorized Overrun Penalty provided for in this Section 12.17 shall apply.
- (ii) Panhandle shall issue notice of an OFO by notifying the affected meter operator(s) and Shippers(s) through the MessengerSM system, the Web Site and by telephone, and facsimile or via e-mail communication. Panhandle will post notification of, and provide updated information concerning, the OFO on the MessengerSM System and the Web Site at the commencement of each Day until Panhandle notifies the affected meter operator(s) and the Shipper(s) that the OFO has ended.
- (iii) Each notice will provide the time and date of issuance of the OFO, the time the OFO is to become effective, the time the OFO is expected to remain in effect, the action required to be taken by the Shipper(s), the reason for issuing the OFO, together with the operating variables providing the basis for issuing the

FERC Docket: RP00-395-003

Original Sheet No. 272D Original Sheet No. 272D : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

OFO, and any other information which may be required in the circumstances.

- (iv) Within a reasonable period of time following the end of the OFO, Panhandle will post on the MessengerSM System and the Web Site a report detailing the conditions that required the issuance and termination of the OFO.
- (e) Remedial Action to Be Taken

The OFO shall require the Shipper(s) subject to the OFO to take such action(s) or refrain from such action(s) as necessary to alleviate the conditions set forth in Sections 12.17(a) and 12.17(b). actions include, but are not limited to, the elimination of Unauthorized Overruns and the reduction of scheduling variances. If the OFO requires the elimination of Unauthorized Overruns and the reduction of scheduling variances, the OFO shall reduce the daily tolerance level for Unauthorized Overruns to zero and shall reduce the tolerance level for scheduling variances from 10% to 5% and require Shipper(s) to maintain scheduling variances to the tolerance level stated in the notice. For Shipper(s) under Rate Schedule SCT, the tolerance level shall be reduced from 10% to 5%, or 500 Dt., whichever is greater.

(f) Penalties for Noncompliance With the OFO

Except as provided in Section 12.17(d)(i), if a Shipper does not comply with the OFO, then:

(i) for an OFO requiring Shipper(s) to reduce scheduling variances, Shipper shall pay for variances in excess of the reduced tolerance level of 5% or 500 Dt. for Shippers under Rate Schedule SCT and 5% for Shipper(s) under other Rate Schedules, the greater of (a) two (2) times the highest daily price published in Gas Daily, Daily Price Survey, Citygates -Chicago LDCs or Citygates - Mich.-Mich Con, whichever is greater, for the day (Saturday and Sunday shall be the preceding Friday price) that Daily Scheduling Charges are incurred, or (b) the following:

Varian	ce		Rat	te	
5% - :>10% - :>15% - :>50%	15%	\$ 50 \$100	0.00	per per per per	Dt. Dt.

FERC Docket: RP00-395-003

Original Sheet No. 272E Original Sheet No. 272E : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(ii) for an OFO requiring Shipper(s) to eliminate Unauthorized Overruns, Shipper shall pay for all Quantities taken in excess of the reduced daily tolerance of zero, the greater of (a) two (2) times the highest daily price published in Gas Daily, Daily Price Survey, Citygates - Chicago LDCs or Citygates -Mich.-Mich Con, whichever is greater, for the day (Saturday and Sunday shall be the preceding Friday price) that Unauthorized Overrun Penalties are incurred, or (b) the following:

Overrun		Kate	Rate			
0% -	5%	\$ 15.00 pe	er Dt.			
> 5% -	10%	\$ 25.00 pe	er Dt.			
>10% -	15%	\$ 50.00 pe	er Dt.			
>15% -	50%	\$100.00 pe	er Dt.			
>50%		\$200.00 pc	er Dt.			

The above overrun tiers shall be determined by comparing the aggregate amount of contract entitlement nominated to a Point of Delivery with the aggregated amount of Gas flowing through such Point of Delivery. It shall be the responsibility of the meter operator at a Point of Delivery to provide in writing to Panhandle an allocation by contract of the gas flowing through such meter.

If a Shipper takes Quantities of Gas in excess of MDCQ which cause a Shipper to incur a penalty under this Section 12.17(f)(ii), such Quantities of Gas will not be considered in any calculation of an otherwise applicable penalty pursuant to Section 12.17(f)(i).

For purposes of calculating the Unauthorized Overrun Penalty under this Section 12.17, a Shipper shall be permitted to aggregate the MDCQs of its firm Transportation Agreements at a designated Point of Delivery. For purposes of aggregation under this Section 12.17(f), Shipper shall have scheduled quantities under the firm Transportation Agreements to be aggregated to no other Point of Delivery.

- (iii) for any other OFO, Shipper shall, for all Quantities tendered or taken in violation of the OFO, pay two (2) times the greater of the highest daily price published in Gas Daily, Daily Price Survey, Citygates - Chicago LDCs or Citygates - Mich.-Mich Con for the day (Saturday and Sunday shall be the preceding Friday price) that OFO penalties are incurred.
- (g) Indemnification

Compliance with an OFO and the other terms and conditions of Panhandle's FERC Gas Tariff is essential to Panhandle's ability to provide deliveries and services under all rate

FERC Docket: RP00-395-003

Original Sheet No. 272F Original Sheet No. 272F : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

schedules. A failure by one or more Shippers to comply with an OFO may affect Panhandle's ability to provide such deliveries and services. In such event and in addition to other provisions hereof and not in lieu of any other remedies or defenses available in law or at equity with respect to any person, Panhandle will have no liability or responsibility for its inability to provide deliveries and services to any Shipper failing to comply with an OFO and will be indemnified and held harmless by the Shipper(s) failing to comply with Panhandle's FERC Gas Tariff and in particular the provisions of this Section 12.17 against any claims related to the failure to provide deliveries and services, except to the extent such claims are the result of Panhandle's negligence, bad faith or wilful misconduct.

12.18 Third-Party Management Services

Subject to the conditions set forth in this Section, a Shipper may obtain services from a third-party provider to manage imbalances between actual receipts and deliveries; to manage variances between scheduled and actual deliveries; and to supply gas for overruns.

- Panhandle and the third-party provider have entered into an agreement which defines how such provider will accommodate Shipper's imbalances, scheduling variances, or overruns, how the provider is to make the corresponding operational changes, the limitations on the level of imbalances, scheduling variances and overruns to be accommodated and the consequences if such levels are exceeded or operational changes are not made. The agreement must provide Panhandle with the ability to call on the third-party provider on a basis consistent with service offered by the third-party provider to the Shipper. The agreement must also specify a predetermined allocation methodology and shall specify the extent to which and the conditions under which the Shipper shall be kept whole because the third-party provider is agreeing to take the imbalance, scheduling variance or overrun. If there is an operational balancing agreement at the point at which the imbalance management service is to be provided, the agreement must also provide that Panhandle shall not be responsible for balancing within the agreed limits of the management service.
- (b) Panhandle and the Shipper have entered into an agreement designating the Service Agreements for which the third-party provider will take the imbalance, scheduling variance, or overrun and designating the point(s) at which the third-party provider will provide the imbalance management service. The point(s) designated must have electronic real-time metering or must be otherwise agreeable to Panhandle.

FERC Docket: RP00-395-003

Original Sheet No. 272G Original Sheet No. 272G: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

The conditions set forth in this Section are minimum conditions that all third-party providers and Shippers utilizing such services must satisfy. When a specific third-party management service is proposed, Panhandle may require the third-party provider and Shipper to satisfy additional conditions. Panhandle shall not be obligated to enter into any agreement to accept third-party imbalance management services which would, in Panhandle's reasonable judgment, impair its ability to meet its existing system requirements or which would not relieve Panhandle of the need to manage (to the extent of the third-party service) the Shipper's imbalances, scheduling variances and overruns.

FERC Docket: RP97- 29-004

Second Revised Sheet No. 273 Second Revised Sheet No. 273: Superseded

Superseding: Sub First Revised Sheet No. 273
GENERAL TERMS AND CONDITIONS
(Continued)

13. POLICY FOR CONSTRUCTION OF NEW RECEIPT OR DELIVERY FACILITIES

All requests for new interconnects must be made in writing. The party requesting the new interconnect shall reimburse Panhandle or cause Panhandle to be reimbursed for any and all reasonable costs and expenses incurred in constructing, establishing or modifying the facilities required to establish a new interconnection on existing facilities for receipt or delivery of Gas hereunder. In the alternative, the party requesting the interconnect may perform the construction at its own cost in compliance with Panhandle's technical requirements. In this regard, all new receipt and delivery meters shall require EGM which shall be owned by Panhandle and all new delivery meters shall require a Flow Control Device with pressure override features that can be remotely operated by Panhandle. Panhandle also must be the custody transfer party at the proposed facility.

In addition to the above requirements, Panhandle will agree to construct a new interconnect or modify an existing interconnect on the following terms:

- (1) The construction of the new interconnect will not create any significant operational problems for Panhandle;
- (2) The proposed interconnect will not adversely affect the rendition of existing service or adversely alter the operation of the pipeline system;
- (3) The new interconnect must be at a mutually agreeable location on the Panhandle system; provided, however, that Panhandle may not deny a customer's request for specific placement of the interconnect without adequate operational, environmental, or legal justification;
- (4) In order to properly and prudently design and size the new interconnect, Panhandle must be provided with reasonable and reliable data concerning the interconnecting facilities, including delivery pressures, and anticipated hourly, daily, monthly and annual volume levels of the service that supports the new interconnect and such other data as is reasonably required to construct the interconnect facility;

FERC Docket: RP97- 29-005

Second Revised Sheet No. 273A Second Revised Sheet No. 273A: Superseded

Superseding: First Revised Sheet No. 273A

GENERAL TERMS AND CONDITIONS

(Continued)

(5) The new interconnect must not result in any minimum pressure receipt or delivery requirement by Panhandle, unless Panhandle agrees otherwise, such agreement shall not be unreasonably withheld, and Panhandle shall not be responsible for any downstream parties' facilities, the operation or maintenance of such facilities, or the delivery of any unauthorized volumes to the facilities;

- (6) The service supporting the interconnect as well as the construction of the new facilities must conform with the provisions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 as well as applicable regulatory requirements;
- (7) The proposed interconnect must not cause Panhandle to be in violation of any applicable environmental or safety laws or regulations with respect to the facilities required to establish an interconnect with Panhandle's existing facilities; and
- (8) The proposed interconnect must not cause Panhandle to be in violation of its right-of-way agreements or any other contractual obligations with respect to the interconnect facilities.

Panhandle will respond to each request for interconnect facilities within sixty (60) days after receiving a written request containing reasonably sufficient and reliable information necessary for Panhandle to make a determination pursuant to this Section 13.

FERC Docket: RP00-395-003

First Revised Sheet No. 274 First Revised Sheet No. 274: Superseded

Superseding: Original Sheet No. 274

GENERAL TERMS AND CONDITIONS (Continued)

14. POOLING AND IN-FIELD TRANSFERS OF STORAGE INVENTORY

14.1 Pooling

(a) A Transportation Agreement may become a
Corresponding Transportation Agreement by:
(1) designating a Pool Point as a Point of Receipt,
and (2) designating a Pooling Transportation
Agreement(s) as the Transportation Agreement
delivering Gas to the Corresponding Transportation
Agreement at the Pool Point. The Maximum Daily
Contract Quantity of the Corresponding
Transportation Agreement assigned to the Pool Point
shall not be less than the Maximum Daily Contract
Quantities of the firm Pooling Transportation
Agreement(s) designated in the Corresponding
Transportation Agreement.

Deliveries under the Pooling Transportation Agreement(s) designated in the Corresponding Transportation Agreement on any Day shall be equivalent to the total Quantities of Gas nominated for receipt at the Pool Point under the Corresponding Transportation Agreement.

(b) Quantities of Gas delivered at the Pool Point under a Pooling Transportation Agreement shall be deemed to have been delivered to the extent of the total amount of Corresponding Shipper's nominated Quantities of receipt at the Pool Point. It shall be the responsibility of the Shipper under the Pooling Transportation Agreement to assure that the total nominations of Gas to be received at the Pool Point under the Corresponding Transportation Agreement is in balance with the total Quantities of Gas such Shipper causes to be received into Panhandle's system, adjusted for Fuel Reimbursement. If the Pooling Shipper has executed a Gas Parking Service Agreement and if the Pooling Shipper has so elected in accordance with the terms of Rate Schedule GPS, Panhandle shall utilize the Pooling Shipper's Gas Parking account to balance the total Quantities of Gas delivered to the Pool Point with the total nominations of gas to be received at the Pool Point under the Corresponding Transportation Agreement. To facilitate scheduling of Corresponding Transportation Agreements, Panhandle shall accept from the Pooling Shipper a prescribed methodology for ranking of Corresponding Shippers' nominated Quantities for receipt at the Pool Point. Such methodology shall be provided by the Pooling Shipper to Panhandle by the nomination deadline for the applicable nomination cycle. In the absence of such methodology, scheduling shall be pro-rata.

FERC Docket: RP03-278-000

Second Revised Sheet No. 275 second Revised Sheet No. 275: Superseded

Superseding: First Revised Sheet No. 275

GENERAL TERMS AND CONDITIONS (Continued)

Each Shipper under a Corresponding Transportation Agreement shall be responsible for assuring that its receipts are in balance with the amounts of its deliveries, adjusted for Fuel Reimbursement.

(c) In addition to the other requirements herein, the Shipper under the Pooling Transportation Agreement at the time of nomination shall identify each Corresponding Shipper, the Corresponding Transportation Agreement and shall state the Quantities of Gas to be delivered to each Corresponding Shipper under each Corresponding Transportation Agreement at the Pool Point.

14.2 In-Field Title Transfers of Storage Inventory

- (a) A Shipper under a Storage Rate Schedule may transfer by sale or otherwise Stored Volumes to any Shipper under another Service Agreement that receives Storage service from the same Storage area, either Market Area Storage Facilities, or Field Area Storage Facilities, if:
 - (i) Both the transferee and transferor of the Stored Volumes provides Panhandle with verification of the transfer in writing;
 - (ii) The transfer does not cause the Stored Volumes to exceed the Maximum Stored Quantity specified in any Service Agreement; and
 - (iii) The transfer does not cause the transferor or transferee to be in violation of any requirement of Rate Schedule PS.
 - (iv) The transfer does not result in an increase in Panhandle's injection or withdrawal obligations that would impair Panhandle's ability to meet all of its service obligations of equal or higher priority.
- (b) Panhandle will recognize the transfer for purposes of computing available Stored Volumes and applicable Withdrawal and Injection Quantities on a prospective basis within one Business Day after receiving the written verification required by Section 14.2(a)(i).

FERC Docket: RP00-395-003

Eighth Revised Sheet No. 276 Eighth Revised Sheet No. 276: Superseded

Superseding: Seventh Revised Sheet No. 276

GENERAL TERMS AND CONDITIONS (Continued)

15. PROCEDURES FOR ASSIGNMENT OF SERVICE RIGHTS

- 15.1 Service Rights That May Be Assigned
 - (a) The Service Agreements of a Releasing Shipper with Service Agreements under Rate Schedule FT, EFT, IOS, WS, PS, FS, LFT or HFT may be assigned, in whole or in part, to a Replacement Shipper for any time period up to the remaining term of the Service Agreement.
 - (b) A Releasing Shipper is a Shipper that has executed a Release Notice in accordance with Section 15.2.
 - (c) A Replacement Shipper is a party that (1) meets the eligibility criteria for Shippers set forth in Section 6, including the credit standards set forth in Section 6.2, except that Panhandle shall not require the Replacement Shipper to make the prepayment required by Section 6.9, (2) accepts assignment of the Service Agreement of a Releasing Shipper pursuant to this Section 15, and, (3) executes a Capacity Release Service Agreement as prescribed by this Tariff. Panhandle will issue and execute an Addendum to the Capacity Release Service Agreement when the Releasing Shipper's Service Agreement is assigned to the Replacement Shipper.
 - (d) A Replacement Shipper may assign, in whole or in part, the Addendum to its Capacity Release Service Agreement as a Releasing Shipper pursuant to this Section 15.
- 15.2 Initiation of the Assignment of Service Rights
 - (a) A Releasing Shipper may initiate the assignment of its Service Agreement by electronically submitting a Release Notice in the form prescribed by this Tariff. Such Release Notice shall be posted on the Messenger system. Panhandle shall accept the upload of prearranged capacity release transactions. A Releasing Shipper may withdraw its Release Notice at any time prior to the close of the posting period where unanticipated circumstances justify and no minimum bid has been made.
 - (b) In the Release Notice the Releasing Shipper must specify the term, quantity and Minimum Rate expressed in dollars and cents or percent of Maximum Rate the Releasing Shipper is willing to accept and whether the assignment is subject to recall and the applicable recall conditions as well

Effective Date: 10/18/1999 Status: Effective

FERC Docket: RP99-482-000

Fourth Revised Sheet No. 277 Fourth Revised Sheet No. 277: Superseded

Superseding: Sub Third Revised Sheet No. 277
GENERAL TERMS AND CONDITIONS
(Continued)

as methods and rights associated with returning the previously recalled capacity to the Replacement Shipper. The Releasing Shipper may specify further objective and nondiscriminatory conditions in its Release Notice, such as alternative economic criteria for evaluating bids which may include highest rate, net revenues or present value, whether volumetrically stated bids will be accepted, whether a volumetric commitment is required, alternative methods for breaking ties, whether contingent bids will be accepted and, if so, the method by which contingent bids will be evaluated and when the contingency must be removed, and whether bids for a portion of the capacity offered will be accepted or whether a prepayment deposit, not to exceed the prepayment deposit set forth in Section 6.9, will be required to be paid to the Releasing Shipper; provided, however, the Releasing Shipper may not specify conditions which are in conflict with the Service Agreement, the applicable Rate Schedule, or these General Terms and Conditions.

Panhandle shall support volumetric releases with volumetric commitments by fully accounting for volumetric and reservation components, consistent with the rules and regulations enunciated by the Commission.

(c) The Releasing Shipper may arrange for the assignment of its Service Agreement to a Designated Replacement Shipper prior to submitting its Release Notice to Panhandle. Such Designated Replacement Shipper must be on the approved bidders list and shall acknowledge the Release Notice through the Messengerb system as a prerequisiteto the posting of the Release Notice. However, the assignment will not become effective except as provided in this Section 15.

15.3 Posting

(a) Shipper will post the information provided in each Release Notice on the Messengerp system. The posting will include the MDCQ or MSQ available for assignment, the Points of Receipt and Delivery and the released quantity for each point, the term of the assignment, whether the assignment is firm or recallable and the recall conditions, if applicable, the Minimum Rate expressed in dollars and cents or percent of Maximum Rate that the Releasing Shipper will accept, and any other conditions the Releasing Shipper specifies for the assignment in accordance with Section 15.2(b).

FERC Docket: RP03-479-000

Ninth Revised Sheet No. 278 Ninth Revised Sheet No. 278: Superseded

Superseding: Eighth Revised Sheet No. 278
GENERAL TERMS AND CONDITIONS
(Continued)

(b) The posting will commence upon receipt of the Release Notice; unless otherwise specified by the Releasing Shipper.

15.4 Bidding For Assignments

- (a) Except as provided in Section 15.4(b), a party must bid on the Messenger system to accept assignment of the Service Agreement under the conditions posted in conjunction with a Release Notice. Bids must be expressed in dollars and cents or percents of Maximum Rate, whichever is stated in the Release Notice. The maximum reservation rate that may be bid shall not exceed the Maximum Rate for the applicable service being released as set forth on the applicable currently effective tariff sheets. In order to submit a valid bid, any party, including a Designated Replacement Shipper, must be on the approved bidders list. To be on the approved bidders list, a party must meet the provisions of Section 6 of these General Terms and Conditions and must have executed a Capacity Release Service Agreement with Panhandle in the form as set forth in this Tariff (Capacity Release Service Agreement). A party shall remain on the approved bidders list until such party notifies Panhandle to the contrary, no longer meets the credit qualifications in Section 6.2 of these General Terms and Conditions or is suspended from the approved bidders list in the event and for such time as such party fails to pay part or all of the amount of any bill for service in accordance with Section 17 of these General Terms and Conditions.
- (b) No bids are required for Service Agreements where the Releasing Shipper notifies Panhandle in its Release Notice that it has arranged for the assignment of its Service Agreement to a Designated Replacement Shipper that is on the approved bidders list and:
 - the Designated Replacement Shipper has agreed to a term for the assignment of not more than thirty-one (31) Days, or

FERC Docket: RP03-479-000

Sixth Revised Sheet No. 278A sixth Revised Sheet No. 278A: Superseded Superseding: Fifth Revised Sheet No. 278A

GENERAL TERMS AND CONDITIONS

ENERAL TERMS AND CONDITION: (Continued)

(ii) the Designated Replacement Shipper has agreed to pay the applicable Maximum Rate for the service and has agreed to all other conditions applicable to the assignment, unless Section 15.4(b)(i) applies.

Such assignments for which no bids are required may be submitted electronically and shall be posted on the Messenger system upon confirmation by the Replacement Shipper. The Capacity Release Service Agreement Addendum number shall be provided when such assignment is posted.

- (c) Bids to accept assignment of the Service Agreement must specifically identify any contingencies and may be made through the Messenger system only. If the Releasing Shipper requires a prepayment deposit, the party submitting the bid must concurrently submit the prepayment deposit through Electronic Funds Transfer to a designated bank account established by Panhandle.
- (d) The capacity release procedure is applicable to all parties involved in the Capacity Release process; however, it is only applicable if 1) all information provided by the parties to the transaction is valid and 2) there are no special terms or conditions of the release. All stated times are Central Time.
 - (i) For biddable releases (less than 1 year)
 - 12:00 p.m. Time by which Releasing Shipper shall post the Release Notice on the Messenger system
 - 1:00 p.m. Bidding period ends
 Evaluation period begins during which any contingency is eliminated, determination of best bid is made and ties are broken
 - 2:00 p.m. Evaluation period ends and award posting if no match required Match or award is communicated to Designated Replacement Shipper
 - 2:30 p.m. Match response is due Designated Replacement Shipper
 - 3:00 p.m. Award posting where match is required to Replacement Shipper or Designated Replacement Shipper

FERC Docket: RP02-466-000

Second Revised Sheet No. 278B Second Revised Sheet No. 278B : Superseded

Superseding: First Revised Sheet No. 278B
GENERAL TERMS AND CONDITIONS
(Continued)

(ii) For biddable releases (1 year or more)

The timeline is the same as biddable releases for less than 1 year except that the Releasing Shipper shall post the Shipper Notice on the Messenger system by 12:00 p.m. Central Time four (4) Business Days before the award. Open season ends no later than 1:00 p.m. Central Time on the Business Day before timely nominations are due (open season is three Business Days).

FERC Docket: RP00-395-003

Sixth Revised Sheet No. 279 Sixth Revised Sheet No. 279: Superseded

Superseding: Fifth Revised Sheet No. 279

GENERAL TERMS AND CONDITIONS (Continued)

(iii) If the Releasing Shipper specifies a bid evaluation methodology other than highest rate, net revenue or present value, or a permanent release or any other special conditions, the above timelines shall apply; provided, however, one additional Business Day will be added to the evaluation period. Such extended evaluation period will cause Gas flow to be at least one day later than Gas could flow under the timeline set forth in Section 15.4(d) (i) or Section 15.4(d) (ii).

- (e) A party may submit only one bid at a time in response to a Release Notice. Once a party has submitted a bid, it may withdraw the bid through the Messenger system at any time prior to the close of the posting period specified in Section 15.3(c) without prejudice to its submitting another bid with an economic value equal to or greater than the economic value of the withdrawn bid. If a bid is not withdrawn prior to the close of the posting period, the bid shall be binding on the party submitting the bid, subject to the satisfaction of any contingencies the Releasing Shipper permits to be satisfied after the close of the bidding period.
- (f) Bids to accept assignment will be posted on the Messenger system with any contingencies identified and with the bidder's identity deleted.

15.5 Selection of Replacement Shipper

(a) The Addendum to the Capacity Release Service Agreement will be issued to the Replacement Shipper whose bid provides the greatest economic value, as defined by the Releasing Shipper in accordance with Section 15.2(b) or, if the Releasing Shipper does not specify a method of determining economic value, the economic value will be determined by utilizing the present value method, and whose bid meets all the conditions placed upon the assignment by the Releasing Shipper as provided for in Section 15.2(b), provided that if the Releasing Shipper has permitted contingent bids, the selection of a Replacement Shipper may be delayed in accordance with the contingencies permitted in the Release Notice. If the bids of two or more Replacement Shippers provide the same economic value and both meet all the conditions placed upon the

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 280 Fourth Revised Sheet No. 280: Superseded

Superseding: Third Revised Sheet No. 280

GENERAL TERMS AND CONDITIONS (Continued)

assignment, Panhandle will tender the Addendum to the Capacity Release Service Agreement to the Replacement Shipper who submitted its bid first unless, in accordance with Section 15.2(b), the Releasing Shipper has specified an alternative method for breaking ties, in which event the method specified by the Releasing Shipper shall be used. If there are multiple bids meeting the minimum conditions, Panhandle shall rank the bids and award the bids, best bid first, until all offered capacity is awarded. The Capacity Release Service Agreement Addendum number shall be assigned when capacity is awarded.

- (b) If the Releasing Shipper has notified Panhandle in the Release Notice that it has arranged a Designated Replacement Shipper and no party is eligible to submit a bid for the assignment, Panhandle will issue the Addendum to the Capacity Release Service Agreement to the Designated Replacement Shipper.
- (c) If the Releasing Shipper has notified Panhandle in the Release Notice that it has arranged a Designated Replacement Shipper who has agreed to pay less than the applicable Maximum Rate for the service, the Designated Replacement Shipper may match the bid that has the greatest economic value within the time period specified in Section 15.4.
- (d) Panhandle will issue the Addendum to the Capacity Release Service Agreement within one hour of the posting of the award of a release. If all the information provided by the Releasing Shipper, bidder, and Designated Replacement Shipper is valid, the Replacement Shipper is creditworthy, and there are no special terms and conditions, Panhandle will issue and execute the Addendum at the same time.
- (e) Panhandle shall post the winning bid and bidder on the Messenger system for ninety (90) Days.
- (f) If an Addendum to the Capacity Release Service Agreement is not tendered to a party submitting a prepayment deposit with its bid, the prepayment deposit will be refunded with carrying charges to such party within two (2) Business Days.

FERC Docket: RP03-479-000

Fifth Revised Sheet No. 281 Fifth Revised Sheet No. 281: Superseded

Superseding: Fourth Revised Sheet No. 281

GENERAL TERMS AND CONDITIONS
(Continued)

15.6 Rights and Obligations of Releasing Shipper

- (a) Except as provided in Section 15.6(f), the Releasing Shipper shall continue to be responsible to Panhandle for all reservation charges and reservation surcharges for Transportation or Storage service provided pursuant to the Shipper's applicable Service Agreement and all other charges and surcharges that would be charged to the Releasing Shipper if the Service Agreement had not been assigned. The Releasing Shipper will not be responsible for Commodity charges or volumetric surcharges which are applicable under Panhandle's Tariff to Quantities of Gas actually transported except as provided in Section 15.6(b) below for a Releasing Shipper with a Negotiated Rate.
- (b) A credit will be included on each of the Releasing Shipper's monthly bills to reflect the reservation fee invoiced to Replacement Shippers less any negotiated marketing fee retained by Panhandle pursuant to Section 15.8 provided, however, that Panhandle and Releasing Shipper may, in connection with a Negotiated Rate based on a rate design other than straight fixed variable, agree upon a payment obligation and crediting mechanism that varies from or is in addition to the provisions of this Section 15.6in order to establish the basis of accounting for revenues from a Replacement Shipper as a means of preserving the economic bases of the Negotiated Rate. In the event of a release with a volumetrically stated rate, the rate shall be no greater than the 100% load factor equivalent of Panhandle's maximum reservation charge currently applicable to the service released and shall be credited to the Releasing Shipper's monthly bill for reservation charges applicable to such service.

If a Replacement Shipper fails to pay all or any part of the reservation fee so credited within thirty (30) days of its due date, then such unpaid amount will be charged to the Releasing Shipper's next monthly bill and will be due and payable by the Releasing Shipper in accordance with Section 17 of these General Terms and Conditions.

The payment of the reservation charge by the Replacement Shipper on a one-part volumetric basis shall not relieve Panhandle, the releasing Shipper or the Replacement Shipper from their obligations to conform to all of the terms and conditions of this Section 15.

FERC Docket: RP03-449-000

Ninth Revised Sheet No. 282 Ninth Revised Sheet No. 282: Superseded

Superseding: Eighth Revised Sheet No. 282
GENERAL TERMS AND

GENERAL TERMS AND CONDITIONS (Continued)

(c) When a Releasing Shipper assigns Transportation service rights under Rate Schedule FT, EFT, LFT or HFT, the MAIQ under the Transportation Agreement will be reduced proportionately by the amount of MDCQ assigned. Any imbalance in excess of this reduced MAIQ must be reduced to the new tolerance level in the month following the completion of the assignment to avoid balancing charges pursuant to Section 12.11.

- (d) When a Releasing Shipper partially assigns its service rights under a Service Agreement by assigning service rights between specific Point(s) of Receipt and Delivery or by assigning only a portion of the Maximum Daily Contract Quantity, the Releasing Shipper's Service Agreement shall be deemed to be modified in accordance with the assignment and the Releasing Shipper may not utilize the service rights assigned during the term of the assignment.
- (e) If the Releasing Shipper's Service Agreement has been assigned to a Designated Replacement Shipper for a period of not more than thirty-one (31) Days, the Releasing Shipper may not assign the Service Agreement to the same Replacement Shipper for a term of not more than thirty-one (31) Days until twentyeight (28) Days after the termination of the assignment unless:
 - (i) the Service Agreement is posted on the Messenger system for bidding in accordance with this Section 15 or
 - (ii) the capacity release becomes effective after September 30, 2002 and the Replacement Shipper has agreed to pay the applicable Maximum Rate for the service, is on the approved bidders list and all the other conditions applicable to the assignment.
- (f) If the Releasing Shipper's Service Agreement is assigned permanently for the remaining term of the agreement, subject to Panhandle's agreement, the Releasing Shipper shall have no further responsibility or obligations under the Service Agreement.
- (g) Releasing Shippers may, to the extent permitted as a condition of the capacity release, recall released capacity. The recall notification to Panhandle shall specify the Quantity in terms of total released capacity entitlements.

FERC Docket: RP03-449-000

First Revised Sheet No. 282A First Revised Sheet No. 282A: Superseded Superseding: Original Sheet No. 282A

RESERVED FOR FUTURE USE.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 283 Fourth Revised Sheet No. 283: Superseded

Superseding: Third Revised Sheet No. 283

GENERAL TERMS AND CONDITIONS (Continued)

15.7 Rights and Obligations of the Replacement Shipper

- (a) A Replacement Shipper shall have all the rights and obligations specified under the applicable Rate Schedule, Service Agreement, and General Terms and Conditions, including the right to reserve primary point capacity up to its contract demand, subject to available capacity, unless the conditions prescribed for the assignment require otherwise.
- (b) When the Capacity Release Service Agreement Addendum is issued, a Replacement Shipper may submit nominations pursuant to Section 8 of these General Terms and Conditions for the next available nomination cycle and any nomination cycle thereafter for which the Addendum to the Capacity Release Service Agreement is effective.
- (c) The Replacement Shipper will be responsible for any imbalance between receipts of Gas and deliveries of Gas, and for all costs and all rates, charges, penalties and fees for Transportation or Storage service provided under the Capacity Release Service Agreement.
- (d) Commodity charges and volumetric surcharges which are applicable under Panhandle's Tariff to Quantities of Gas actually transported shall be a separately stated charge on the Replacement Shipper's Monthly Bill and shall be at the Maximum Rate contained on the currently effective Tariff sheet unless Panhandle agrees to a lesser rate.

15.8 Marketing Fee

The Releasing Shipper and Panhandle may agree that, in return for Panhandle actively marketing the Releasing Shipper's Service Agreement, Panhandle will reduce the Releasing Shipper's invoice credit by a negotiated marketing fee.

Effective Date: 10/18/1999 Status: Effective

FERC Docket: RP99-482-000

First Revised Sheet No. 284 First Revised Sheet No. 284: Superseded

Superseding: Original Sheet No. 284

GENERAL TERMS AND CONDITIONS
(Continued)

16. ASSIGNMENT OF UPSTREAM SERVICE RIGHTS

This section sets forth the terms and conditions under which Panhandle will assign in whole or in part, the rights under executed Transportation Service Agreements between Panhandle and upstream pipelines in accordance with 18 C.F.R. 284.242. The Transportation Agreements subject to assignment pursuant to this Section are:

Colorado Interstate Gas Company - Rate Schedule X-38 Trunkline Gas Company - Rate Schedule PT-Firm Stingray Pipeline Company - Rate Schedule FT

Assignments of Panhandle's service rights under the above Transportation Service Agreement shall occur as follows:

- 16.1 All assignments pursuant to Section 16 shall be permanent.
- 16.2 Panhandle shall give notice on the Messengerb system of its upstream firm service available for assignment under this Section for at least thirty (30) Days commencing February 1, 1993, with initial assignments to become effective on April 1, 1993, or the effective date of this Section 16, whichever is later, to those making requests for assignment to Panhandle in writing within such thirty (30) Day period. Thereafter, any of Panhandle's service rights that have not been assigned during such thirty (30) Day period shall remain posted on the Messengerb system and available for additional assignment until all such capacity has been assigned or otherwise disposed of.
- 16.3 If upstream firm capacity is not assigned and Panhandle is unable to release available upstream capacity pursuant to the upstream pipeline's capacity assignment program, it shall recover costs associated with such "stranded" upstream capacity as a transition cost, pursuant to Section 18.9.

Effective Date: 04/01/1997 Status: Effective

FERC Docket: RP97- 4-002

First Revised Sheet No. 285 First Revised Sheet No. 285: Superseded

Superseding: Original Sheet No. 285
GENERAL TERMS AND CONDITIONS
(Continued)

17. STATEMENTS AND PAYMENTS

17.1 Panhandle shall send to Shipper, on or before the 9th Business Day after the end of the month, a billing of charges for service rendered in the preceding Month. Such charges may be based on estimated Quantities of Gas if actual Quantities of Gas are unavailable in time to prepare the billing. In that event, Panhandle shall provide, in the succeeding Month's billing, an adjustment based on any difference between actual Quantities of Gas and estimated Quantities of Gas. If Shipper has contracted for multiple services with Panhandle, then Shipper may receive a master bill which shall contain multiple charges for the multiple services and/or contracts for which Shipper has contracted. Panhandle may waive the applicability of any fixed Monthly charges during periods prior to the completion of Panhandle's, Shipper's or third party's facilities required to perform the service involved.

17.2 (a) Payment

Shipper shall pay Panhandle by Electronic Funds Transfer to a designated bank account established by Panhandle for billed amounts equal to or greater than \$100,000. For billed amounts less than \$100,000 Shipper shall pay Panhandle by check to Panhandle's designated Post Office Box or at Shipper's election by Electronic Funds Transfer to a designated bank account established by Panhandle. Payments shall be made by Shipper to Panhandle within 10 Days from the date on which the bill is sent (Due Date) for all charges billed in accordance with the provisions of the applicable Rate Schedules. Payments made by Electronic Funds Transfer shall be considered to have been made on the date when such payment of good funds is received by Panhandle.

Effective Date: 04/01/1997 Status: Effective

FERC Docket: RP97- 4-005

Sub Second Revised Sheet No. 286 Sub Second Revised Sheet No. 286: Superseded

Superseding: First Revised Sheet No. 286
GENERAL TERMS AND CONDITIONS
(Continued)

(b) Unpaid and Disputed Bills

Should Shipper fail to pay part or all of the amount of any such bill, interest thereon shall accrue at an average prime interest rate computed in a manner consistent with Section 154.501(d) of the Commission's Regulations, from the Due Date until date of receipt of payment by Panhandle. If such failure to pay continues 30 Days after payment is due, Panhandle, in addition to any other remedy it may have, may suspend further service to Shipper until such amount is paid after Panhandle provides Shipper with 20 Days prior written notice; provided, however, that if Shipper in good faith disputes in writing the amount of any such bill or parts thereof and pays to Panhandle such amounts as it concedes to be correct, and at any time thereafter within 30 Days of demand made by Panhandle shall furnish a good and sufficient surety bond guaranteeing payment to Panhandle of the amount ultimately found due upon such bills after a final determination, which may be reached either by agreement or judgment of the courts as may be the case, then Panhandle shall not be entitled to suspend further service unless and until default be made in the conditions of such bond. If resolution of the dispute is in favor of Shipper and the Shipper furnished a surety bond instead of paying the disputed amount, then Panhandle shall reimburse Shipper for the cost of securing that surety bond. No payment by Shipper of the amount of a disputed bill shall prejudice the right of Shipper to claim an adjustment of the disputed bill.

- 17.3 In the event that an error is discovered in the invoiced amount hereunder, such error shall be adjusted within 30 Days of the determination thereof, provided that claim therefor shall have been made in writing.
 - (a) Measurement data corrections should be processed within 6 months of the production month with a 3 month rebuttal period.
 - (b) The time limitation for disputes of allocations should be 6 months from the date of the initial month-end allocation with a 3-month rebuttal period.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-005

Original Sheet No. 286A Original Sheet No. 286A: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(c) Prior period adjustment time limits should be 6months from the date of the initial transportation invoice and 7 months from date of initial sales invoice with a 3-month rebuttal period, excluding government-required rate changes.

The above shall not apply in the case of deliberate omission or misrepresentation or mutual mistake of fact. Parties' other statutory or contractual rights shall not otherwise be diminished by these standards. Effective Date: 05/25/1996 Status: Effective

FERC Docket: RP96-214-000

First Revised Sheet No. 287 First Revised Sheet No. 287: Superseded

Superseding: Original Sheet No. 287
GENERAL TERMS AND CONDITIONS
(Continued)

17.4 Shipper agrees to reimburse Panhandle for all filing or other fees, in connection with any service subject to these General Terms and Conditions, that Panhandle is obligated to pay to the Commission or to any other governmental authority having jurisdiction. The term "fees" as used herein, shall mean any fee or charge now or hereafter levied, assessed or made by any governmental authority on the Gas itself or on the act, right or privilege of producing, severing, gathering, transporting, handling, selling, receiving or delivering Gas, however such fees or charges are measured.

17.5 Refunds

Refunds due Shipper pursuant to the terms of this tariff or orders of the Commission shall be paid by Electronic Funds Transfer to a bank account established by Shipper if:

- (a) Shipper has transmitted payment for services to Panhandle by Electronic Funds Transfer within the twelve Month period preceding the date of the refunds; and
- (b) Shipper has designated in writing a bank account for the receipt of Electronic Funds Transfers at least 30 Days prior to the date established for refunds; and
- (c) The amount of the refund, including interest, equals or exceeds \$100,000.

Refunds not paid by Electronic Funds Transfer shall be paid by check.

17.6 Order of Discounting

If and when Panhandle discounts the rates applicable for service under any Service Agreement under Rate Schedules included in Panhandle's FERC Gas Tariff, the components of the currently applicable Maximum Rate shall be discounted in the following order: the first item of the overall charge discounted will be the GRI surcharge, followed by the base rate charge, stranded transportation cost surcharges and other transition cost surcharges, and last, the GSR surcharges. Other non-transition cost surcharges will be attributed in accordance with the applicable section of the General Terms and Conditions which provides for the surcharge as agreed by Panhandle and its shippers in individual proceedings.

Effective Date: 05/01/1993 Status: Effective

FERC Docket: RS92- 22-005

Original Sheet No. 288 Original Sheet No. 288: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

18. PERIODIC RATE ADJUSTMENTS AND SURCHARGES

Periodically rates and charges under Panhandle's FERC Gas Tariff shall be adjusted to reflect changes in Panhandle's expenditures for Take-or-Pay, the Gas Research Institute, the Annual Charge Adjustment, the Applicable Surcharge Adjustments and the Transition Cost Recovery Mechanism. Such expenditures shall include both expenditures chargeable to operations as well as expenditures includable in rate base. Panhandle and Shipper recognize that Panhandle shall from time to time experience changes in costs related to providing service under this FERC Gas Tariff, including, but not limited to, changes in the cost of labor, benefits, materials and supplies, taxes, and required rate of return, costs associated with the resolution of past disputes or outstanding uncertainties concerning amounts owed by Panhandle or attributable to Panhandle, and costs generated by decisions by the Commission, the courts or by an arbitration panel or other body having jurisdiction over Panhandle. It may be appropriate, equitable and consistent with cost responsibility to allocate such costs among Shippers based on or taking into account past period factors, such as contract demand levels, purchases or other factors related to a prior period of time. Panhandle shall have the right from time to time to make rate change filings which may include such costs and utilize an allocation methodology based in whole or in part on factors related to past periods. Specific allocation methodologies will be determined when Panhandle submits a filing to recover any of the aforesaid costs. Shipper shall have the right to intervene and protest any such filings.

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 289 Fourth Revised Sheet No. 289: Superseded

Superseding: Substitute Second Revised Sheet No. 289
GENERAL TERMS AND CONDITIONS

(Continued)

18.1 Gas Research Institute (GRI) Funding Unit

(a) Purpose

GRI, an Illinois not for profit corporation, has been organized for the purpose of sponsoring Research, Development and Demonstration (RD&D) programs in the field of natural and manufactured Gas for the purpose of assisting all segments of the Gas industry in providing adequate, reliable, safe, economic and environmentally acceptable Gas service for the benefit of Gas consumers and the general public in the United States. Panhandle is a collection agent for GRI

For the purpose of funding GRI's approved expenditures, and subject to the further terms and conditions set forth in the Stipulation and Agreement Concerning Post-1993 GRI Funding Mechanism and the Orders approving such Stipulation and Agreement found at Gas Research Institute, 62 F.E.R.C. 61,280 (1993), reh'q 63 F.E.R.C. 61,316 (1993), this Section 18.1 establishes a GRI Funding Unit which shall be collected for Quantities of Gas transported under Panhandle's transportation Rate Schedules; provided, however, such charge shall not be applicable (1) to discounted transactions except as expressly provided below, (2) to Shippers which are interstate pipelines and which include in their rates a charge for RD&D by GRI, (3) to the transportation of liquids and/or liquefiables, or (4) to transportation which is solely gathering service and provided, further, that a GRI Funding Unit shall not be applied to the same Quantity of Gas more than once.

- (b) Basis of the GRI Funding Unit
 - (1) Panhandle's Rate Schedules FT, EFT, SCT, IT, EIT, LFT and HFT shall include an increment for a GRI Reservation Funding Unit and a GRI Commodity Funding Unit, as applicable, for non-discounted transactions as stated on the effective Tariff Sheet Nos. 4, 5, 6, 7, 8, 15 and 17. Such GRI Funding Unit shall be that increment, adjusted to Panhandle's

Effective Date: 01/01/1994 Status: Effective

FERC Docket: TM94-3-28-001

Substitute First Revised Sheet No. 290 Substitute First Revised Sheet No. 290: Superseded

Superseding: Original Sheet No. 290
GENERAL TERMS AND CONDITIONS
(Continued)

measurement basis (dekatherms), which has been approved by Federal Energy Regulatory Commission Orders approving GRI's RD&D expenditures.

- (2) As to discounted transactions, if the discounted rate is less than the GRI Funding Unit, Panhandle shall remit that portion of the GRI Funding Unit actually collected. For purposes of discounted transactions, any GRI Funding Unit shall be considered to be the first component of rates discounted.
- (3) The applicable GRI Funding Unit shall be determined for high load factor firm shippers (load factor exceeding fifty (50) percent) and low load factor firm shippers (load factor of fifty (50) percent or less). The shipper's load factor for each firm service agreement shall be determined annually. The load factor shall be calculated from actual firm volumes using the most recent twelve (12) months of volumes available (including volumes from assignment of service rights pursuant to Section 15 herein) as a percent of firm annual contract entitlements. The Shipper's load factor determination shall remain in effect for the ensuing calendar year. For firm shippers without a historical load factor, for purposes of the GRI Funding Unit, the GRI Funding Unit shall be assessed each month based on actual throughput for each prior month of service until a twelve (12) month history is established.

(c) Filing Procedure

The initial GRI Funding Unit or any subsequent changes in such Funding Unit shall be filed by Panhandle at least thirty (30) Days prior to the proposed effective date.

The proposed effective date of filings pursuant to this subsection shall be the effective date prescribed by the Commission. Any such filing shall not become effective unless it becomes effective without suspension or refund obligation.

Effective Date: 01/01/1999 Status: Effective

FERC Docket: RP99-138-000

First Revised Sheet No. 290A First Revised Sheet No. 290A: Superseded

Superseding: Original Sheet No. 290A

GENERAL TERMS AND CONDITIONS (Continued)

(d) Remittance to GRI

Panhandle shall remit to GRI, not later than fifteen (15) Days after the receipt thereof, all monies actually collected and received by virtue of the GRI Funding Unit and the voluntary contribution mechanism, less any amounts properly payable to a Federal, State or Local authority relating to the monies received hereunder. In no event will Panhandle be required to remit to GRI amounts not actually collected, except as specified in this Section 18.1. For the amounts received through the voluntary contribution mechanism, Panhandle shall indicate to GRI the amounts applicable to specific projects and project areas, if so indicated by Shipper.

(e) Payment by Shipper

The amount of applicable Shipper's GRI Funding Unit shall be due and payable with the monthly billing in accordance with the General Terms and Conditions.

(f) Flowthrough of Refunds from GRI

To the extent Panhandle receives refunds from GRI because (1) Panhandle has remitted to GRI more than its target level, or (2) GRI has met its current year cash requirements, Panhandle will, in turn, flowback such amounts to its non-discounted firm Shippers in accordance with the Stipulation and Agreement previously described in Section 18.1(a).

(g) Voluntary Contribution Mechanism

Shipper shall be permitted to make voluntary contributions to GRI in such amounts and for such GRI project(s) or project area(s) as Shipper specifies. To facilitate the voluntary collection mechanism, Panhandle will include with Shipper's invoice a procedure to permit Shipper to designate contributions for GRI RD&D Programs. The voluntary contribution invoice mechanism is not a pipeline rate, rate provision nor term or condition of service. It is included in the invoice in accordance with the January 21, 1998, Stipulation and Agreement Concerning GRI Funding approved by the Commission in Gas Research Institute, 83 F.E.R.C. 61,093 (1998), order on reh'g, 83 F.E.R.C. 61,331 (1998). Any amounts remitted to Panhandle pursuant to the voluntary contribution mechanism shall be forwarded to GRI by Panhandle as described in Section 18.1(d) herein and are not refundable by Panhandle.

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 291 Fourth Revised Sheet No. 291: Superseded

Superseding: Third Revised Sheet No. 291
GENERAL TERMS AND CONDITIONS
(Continued)

18.2 Annual Charge Adjustment Provision (ACA)

(a) Purpose

Annual charges are assessed Gas pipelines by the Commission under Part 382 of the Commission's Regulations prior to each fiscal year in order to cover the cost of the operation of the FERC. For the purpose of recovering such charges assessed Panhandle by the Commission, this Section 18.2 establishes pursuant to Section 154.402 of the Commission's Regulations an ACA Surcharge to be applicable to quantities transported under Panhandle's Rate Schedules FT, EFT, SCT, IT, EIT, LFT and HFT. This ACA Surcharge is in addition to any amounts otherwise payable to Panhandle under said Rate Schedules.

(b) Basis of the ACA Unit Charge Rate

The ACA Unit Charge Rate shall be that increment, adjusted to Panhandle's measurement base (Dekatherm) and pressure base, if required, which has been approved by valid Commission Orders approving an annual charge unit rate.

(c) Filing Procedure

The ACA Surcharge or any subsequent changes in such surcharge shall be filed by Panhandle at least thirty (30) Days prior to the proposed effective date unless for good cause shown a lesser notice period is allowed by valid Commission Orders or the establishment by the Commission of the annual charge unit rate for a particular fiscal year prohibits Panhandle from making a timely filing. The proposed effective date of filings pursuant to this subsection shall be the effective date prescribed by the Commission. Such filing shall become effective without suspension or refund obligation.

FERC Docket: RP03-479-000

First Revised Sheet No. 292 First Revised Sheet No. 292 : Superseded Superseding: Original Sheet No. 292

Notice of Cancellation Section 18.3

Notice is hereby given that effective June 15, 2003, Section 18.3 constituting Original Sheet Nos. 292-294 and Third Revised Sheet No. 295 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled. Effective Date: 05/01/1993 Status: Effective

FERC Docket: RS92- 22-005

Original Sheet No. 293 Original Sheet No. 293: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(c) Applicability

The TOP Fixed Surcharge amounts as shown in column (1) on Tariff Sheet Nos. 20, 21 and 22 shall apply, as specified herein below, upon the effectiveness of this Section 18.3 to those Buyers/Shippers identified on such tariff sheets.

(d) Recovery of TOP Fixed Surcharge

- (1) The amount to be recovered from Buyer or Shipper through the TOP Fixed Surcharge shall be reflected on Tariff Sheet Nos. 20, 21 and 22. Carrying charge reconciliation amounts for the period February 1, 1991 until the effective date of this tariff sheet of the cost recovery procedures herein shall be netted with the total Principal Amounts and reflected on Tariff Sheet Nos. 20, 21 and 22.
- (2) Each Buyer/Shipper shall have the option, in lieu of payment of the total TOP Fixed Surcharge amount, of paying thirty-six (36) monthly consecutive payments equal to 1/36th of such amount. Carrying charges on the total TOP Fixed Surcharge amounts shall be calculated and included on each monthly invoice for those Buyers/Shippers electing the thirty-six (36) month payment option. Notwithstanding such election, any Buyer/Shipper may, at any time, pay the entire amount of its unpaid share of TOP Fixed Surcharges to Panhandle, with no further obligation for carrying charges. Notwithstanding any such election, if Buyer's/Shipper's contract terminates during the period in which such payments are to be made, Panhandle shall invoice Buyer/Shipper for the then current unpaid balance due which shall become due and payable one day

Effective Date: 05/01/1993 Status: Effective

FERC Docket: RS92- 22-005

Original Sheet No. 294 Original Sheet No. 294: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

prior to the expiration of Buyer's/Shipper's contract. Upon receipt of such invoice, Buyer/Shipper may elect to continue making payments according to its prior elections by continuing to do so. Such action shall constitute Buyer's/Shipper's irrevocable agreement to pay such sums without condition or qualification.

(3) Carrying charges on unpaid principal amounts for those Buyers/Shippers electing the thirty-six (36) month amortization option shall be determined in accordance with Section 154.67 of the Commission's Regulations.

(e) Billing and Payment

Panhandle shall render bills to Buyer/Shipper for TOP Fixed Surcharges on or after the tenth (10th) day of each month. Buyer/Shipper shall pay Panhandle, at its designated office, on or before the tenth (10th) day after such billing. Should Buyer/Shipper fail to pay any amount which shall become due hereunder when such amount is due, interest thereon shall accrue at the rate computed in accordance with Section 154.67(c) of the Commission's Regulations, until the entire amount is paid or collected. The TOP Fixed Surcharge amount shall be payable in a single lump sum or as a fixed monthly charge to those Buyers/Shippers who have elected the thirty-six (36) month payment option.

(f) TOP Volumetric Surcharge

The amount to be recovered from each Buyer/Shipper for the TOP Volumetric Surcharge shall be allocated to all throughput and charged monthly as a surcharge on rates applicable under the contracts of each

Effective Date: 01/09/1997 Status: Effective

FERC Docket: RP97-185-000

Third Revised Sheet No. 295 Third Revised Sheet No. 295: Superseded

Superseding: Substitute First Revised Sheet No. 295 GENERAL TERMS AND CONDITIONS

GENERAL TERMS AND CONDITION (Continued)

Buyer/Shipper under Panhandle's First Revised Volume No. 1 Tariff under Rate Schedules FT, EFT, SCT, IT, EIT and LFT, various rate schedules under Panhandle's FERC Gas Tariff, Original Volume No. 2 (as shown thereon), with the exception of volumes redelivered from a storage field to the extent that the Volumetric Surcharge was applicable to the corresponding volume of gas delivered to the storage field, or if the injection of such gas into storage occurred prior to the effective date of this tariff sheet and with the exception of volumes of gas transported on any portion of Panhandle's System when title to such gas passes to another Shipper of gas for immediate redelivery and subsequent transportation on Panhandle's System. The amount of the Volumetric Surcharge shall be as stated in Tariff Sheet No. 19 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, and the applicable rate schedules of Panhandle's FERC Gas Tariff, Original Volume No. 2. The Volumetric Surcharge shall be added to and become a part of the stated Minimum Rates and stated Maximum Rates for transportation service provided under Rate Schedules FT, EFT, SCT, IT, EIT and LFT.

(q) Reservations

If the cost recovery procedures set forth in this Section 18.3 are made applicable to a Buyer/Shipper which has filed a petition for review of the orders of the Commission approving the provisions of this Section 18.3 or to a Buyer/Shipper who has intervened in such review proceeding in support of such petition, the right of Panhandle to assess such additional charges as may be necessary to assure that Panhandle may recover the fully allocated share of the costs encompassed by this Section 18.3 and applicable carrying costs from such party as may result from any such litigation is reserved and the right of Panhandle to use such means of collection of such costs from such a party is also reserved.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Third Revised Sheet No. 296 Third Revised Sheet No. 296 : Superseded Superseding: Second Revised Sheet No. 296

Notice of Cancellation Section 18.4

Notice is hereby given that effective June 15, 2003, Section 18.4 constituting First Revised Sheet Nos. 297-298A and Second Revised Sheet No. 296 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP96-259-000

First Revisd Sheet No. 297 First Revisd Sheet No. 297 : Superseded

Superseding: Original Sheet No. 297

GENERAL TERMS AND CONDITIONS (Continued)

(4) The Section 18.4 Recovery Period shall commence on July 1, 1996 and shall terminate on June 30, 1997. There also shall be a Section 18.4 Reconciliation Recovery Period consisting of the twelve months commencing September 1, 1997.

(5) Levelized Interest - The interest amount resulting from the projected amortization of the Total Second Supplemental Surcharge Amount over the Section 18.4 Recovery Period in accordance with Section 154.501(d) of the Commission's Regulations.

(c) Applicability

The amount to be recovered from each Shipper for the Second Supplemental TOP Volumetric Surcharge shall be allocated to all throughput (including volumes transported under Storage Related Transportation) and charged Monthly as a surcharge on rates applicable under the contracts of each Shipper under Panhandle's FERC Gas Tariff, First Revised Volume No. 1 under Rate Schedules FT, EFT, SCT, LFT, IT and EIT, and any rate schedule which shall replace any of the foregoing with the exception of volumes redelivered from Storage to the extent that the Second Supplemental Volumetric Surcharge was applicable to the corresponding volume of Gas delivered to Storage, and with the exception of volumes of Gas transported on any portion of Panhandle's system when title to such Gas passes to another Shipper of Gas for immediate redelivery and subsequent Transportation on Panhandle's system. The amount of the Second Supplemental Volumetric Surcharge shall be as stated in Tariff Sheet Nos. 4, 5, 6, 15, 7 and 8 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1. The Second Supplemental Volumetric Surcharge shall be added to and become a part of the stated rates for Transportation service provided under Rate Schedules FT, EFT, SCT, LFT, IT and EIT.

FERC Docket: RP96-259-000

First Revised Sheet No. 298 First Revised Sheet No. 298: Superseded

Superseding: Orignal Sheet No. 298
GENERAL TERMS AND CONDITIONS
(Continued)

(d) Determination of Second Supplemental TOP Volumetric Surcharge

Subject to the provisions of Section 18.4(b), the Second Supplemental TOP Volumetric Surcharge for the Section 18.4 Recovery Period shall be based on the Total Second Supplemental Surcharge Amount including the levelized interest amount divided by the most recent twelve months of throughput data available at the time of filing.

(e) Billing and Payment

Panhandle shall render bills to Shipper for the Second Supplemental TOP Volumetric Surcharge in accordance with the applicable Rate Schedules referenced in Section 18.4(c) above.

(f) Recovery of Panhandle's Second Supplemental Take-or-Pay Settlement Costs

In addition to the other charges calculated pursuant to this Section 18.4, in the event that Shipper or any customer of Shipper, whether direct or indirect, or any state agency participating in the proceedings initiated by the filing of this Section 18.4 shall select to challenge the prudence of Panhandle's Second Supplemental Take-or-Pay Settlement Costs and shall be unsuccessful in its prosecution of such challenge, Panhandle shall be permitted to direct bill to the Shipper affected by such challenge up to 100% of such customer's full proportionate share of Panhandle's Second Supplemental Take-or-Pay Settlement Costs to reflect the challenging party's or parties' cost responsibility, resulting from such prudence litigation. Invoices for such direct bills, including carrying charges calculated in accordance with Section 154.501(d) of the Commission's Regulations, shall be issued within thirty (30) Days of the Commission's determination of any such prudence challenge and shall be due and payable within twenty (20) Days of the issuance of such invoice.

FERC Docket: RP97-185-000

First Revised Sheet No. 298A First Revised Sheet No. 298A: Superseded

Superseding: Original Sheet No. 298A
GENERAL TERMS AND CONDITIONS
(Continued)

(g) Reconciliation of Amounts to be Recovered

For the Section 18.4 Recovery Period, Panhandle shall calculate the amount of the Second Supplemental TOP Surcharge which it shall be deemed to have recovered by multiplying throughput for each month by the Second Supplemental TOP Volumetric Surcharge applicable for such month. For purposes of determining collections for the Section 18.4 Recovery Period, any discounts applied by Panhandle to its rates during such periods shall be deemed to have been applied first to the Second Supplemental TOP Volumetric Surcharge.

If the sum of the amounts recovered, inclusive of the levelized interest component, for the Section 18.4 Recovery Period shall be calculated to be greater than the Total Second Supplemental Surcharge Amount, the amount of such excess divided by the most recent twelve months of throughput data available, shall be subtracted from the Maximum Rate to be collected under each of Panhandle's services in effect during the Section 18.4 Reconciliation Amount Recovery Period, with carrying charges computed in the same manner as provided for in connection with the levelized interest provisions hereof.

If the sum of the amounts recovered, inclusive of the levelized interest component, for the Section 18.4 Recovery Period shall be calculated to be less than the Total Second Supplemental Surcharge Amount, the amount of such deficiency divided by the most recent twelve months of throughput data available, shall be added to the Maximum Rate to be collected under each of Panhandle's services in effect during the Section 18.4 Reconciliation Amount Recovery Period, with carrying charges computed in the same manner as provided for in connection with the levelized interest provisions hereof.

Panhandle shall not be obligated to implement procedures for further reconciliations beyond the Section 18.4 Reconciliation Amount Recovery Period for which provision is made herein.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Third Revised Sheet No. 299 Third Revised Sheet No. 299: Superseded

Superseding: Substitute First Revised Sheet No. 299

Notice of Cancellation Section 18.5

Notice is hereby given that effective June 15, 2003, Section 18.5 constituting Original Sheet No. 300; First Revised Sheet Nos. 301 and 305; Substitute First Revised Sheet Nos. 299, 303 and 304; and Third Revised Sheet No. 302 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RS92- 22-005

Original Sheet No. 300 Original Sheet No. 300: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(c) Definitions

The following defines certain terms as they are used in this Section $18.5\colon$

- (1) Settlement Direct Bill Surcharge Amounts shall mean the amounts specified on effective Tariff Sheet No. 23 for each Buyer and Shipper identified therein.
- (2) Settlement Demand (or Reservation) Surcharge Unit Amounts shall mean the amounts which shall be added to the demand (or reservation charges) for firm transportation services, as further described in Section 18.5(f) hereof.
- (3) Settlement Volumetric Surcharge Unit Amounts shall mean the amounts which shall be added to the commodity component of firm and interruptible transportation services, as further described in Section 18.5(g) hereof.
- (4) As used for purposes of this Section 18.5, the term Seller and Transporter are interchangeable as indicated by the context.
- (d) Recovery of Settlement Direct Bill Surcharge Amounts

Panhandle shall recover from those Buyers and those Shippers identified on effective Tariff Sheet No. 23 Settlement Direct Bill Surcharge Amounts in the manner prescribed in Section $18.5\,(\mathrm{e})$.

- (e)Billing and Payment of Settlement Direct Bill Surcharge Amounts $\,$
 - (1) The entire amount of each Buyer's and each affected Shipper's Settlement Direct Bill Surcharge Amount shall be due and payable ten days after Panhandle submits an invoice to Buyer

FERC Docket: RP96-214-000

First Revised Sheet No. 301 First Revised Sheet No. 301: Superseded

Superseding: Original Sheet No. 301
GENERAL TERMS AND CONDITIONS
(Continued)

or Shipper for such amounts. Panhandle shall render bills hereunder on or after the tenth day of each month. Buyer/Shipper shall pay Panhandle by electronic funds transfer to a designated bank account established by Panhandle for amounts equal to or greater than \$100,000. For amounts less than \$100,000 Buyer/Shipper shall pay Panhandle by check to Panhandle's designated office or at Buyer/Shipper's election by electronic funds transfer to a designated bank account established by Panhandle. Buyer/Shipper shall pay Panhandle, at its designated office, on or before the tenth day after such billing the full amount so invoiced. Should Buyer/Shipper fail to pay any amount which shall become due hereunder interest thereon shall accrue at the rate computed using the factors specified in Section 154.501(d) of the Commission's Regulations, until such time as the full amount due has been paid or collected.

(2) Each Buyer or Shipper shall have the option, in lieu of a lump sum payment of such amount, of paying sixty (60) monthly consecutive payments equal to 1/60th of such amount. Carrying charges on such amounts shall commence on the effective date of Section 32 of Transporter's Former Tariff and shall be calculated and included on each monthly invoice for those Buyers and Shippers electing the sixty (60) month payment option. The amount of Buyer's or Shipper's Settlement Direct Bill Surcharge Amount which Buyer or Shipper may elect to pay in lieu of paying a lump-sum payment, as provided for herein shall be stated on effective Tariff Sheet No. 23 hereof as the same may be changed from time to time. Any Buyer or Shipper may, at any time, pay all or a portion of its unpaid share of the Settlement Direct Bill Surcharge Amount for which it is responsible pursuant to this Section 18.5, and in such event its obligation for carrying charges shall be applicable only to amounts unpaid.

FERC Docket: RP96-214-000

Third Revised Sheet No. 302 Third Revised Sheet No. 302: Superseded

Superseding: Substitute First Revised Sheet No. 302
GENERAL TERMS AND CONDITIONS
(Continued)

(3) Carrying charges on unpaid principal amounts for those Buyers or Shippers electing to extend their payments hereunder shall be determined using the methods specified in Section 154.501(d) of the Commission's Regulations.

(4) Notwithstanding an election by a Buyer or a Shipper pursuant to 18.5(e)(2) hereof, if Buyer's or Shipper's contract terminates during the period in which such payments are to be made, Seller shall invoice Buyer or Shipper for the then current unpaid balance due and all of such amounts shall become due and payable not later than one day prior to the expiration of Buyer's or Shipper's contract.

Upon receipt of such invoice, Buyer or Shipper may elect to continue making payments according to its prior elections by continuing to do so. Such action shall constitute Buyer's or Shipper's irrevocable agreement to pay such sums without condition or qualification.

- (5) For purposes of 18.5(e)(4), the term "without condition or qualification" shall mean Buyer's or Shipper's irrevocable disavowal of any attempt to avoid payment of the remaining balance and applicable carrying charges, after ceasing the contractual relationship and to continue to make payments pursuant to the payment schedule. The term amounts to an irrevocable promise to pay the full Settlement Direct Bill Surcharge Amount.
- (f) Settlement Demand (or Reservation) Surcharge
 - (1) Applicability

This Section 18.5(f) is applicable to Shippers served under Panhandle's Rate Schedules FT, EFT, SCT and LFT and to Shippers served under Panhandle's FERC Gas Tariff Original Volume No. 2, except that it shall

FERC Docket: RP95-397-003

Substitute First Revised Sheet No. 303 Substitute First Revised Sheet No. 303: Superseded

Superseding: Original Sheet No. 303
GENERAL TERMS AND CONDITIONS
(Continued)

not apply to firm service for the transportation of volumes of gas transported on any portion of the system for one Shipper, when title to such gas passes to another Shipper of gas for immediate redelivery and subsequent transportation on Panhandle's System.

(2) Recovery of Settlement Demand (or Reservation) Surcharge Unit Amounts

Panhandle shall recover from affected Buyers/Shippers \$0.15 per dekatherm of monthly contract demand, which amount will be specified on the effective Tariff Sheet Nos. 4, 5, 6 and 15 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and the applicable rate schedules of Panhandle's FERC Gas Tariff, Original Volume No. 2 for a period of sixty (60) months commencing with the effective date of Section 32 of Panhandle's Former Tariff or the first day of the month commencing after the effectiveness of the Settlement identified in Section 18.5(a), whichever shall later occur.

(3) Billing and Payment of Settlement Demand (or Reservation) Surcharge Unit Amounts

Panhandle shall include charges for applicable Settlement Demand (or Reservation) Surcharge Unit Amounts in its regularly scheduled invoices to affected Shippers to whom the Settlement Demand (or Reservation) Surcharge Unit Amounts are applicable.

(4) Survivability of Settlement Demand (or Reservation) Surcharges

The provisions of this Section 18.5(f) respecting Settlement Demand (or Reservation) Surcharge Unit Amounts shall be fully applicable to successor and replacement rate schedules to the extent the period of the effectiveness of these charges has not expired.

FERC Docket: RP95-397-003

Substitute First Revised Sheet No. 304 Substitute First Revised Sheet No. 304: Superseded

Superseding: Original Sheet No. 304

GENERAL TERMS AND CONDITIONS
(Continued)

(g) Settlement Volumetric Surcharge

(1) Applicability

This Section 18.5(g) is applicable to Shippers served under Panhandle's Rate Schedules FT, EFT, SCT, IT, EIT and LFT and to Shippers served under Panhandle's FERC Gas Tariff, Original Volume No. 2 except that it shall not apply to:

- (i) volumes of gas withdrawn from storage, to the extent the Settlement Volumetric Surcharge Unit Amount was applicable to the transportation of the corresponding volumes of gas into storage or if the injection into storage occurred prior to the commencement of charges hereunder; and
- (ii) volumes of gas transported on any portion of Panhandle's system for one Shipper, when title to such gas passes to another Shipper of gas for immediate redelivery and subsequent transportation on Panhandle's system.
- (2) Recovery of Settlement Volumetric Surcharge Unit

Panhandle shall recover from affected Shippers 0.60 > per dekatherm of all gas transported on Panhandle's system, which amount will be specified on the effective Tariff Sheet Nos. 4, 5, 6, 7, 8 and 15 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and the applicable rate schedules of Panhandle's FERC Gas Tariff, Original Volume No. 2 for a period of sixty (60) months commencing with the effective date of Section 32 of Panhandle's Former Tariff or the first day of the month commencing after the effectiveness of the Settlement identified in 18.5(a), whichever shall later occur.

FERC Docket: RP99-192-000

First Revised Sheet No. 305 First Revised Sheet No. 305 : Superseded

Superseding: Original Sheet No. 305
GENERAL TERMS AND CONDITIONS
(Continued)

(3)Billing and Payment of Settlement Volumetric Surcharge Unit Amounts

Panhandle shall include charges for applicable Settlement Volumetric Surcharge Unit Amounts in its regularly scheduled invoices to affected Shippers to whom the Settlement Volumetric Surcharge Unit Amounts are applicable.

(4) Survivability of Settlement Volumetric Surcharges

The provisions of this Section 18.5(g) respecting Settlement Volumetric Surcharge Unit Amounts shall be fully applicable to successor and replacement rate schedules to the extent the period of the effectiveness of these charges has not expired.

(h) Reservations

In addition to the other charges calculated pursuant to this Section 18.5, in the event that Shipper or any customer of Shipper, whether direct or indirect, or any state agency participating in the proceedings initiated by the filing of this Section 18.5 shall elect to challenge the prudence of Panhandle's Settlement Surcharges and shall be unsuccessful in its prosecution of such challenge, Panhandle shall be permitted to direct bill to the Shipper affected by such challenge up to 100% of such customer's full proportionate share of Panhandle's Costs as reflected in Section 32 of Panhandle's Former Tariff, the challenging's party's or parties' cost responsibility, resulting from such prudence litigation. Invoices for such direct bills, including carrying charges calculated in accordance with Section 154.501(d) of the Commission's Regulations, shall be issued within thirty days of the Commission's determination of any such prudence challenge and shall be due and payable within twenty days of the issuance of such invoice.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

First Revised Sheet No. 306 First Revised Sheet No. 306 : Superseded Superseding: Original Sheet No. 306

Notice of Cancellation Section 18.6

Notice is hereby given that effective June 15, 2003, Section 18.6 constituting Original Sheet Nos. 306, 308 and 309; and Substitute First Revised Sheet No. 307 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP95-397-003

Substitute First Revised Sheet No. 307 Substitute First Revised Sheet No. 307: Superseded

Superseding: Original Sheet No. 307
GENERAL TERMS AND CONDITIONS
(Continued)

(1) Canadian Resolution Demand (or Reservation) Surcharges shall mean the amounts which shall be added to the demand (or reservation charges) for firm services, as further described in Section 18.6(c) hereof.

- (2) Canadian Resolution Volumetric Surcharges shall mean the amounts which shall be added to the rates for each unit of services, as further described in Sections 18.6(c) and 18.6(d) hereof.
- (c) Canadian Resolution Surcharges Applicable to Firm Services
 - (1) Applicability

This Section 18.6 is applicable to all Buyers and Shippers served under Panhandle's Former Tariff, to Shipper's served under Rate Schedules FT, EFT, SCT and LFT and to those Shippers served under Panhandle's FERC Gas Tariff, Original Volume No. 2, as described therein, except that it shall not apply to any contract for service described in Section 18.6(e) hereof.

(2) Recovery of Canadian Resolution Surcharges from Firm Services

For 65 months commencing April 1, 1993 Panhandle shall recover from Buyers and Shippers served under:

(i) Panhandle's Rate Schedules FT, EFT and LFT and to those Shippers served under Panhandle's FERC Gas Tariff, Original Volume No. 2, a Canadian Resolution Demand (or Reservation) Surcharge of \$0.39 for each dekatherm of monthly contract billing demand.

FERC Docket: RS92- 22-005

Original Sheet No. 308 Original Sheet No. 308: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(ii) Panhandle's Rate Schedule SCT, a Canadian Resolution Volumetric surcharge of 2.44> per dekatherm of service.

(d)Canadian Resolution Surcharges Applicable to Interruptible Services

(1) Applicability

This Section 18.6(d) is applicable to all Buyers and Shippers served under Panhandle's Former Tariff, to Shippers served under Rate Schedules IT and EIT and to those Shippers served under Panhandle's FERC Gas Tariff, Original Volume No. 2, as described therein.

(2) Recovery of Canadian Resolution Surcharges From Interruptible Services

For 65 months commencing April 1, 1993, Panhandle shall recover from Buyers and Shippers served under the Rate Schedules identified in Section 18.6(d) hereof a Canadian Resolution Volumetric Surcharge of 1.75 > per dekatherm of service for all interruptible transportation services, except as described in Section 18.6(e) hereof.

(e)Contracts to which the Canadian Resolution Surcharges shall not be applied

The Canadian Resolution Surcharges shall not be applied to a firm or interruptible transportation contract to the extent that volumes thereunder are transported for immediate downstream redelivery and subsequent transportation on Panhandle's system under another transportation contract to which the Canadian Resolution Surcharges are applicable; nor shall it apply to any contract storage rate schedule, including any storage rate schedule to

FERC Docket: RS92- 22-005

Original Sheet No. 309 Original Sheet No. 309: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

which transmission costs have been allocated; nor shall it apply to transportation associated with the withdrawal of contract storage volumes stored on Panhandle's system (including third-party storage for which Panhandle has contracted), when the Canadian Resolution Surcharge is applicable to the transportation of such gas for injection into storage.

(f) Nature of the Canadian Resolution Surcharges

The Canadian Resolution Surcharges shall be added to the currently effective rates for all rate schedules to which such surcharge is or may become applicable.

(g) Billing and Payment of Canadian Resolution Surcharges

Panhandle shall include charges for applicable Canadian Resolution Surcharges in its regularly scheduled invoices to affected Shippers to whom the Canadian Resolution Surcharges are applicable.

(h) Survivability of Canadian Resolution Surcharges

To the extent the period of the effectiveness of the charges described in Sections $18.6\,(c)$ and $18.6\,(d)$ has not expired, the provisions of this Section 18.6 respecting Canadian Resolution Surcharges shall be fully applicable to successor, replacement and new rate schedules, other than those described in Section $18.6\,(e)$ hereof.

FERC Docket: RP96-214-000

First Revised Sheet No. 310 First Revised Sheet No. 310 : Superseded

Superseding: Original Sheet No. 310
GENERAL TERMS AND CONDITIONS
(Continued)

18.7 GAS SUPPLY REALIGNMENT COSTS

(a) Purpose

This Section 18.7 establishes the means by which Panhandle shall recover its Gas Supply Realignment (GSR) Costs.

(b) Definitions

The following defines certain terms as they are used in this Section 18.7:

- (1) Gas Supply Realignment (GSR) Costs are actual payments made in cash by Panhandle, or amounts for which Panhandle has incurred a written obligation or commitment to pay for buyouts, buydowns, contract reformation, payments to resolve or respond to litigation or arbitration proceedings, or payments to assign contracts that are attributable to customer elections made pursuant to 18 C.F.R. 284.14(d) and carrying charges, calculated in accordance with Section 154.501(d) of the Commission's Regulations, on actual payments from the date such payments are made to the effective date of the rate filing made to recover such payments.
- (2) Initial GSR Cost Collection Period will consist of the three (3) years commencing with the effective date of the rate filing to recover GSR Costs.
- (3) Carryover GSR Cost Collection Period will consist of each annual extension in accordance with Section 18.7(f) hereof to complete the full recovery of Panhandle's GSR Costs.

Effective Date: 07/10/2000 Status: Effective

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 311 Fourth Revised Sheet No. 311 : Superseded

Superseding: Third Revised Sheet No. 311
GENERAL TERMS AND CONDITIONS
(Continued)

(c) Mechanisms for Recovery of GSR Costs

Ninety (90) percent of Panhandle's GSR Costs, including carrying charges, shall be recovered by means of a GSR Reservation Surcharge applicable to service under Rate Schedules FT, EFT, SCT, LFT and HFT. The GSR Reservation Surcharge applicable to service under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor rate. The GSR Reservation Surcharge shall be stated on effective Tariff Sheet Nos. 4, 5, 6, 15 and 17 of Panhandle's FERC Gas Tariff First Revised Volume No. 1, as the same may change from time to time, and shall become a part of the stated rates for services provided under such Rate Schedules.

Ten (10) percent of Panhandle's GSR Costs, including carrying charges, shall be recovered through the rates applicable to interruptible service under Rate Schedules IT and EIT as the same may change from time to time, and shall become part of the stated rates for services provided under such Rate Schedules.

Carrying Charges shall be calculated in accordance with Section 154.501(d) of the Commission's Regulations.

The provisions of this Section 18.7 shall be fully applicable to successor, replacement and new rate schedules.

- (d) Filing Procedure For Initial GSR Cost Collection $\mbox{\sc Period}$
 - (1) Panhandle may make one or more rate filings to recover GSR Costs from Shippers served under Rate Schedules FT, EFT, SCT, IT, EIT, LFT and HFT. An Initial GSR Cost Collection Period shall apply to each rate filing Panhandle makes to recover GSR Costs.

Effective Date: 07/10/2000 Status: Effective

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 312 Fourth Revised Sheet No. 312 : Superseded

Superseding: Third Revised Sheet No. 312
GENERAL TERMS AND CONDITIONS
(Continued)

(2) The GSR Reservation Surcharge during the Initial GSR Cost Collection Period shall be designed to recover the applicable GSR Costs over the three (3) year Initial GSR Cost Collection Period using the applicable firm service design determinants underlying Panhandle's then effective rates for service under Rate Schedules FT, EFT, SCT, LFT and HFT.

The GSR Reservation Surcharge shall be based on one-third (1/3) of the applicable GSR Costs to be recovered pursuant to this Section 18.7. Such GSR Costs shall include carrying charges for the Initial GSR Cost Collection Period.

The GSR Reservation Surcharge shall be adjusted annually during the Initial GSR Cost Collection Period to reflect changes in the effective billing determinants under Rate Schedules FT, EFT, SCT, LFT and HFT. Such revisions to the GSR Reservation Surcharge will become effective twelve (12) months after the effective date of the rate filing to establish the Initial GSR Cost Collection Period or subsequent adjustment to the GSR Reservation Surcharge. At least thirty (30) Days prior to the effective date of the annual adjustment to the GSR Reservation Surcharge, Panhandle shall file with the Federal Energy Regulatory Commission and post, as defined in Section 154.2(d) of the Commission's Regulations, Tariff Sheet Nos. 4, 5, 6, 15 and 17, together with supporting computations.

(3) Rates for interruptible service under Rate Schedules IT and EIT shall be adjusted to permit the recovery of the applicable GSR Costs during the three (3) year Initial GSR Cost Collection Period using the applicable interruptible service design determinants underlying Panhandle's

FERC Docket: RP96-214-000

First Revised Sheet No. 313 First Revised Sheet No. 313 : Superseded

Superseding: Substitute Original Sheet No. 313
GENERAL TERMS AND CONDITIONS
(Continued)

then-effective rates. The adjustment to rates for interruptible service shall be based on one-third (1/3) of the applicable GSR Costs. Such GSR Costs shall include carrying costs for the Initial GSR Cost Collection Period.

(e) GSR Exit Fee

If at any time prior to the conclusion of the Initial GSR Cost Collection Period or any Carryover GSR Cost Collection Period, a Shipper under a firm Service Agreement affected by this Section 18.7 requests a modification to the Service Agreement to reduce the MDCQ or requests a termination of the Service Agreement and if Panhandle agrees to such reduction or termination, then the affected Shipper shall immediately become obligated to Panhandle for the ${\tt GSR}$ Exit Fee. The GSR Exit Fee shall equal the amount which would have been collected during the remainder of the Initial or the Carryover Cost Collection Period, as applicable, of this Section 18.7 at the level of firm service applicable to such Shipper at the effective date hereof, or the effective date of the Service Agreement whichever is later. Panhandle will invoice such amount to the affected Shipper in a direct bill within thirty (30) days after the termination or reduction in firm service rights. Invoices pursuant to this Section 18.7(e), including carrying charges calculated in accordance with Section 154.501(d) of the Commission's Regulations shall be due and payable within ten (10) days of the issuance of such invoices.

(f) Reconciliation

(1) Within ninety (90) days of the conclusion of a GSR Cost Collection Period, Panhandle shall file a limited rate filing under Section 4 of the Natural Gas Act to reconcile its GSR Costs with the actual amounts of GSR Costs recovered during such period. The actual amounts of GSR costs recovered through GSR Reservation Surcharges

Effective Date: 07/10/2000 Status: Effective

FERC Docket: RP00-162-000

Third Revised Sheet No. 314 Third Revised Sheet No. 314: Superseded

Superseding: Substitute First Revised Sheet No. 314 $_{\mbox{\scriptsize GENERAL}}$ TERMS AND CONDITIONS

(Continued)

under Rate Schedules FT, EFT, SCT, LFT and HFT shall be determined separately from the actual amounts of GSR Costs recovered through interruptible transportation rates under Rate Schedules IT and EIT. For purposes of these calculations, any amount by which interruptible transportation rates have been discounted (up to the amounts of GSR costs to be recovered through such rates) will serve to decrease the actual amounts of GSR Costs collected through interruptible rates; and any amount by which GSR Reservation Surcharges charged to any customer affected by this Section 18.7 have been discounted (up to the amounts of GSR costs to be recovered through such rates) pursuant to a discount agreement entered into prior to the effective date of this Section 18.7, will serve to decrease the amount of actual GSR Costs collected through GSR Reservation Surcharges. Any GSR Exit Fees will be added to the actual amounts of GSR Costs recovered through GSR Reservation Surcharges.

- (2) If Panhandle's collections hereunder shall equal or exceed its GSR Costs, Panhandle shall file to terminate further collections hereunder and to refund the amount of any excess collected to all Shippers affected hereby in proportion to the principal amount of actual GSR costs which they have paid during the GSR cost collection period. Within thirty (30) days of the Commission's approval of such filing, refunds, with applicable carrying charges, shall be paid. In addition, Panhandle will submit a report to the Commission setting out a comparison of its GSR costs and the amounts collected hereunder and any repayments actually made.
- (3) If Panhandle's collections hereunder are less than its GSR Costs, Panhandle shall file to recover such deficiency, including carrying charges, during the Carryover GSR Cost Collection Surcharge Period by filing with the Commission GSR Reservation Surcharges and interruptible

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Fourth Revised Sheet No. 315 Fourth Revised Sheet No. 315: Superseded

Superseding: Third Revised Sheet No. 315

GENERAL TERMS AND CONDITIONS (Continued)

Transportation rates. The GSR Reservation Charge will be determined by dividing the remaining GSR Costs attributable to firm services to be recovered through the GSR Reservation Surcharge by the design billing determinants underlying Panhandle's then-effective rates applicable to Rate Schedules FT, EFT, SCT, LFT and HFT. The remaining GSR Costs to be recovered in the interruptible transportation rates under Rate Schedules IT and EIT shall be determined by dividing the remaining GSR Costs attributable to interruptible services by the applicable interruptible service design determinants underlying Panhandle's then-effective rates. The Carryover GSR Cost Collection Period shall commence on the first day of the month following Commission approval of such filing.

Notice of Cancellation Section 18.8

Notice is hereby given that effective June 15, 2003, Section 18.8 constituting Original Sheet No. 317; First Revised Sheet Nos. 316 and 320; Second Revised Sheet Nos. 318 and 319; and Sections 18.8(a) and (b) on Third Revised Sheet No. 315 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP94-238-000

First Revised Sheet No. 316 First Revised Sheet No. 316 : Superseded

Superseding: Original Sheet No. 316

GENERAL TERMS AND CONDITIONS (Continued)

adjustments which Panhandle shall become obligated to pay, if such prior period adjustments arise from services provided or Gas purchased at any time commencing July 31, 1991 through the end of the day prior to the effectiveness of this Section 18.8. Upon the effectiveness of this Section 18.8, the unrecovered Account No. 191 Deferred Account Balance shall be adjusted to include a final reconciliation of amounts for exchange transactions and transportation imbalances recorded in Account No. 806. If the balance of Panhandle's FERC Account No. 191 shall be a credit balance, or Panhandle later receives refunds from its suppliers for services provided prior to the effectiveness of this Section 18.8, Panhandle shall refund a proportionate share of such balance or refunds to the Buyers affected hereunder.

(c) Amount of Direct Bills and Refund

The amount of the Direct Bill and Refunds to any Buyer shall consist of a Deferred Account Component, as described in Section 18.8(d) hereof, and a prior Period Adjustment Component, as described in Section 18.8(e) hereof. Each component shall reflect demand and commodity charges, as may be appropriate.

- (d) Calculation of the Deferred Account Component
 - (1) The Deferred Account Component of Buyer's Direct Bill shall be computed for each Buyer by adding Buyer's proportionate share of Panhandle's commodity, including carrying charges, deferred account balance to Buyer's proportionate share of Panhandle's demand, including carrying charges, deferred account balance.

FERC Docket: RS92- 22-005

Original Sheet No. 317 Original Sheet No. 317: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

- (2) Buyer's proportionate share of Panhandle's commodity deferred account balance shall be determined by multiplying the total commodity deferred account balance, whether the same shall be a debit or credit, on the day prior to the effective date of this Section 18.8, by a fraction consisting of the amount of Panhandle's commodity sales volumes to Buyer during the twelve months prior to the effectiveness of this Section 18.8 as the numerator and Panhandle's total Buyer's commodity sales volumes during the same period as the denominator.
- (3) Buyer's proportionate share of Panhandle's demand deferred account balance shall be determined by multiplying the total demand deferred account balance, whether the same shall be a debit or credit, on the day prior to the effective date of this Section 18.8, by a fraction consisting of the Buyer's demand billing determinants during the twelve months prior to the effectiveness of this Section 18.8 as the numerator and Panhandle's total Buyer's sales demand billing determinants during the same period as the denominator.
- (e) Calculation of Prior Period Adjustment Component
 - (1) The Prior Period Adjustment Component of Buyer's Direct Bill shall be computed for each Buyer by adding Buyer's proportionate shares of the commodity and demand portions of each prior period adjustment which has been charged or refunded to Panhandle, as the case may be, and which have not been reflected in Panhandle's deferred account prior to application of this Section 18.8. The Prior Period Adjustment

FERC Docket: RP96-214-000

Second Revised Sheet No. 318 Second Revised Sheet No. 318 : Superseded

Superseding: First Revised Sheet No. 318
GENERAL TERMS AND CONDITIONS
(Continued)

Component shall be limited to a nine month period which shall commence on the date this Tariff becomes effective, except to the extent Panhandle receives refunds of amounts previously charged to Account No. 191 for which there will be no such time limitation.

- (2) Buyer's proportionate share of prior period commodity charges or refunds shall be determined by multiplying, for each prior period adjustment, the total commodity portion of such adjustment by a fraction which shall be determined in Section 18.8(d)(2).
- (3) Buyer's proportionate share of prior period demand charges shall be determined by multiplying, for each such prior period adjustment, the total demand portion of such adjustment by a fraction which shall be determined in Section 18.8(d)(3).
- (4) Carrying charges on all such amounts shall be calculated using the methods specified in Section 154.501(d) of the Commission's Regulations.
- (f) Nature of Obligations
 - (1) The entire amount of Buyer's obligation to Panhandle as described in this Section 18.8, including its subsections, shall be deemed to be due on the day prior to the date this Section becomes effective.
 - (2) Panhandle shall invoice the Buyer for the Direct Bill component hereunder on or after the tenth day of the month following the Commission's approval of Panhandle's disposition of Account No. 191 amounts. The entire amount of each Buyer's unrecovered Account No. 191 Direct Bill Amount shall be payable ten (10) days thereafter.

FERC Docket: RP96-214-000

Second Revised Sheet No. 319 Second Revised Sheet No. 319: Superseded

Superseding: First Revised Sheet No. 319
GENERAL TERMS AND CONDITIONS
(Continued)

Buyer shall pay Panhandle by electronic funds transfer to a designated bank account established by Panhandle for amounts equal to or greater than \$100,000. For amounts less than \$100,000, Buyer shall pay Panhandle by check to Panhandle's designated office or at Buyer's election by electronic funds transfer to a designated bank account established by Panhandle. Buyer shall pay Panhandle, at its designated office, on or before the tenth day after such billing the full amount so invoiced. Should Buyer fail to pay any amount which shall become due hereunder interest thereon shall accrue at the rate computed using the factors specified in Section 154.501(d) of the Commission's Regulations, until such time as the full amount due has been paid or collected.

- (3) Each Buyer shall have the option, in lieu of a lump sum payment of the total Direct Bill for Buyer's obligation for unrecovered Account No. 191 amounts, of paying twelve (12) consecutive monthly payments equal to 1/12th of such amount. Carrying charges on the total unrecovered Account No. 191 Direct Bill amount shall commence on the effective date of this Section 18.8 and shall be calculated and included on each monthly invoice for those Buyers electing the twelve (12) month payment option. Notwithstanding such election, a Buyer may, at any time, pay the entire amount of its unpaid share of the unrecovered Account No. 191 Direct Bill amount to Panhandle, with no further obligation for carrying charges. If any Buyer elects the twelve (12) month payment option, Panhandle shall file, within thirty (30) days of when the final payment is due, a report showing the amount billed and the amount received.
- (4) The Prior Period Adjustment component shall be filed twelve (12) months after the effective date of this Section 18.8. Additional unrecovered

FERC Docket: RP96-214-000

First Revised Sheet No. 320 First Revised Sheet No. 320 : Superseded

Superseding: Original Sheet No. 320
GENERAL TERMS AND CONDITIONS
(Continued)

Account No. 191 amounts will be direct billed in accordance with Section 18.8(f)(2), and refunds of Account No. 191 amounts will be paid by Panhandle to affected Buyers after approval by the Commission. The filing made twelve (12) months after the effective date of this Section 18.8 shall constitute Panhandle's final flow through of the Prior Period Adjustment component.

- (5) Carrying charges on unpaid unrecovered Account No. 191 Direct Bill amounts for those Buyers electing to extend their payments in accordance with Section 18.8(f)(3) for the Prior Period component shall be calculated using the methods specified in Section 154.501(d) of the Commission's Regulations.
- (6) The issuance by Panhandle of direct bills or the payment by Panhandle of refunds pursuant to this Section 18.8 is subject to prior filing for approval of the Federal Energy Regulatory Commission.

(q) Reservations

This Section 18.8 shall not be construed to affect Panhandle's right to recover from Buyer such transition or other costs as it may incur in connection with contracts with suppliers of natural gas or otherwise, such rights being expressly reserved, subject to applicable Commission orders and regulations. Panhandle reserves the right to reallocate the charges for which provision is made hereunder, in the event these charges shall be determined to be ineffective.

FERC Docket: RS92- 22-005

Original Sheet No. 321 Original Sheet No. 321: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

18.9 Treatment of Other Transition Costs

This Section 18.9 establishes procedures by which Panhandle will recover the costs of restructuring its services in accordance with FERC Order No. 636, Order No. 636-A and Order No. 636-B and 18 C.F.R. Part 284 other than Gas Supply Realignment Costs, the recovery of which shall be governed by Section 18.7 of these General Terms and Conditions, and Unrecovered FERC Account No. 191 balances, the recovery of which shall be governed by Section 18.8 of the General Terms and Conditions.

(a) Costs of New Facilities

Panhandle shall be entitled to file under Section 4 of the Natural Gas Act and the Commission's Regulations and recover the capital costs and other costs incurred for, and in connection with providing the restructured services in accordance with Order No. 636, Order No. 636-A and Order No. 636-B.

(b) Stranded Costs

Panhandle has incurred costs associated with facilities and contracts to provide services under its Former Tariff. In the event the facilities and contracts related to these costs cease to be required as a result of the restructuring of service, Panhandle shall be entitled to file under Section 4 of the Natural Gas Act and the Commission's Regulations and recover such costs.

(c) Upstream Transition Costs

Panhandle shall be entitled to file under Section 4 of the Natural Gas Act and the Commission's Regulations and recover the amounts it has been billed by other pipelines attributable to their service restructurings required by Order No. 636, Order No. 636-A and Order No. 636-B to the extent such costs are not recovered pursuant to Section 18.7 of these General Terms and Conditions.

Effective Date: 06/01/2000 Status: Effective

FERC Docket: RP91-229-028

Second Revised Sheet No. 321A Second Revised Sheet No. 321A: Superseded

Superseding: First Revised Sheet No. 321A

GENERAL TERMS AND CONDITIONS (Continued)

18.10 Trunkline/TLC Settlement Surcharge Credit

(a) Purpose

This Section 18.10 establishes Settlement Surcharges as further described in this Section 18.10 as the means by which Panhandle shall flow-through amounts received pursuant to Article VIII, Section 4 of a Stipulation and Agreement filed in Panhandle Eastern Pipe Line Co., Docket No. RP91-229-000; Panhandle Eastern Pipe Line Co., Docket No. RP92-118-000; Panhandle Eastern Pipe Line Co., Docket No. RP92-125-000; Panhandle Eastern Pipe Line Co., Docket No. RP92-127-000; Panhandle Eastern Pipe Line Co., Docket No. RP92-128-000; Panhandle Eastern Pipe Line Co., Docket No. RP92-166-000; Panhandle Eastern Pipe Line Co., Docket No. TA91-1-28-000 et al.; Panhandle Eastern Pipe Line Co., Docket No. TA92-1-28-000 et al.; Panhandle Eastern Pipe Line Co., Docket No. TM91-9-28-000; Panhandle Eastern Pipe Line Co., Docket No. TM92-3-28-000, -001; Panhandle Eastern Pipe Line Co., Docket No. TQ91-1-28-000; Panhandle Eastern Pipe Line Co., Docket No. TQ91-2-28-000; Panhandle Eastern Pipe Line Co., Docket No. TQ91-3-28-000 et al.; Panhandle Eastern Pipe Line Co., Docket No. TQ91-4-28-000 et al.; Panhandle Eastern Pipe Line Co., Docket No. TQ92-1-28-000, -001; Panhandle Eastern Pipe Line Co., Docket No. TQ92-2-28-000, -001; Panhandle Eastern Pipe Line Co., Docket No. TQ92-3-28-000; Panhandle Eastern Pipe Line Co., Docket No. TQ92-4-28-000, (NOT CONSOLIDATED), and approved by Order of the Federal Energy Regulatory Commission.

(b) Applicability

This Section 18.10 is applicable to Shippers served under Panhandle's Rate Schedules FT, EFT, SCT, IT, EIT and LFT.

The flow-through of the amounts received through the Trunkline/TLC Settlement Reservation and Volumetric credit surcharges shall apply as specified herein below commencing with the effectiveness of this Section 18.10 and continuing for a period of sixty (60) months.

Effective Date: 06/01/2000 Status: Effective

FERC Docket: RP91-229-028

Third Revised Sheet No. 321B Third Revised Sheet No. 321B: Superseded Superseding: Second Revised Sheet No. 321B

GENERAL TERMS AND CONDITIONS (Continued)

(c) Definitions

The following defines certain terms as they are used in this Section 18.10:

- (1) The "Trunkline/TLC Settlement Amount" shall mean that portion of the refund received by Panhandle to be credited to customers subject to the reservation and volumetric surcharges.
- (2) "Trunkline/TLC Settlement Reservation Surcharge Credit", shall mean the currently effective surcharge applicable to service under Rate Schedules FT, EFT, SCT and LFT, as shown on Tariff Sheets Nos. 4, 5, 6, and 15 of First Revised Volume No. 1.
- (3) "Trunkline/TLC Settlement Volumetric Surcharge Credit" shall mean the currently effective surcharge applicable to service under Rate Schedules FT, EFT, SCT, IT, EIT and LFT, as shown on each of the Tariff Sheets Nos., 4, 5, 6, 7, 8 and 15 of First Revised Volume No. 1.
- (4) "Levelized Interest" shall mean the interest amount resulting from the projected amortization of the Trunkline/TLC Settlement Amount over the Section 18.10 Recovery Period in accordance with Section 154.501(d) of the Commission Regulations.
- (d) Trunkline/TLC Settlement Reservation Surcharge Credit

Twenty percent (20%) of the total Trunkline/TLC Settlement Amount, including Levelized Interest, shall be returned by means of a Trunkline/TLC Settlement Reservation Surcharge Credit. The Trunkline/TLC Settlement Reservation Surcharge Credit applicable to service under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor. The Trunkline/TLC Settlement Reservation Surcharge Credit shall be stated on Tariff Sheet No. 19 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated maximum rates for service provided under Rate Schedules FT, EFT, SCT and LFT.

Effective Date: 06/01/2000 Status: Effective

FERC Docket: RP91-229-028

Second Revised Sheet No. 321C second Revised Sheet No. 321C : Superseded Superseding: First Revised Sheet No. 321C

GENERAL TERMS AND CONDITIONS (Continued)

(e) Trunkline/TLC Settlement Volumetric Surcharge Credit

Twenty percent (20%) of the total Trunkline/TLC Settlement Amount including Levelized Interest, shall be returned by means of a Trunkline/TLC Settlement Volumetric Surcharge Credit. The Trunkline/TLC Settlement Volumetric Surcharge Credit shall be stated on Tariff Sheet No. 19 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated maximum rates for service provided under Rate Schedules FT, EFT, SCT, IT, EIT and LFT.

FERC Docket: RP97-358-000

Second Revised Sheet No. 321D second Revised Sheet No. 321D : Effective Superseding: First Revised Sheet No. 321D

RESERVED FOR FUTURE USE.

FERC Docket: RP97-358-000

Second Revised Sheet No. 321E second Revised Sheet No. 321E : Effective Superseding: First Revised Sheet No. 321E

RESERVED FOR FUTURE USE.

FERC Docket: RP96-214-000

First Revised Sheet No. 321F First Revised Sheet No. 321F : Superseded

Superseding: Original Sheet No. 321F

GENERAL TERMS AND CONDITIONS (Continued)

18.11 Dakota Gasification Transition Cost Surcharge

(a) Applicability

This Section 18.11 establishes the mechanism for the recovery by Panhandle of certain costs which are applicable to services provided to Panhandle by ANR Pipeline Company (ANR), under ANR's Rate Schedule FTS-1. To recover these Transition Costs, Panhandle shall be entitled to surcharge Shippers utilizing its Market Area Storage Facilities under Rate Schedules WS, PS, FS and IWS to flow through all Dakota Gasification Costs that Panhandle is required to pay to ANR.

(b) Definition

As used in this Section 18.11, Dakota Gasification Transition Costs are those capacity and supply costs associated with ANR's obligations to Dakota Gasification Company which the Commission has permitted ANR to recover from Panhandle pursuant to Section 28.1(c) of ANR's FERC Gas Tariff, Second Revised Volume No. 1, or successor tariff.

(c) Filings

- (1) The effective date of each Dakota Gasification Transition Cost Surcharge shall be June 1, September 1, December 1 and March 1.
- (2) Prior to the effective date of each Dakota Gasification Transition Cost Surcharge, Panhandle shall file with the Commission and post, as defined in Section 154.2(d) of the Commission's Regulations, a statement of the effective Dakota Gasification Transition Cost Surcharge as determined in accordance with this Section 18.11. In each filing, Panhandle shall include workpapers detailing the basis used to determine the amount of the surcharge. In the event any such change is precluded by law from going into effect on such date, it shall be made effective as soon thereafter as permitted by law.

FERC Docket: RP94-231-000

Original Sheet No. 321G original Sheet No. 321G: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(d) Determination of the Dakota Gasification Transition Cost Surcharge

The Dakota Gasification Transition Cost Surchage shall be determined on the basis of (1) the design units underlying Panhandle's currently effective Deliverability Charge for firm service and the Monthly Inventory Charge for interruptible service for the Market Area component of Rate Schedules WS, PS and FS and the Market Area component of Rate Schedule IWS, respectively, and (2) the Dakota Gasification Transition Costs for the three (3) Months preceding the effective date of the Surcharge on an "as-billed" basis to reflect the basis of the billings submitted to Panhandle by ANR. In addition, commencing with the Surcharge to be effective on June 1, 1995, each Dakota Gasification Transition Cost Surcharge shall be adjusted as necessary to recover or to return the balance in Panhandle's Unrecovered Dakota Gasification Transition Cost Account as of two (2) Months prior to June 1 over twelve (12) billing periods beginning each June 1.

(e) Unrecovered Dakota Gasification Transition Cost Account

Panhandle shall maintain an Unrecovered Dakota Gasification Transition Cost Account to record the cummulative monthly difference between actual Dakota Gasification Transition Costs incurred and the actual amounts recovered by Panhandle for Dakota Gasification Transition Costs.

(f) Crediting of Refunds Received, Debiting of Additional Surcharges

Commencing with the effective date of this Section 18.11, Panhandle shall credit all refunds received which are related to its Dakota Gasification Transition Costs to its Unrecovered Dakota Gasification Transition Cost Account. Commencing with the effective date of this Section 18.11, Panhandle shall debit any similarly attributable additional surcharge for its Dakota Gasification Transition Costs to its Unrecovered Dakota Gasification Transition Cost

FERC Docket: RP96-214-000

First Revised Sheet No. 321H First Revised Sheet No. 321H : Superseded

Superseding: Original Sheet No. 321H

GENERAL TERMS AND CONDITIONS (Continued)

(g) Carrying Charges

Each month Panhandle shall credit or debit its Unrecovered Dakota Gasification Transition Cost Account, as appropriate, with carrying charges on the prior month's ending balance. Carrying charges shall be calculated in a manner consistent with the procedures set forth in Section 154.501(d) of the Commission's Regulations.

(h) Billing and Payment

In addition to its other applicable charges and surcharges for Market Area Storage service under Rate Schedules WS, PS, FS and IWS, Panhandle's invoices to Shippers under Rate Schedules WS, PS, FS and IWS shall include the amount for the Dakota Gasification Transition Cost Surcharge shown on Tariff Sheet Nos. 11, 12, 13 and 14. The Dakota Gasification Transition Cost Surcharge shall be billed on the basis of the Maximum Daily Withdrawal Quantity for Rate Schedules WS, PS and FS and the volumes for the Monthly Inventory Charge for Rate Schedule IWS. Shipper's obligation to pay for services provided under such Rate Schedules shall include the obligation to pay the ${\tt Dakota\ Gasification\ Transition\ Cost\ Surcharge.} \ \ {\tt The}$ provisions of Section 17 hereof also shall apply to Shipper's payments of the Dakota Gasification Transition Cost Surcharge.

FERC Docket: RP94-238-000

Original Sheet No. 3211 Original Sheet No. 3211: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

18.12 Unrecovered PGA Costs

(a) Purpose

This Section 18.12 establishes the means by which Panhandle shall recover its Unrecovered PGA Costs as defined below, in accordance with FERC Order No. 636, et seq. using reservation and volumetric surcharges

(b) Definitions

The following defines certain terms as they are used in this Section 18.12:

- (1) Unrecovered PGA Costs Those costs which remained in Panhandle's Account No. 191 balance on April 30, 1993, but which the Commission has determined were applicable to unrecovered balances which existed at July 31, 1991 and in accordance with the Commission's March 24, 1994 Order On Technical Conference And Accepting Compliance Filing in Docket Nos. RP93-172-000 and RP93-172-001 became unrecoverable through the provisions of Section 18.8 of these General Terms and Conditions, including carrying charges.
- (2) Section 18.12 Recovery Period The three year period which shall commence on July 1, 1994 and shall terminate on June 30, 1997.
- (3) Section 18.12 Reconciliation Recovery Period-The twelve months commencing on the first day of the month following Commission approval of Panhandle's Reconciliation filing as provided for in Section (e) herein.

FERC Docket: RP95-397-000

First Revised Sheet No. 321J First Revised Sheet No. 321J : Superseded

Superseding: Sub Original Sheet No. 321J GENERAL TERMS AND CONDITIONS (Continued)

> (4) Levelized Interest - The interest amount resulting from the projected amortization of the Unrecovered PGA Costs over the Section 18.12 Recovery Period in accordance with Section 154.67(c) of the Commission's Regulations.

(c) Mechanism for Recovery of Unrecovered PGA Costs

Ninety percent of Panhandle's Unrecovered PGA Costs, including Levelized Interest, shall be recovered by means of a Reservation Surcharge applicable to service under Rate Schedules FT, EFT, SCT and LFT. The Reservation Surcharge applicable to service under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor rate. The Reservation Surcharge shall be stated on effective Tariff Sheet Nos. 4, 5, 6 and 15 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated rates for service provided under such Rate Schedules.

Ten percent of Panhandle's Unrecovered PGA Costs, including Levelized Interest, shall be recovered by means of a Volumetric Surcharge applicable to interruptible service under Rate Schedules IT and EIT. The Volumetric Surcharge shall be stated in Tariff Sheet Nos. 7 and 8 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become part of the stated rates for service provided under such Rate Schedules.

On or before June 1 of each year during the Section 18.12 Recovery Period, Panhandle will file with the Commission such updates as otherwise are required by this Section 18.12 to the Reservation Surcharge applicable to service under Rate Schedules FT, EFT, SCT and LFT and the Volumetric Surcharge applicable to service under Rate Schedules IT and EIT with such updated Surcharges to be effective on the immediately following July 1. Panhandle will provide 30 days notice of such changes to the Reservation and Volumetric Surcharge.

The provisions of this Section 18.12 shall be fully applicable to successor, replacement and new rate schedules.

FERC Docket: RP95-397-000

First Revised Sheet No. 321K First Revised Sheet No. 321K : Superseded

Superseding: Original Sheet No. 321K
GENERAL TERMS AND CONDITIONS
(Continued)

(d) Filing Procedures For Unrecovered PGA Costs

The Reservation Surcharge during the Section 18.12 Recovery Period shall be designed to recover ninety percent of the Unrecovered PGA Costs over three years using the applicable firm service design determinants underlying Panhandle's then effective rates for service under Rate Schedules FT, EFT, SCT and LFT.

The Volumetric Surcharge during the Section 18.12 Recovery Period shall be designed to recover ten percent of the Unrecovered PGA Costs over three years using the applicable interruptible service design determinants underlying Panhandle's then effective rates for service under Rate Schedules IT and EIT.

(e) Reconciliation

(1) Within ninety (90) days of the conclusion of the Section 18.12 Recovery Period, Panhandle shall reconcile its Unrecovered PGA Costs with the actual amounts of Unrecovered PGA Costs recovered during such period. The actual amounts of Unrecovered PGA costs recovered through Reservation Surcharges under Rate Schedules FT, EFT, SCT and LFT shall be determined separately from the actual amounts of Unrecovered PGA Costs recovered through interruptible transportation rates under Rate Schedules IT and EIT.

FERC Docket: RP95-397-000

First Revised Sheet No. 321L : Superseded

Superseding: Sub Original Sheet No. 321L GENERAL TERMS AND CONDITIONS (Continued)

> (2) If Panhandle's actual collections hereunder shall equal or exceed its Unrecovered PGA Costs, Panhandle shall file to terminate further collections hereunder and to refund the amount of any excess collected to all Shippers affected hereby in proportion to the principal amount of actual Unrecovered PGA costs which they have paid during the Section 18.12 Recovery Period. Within thirty (30) days of the Commission's approval of such filing, refunds, with applicable carrying charges calculated in accordance with Section 154.67(c) of the Commission's Regulations, shall be paid. In addition, Panhandle will submit a report to the Commission setting out a comparison of its Unrecovered PGA costs and the amounts collected hereunder and any repayments actually made.

> (3) If Panhandle's actual collections hereunder are less than its Unrecovered PGA Costs, Panhandle shall file to recover such deficiency, including carrying charges calculated in accordance with Section 154.67(c) of the Commission's Regulations, during the Section 18.12 Reconciliation Recovery Period by filing with the Commission Unrecovered PGA Reservation Surcharges and interruptible Volumetric Surcharges. The Unrecovered PGA Cost Reservation Surcharge will be determined by dividing the remaining Unrecovered PGA Costs attributable to firm services to be recovered through the Unrecovered PGA Reservation Surcharge by the design billing determinants underlying Panhandle's then-effective rates applicable to Rate Schedules FT, EFT, SCT and LFT. The Unrecovered PGA Costs Volumetric Surcharge shall be determined by dividing the remaining Unrecovered PGA Costs attributable to interruptible services by the applicable interruptible service design determinants underlying Panhandle's then-effective rates.

FERC Docket: RP03-479-000

Third Revised Sheet No. 321M Third Revised Sheet No. 321M : Superseded

Superseding: Substitute First Revised Sheet No. 321M
Notice of Cancellation
Section 18.13

Notice is hereby given that effective June 15, 2003, Section 18.13 constituting Substitute First Revised Sheet Nos. 321M, 3210 and 321P; Second Revised Sheet No. 321N and Third Revised Sheet No. 321Q of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP99-192-000

Second Revised Sheet No. 321N Second Revised Sheet No. 321N : Superseded

Superseding: Substitute First Revised Sheet No. 321N GENERAL TERMS AND CONDITIONS (Continued)

(3) "Stranded Transportation Cost Volumetric Surcharge" or "ST Volumetric Surcharge" shall mean the currently effective surcharge applicable to service under Rate Schedules IT and EIT, as shown on Tariff Sheet Nos. 7 and 8 of First Revised Volume No. 1.

- (4) "Section 18.13 Recovery Period" shall mean the three (3) year period which shall commence on July 1, 1994 and shall terminate on June 30, 1997.
- (5) "Section 18.13 Reconciliation Recovery Period" shall mean the twelve (12) months commencing on the first day of the month following Commission approval of Panhandle's Reconciliation filing as provided for in Section 18.13(g).
- (6) "Levelized Interest" shall mean the interest amount resulting from the projected amortization of the ST Costs over the Section 18.13 Recovery Period in accordance with Section 154.501(d) of the Commission's Regulations.
- (d) Mechanism for Recovery of Stranded Transportation Costs

Ninety percent (90%) of Panhandle's ST Costs, including Levelized Interest, shall be recovered by means of the ST Reservation Surcharge. The ST Reservation Surcharge applicable to service under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor. The ST Reservation Surcharge shall be stated on effective Tariff Sheet Nos. 4, 5, 6 and 15 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated rates for service provided under such Rate Schedules.

FERC Docket: RP95-397-003

Substitute First Revised Sheet No. 3210 Substitute First Revised Sheet No. 3210: Superseded

Superseding: Sub Original Sheet No. 3210

GENERAL TERMS AND CONDITIONS (Continued)

Ten percent (10%) of Panhandle's ST Costs, including Levelized Interest, shall be recovered by means of the ST Volumetric Surcharge applicable to service under Rate Schedules IT and EIT. The ST Volumetric Surcharge shall be stated on effective Tariff Sheet Nos. 7 and 8 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated rates for service provided under such Rate Schedules.

On or before June 1 of each year during the Section 18.13 Recovery Period, Panhandle will file with the Commission such updates as otherwise are required by this Section 18.13 to the ST Reservation Surcharge applicable to service under Rate Schedules FT, EFT, SCT and LFT and the ST Volumetric Surcharge applicable to service under Rate Schedules IT and EIT with such updated Surcharges to be effective on the immediately following July 1. Panhandle will provide 30 days notice of such changes to the ST Reservation and ST Volumetric Surcharge.

The provisions of this Section 18.13 shall be fully applicable to successor, replacement and new rate schedules.

(e) Filing Procedures for Stranded Transportation ${\tt Costs}$

The ST Reservation Surcharge during the Section 18.13 Recovery Period shall be designed to recover ninety percent (90%) of the ST Costs over three (3) years using the applicable firm service design determinants underlying Panhandle's then effective rates for service under Rate Schedules FT, EFT, SCT and LFT.

The ST Volumetric Surcharge during the Section 18.13 Recovery Period shall be designed to recover ten percent (10%) of the ST Costs over three (3) years using the applicable interruptible service design determinants underlying Panhandle's then effective rates for service under Rate Schedules IT and EIT.

FERC Docket: RP95-397-003

Substitute First Revised Sheet No. 321P Substitute First Revised Sheet No. 321P: Superseded

Superseding: Sub Original Sheet No. 321P

GENERAL TERMS AND CONDITIONS (Continued)

(f) ST Exit Fee

If at any time prior to the conclusion of recoveries of costs pursuant to this Section 18.13, a Shipper under a firm Service Agreement affected by this Section 18.13 requests a modification to the Service Agreement to reduce the MDCQ or requests a termination of the Service Agreement and if Panhandle agrees to such reduction or termination, then such Shipper shall become obligated to Panhandle for payment of the ST Exit Fee, as mutually agreed to by Panhandle and such Shipper. Within thirty (30) days after any such termination, Panhandle shall send an invoice for amounts due hereunder from a Shipper affected by this Section 18.13(f) and payment shall be due within ten (10) days of the issuance of such invoices. Panhandle shall include amounts due hereunder from a Shipper with reduced MDCQ with the Shipper's first invoice for continuing service after the effectiveness of such reduction. Payment of an ST Exit Fee by a Shipper reducing its MDCQ shall not serve to extinguish the obligation of such Shipper for ST Reservation Surcharges applicable to the remainder of its MDCQ. A firm Shipper which exercises a contractual right to reduce or terminate its contract will not be subject to an ST Exit Fee to the extent of such reduction or termination.

(g) Reconciliation Provisions

(1) Within ninety (90) days of the conclusion of the Section 18.13 Recovery Period, Panhandle shall reconcile its ST Costs with the actual amounts of ST Costs recovered during such period. The actual amounts of ST Costs recovered through ST Reservation Surcharges under Rate Schedule FT, EFT, SCT and LFT shall be determined separately from the actual amounts of ST Costs recovered through ST Volumetric Surcharges under Rate Schedules IT and EIT.

FERC Docket: RP96-214-000

Third Revised Sheet No. 321Q Third Revised Sheet No. 321Q: Superseded

Superseding: Substitute First Revised Sheet No. 321Q GENERAL TERMS AND CONDITIONS (Continued)

- (2) If Panhandle's actual collections hereunder shall equal or exceed its ST Costs, Panhandle shall file to terminate further collections hereunder and to refund the amount of any excess collected to all Shippers affected hereby in proportion to the principal amount of actual ST Costs which they have paid during the Section 18.13 Recovery Period. Within thirty (30) days of the Commission's approval of Panhandle's filing, refunds, with applicable carrying costs computed in accordance with Section 154.501(d) of the Commission's Regulations, shall be paid. In addition, Panhandle will submit a report to the Commission setting out a comparison of its ST Costs and the actual amounts collected hereunder and any repayments actually made.
- (3) If Panhandle's actual collections hereunder are less than its ST Costs, Panhandle shall file to recover such deficiency, including carrying charges computed in accordance with Section 154.501(d) of the Commission's Regulations, during the Section 18.13 Reconciliation Period by means of ST Reservation Surcharges and ST Volumetric Surcharges. The ST Reservation Surcharge shall be determined by dividing the remaining ST Costs attributable to firm services by the design billing determinants underlying Panhandle's then-effective rates applicable to service under Rate Schedules FT, EFT, SCT and LFT. The ST Reservation Surcharge under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor. The ST Volumetric Surcharge shall be determined by dividing the remaining ST Costs attributable to interruptible services by the design billing determinants underlying Panhandle's then-effective rates for service under Rate Schedules IT and EIT.

FERC Docket: RP00-162-000

Third Revised Sheet No. 321R Third Revised Sheet No. 321R : Superseded

Superseding: Substitute First Revised Sheet No. 321R GENERAL TERMS AND CONDITIONS (Continued)

18.14 Recovery of Miscellaneous Stranded Costs

(a) Purpose

This Section 18.14 establishes the means by which Panhandle shall recover Miscellaneous Stranded Costs in accordance with FERC Order No. 636 et seq., using reservation and volumetric surcharges.

(b) Applicability

The Miscellaneous Stranded Cost Reservation and Volumetric Surcharges shall apply as specified herein below commencing with the effectiveness of this Section 18.14 or the effective date of the filing made to recover the Miscellaneous Stranded Costs and continuing for a period of three (3) years to those shippers served under Rate Schedules FT, EFT, SCT, IT, EIT, LFT and HFT and thereafter for such time as is described in Section 18.14(g).

(c) Definitions

The following defines certain terms as they are used in this Section 18.14:

(1) "Miscellaneous Stranded Cost Reservation Surcharge" shall mean the currently effective surcharge applicable to service under Rate Schedules FT, EFT, SCT, LFT and HFT as shown on each of Tariff Sheet Nos. 4, 5, 6, 15 and 17 of First Revised Volume No. 1.

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 321S Fourth Revised Sheet No. 321S: Superseded

Superseding: Third Revised Sheet No. 321S

GENERAL TERMS AND CONDITIONS (Continued)

"Miscellaneous Stranded Cost Volumetric Surcharge" shall mean the currently effective surcharge applicable to service under Rate Schedules IT and EIT, as shown on Tariff Sheet Nos. 7 and 8 of First Revised Volume No. 1.

- (3) "Section 18.14 Recovery Period" shall mean the three (3) year period which shall commence upon the effective date of the filing to recover the eligible Miscellaneous Stranded Costs.
- (4) "Section 18.14 Reconciliation Recovery Period" shall mean the twelve (12) months commencing on the first day of the month following Commission approval of Panhandle's Reconciliation filing as provided for in Section 18.14(g).
- (5) "Levelized Interest" shall mean the interest amount resulting from the projected amortization of the Miscellaneous Stranded Costs over the Section 18.14 Recovery Period in accordance with Section 154.501(d) of the Commission's Regulations.
- (d) Mechanism for Recovery of Miscellaneous Stranded Costs

Ninety percent (90%) of Panhandle's Miscellaneous Stranded Costs, including Levelized Interest, shall be recovered by means of the Miscellaneous Stranded Costs Reservation Surcharge. The Miscellaneous Stranded Costs Reservation Surcharge applicable to service under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor. The Miscellaneous Stranded Costs Reservation Surcharge shall be stated on effective Tariff Sheet Nos. 4, 5, 6, 15 and 17 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated rates for service provided under such Rate Schedules.

FERC Docket: RP00-162-000

Third Revised Sheet No. 321T Third Revised Sheet No. 321T : Superseded

Superseding: Substitute First Revised Sheet No. 321T GENERAL TERMS AND CONDITIONS (Continued)

Ten percent (10%) of Panhandle's Miscellaneous Stranded Costs, including Levelized Interest, shall be recovered by means of the Miscellaneous Stranded Cost Volumetric Surcharge applicable to service under Rate Schedules IT and EIT. The Miscellaneous Stranded Cost Volumetric Surcharge shall be stated on effective Tariff Sheet Nos. 7 and 8 of Panhandle's FERC Gas Tariff, First Revised Volume No. 1 and shall become a part of the stated rates for service provided under such Rate Schedules.

On or before September 1 of each year during the Section 18.14 Recovery Period, Panhandle will file with the Commission such updates as otherwise are required by this Section 18.14 to the Miscellaneous Stranded Cost Reservation Surcharge applicable to service under Rate Schedules FT, EFT, SCT, LFT and HFT and the Miscellaneous Stranded Cost Volumetric Surcharge applicable to service under Rate Schedules IT and EIT with such updated Surcharges to be effective on the immediately following October 1. Panhandle will provide 30 days notice of such changes to the Miscellaneous Stranded Cost Reservation and Miscellaneous Stranded Cost Volumetric Surcharge.

The provisions of this Section 18.14 shall be fully applicable to successor, replacement and new rate schedules.

(e) Filing Procedures for Miscellaneous Stranded Costs

The Miscellaneous Stranded Cost Reservation Surcharge during the Section 18.14 Recovery Period shall be designed to recover ninety percent (90%) of the Miscellaneous Stranded Costs over three (3) years using the applicable firm service design determinants underlying Panhandle's then effective rates for service under Rate Schedules FT, EFT, SCT, LFT and HFT.

The Miscellaneous Stranded Cost Volumetric Surcharge during the Section 18.14 Recovery Period shall be designed to recover ten percent (10%) of the Miscellaneous Stranded Costs over three (3) years using the applicable interruptible service design determinants underlying Panhandle's then effective rates for service under Rate Schedules IT and EIT.

FERC Docket: RP00-162-000

Third Revised Sheet No. 321U Third Revised Sheet No. 321U: Superseded

Superseding: Substitute First Revised Sheet No. 321U GENERAL TERMS AND CONDITIONS (Continued)

(f) Miscellaneous Stranded Cost Exit Fee

If at any time prior to the conclusion of recoveries of costs pursuant to this Section 18.14, a Shipper under a firm Service Agreement affected by this Section 18.14 requests a modification to the Service Agreement to reduce the MDCQ or requests a termination of the Service Agreement and if Panhandle agrees to such reduction or termination, then such Shipper shall become obligated to Panhandle for payment of the Miscellaneous Stranded Cost Exit Fee, as mutually agreed to by Panhandle and such Shipper. Within thirty (30) days after any such termination, Panhandle shall send an invoice for amounts due hereunder from a Shipper affected by this Section 18.14(f) and payment shall be due within ten (10) days of the issuance of such invoices. Panhandle shall include amounts due hereunder from a Shipper with reduced MDCQ with the Shipper's first invoice for continuing service after the effectiveness of such reduction. Payment of a Miscellaneous Stranded Cost Exit Fee by a Shipper reducing its MDCQ shall not serve to extinguish the obligation of such Shipper for Miscellaneous Stranded Cost Reservation Surcharges applicable to the remainder of its MDCQ. A firm Shipper which exercises a contractual right to reduce or terminate its contract will not be subject to a Miscellaneous Stranded Cost Exit Fee to the extent of such reduction or termination.

(g) Reconciliation Provisions

(1) Within ninety (90) days of the conclusion of the Section 18.14 Recovery Period, Panhandle shall reconcile its Miscellaneous Stranded Costs with the actual amounts of Miscellaneous Stranded Costs recovered during such period. The actual amounts of Miscellaneous Stranded Costs recovered through Miscellaneous Stranded Cost Reservation Surcharges under Rate Schedule FT, EFT, SCT, LFT and HFT shall be determined separately from the actual amounts of Miscellaneous Stranded Costs recovered through Miscellaneous Stranded Cost Volumetric Surcharges under Rate Schedules IT and EIT.

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 321V Fourth Revised Sheet No. 321V: Superseded

Superseding: Third Revised Sheet No. 321V

GENERAL TERMS AND CONDITIONS (Continued)

- (2) If Panhandle's actual collections hereunder shall equal or exceed its Miscellaneous Stranded Costs, Panhandle shall file to terminate further collections hereunder and to refund the amount of any excess collected to all Shippers affected hereby in proportion to the principal amount of actual Miscellaneous Stranded Costs which they have paid during the Section 18.14 Recovery Period. Within thirty (30) days of the Commission's approval of Panhandle's filing, refunds, with applicable carrying costs computed in accordance with Section 154.501(d) of the Commission's Regulations, shall be paid. In addition, Panhandle will submit a report to the Commission setting out a comparison of its Miscellaneous Stranded Costs and the actual amounts collected hereunder and any repayments actually made.
- (3) If Panhandle's actual collections hereunder are less than its Miscellaneous Stranded Costs, Panhandle shall file to recover such deficiency, including carrying charges computed in accordance with Section 154.501(d) of the Commission's Regulations, during the Section 18.14 Reconciliation Period by means of Miscellaneous Stranded Cost Reservation Surcharges and Miscellaneous Stranded Cost Volumetric Surcharges. The Miscellaneous Stranded Cost Reservation Surcharge shall be determined by dividing the remaining Miscellaneous Stranded Costs attributable to firm services by the design billing determinants underlying Panhandle's then-effective rates applicable to service under Rate Schedules FT, EFT, SCT, LFT and HFT. The Miscellaneous Stranded Cost Reservation Surcharge under Rate Schedule SCT shall be converted to a Volumetric Surcharge at a 52.5% load factor. The Miscellaneous Stranded Cost Volumetric Surcharge shall be determined by dividing the remaining Miscellaneous Stranded Costs attributable to interruptible services by the design billing determinants underlying Panhandle's theneffective rates for service under Rate Schedules IT and EIT.

FERC Docket: RP03-479-000

First Revised Sheet No. 321W First Revised Sheet No. 321W : Superseded Superseding: Original Sheet No. 321W

Notice of Cancellation Section 18.15

Notice is hereby given that effective June 15, 2003, Section 18.15 constituting Original Sheet Nos. 321W and 321X of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP94-393-000

Original Sheet No. 321X Original Sheet No. 321X: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(3) As used for purposes of this Section 18.15, the terms Buyer/Shipper shall mean the entities identified on Effective Tariff Sheet Nos. 25 through 25R. Buyers/Shippers are customers that were served under Panhandle's sales Rate Schedules G-1, G-2, G-3, SG-1, SG-2, SG-3, LS-1 LS-2, SS-1 and CS-1 under the Former Tariff, customers with Transportation Agreements under Rate Schedules PT-Firm and PT-Interruptible of the Former Tariff, and customers served under various rate schedules pursuant to Original Volume No. 2 of Panhandle's Tariff.

(d) Recovery of Settlement Direct Bill Amounts

Panhandle shall recover from those Buyers and Shippers identified on effective Tariff Sheet Nos. 25 through 25R Settlement Direct Bill Amounts in the manner prescribed in Section 18.15(e). Such reimbursement to Panhandle shall be divided equally between the sales customers as a class and the transportation customers as a class.

(e) Billing and Payment

The entire amount of each affected Buyer's and Shipper's Settlement Direct Bill Amounts shall be due and payable ten (10) days after Panhandle submits an invoice to Buyer or Shipper for such amount. Panhandle shall render such invoice on or after the tenth day of the month following the effectiveness of this Section 18.15. Buyer/Shipper shall pay Panhandle by check to Panhandle's designated office or at Buyer/Shipper's election by electronic funds transfer to a designated bank account established by Panhandle.

FERC Docket: RS92- 22-005

Original Sheet No. 322 Original Sheet No. 322: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

19. WARRANTY AND ASSIGNMENTS

- 19.1 Except as otherwise provided in the Rate Schedule under which Shipper is receiving service, both Shipper and Panhandle warrant that it shall, at the time of delivery of Gas to the other, have good title to or good right to deliver all such Gas, and that it shall deliver, or cause to be delivered, such Gas free from all liens, encumbrances and claims whatsoever. Except as provided in Section 17.4 above, both Panhandle and Shipper shall, as to the Gas it delivers or causes to be delivered to the other, indemnify and save the other harmless from all suits, actions, debts, accounts, damages, costs, losses and expenses arising from or out of any adverse claims of any and all persons to said Gas and/or to royalties, taxes, fees or charges thereon.
- 19.2 As to all matters within its actual or imputed control, Shipper represents and warrants that service hereunder and all arrangements incident thereto conform to applicable regulations. Shipper hereby agrees to indemnify and hold Panhandle harmless against any and all actions, suits or proceedings concerning such service or arrangements, which are brought before or instituted by any authority having jurisdiction if and to the extent that such actions result from or are based upon allegations of fact that are inconsistent with the representations herein made by Shipper or any failure of the warranties herein given by Shipper.
- 19.3 Both Panhandle and Shipper may assign or pledge Service Agreements and all rights and obligations thereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except in accordance with Sections 15 and 16 herein, neither Panhandle nor Shipper shall assign a Service Agreement or any of its rights thereunder unless it shall first have obtained the written consent of the other(s). Such consent shall not be unreasonably withheld.

FERC Docket: RS92- 22-005

Original Sheet No. 323 Original Sheet No. 323: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

20. FORCE MAJEURE

In the event, to the extent, and for so long as either Panhandle or Shipper is unable, by reason of force majeure, to carry out its obligations hereunder, in whole or in part, the obligations of either of Panhandle or Shipper, other than to make payment of reservation fees or capacity charges which shall become due, shall be suspended, in whole or in part, and neither Panhandle nor Shipper shall be liable in damages to the other as a result of Force Majeure. "Force Majeure," as employed herein, shall mean any cause, whether of the kind herein enumerated or otherwise, not within the control of either Panhandle or Shipper claiming suspension, and which by the exercise of due diligence, either Panhandle or Shipper has been unable to prevent or overcome, including without limitation acts of God, the government including the issuance of rules or orders which serve to frustrate or prevent the performance of Panhandle, or a public enemy; strikes, lockouts, or other industrial disturbances; wars, blockades, or civil disturbances of any kind; epidemics, landslides, hurricanes, washouts, tornadoes, storms, earthquakes, lightning, fires, explosions, arrests, and restraints of governments or people; freezing of, breakage or accident to, or the necessity for making repairs or alterations to wells, machinery or lines of pipe; partial or entire failure of wells; and the inability of either Panhandle or Shipper to acquire, or the delays on the part of either of Panhandle or Shipper in acquiring, at reasonable cost and after the exercise of reasonable diligence: (a) any servitudes, rights of way grants, permits, or licenses; (b) any materials or supplies for the construction or maintenance of facilities; or (c) any permits or permissions from any governmental agency; if such are required to enable either of Panhandle or Shipper to fulfill its obligations hereunder. Additionally, Panhandle shall be excused in whole or in part, from its performance for inability to obtain Transportation from or through third party pipelines. Either Panhandle or Shipper claiming force majeure shall give to the other(s) notice and full particulars of such force majeure by telephone or facsimile as soon as reasonably possible after the occurrence of the case relied on, and shall remedy such inability to perform with all reasonable dispatch; provided, however, that such requirement or remedy shall not require the settlement of strikes or lockouts by accession to the demands of those opposing either of Panhandle or Shipper when such course is inadvisable in the discretion of either of Panhandle or Shipper.

FERC Docket: RS92- 22-005

Original Sheet No. 324 Original Sheet No. 324: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

21. TERMINATION FOR DEFAULT

If either Panhandle or Shipper shall fail to perform any of the covenants or obligations imposed upon it or them under and by virtue of a Service Agreement hereunder, which obligations have not been suspended pursuant to Section 20 herein, then in such event the other party may at its option terminate such Service Agreement by proceeding as follows: The party not in default shall cause a written notice to be served on the party in default stating specifically the cause for terminating the Service Agreement and declaring it to be the intention of the party giving the notice to terminate the same; thereupon the party in default shall have 30 Days after the service of the aforesaid notice in which to remedy or remove the cause or causes stated in the notice for terminating the agreement, and if within said period of 30 Days the party in default does so remove and remedy said cause or causes and fully indemnify the party not in default for any and all consequences of such breach, by a good and sufficient indemnity bond or otherwise, then such notice shall be withdrawn and the Service Agreement shall continue in full force and effect. In case the party in default does not so remedy and remove the cause or causes or does not so indemnify the party giving the notice for any and all consequences of such breach, within said period of 30 Days, then, after any necessary authorization by regulatory bodies having jurisdiction, the Service Agreement shall become null and void from and after the expiration of said period, provided that notice of termination has not been withdrawn prior thereto. Any cancellation of such Service Agreement pursuant to the provisions of this paragraph shall be without prejudice to the right of Panhandle to collect any amounts then due to it prior to the time of cancellation, and shall be without prejudice to the obligation of Panhandle and Shipper to balance receipts and deliveries of Gas pursuant to the Service Agreement, and without waiver of any remedy to which the party not in default may be entitled for violations of such Service Agreement.

FERC Docket: RS92- 22-005

Original Sheet No. 325 Original Sheet No. 325 : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

22. REGULATIONS

These General Terms and Conditions, the Rate Schedules to which they apply and any executed Service Agreement for service, as well as the respective obligations of the parties hereunder are subject to valid laws, orders, rules and regulations of duly constituted authorities having jurisdiction and are subject to change from time to time by addition, amendment or substitution as provided by law.

FERC Docket: RP00-522-000

Fourth Revised Sheet No. 326 Fourth Revised Sheet No. 326: Superseded

Superseding: Third Revised Sheet No. 326

GENERAL TERMS AND CONDITIONS

(Continued)

23. OPERATION OF RATE SCHEDULES IN CONJUNCTION WITH MARKETING AND GATHERING AFFILIATES

23.1 Operation of Rate Schedules in Conjunction With Marketing Affiliates

All terms and conditions contained herein shall be applied in a uniform and nondiscriminatory manner without regard to whether the Gas transported is sold by Panhandle, Panhandle's marketing affiliate or any other Seller of

Information on facilities and operating personnel that Panhandle shares with its marketing affiliates will be available on the Web Site.

23.2 Operation of Rate Schedules in Conjunction With Gathering Affiliates

Panhandle will provide nondiscriminatory access to all sources of supply in accordance with Part 284 of the Commission's regulations and will not give Shippers of any gathering affiliate undue preference over Shippers of nonaffiliated gatherers or other customers in scheduling, transportation, storage or curtailment priority.

Panhandle will not condition or tie its agreement to provide transportation service to an agreement by the producer, customer, end-user, or Shipper relating to any service by any gathering affiliate, any services by it on behalf of any gathering affiliate, or any services in which a gathering affiliate is involved.

FERC Docket: RP00-162-000

Fifth Revised Sheet No. 327 Fifth Revised Sheet No. 327: Superseded

Superseding: Fourth Revised Sheet No. 327
GENERAL TERMS AND CONDITIONS
(Continued)

24. FUEL REIMBURSEMENT ADJUSTMENT

The Fuel Reimbursement Percentages under Rate Schedules FT, EFT, SCT, IT, EIT, IOS, WS, PS, FS, IIOS, IWS, LFT and HFT shall be adjusted downward to reflect reductions and may be adjusted upward to reflect increases in fuel usage and lost and unaccounted for Gas in accordance with this Section 24.

- 24.1 Filing of Fuel Reimbursement Adjustment
 - (a) Effective Date of Adjustment

The effective date of each Fuel Reimbursement Adjustment shall be November 1 and April 1.

(b) Filing Procedure

At least thirty 30 Days prior to the effective date of adjustment, Panhandle shall file with the Commission and post, as defined by Section 154.2(d) of the Commission's Regulations, a schedule of effective Fuel Reimbursement Quantities, as determined in accordance with Section 24.2 hereof, together with supporting documentation. Such filing shall be in lieu of any other rate change filing required by the Commission's Regulations under the Natural Gas Act.

(c) Fuel Reimbursement Adjustment Period

The Fuel Reimbursement Adjustment Period shall be billing periods beginning with each effective Date of Adjustment.

24.2Computation of Effective Fuel Reimbursement Percentage

The effective Fuel Reimbursement Percentage shall be the sum of the current Fuel Reimbursement Percentage and the Annual Fuel Reimbursement Surcharge.

FERC Docket: RP00-162-000

Fourth Revised Sheet No. 328 Fourth Revised Sheet No. 328: Superseded

Superseding: Substitute Second Revised Sheet No. 328 GENERAL TERMS AND CONDITIONS

(Continued)

24.3 Computation of Current Fuel Reimbursement Percentage

The current Fuel Reimbursement Percentage shall be determined on the basis of (1) the estimated Quantities of Gas delivered to Panhandle for the account of Shippers under Rate Schedules FT, EFT, SCT, IT, EIT, IOS, WS, PS, FS, IIOS, IWS, LFT and HFT and (2) the projected Quantities of Gas that shall be required for fuel usage and the lost and unaccounted for Gas.

- 24.4 Computation of the Annual Fuel Reimbursement Surcharge
 - (a) The Annual Fuel Reimbursement Surcharge shall be computed by dividing the balance four (4) Months prior to November 1 of the Deferred Fuel Reimbursement Account by Panhandle's estimated Quantities of Gas for Transportation and Storage under Rate Schedules FT, EFT, SCT, IT, EIT, IOS, WS, PS, FS, IIOS, IWS, LFT and HFT for the Recovery Period.
 - (b) The Recovery Period for the Annual Fuel Reimbursement Surcharge shall be the 12 billing Months beginning November 1.
 - (c) Panhandle shall maintain a Deferred Fuel Reimbursement Account with appropriate subaccounts, beginning with the effective date of this Section 24. For each billing Month, the applicable subaccounts shall be increased or decreased for a positive or negative change in Fuel Reimbursement for the billing Month.
 - (d) A change in Fuel Reimbursement for each billing Month shall be the difference between (1) the applicable currently effective Fuel Reimbursement percentage for the billing Month multiplied by Panhandle's Transportation Quantity received and Storage Quantities injected and withdrawn during the billing Month and (2) the actual Quantities of Gas expended for fuel usage and lost and unaccounted for Gas during the billing Month.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 329 Fourth Revised Sheet No. 329: Superseded

Superseding: Third Revised Sheet No. 329

GENERAL TERMS AND CONDITIONS (Continued)

25. FLOW THROUGH OF CASH-OUT REVENUES AND PENALTIES IN EXCESS OF

25.1 Flow Through of Cash-Out Revenues in Excess of Costs

- (a) This Section 25.1 of the General Terms and Conditions sets forth the procedures under which Panhandle will flow through to Shippers or carry forward, for each annual billing period any excess of revenues received over costs incurred under the cash-out provisions in Sections 12.11(a) and 12.11(b).
- (b) The annual rate adjustment period applicable to this Section 25.1 shall be the twelve month period commencing each May 1, and ending the following April 30.
- (c) The net revenues received pursuant to the provisions of Section 25.1(a) shall be the amount collected for the twelve (12) month period ending three months prior to May 1. If the revenues received exceed the costs incurred, then Panhandle will calculate a Reservation Charge credit that will reduce the maximum Reservation Charge applicable to Rate Schedules FT, EFT, SCT, LFT and HFT and a Commodity Charge credit that will reduce the maximum Commodity Charge applicable to Rate Schedules IT and EIT for a 12 month period which shall commence May 1.
- (d) If the revenues received are less than the costs incurred, then Panhandle shall carry forward the net underrecoveries to the subsequent annual billing period and may offset such net underrecoveries against any future net overrecoveries that may occur in a subsequent annual billing period.

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 330 Fourth Revised Sheet No. 330: Superseded

Superseding: Third Revised Sheet No. 330

GENERAL TERMS AND CONDITIONS (Continued)

(e) Panhandle will calculate the Reservation Charge Credit by dividing the net revenues calculated above by the sum of (1) MDCQ under Rate Schedules FT, EFT, SCT, LFT and HFT in effect at the end of each annual billing period and (2) the equivalent of MDCQ applicable to the actual IT and EIT volumes for the annual billing period calculated at a 100% load factor rate. The Reservation Charge Credit applicable to service under Rate Schedule SCT shall be converted to a volumetric credit at a 52.5% load factor rate. The Commodity Charge Credit applicable to services under Rate Schedules IT and EIT shall be the Reservation Charge Credit converted to a volumetric credit at a 100% load factor rate. The Reservation Charge Credit shall be computed to the nearest cent. No Reservation Charge Credit shall be made unless the change in rates is at least one (1) cent. The Commodity Charge Credit shall be computed to the nearest one hundredth of a cent. No Commodity Charge Credit shall be made unless the change in rates is at least .01 cents.

- (f) If the computations in Section 25.1(e) above do not produce a credit large enough to effect a change in rates, the net revenues for the annual billing period will be carried over to the next annual billing period and shall accrue interest in a manner consistent with Section 154.501(d) of the Commission's Regulations.
- 25.2 Flow Through of Penalties in Excess of Costs
 - (a) This Section 25.2 of the General Terms and Conditions sets forth the procedures under which Panhandle will flow through to firm Shippers any penalties in excess of costs collected pursuant to Sections 9.5(b), 12.11(g), 12.11(h), 12.16, 12.17 and 27.4(b) of the General Terms and Conditions, Section 3.6(c) of Rate Schedules EFT, SCT and HFT, Section 3.6 of Rate Schedule EIT, Sections 3.6(d) and (e) of Rate Schedule LFT and Section 4(a) of Rate Schedule GDS. Panhandle will reduce such penalties for reasonable incremental out-of-pocket costs incurred as a direct result of the Shipper's conduct which was penalized pursuant to these Sections.
 - (b) The crediting period applicable to this Section 25.2 shall be monthly.

FERC Docket: RP00-395-003

Original Sheet No. 330A Original Sheet No. 330A: Superseded

GENERAL TERMS AND CONDITIONS (Continued)

- (c) Credit to Non-Offending Shippers
 - (i) Panhandle will net all revenues received pursuant to Sections 9.5(b), 12.11(g) 12.11(h), 12.16 and 12.17 of the General Terms and Conditions, Section 3.6(c) of Rate Schedules EFT, SCT and HFT, Section 3.6 of Rate Schedule EIT, Sections 3.6(d) and (e) of Rate Schedule LFT and Section 4(a) of Rate Schedule GDS against the reasonable incremental out-of-pocket costs incurred for such revenues. Panhandle will credit the net amount to those Shippers under Rate Schedules FT, EFT, SCT, LFT and HFT that were not billed pursuant to Sections 9.5(b), 12.11(g) $12.11\,(h)\,\text{,}\ 12.16$ and 12.17 of the General Terms and Conditions, Section 3.6(c) of Rate Schedules EFT, SCT and HFT, Section 3.6 of Rate Schedule EIT, Sections 3.6(d) and (e) of Rate Schedule LFT and Section 4(a) of Rate Schedule GDS, during the applicable month ("Non-Offending Shippers"). Each Non-Offending Shipper's credit shall be based on (a) 0.5 times the ratio of the actual revenues billed for services to the Non-Offending Shipper during the month to the actual revenues billed for services to all Non-Offending Shippers and (b) 0.5 times the ratio of the MDCQ of the Non-Offending Shipper for the month to the total MDCQ of all Non-Offending Shippers. Each Non-Offending Shipper's credit shall be paid with a billing adjustment, including supporting documentation, to the billing of charges for service during the following month. Panhandle shall file a report with the Commission twelve (12) months after this Section 25.2(c)(i) becomes effective showing the penalty revenues, the costs netted against the penalty revenues, and the resulting penalty revenue credits for each of the twelve months. For each type of penalty in such report, Panhandle shall also (1) identify its incremental out-of-pocket costs that were caused by shipper misconduct and the shipper misconduct that caused the costs; (2) account separately for such costs; and (3) provide supporting documentation of the costs and the shipper misconduct that caused them.

FERC Docket: RP00-395-003

Original Sheet No. 330B Original Sheet No. 330B : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

(ii) Panhandle will net all revenues received pursuant to Section 27.4(b) against the reasonable incremental out-of-pocket costs incurred for such revenues. Panhandle will credit the net amount to those Non-Offending Shippers under Rate Schedules WS, PS and FS that were not billed pursuant to Section 27.4(b), during the applicable month. Each Non-Offending Shipper's credit shall be based on (a) 0.5 times the ratio of the actual revenues billed for services to the Non-Offending Shipper during the month to the actual revenues billed for services to all Non-Offending Shippers and (b) 0.5 times the ratio of the MDWQ of the Non-Offending Shipper for the month to the total MDWQ of all Non-Offending Shippers. Each Non-Offending Shipper's credit shall be paid with a billing adjustment, including supporting documentation, to the billing of charges for service during the following month. Panhandle shall file a report with the Commission twelve (12) months after this Section 25.2(c)(ii) becomes effective showing the penalty revenues, the costs netted against the penalty revenues, and the resulting penalty revenue credits for each of the twelve months. In such report, Panhandle shall also (1) identify its incremental out-of-pocket costs that were caused by shipper misconduct and the shipper misconduct that caused the costs; (2) account separately for such costs; and (3) provide supporting documentation of the costs and the shipper misconduct that caused

FERC Docket: RP91-229-023

Fifth Revised Sheet No. 331 Fifth Revised Sheet No. 331 : Superseded Superseding: Fourth Revised Sheet No. 331

Notice of Cancellation Section 26

Notice is hereby given that effective April 1, 1997, Section 26 constituting Fourth Revised Sheet No. 331, Substitute Second Revised Sheet No. 332, Fourth Revised Sheet No. 333, Third Revised Sheet No. 334, Substitute Second Revised Sheet No. 335 and Fourth Revised Sheet No. 336 of the FERC Gas Tariff, First Revised Volume No. 1 of Panhandle Eastern Pipe Line Company is cancelled.

FERC Docket: RP91-229-023

Third Revised Sheet No. 332 Third Revised Sheet No. 332 : Superseded Superseding: Substitute Second Revised Sheet No. 332

FERC Docket: RP91-229-023

Fifth Revised Sheet No. 333 Fifth Revised Sheet No. 333 : Superseded Superseding: Fourth Revised Sheet No. 333

FERC Docket: RP91-229-023

Fourth Revised Sheet No. 334 Fourth Revised Sheet No. 334 : Superseded Superseding: Third Revised Sheet No. 334

FERC Docket: RP91-229-023

Third Revised Sheet No. 335 Third Revised Sheet No. 335 : Superseded Superseding: Substitute Second Revised Sheet No. 335

FERC Docket: RP91-229-023

Fifth Revised Sheet No. 336 Fifth Revised Sheet No. 336 : Superseded Superseding: Fourth Revised Sheet No. 336

FERC Docket: RP94-160-000

First Revised Sheet No. 337 First Revised Sheet No. 337: Superseded

Superseding: Sub Original Sheet No. 337

GENERAL TERMS AND CONDITIONS (Continued)

27. MISCELLANEOUS

- 27.1 Service Agreements shall be governed by and interpreted in accordance with the laws of the State of Texas, except for any conflict of laws provisions thereof which would require the application of the laws of another jurisdiction.
- 27.2 The Service Agreements create no rights in third parties.
- 27.3 Panhandle may waive any rights hereunder or any obligations of Shipper on a basis which is not unduly discriminatory; provided that no waiver shall operate or be construed as a waiver of other or future rights or obligations, whether of a like or different character.
- 27.4(a) Storage service provided by Panhandle under Rate Schedule PS, and Rate Schedules WS, FS and IWS if designated by Shipper as Market Area Storage related, shall be provided from Storage capacity available to Panhandle from its Market Area Storage Facilities, and no injections or withdrawals for the account of any Shipper under such Rate Schedules shall be identified with any particular Market Area Storage Facility provided, however, when a Shipper designates as the WS or PS Point an off-system Market Area Storage Facility Panhandle has leased from a third party, injections and withdrawals will be identified with that particular facility. Storage service provided by Panhandle under Rate Schedules IOS and IIOS and Rate Schedules WS, FS and IWS if designated by Shipper as Field Area Storage related shall be provided from Storage capacity available to Panhandle from its Field Area Storage Facilities, and no injections or withdrawals for the account of any Shipper under such Rate Schedules shall be identified with any particular Field Area Storage Facility.
 - (b) If Shipper designates as a WS or PS Point an off-system Market Area Storage Facility Panhandle has leased from a third party, then, in addition to all the requirements set forth in Rate Schedules WS, PS, FS or IWS, as applicable, Shipper shall, not less than ten (10) Business Days prior to the commencement of each Storage injection and withdrawal cycle, enter into an operating

FERC Docket: RP03-479-000

Fifth Revised Sheet No. 338 Fifth Revised Sheet No. 338: Superseded

Superseding: Fourth Revised Sheet No. 338

GENERAL TERMS AND CONDITIONS (Continued)

effective,

plan with Panhandle to coordinate Storage injections and withdrawals. The operating plan shall include the Shipper's planned injections into Storage during the Summer Period and planned withdrawals from Storage during the Winter Period, which shall be consistent with both the terms and conditions applicable to the Rate Schedule under which Panhandle provides Storage service to the Shipper and the terms and conditions applicable to the agreement under which the third party provides Storage service to Panhandle and such other information concerning the Storage activity which Panhandle deems necessary to efficiently provide service to all Shippers. Subject to the provisions of Rate Schedules WS, PS, FS and IWS, as applicable, Shipper may nominate injections and withdrawals which do not comport with the plan it previously provided Panhandle so long as such deviation does not interfere with Panhandle's ability to provide service to other Shippers. If necessary to maintain operational control of its system, Panhandle may order the Shipper to adjust its nominations so as to inject Gas into Storage, cease injecting Gas into Storage, withdraw Gas from Storage, or cease withdrawing Gas from Storage, as applicable under the circumstances existing at the time such order is issued, and Shipper shall be obligated to comply with such order. Panhandle shall provide Shipper with as much advance notice of the order as is reasonable in the then existing circumstances through the Messenger system, the Web Site and by telephone, and facsimile or via e-mail communication. Generally, the notice will be issued by 11:00 a.m. Central Time on the Day before the order is to become effective. The order will become effective at the commencement of the Day. When operating conditions so require, Panhandle may issue the notice after 11:00 a.m. and may provide that the order will become effective before the commencement of the Day. The notice will provide the time and date the order is to become

FERC Docket: RP00-395-003

Fifth Revised Sheet No. 338A Fifth Revised Sheet No. 338A: Superseded

Superseding: Fourth Revised Sheet No. 338A

GENERAL TERMS AND CONDITIONS

(Continued)

the time the order is expected to remain in effect, the action required of the Shipper, the reason for issuing the order, together with operating variables providing the basis for issuing the order, and any other information which may be required in the circumstances. Panhandle will post notification and provide updated information concerning the need for the order on the Messenger system and the Web Site at the commencement of the Day until Panhandle notifies the Shipper that the order is no longer effective. If Shipper does not adjust its injection or withdrawal nominations as required by the order from Panhandle, a penalty of \$25 per Dt. shall apply to the variance from the order until such time as the Shipper complies with the order; provided, however, in no event will this penalty apply until Shipper has had at least two (2) hours to take the actions required to comply with the order. Within a reasonable time after the order terminates, Panhandle will post on the Messenger system and the Web Site a report detailing the conditions that required the issuance and termination of the order.

(c) For purposes of reporting Storage inventories for state ad valorem taxes, the total inventories of Gas in Market Area Storage Facilities and Field Area Storage Facilities in any particular state shall be determined. Inventories in Market Area Storage Facilities shall be allocated to all Shippers with inventories under Rate Schedules PS, and, if provided from Market Area Storage Facilities, WS, FS and IWS, based on the ratio of total inventories for the state divided by total Storage inventories for all states times the Shipper's total Stored Volume under such Rate Schedules; inventories in Field Area Storage Facilities shall be allocated to all Shippers with inventories under Rate Schedules IOS, IIOS and, if provided from Market Area Storage Facilities, WS, FS and IWS, based on the ratio of total Storage inventories for the state divided by total Storage inventories for all states times the Shipper's total Stored Volume under such Rate Schedules.

FERC Docket: RP00-395-003

Original Sheet No. 338B Original Sheet No. 338B : Superseded

GENERAL TERMS AND CONDITIONS (Continued)

- 27.5 After the effective date of this Tariff, Panhandle may from time to time sell Gas as required by Section 3 of Rate Schedule PS, Sections 12.11 and 12.13(b) of the General Terms and Conditions, or to dispose of excess Storage inventory. The point of sale for all such sales shall be as specified in this Section 27.5:
 - (a) The IOS Point for sales of excess storage inventory, for sales pursuant to Section 3 of Rate Schedule PS and sales pursuant to Section 12.11 of these General Terms and Conditions, if the imbalance is associated with service under Rate Schedules FT, EFT, SCT, GDS, IT, EIT, LFT or HFT;
 - (b) The Point of Delivery for sales of unauthorized deliveries pursuant to Section 12.13(b) of these General Terms and Conditions.

FERC Docket: RP03-449-000

Twelfth Revised Sheet No. 339 Twelfth Revised Sheet No. 339: Superseded Superseding: Sub Eleventh Revised Sheet No. 339

GENERAL TERMS AND CONDITIONS
(Continued)

(c) The point where the Gas to be sold enters Panhandle's system for all other sales.

The purchaser shall be responsible for the Transportation of the \mbox{Gas} from the point of sale.

27.6 Panhandle has adopted all of the Business Practices and Electronic Communication Standards which are incorporated in 18 CFR 284.12(a). In addition to the standards which are reflected in other provisions of this tariff, the following NAESB standards, definitions and data sets are incorporated herein by reference.

Version 1.6: 0.3.1, 1.2.1 through 1.2.3, 1.2.5, 1.2.6, 1.2.8 through 1.2.19, 1.3.2(v) and (vi), 1.3.3, 1.3.4, 1.3.6, 1.3.13 through 1.3.18, 1.3.20 through 1.3.29, 1.3.31 through 1.3.78, 1.3.79, 1.4.1 through 1.4.7, 2.2.1 through 2.2.3, 2.3.1 through 2.3.4, 2.3.6 through 2.3.13, 2.3.15, 2.3.17, 2.3.19 through 2.3.25, 2.3.27, 2.3.28 through 2.3.39, 2.3.42 through 2.3.44, 2.3.48, 2.3.50, 2.4.1 through 2.4.16, 3.3.1 through 3.3.14, 3.3.16 through 3.3.18, 3.3.20 through 3.3.26, 3.4.1 through 3.4.4, 4.2.1 through 4.2.20, 4.3.1 through 4.3.3, 4.3.5 through 4.3.76, 4.3.78 through 4.3.88, 5.2.1, 5.2.2, 5.3.2 (for non-biddable releases), 5.3.12, 5.3.17 through 5.3.25, 5.3.27 through 5.3.43, 5.4.1 through 5.3.25, 5.3.27 through 5.3.43, 5.4.1 through 5.4.22 and standards contained in Recommendations R02002 (Standards 5.3.44 through 5.3.54) and R02002-2 (Standards 5.2.3 and 5.3.55 through 5.3.58).

- 27.7 Panhandle may, from time to time, enter into agreements with other interstate or intrastate pipeline companies for capacity (off-system capacity). In the event that Panhandle acquires off-system capacity, Panhandle will provide service to Shippers with the off-system capacity pursuant to Panhandle's open access tariff and subject to Panhandle's Commission-approved rates, as such tariff and rates may change from time to time. For purposes of transactions entered into subject to this Section 27.7, the "shipper must have title" requirement is waived.
- 27.8 If Panhandle has agreed to a discount with a Shipper receiving Transportation service pursuant to Part 284 of the Commission's Regulations and the discount is limited to specific Point(s) of Receipt or Delivery or both, the Shipper may request that such discount apply to transportation service at a different Point of Receipt

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 340 Second Revised Sheet No. 340: Superseded

Superseding: First Revised Sheet No. 340
GENERAL TERMS AND CONDITIONS
(Continued)

or Delivery at which Panhandle and the Shipper have not specifically agreed to the rate. There is a rebuttable presumption that such discount shall apply at the requested point if Panhandle grants discounts to other Shipper(s) receiving Transportation service utilizing that point. However, Panhandle can rebut this presumption by demonstrating that the proposed service to the Shipper is not similarly situated to the service receiving a discount at the requested point. If Shipper is granted a discount for Transportation at the requested point, Shipper shall pay the higher of its contractual rate or the discount rate provided to another Shipper utilizing the new point.

A Shipper may request a discount for service at a new point by faxing a request to Panhandle. Panhandle shall respond to requests received between 6:30 a.m. Central Time and 4:00 p.m. Central Time within two (2) hours from the time the request is received. Panhandle shall respond to overnight requests received after 4:00 p.m. Central Time by no later than 8:30 a.m. Central Time the next morning. If Shipper has previously requested a discount to the same point and Panhandle has not agreed, Panhandle shall not respond to a renewed discount request unless rates for transportation service applicable to that point have changed.

Effective Date: 02/01/2002 Status: Effective FERC Docket: RP00-395-003

Sheet Nos. 341-499 Sheet Nos. 341-499 : Superseded

SHEET NOS. 341 THROUGH 499 RESERVED FOR FUTURE USE

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 First Revised Sheet No. 500 : Superseded
Superseding: Original Sheet No. 500 RATE SCHEDULE FT FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
This contract is Firm Seasonal Storage Related
In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows:
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a firm basis, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)
Firm Seasonal Storage Related Contracts
Winter Withdrawal MDCQ (Dt./Day) Summer Injection MDCO (Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 501 Original Sheet No. 501: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule FT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the primary Point(s) of Receipt set out on Exhibit A, as well as among the primary Point(s) of Delivery set out on Exhibit A. Such assignment may be changed, subject to the availability of capacity, in accordance with the General Terms and Conditions.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm Transportation service for Shipper pursuant to this Agreement from ______ until _____, when this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fifth Revised Sheet No. 502 Fifth Revised Sheet No. 502: Superseded

Superseding: Sub Fourth Revised Sheet No. 502
RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule FT filed with the Commission, as such rates and charges and Rate Schedule FT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule FT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule FT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit C hereto.

FERC Docket: RP99-199-000

Third Revised Sheet No. 503 Third Revised Sheet No. 503: Superseded

Superseding: Second Revised Sheet No. 503
RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule FT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule FT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule FT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 504 Original Sheet No. 504: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic scheduling of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 505 Second Revised Sheet No. 505: Superseded

Superseding: First Revised Sheet No. 505

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for any other purpose)

Attn: Gas Control P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-4272 or 1-800-275-7375 Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 506 First Revised Sheet No. 506: Superseded Superseding: Original Sheet No. 506

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 507} \; \texttt{Original Sheet No. 507} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Firm Service Under Rate Schedule FT

	and						
	Contract	No					
Maximum	Daily Contact	Quantity	(Dt./Day)	·			
			Firm	Seasonal	Storage	Related	Contracts
	Withdrawal Maxi						

FERC Docket: RS92- 22-005

Original Sheet No. 508 Original Sheet No. 508: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule FT

Primary Point(s) of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.

Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Function-

alization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.

Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 509 Substitute Original Sheet No. 509: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule FT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 510 First Revised Sheet No. 510 : Superseded Superseding: Original Sheet No. 510

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Firm Service Under Rate Schedule FT

and	
Contract No	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 511 Substitute Original Sheet No. 511: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule FT

Primary Point(s) of Delivery

Function-

alization

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.

Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Functionalization

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 512 Substitute Original Sheet No. 512: Superseded

RATE SCHEDULE FT (Continued)
FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule FT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 Second Revised Sheet No. 513 Seco

Second Revised Sheet No. 513 Second Revised Sheet No. 513: Superseded Superseding: First Revised Sheet No. 513

Superseding: First Revised Sheet No. 513

RATE SCHEDULE FT (Continued)

FIRM TRANSPORTATION SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT B

Transportation Agreement
For
Firm Service
Under Rate Schedule FT
Contract No.

CUSTOMIZED RESERVATION PATTERN'D ELECTION

(Shipper) hereby elects the Customized Reservation Patternp option in accordance with Section 3.10 of Panhandle's Rate Schedule FT and hereby notifies Panhandle that it desires to be billed, and agrees to pay, the Reservation Charges elected from time to time for the period commencing November 1, through October 31,
Shipper acknowledges that this election is an alternative to the uniform monthly billing of reservation charges for Rate Schedule FT set forth on effective Tariff Sheet No. 4, as revised from time to time, and that Shipper's election shall constitute its agreement to pay the otherwise applicable charges according to the payment schedule set forth in its election including such additional charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by

Shipper.

PANHANDLE EASTERN PIPE LINE COMPANY

BY:

(Please type or print name)

SHIPPER:

BY:

(Please type or print name)

DATED:

SUPERSEDES EXHIBIT B DATED:

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 Fourth Revised Sheet No. 514 Fourth Revised Sheet No. 514: Superseded Superseding: Third Revised Sheet No. 514 RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF SERVICE AGREEMENT EXHIBIT C Transportation Agreement For Firm Service Under Rate Schedule FT Contract No. NEGOTIATED RATE AGREEMENT Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule FT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule FT. Specification of Negotiated Rate: PANHANDLE EASTERN PIPE LINE COMPANY (Please type or print name) SHIPPER: BY: (Please type or print name)

DATED:

SUPERSEDES EXHIBIT C DATED:

378

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 515 Second Revised Sheet No. 515: Superseded Superseding: First Revised Sheet No. 515

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 516 Second Revised Sheet No. 516: Superseded Superseding: First Revised Sheet No. 516

RESERVED FOR FUTURE USE.

FERC Docket: RP96-224-000

Original Sheet No. 516A Original Sheet No. 516A: Superseded

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF SERVICE AGREEMENT

EXHIBIT D

Transportation Agreement
For
Firm Service
Under Rate Schedule FT

CUSTOMIZED RESERVATION PATTERND ELECTION

(Shipper) hereby elects the Customized Reservation Patternp option in accordance with Section 3.10 of Panhandle's Rate Schedule FT and hereby notifies Panhandle that it desires to be billed, and agrees to pay, the Reservation Charges elected from time to time for the period commencing November 1, 199_ through October 31, 199_.

Shipper acknowledges that this election is an alternative to the uniform monthly billing of reservation charges for Rate Schedule FT set forth on effective Tariff Sheet No. 4, as revised from time to time, and that Shipper's election shall constitute its agreement to pay the otherwise applicable charges according to the payment schedule set forth in its election including such additional charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER: _					
BY:					
DATED:					
SUPERSEDES	S EXHIBI	r d di	ATED:		

FERC Docket: RP97-185-002

Sub Original Sheet No. 516B Sub Original Sheet No. 516B : Superseded

RATE SCHEDULE FT (Continued) FIRM TRANSPORTATION SERVICE FORM OF SERVICE AGREEMENT

EXHIBIT E

Transportation Agreement
For
Firm Service
Under Rate Schedule FT
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule FT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _______, 19__ and continuing until _______, 19__. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule FT set forth on effective Tariff Sheet No. 4, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule FT.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER: _					
BY:					
DATED:					
SUPERSEDES	EXHIBIT	E D.	ATED:		

FERC Docket: RP99-506-000
First Revised Sheet No. 517 First Revised Sheet No. 517 : Superseded Superseding: Original Sheet No. 517 RATE SCHEDULE EFT
ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies)
Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
This contract is Firm Seasonal Storage Related
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a firm basis, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)
Firm Seasonal Storage Related Contracts
Winter Withdrawal MDCQ (Dt./Day) Summer Injection MDCO (Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 518 Original Sheet No. 518: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule EFT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the primary Point(s) of Receipt set out on Exhibit A, as well as among the primary Point(s) of Delivery set out on Exhibit A.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as are defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm Transportation service for Shipper pursuant to this Agreement from _____ until _____, when this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fifth Revised Sheet No. 519 Fifth Revised Sheet No. 519: Superseded

Superseding: Sub Fourth Revised Sheet No. 519
RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule EFT filed with the Commission, as such rates and charges and Rate Schedule EFT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule EFT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule EFT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fourth Revised Sheet No. 520 Fourth Revised Sheet No. 520 : Superseded

Superseding: Third Revised Sheet No. 520
RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit C hereto.

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule EFT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule EFT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule EFT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 521 Original Sheet No. 521: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 522 Second Revised Sheet No. 522 : Superseded

Superseding: First Revised Sheet No. 522

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control

any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 523 First Revised Sheet No. 523 : Superseded Superseding: Original Sheet No. 523

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Original Sheet No. 524 Original Sheet No. 524 : Superseded

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Firm Service Under Rate Schedule EFT

and	
Contract No.	
Maximum Daily Contract Quantity (Dt./Da	у)
Firm	Seasonal Storage Related Contracts
Winter Withdrawal Maximum Daily Contract Summer Injection Maximum Daily Contract	

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 525 Substitute Original Sheet No. 525: Superseded

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule EFT

Primary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Atmos.

Atmos.

Existing/ Operated and Pres. Seq. No. Installed by Maintained by (Psia) Proposed

Secondary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 526 Substitute Original Sheet No. 526: Superseded

RATE SCHEDULE EFT (Continued) ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule EFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Pres.

Existing/ Operated and Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Seq. No. Proposed Installed by Pres.

Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 527 First Revised Sheet No. 527 : Superseded Superseding: Original Sheet No. 527

RATE SCHEDULE EFT (Continued)

(Date)

ENHANCED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Firm Service Under Rate Schedule EFT

and	
Contract No.	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
litle	Title
EXECUTED ,	EXECUTED ,

(Date)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 528 Substitute Original Sheet No. 528: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule EFT

Primary Point(s) of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 529 Substitute Original Sheet No. 529: Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule EFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

SHIPPER:

Second Revised Sheet No. 530 Second Revised Sheet No. 530 : Superseded

Superseding: First Revised Sheet No. 530

RATE SCHEDULE EFT (Continued)

ENHANCED FIRM TRANSPORTATION SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT B

Transportation Agreement For Enhanced Firm Service Under Rate Schedule EFT Contract No.

CUSTOMIZED RESERVATION PATTERN'D ELECTION

(Shipper) hereby elects the Customized
Reservation Patternp option in accordance with Section 3.10
of Panhandle's Rate Schedule EFT and hereby notifies Panhandle
that it desires to be billed, and agrees to pay, the Reservation
Charges elected from time to time for the period commencing
November 1, through October 31,
Shipper acknowledges that this election is an alternative to the
uniform monthly billing of reservation charges for Rate Schedule
EFT set forth on effective Tariff Sheet No. 5, as revised from time
to time, and that Shipper's election shall constitute its agreement
to pay the otherwise applicable charges according to the payment
schedule set forth in its election including such additional
charges as may be required to ensure that the total applicable
annual Reservation Charge obligations are paid in full. Shipper's
election constitutes its waiver of reliance on uniform monthly

billing requirements otherwise applicable for the affected months.

Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE EASTERN PIPE LINE COMPANY

BY:

(Please type or print name)

BY: ____

(Please type or print name)

DATED:

SUPERSEDES EXHIBIT B DATED:

FERC Docket: RP99-506-000

Fourth Revised Sheet No. 531 Fourth Revised Sheet No. 531 : Superseded

Superseding: Third Revised Sheet No. 531

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT C

Transportation Agreement For Enhanced Firm Service Under Rate Schedule EFT Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule EFT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, and continuing until _____, ___. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule EFT set forth on effective Tariff Sheet No. 5, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are

Specification of Negotiated Rate:

available to it under the Rate Schedule EFT.

PANHANDLE	EASTERN PIPE LINE COMPANY
BY:	
	(Please type or print name)
SHIPPER:	
BY:	
•	(Please type or print name)
DATED:	
SUPERSEDE	S EXHIBIT C DATED:

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 532 Second Revised Sheet No. 532: Superseded Superseding: First Revised Sheet No. 532

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 533 Second Revised Sheet No. 533 : Superseded Superseding: First Revised Sheet No. 533

RESERVED FOR FUTURE USE.

FERC Docket: RP96-224-000

Original Sheet No. 533A Original Sheet No. 533A : Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT D

Transportation Agreement
For
Enhanced Firm Service
Under Rate Schedule EFT

CUSTOMIZED RESERVATION PATTERND ELECTION

(Shipper) hereby elects the Customized Reservation Patternp option in accordance with Section 3.10 of Panhandle's Rate Schedule EFT and hereby notifies Panhandle that it desires to be billed, and agrees to pay, the Reservation Charges elected from time to time for the period commencing November 1, 199 through October 31, 199.

Shipper acknowledges that this election is an alternative to the uniform monthly billing of reservation charges for Rate Schedule EFT set forth on effective Tariff Sheet No. 5, as revised from time to time, and that Shipper's election shall constitute its agreement to pay the otherwise applicable charges according to the payment schedule set forth in its election including such additional charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER:					
BY:					
DATED:					
SUPERSEDES	SEXHIBIT	מ חיז	ATED:		

FERC Docket: RP97-185-002

Sub Original Sheet No. 533B Sub Original Sheet No. 533B : Superseded

RATE SCHEDULE EFT (Continued)
ENHANCED FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT E

Transportation Agreement
For
Enhanced Firm Service
Under Rate Schedule EFT
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule EFT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing ______, 19__ and continuing until ______, 19__. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule EFT set forth on effective Tariff Sheet No. 5, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule EFT.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY
BY:				
SHIPPER: _				
BY:				
DATED:				
SUPERSEDES	EXHIBI	r e da	ATED:	

FERC Docket: RP99-506-000
First Revised Sheet No. 534 First Revised Sheet No. 534 : Superseded Superseding: Original Sheet No. 534 RATE SCHEDULE SCT
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies)
Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
This contract is Firm Seasonal Storage Related
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a firm basis, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)
Firm Seasonal Storage Related Contracts
Winter Withdrawal MDCQ (Dt./Day) Summer Thjection MDCO (Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 535 Original Sheet No. 535: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule SCT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the primary Point(s) of Receipt set out on Exhibit A, as well as among the primary Point(s) of Delivery set out on Exhibit A.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities, as are defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm Transportation service for Shipper pursuant to this Agreement from _____ until ____, when this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Third Revised Sheet No. 536 Third Revised Sheet No. 536: Superseded

Superseding: Sub Second Revised Sheet No. 536
RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule SCT filed with the Commission, as such rates and charges and Rate Schedule SCT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule SCT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule SCT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

First Revised Sheet No. 537 First Revised Sheet No. 537: Superseded

Superseding: Original Sheet No. 537

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule SCT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule SCT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule SCT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 538 Original Sheet No. 538: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic scheduling of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 539 Second Revised Sheet No. 539 : Superseded

Superseding: First Revised Sheet No. 539

RATE SCHEDULE SCT (Continued) SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for

Attn: Gas Control P. O. Box 4967 any other purpose)

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 540 First Revised Sheet No. 540 : Superseded Superseding: Original Sheet No. 540
RATE SCHEDULE SCT (Continued)

SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 541} \; \texttt{Original Sheet No. 541} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Firm Service
Under Rate Schedule SCT

and	
Contract No	
Maximum Daily Contract Quantity	(Dt./Day)
	Firm Seasonal Storage Related Contracts
Winter Withdrawal Maximum Daily Summer Injection Maximum Daily	Contract Quantity (Dt./Day)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 542 Substitute Original Sheet No. 542: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule SCT

Primary Point(s) of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.

Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Function-

alization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 543 Substitute Original Sheet No. 543: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule SCT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Pres.

Seq. No. Proposed Maintained by Installed by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Installed by Pres. Seq. No. Proposed Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 544 First Revised Sheet No. 544: Superseded Superseding: Original Sheet No. 544

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Firm Service Under Rate Schedule SCT

and	
Contract No	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,
(Date)	(Date)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 545 Substitute Original Sheet No. 545: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule SCT

Primary Point(s) of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.

Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Function-

alization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 546 Substitute Original Sheet No. 546: Superseded

RATE SCHEDULE SCT (Continued)
SMALL CUSTOMER FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule SCT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

Surcharges

 $\textbf{Original Sheet No. 547} \; \texttt{Original Sheet No. 547} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE SCT (Continued) SMALL CUSTOMER FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT B

Transportation Agreement For Firm Service Under Rate Schedule SCT

RATES

Contract No			
Shipper:			
Commodity Charge	Rate Per Dt.	Delivered	Fuel Reimbursement
Gathering			
Field Zone			
Market Zone Access Mileage (per 100 miles)		

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000
First Revised Sheet No. 548 First Revised Sheet No. 548: Superseded Superseding: Original Sheet No. 548 RATE SCHEDULE IT INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows:
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a capacity available basis, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 549 Original Sheet No. 549: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule IT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating an interruptible Point(s) of Receipt or interruptible Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such interruptible Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide interruptible Transportation service for Shipper pursuant to this Agreement from _____ until ____ , when the term of this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Third Revised Sheet No. 550 Third Revised Sheet No. 550: Superseded

Superseding: Sub Second Revised Sheet No. 550
RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule IT filed with the Commission, as such rates and charges and Rate Schedule IT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule IT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule IT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

First Revised Sheet No. 551 First Revised Sheet No. 551 : Superseded

Superseding: Original Sheet No. 551

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule IT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule IT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule IT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 552 Original Sheet No. 552: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 553 Second Revised Sheet No. 553 : Superseded

Superseding: First Revised Sheet No. 553

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Attn: Marketing Operations P. O. Box 4967 Scheduling:

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 554 First Revised Sheet No. 554 : Superseded Superseding: Original Sheet No. 554 RATE SCHEDULE IT (Continued)

INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 555} \; \texttt{Original Sheet No. 555} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Interruptible Service
Under Rate Schedule IT

	á	and			 _
	(Contract N	No	 	
Maximum	Daily	Contract	Quantity	(Dt./Day)	

FERC Docket: RS92- 22-005

Original Sheet No. 556 Original Sheet No. 556: Superseded

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Interruptible Service Under Rate Schedule IT

Point(s) of Delivery

Function-

alization/
Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Proposed Installed by Pres. Seq. No. Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

Original Sheet No. 557 Original Sheet No. 557: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Interruptible Service Under Rate Schedule IT

Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 558 First Revised Sheet No. 558 : Superseded Superseding: Original Sheet No. 558

RATE SCHEDULE IT (Continued) INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Interruptible Service Under Rate Schedule IT

and	
Contract No	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

Original Sheet No. 559 Original Sheet No. 559: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Interruptible Service
Under Rate Schedule IT

Point of Delivery

Function-

alization/

Seq. No. Redelivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

Original Sheet No. 560 Original Sheet No. 560: Superseded

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Interruptible Service
Under Rate Schedule IT

Point of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 561} \; \texttt{Original Sheet No. 561} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE IT (Continued)
INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT B

Transportation Agreement For Interruptible Service Under Rate Schedule IT

RATES

Reimbursement

Contract No		
Shipper:		
Commodity Charges	Rate Per Dt. Delivered	Fuel
Gathering		

Field Zone

Market Zone
Access
Mileage (per 100 miles)

Surcharges

FERC Docket: RP99-506-000
First Revised Sheet No. 562 First Revised Sheet No. 562: Superseded Superseding: Original Sheet No. 562 RATE SCHEDULE EIT
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies)
Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows:
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point of Delivery, on a capacity available basis, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 563 Original Sheet No. 563: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule EIT.

Exhibit A hereto states the Point(s) of Receipt and the Point of Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating an interruptible Point(s) of Receipt or interruptible Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such interruptible Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide interruptible Transportation service for Shipper pursuant to this Agreement from _____ until ____ , when the term of this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Third Revised Sheet No. 564 Third Revised Sheet No. 564: Superseded

Superseding: Sub Second Revised Sheet No. 564
RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule EIT filed with the Commission, as such rates and charges and Rate Schedule EIT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule EIT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule EIT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

First Revised Sheet No. 565 First Revised Sheet No. 565 : Superseded

Superseding: Original Sheet No. 565

RATE SCHEDULE EIT (Continued)

ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE

FORM OF TRANSPORTATION AGREEMENT

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule EIT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule EIT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule EIT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 566 Original Sheet No. 566: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 567 Second Revised Sheet No. 567: Superseded

Superseding: First Revised Sheet No. 567

RATE SCHEDULE EIT (Continued) ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967

Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 568 First Revised Sheet No. 568 : Superseded Superseding: Original Sheet No. 568

RATE SCHEDULE EIT (Continued)

ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several $\,$ counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 569} \; \texttt{Original Sheet No. 569} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Interruptible Service
Under Rate Schedule EIT

and	
Contract No.	
Maximum Daily Contract Quanti	ty (Dt./Day)

FERC Docket: RS92- 22-005

Original Sheet No. 570 Original Sheet No. 570: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Interruptible Service Under Rate Schedule EIT

Point of Delivery

Functionalization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

Original Sheet No. 571 original Sheet No. 571 : Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Interruptible Service Under Rate Schedule EIT

Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 572 First Revised Sheet No. 572 : Superseded Superseding: Original Sheet No. 572

RATE SCHEDULE EIT (Continued) ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement Between Panhandle Eastern Pipe Line Company For Interruptible Service Under Rate Schedule EIT

and	
Contract No.	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

Original Sheet No. 573 Original Sheet No. 573: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Interruptible Service
Under Rate Schedule EIT

Point(s) of Delivery

Function-

alization

Seq. No. Redelivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

Original Sheet No. 574 Original Sheet No. 574: Superseded

RATE SCHEDULE EIT (Continued)
ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Interruptible Service
Under Rate Schedule EIT

Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RS92- 22-005

Original Sheet No. 575 Original Sheet No. 575: Superseded

RATE SCHEDULE EIT (Continued) ENHANCED INTERRUPTIBLE TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT B

Transportation Agreement For Interruptible Service Under Rate Schedule EIT

RATES

Contract	No.		
Shipper:		 	

Commodity Charge Rate Per Dt. Delivered Fuel Reimbursement

Gathering

Field Zone Market Zone Access Mileage (per 100 miles)

Surcharges

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 First Revised Sheet No. 576 First Revised Sheet No. 576: Superseded Superseding: Original Sheet No. 576 RATE SCHEDULE IOS IN/OUT STORAGE SERVICE FORM OF GAS STORAGE AGREEMENT THIS AGREEMENT, made and entered into this day of ("Shipper"), WITNESSETH: WHEREAS, Panhandle has available underground Gas storage capacity; and WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage service. NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows: This contract is for Market Area Storage ___ This contract is for Field Area Storage ARTICLE 1 DELIVERY OF THE STORED VOLUME Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and desires to have certain volumes hereinafter set forth (Stored Volume) delivered into storage for Shipper's account during the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may maintain in storage hereunder at any given time. Injections shall be made in accordance with Shipper's nominations, but Panhandle shall not be required to inject Gas at a rate in excess of 1/200th of Shipper's Maximum Stored Quantity on any Day. At no time shall Shipper's Stored Volume exceed its Maximum Stored Quantity. Stored Volumes are to be delivered in accordance with Shipper's nominations pursuant to the terms and conditions of this Agreement, all in accordance with Section 2 of Rate Schedule IOS. Maximum Stored Quantity

Maximum Daily Injection Quantity _____ Dt./Day
Maximum Daily Withdrawal Quantity _____ Dt./Day

FERC Docket: RS92- 22-005

Original Sheet No. 577 Original Sheet No. 577: Superseded

RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE
FORM OF GAS STORAGE AGREEMENT

1.2 Shipper shall have the right to have delivered to it at the IOS Point, as described in Rate Schedule IOS, any portion of its Stored Volume, up to a maximum withdrawal of 1/100 of its Maximum Stored Quantity, on any Day during the term hereof. In no event shall volumes delivered hereunder to Shipper ever exceed Shipper's Stored Volume as defined in Section 1.1. Panhandle shall not be required to deliver quantities of Gas on any Day which exceed the Maximum Daily Contract Quantities under firm Transportation Agreements to which Shipper has a right to deliver Gas at the IOS Point.

1.3 Transporter shall retain, as Fuel Reimbursement, the percentages of Gas injected and withdrawn hereunder, as set forth on effective Tariff Sheet No. 9.

ARTICLE 2 BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule IOS filed with the Commission, as such rates and charges and Rate Schedule IOS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule IOS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fifth Revised Sheet No. 578 Fifth Revised Sheet No. 578: Superseded

Superseding: Sub Fourth Revised Sheet No. 578
RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE
FORM OF GAS STORAGE AGREEMENT

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule IOS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually injected, withdrawn or stored (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually injected, withdrawn or stored); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit A hereto.

ARTICLE 3 POINT OF DELIVERY

3.1 All volumes delivered to Shipper hereunder shall be delivered at the IOS Point, as described in Rate Schedule IOS.

ARTICLE 4

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide firm storage service for Shipper pursuant to this Agreement from until , when the term of this $\overline{\text{Agreement shall}}$ expire.

FERC Docket: RP97- 4-002

Second Revised Sheet No. 579 Second Revised Sheet No. 579: Superseded

Superseding: First Revised Sheet No. 579
RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE
FORM OF GAS STORAGE AGREEMENT

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule IOS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule IOS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6

SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 580 Original Sheet No. 580: Superseded

RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE
FORM OF GAS STORAGE AGREEMENT

ARTICLE 7

Except as herein otherwise provided, any notice, request, demand, statement or bill provided for in this Agreement and in the General Terms and Conditions of the Tariff and the Rate Schedules to which they apply, or any notices which either Panhandle or Shipper may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered mail to the post office address of Panhandle or Shipper, or at such other address as either shall designate by formal written notice. Routine communications, including monthly statements and payments, shall be considered as delivered when mailed by either registered or ordinary mail. The post office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 581 Second Revised Sheet No. 581 : Superseded

Superseding: First Revised Sheet No. 581

RATE SCHEDULE IOS (Continued) IN/OUT STORAGE SERVICE FORM OF GAS STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967

Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 582 First Revised Sheet No. 582 : Superseded Superseding: Original Sheet No. 582

RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE FORM OF GAS STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first above written:

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RP99-506-000

Second Revised Sheet No. 583 Second Revised Sheet No. 583 : Superseded

Superseding: First Revised Sheet No. 583

RATE SCHEDULE IOS (Continued)

IN/OUT STORAGE SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT A

Storage Agreement
For
Firm Service
Under Rate Schedule IOS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule IOS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, ___ and continuing until _____, ___ and continuing until _____, ___ alternative to the billing of charges for Rate Schedule IOS set forth on effective Tariff Sheet No. 9, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule IOS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN PIPE LINE COMPANY		
BY:			
_			
	(Please type or print name)		
SHIPPER:			
BY:			
-	(Please type or print name)		
DATED:			
SUPERSEDES EXHIBIT A DATED:			

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 584 second Revised Sheet No. 584 : Superseded Superseding: First Revised Sheet No. 584

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 585 second Revised Sheet No. 585 : Superseded Superseding: First Revised Sheet No. 585

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 586 Second Revised Sheet No. 586: Superseded Superseding: First Revised Sheet No. 586

RESERVED FOR FUTURE USE.

FERC Docket: RP97-185-002

Sub Original Sheet No. 586A Sub Original Sheet No. 586A: Superseded

RATE SCHEDULE IOS (Continued)
IN/OUT STORAGE SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT C

Storage Agreement
For
Firm Service
Under Rate Schedule IOS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule IOS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing ______, 19___ and continuing until _______, 19___. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule IOS set forth on effective Tariff Sheet No. 9, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule IOS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER:					
BY:					
DATED:					_
SUPERSEDES	S EXHIBIT	r c da	ATED:		

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 First Revised Sheet No. 587 First Revised Sheet No. 587: Superseded Superseding: Original Sheet No. 587 RATE SCHEDULE IIOS INTERRUPTIBLE IN/OUT STORAGE SERVICE FORM OF INTERRUPTIBLE STORAGE AGREEMENT THIS AGREEMENT, made and entered into this _, ____, by and between PANHANDLE EASTERN PIPE LINE COMPANY (" Panhandle") and ("Shipper"), WITNESSETH: WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage service on a capacity available NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows: ARTICLE 1 DELIVERY OF THE STORED VOLUME Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and desires to have certain volumes hereinafter set forth (Stored Volume) delivered into storage for Shipper's account during the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may maintain in storage hereunder at any given time. Injections shall be made in accordance with Shipper's nominations, but Panhandle will not be required to inject Gas in excess of the Maximum Daily Injection Quantity. At no time shall Shipper's Stored Volume exceed its Maximum Stored Quantity. Stored Volumes are to be delivered in accordance with Shipper's nominations pursuant to the terms and conditions of this Agreement, all in accordance with Section 2 of Rate Schedule IIOS. Maximum Stored Quantity Maximum Daily Injection Quantity _____ Dt./Day Maximum Daily Withdrawal Quantity Dt./Day

FERC Docket: RS92- 22-005

Original Sheet No. 588 Original Sheet No. 588: Superseded

RATE SCHEDULE IIOS (Continued)
INTERRUPTIBLE IN/OUT STORAGE SERVICE
FORM OF INTERRUPTIBLE STORAGE AGREEMENT

1.2 Shipper shall have the right to nominate for delivery to it at the IOS Point, as described in Rate Schedule IOS, on an interruptible basis, any portion of its Stored Volume, up to the Maximum Daily Withdrawal Quantity, on any Day during the term hereof. In no event shall volumes delivered hereunder to Shipper ever exceed Shipper's Stored Volume as defined in Section 1.1. Panhandle shall not be required to deliver quantities of Gas on any Day for which there is insufficient available capacity under Transportation Agreements to which Shipper has a right to deliver Gas at the IOS Point.

1.3 Transporter shall retain, as Fuel Reimbursement, the percentages of Gas injected and withdrawn hereunder, as set forth on effective Tariff Sheet No. 10.

ARTICLE 2 BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule IIOS filed with the Commission, as such rates and charges and Rate Schedule IIOS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule IIOS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

For the information of Shipper, Exhibit A hereto states the rates and charges anticipated to be effective and applicable to the service agreed to hereunder as of the date first stated above; but, the terms of Rate Schedule IIOS, other applicable provisions of Panhandle's Tariff, and the Regulations and Orders of the Commission shall control in the event of any conflict with the rates and charges stated on Exhibit A hereto.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Second Revised Sheet No. 589 Second Revised Sheet No. 589: Superseded

Superseding: Sub First Revised Sheet No. 589
RATE SCHEDULE IIOS (Continued)
INTERRUPTIBLE IN/OUT STORAGE SERVICE
FORM OF INTERRUPTIBLE STORAGE AGREEMENT

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule IIOS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually stored (i.e., that the rates shall be adjusted in a specified relationship to the actual Stored Quantity); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

ARTICLE 3 POINT OF DELIVERY

3.1 All volumes delivered to Shipper hereunder shall be delivered at the IOS Point, as described in Rate Schedule IOS.

ARTICLE 4

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide interruptible storage service for Shipper pursuant to this Agreement from until , when the term of this $\overline{\text{Agreement shall expire}}$.

FERC Docket: RP99-199-000

First Revised Sheet No. 590 First Revised Sheet No. 590 : Superseded

Superseding: Original Sheet No. 590

RATE SCHEDULE IIOS (Continued)
INTERRUPTIBLE IN/OUT STORAGE SERVICE
FORM OF INTERRUPTIBLE STORAGE AGREEMENT

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule IIOS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule IIOS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 MISCELLANEOUS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7 NOTICES

7.1 Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 591 Second Revised Sheet No. 591 : Superseded

Superseding: First Revised Sheet No. 591

RATE SCHEDULE IIOS (Continued) INTERRUPTIBLE IN/OUT STORAGE SERVICE FORM OF INTERRUPTIBLE STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control

any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 592 First Revised Sheet No. 592 : Superseded Superseding: Original Sheet No. 592

RATE SCHEDULE IIOS (Continued) INTERRUPTIBLE IN/OUT STORAGE SERVICE FORM OF INTERRUPTIBLE STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 593} \; \texttt{Original Sheet No. 593} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE IIOS (Continued) INTERRUPTIBLE IN/OUT STORAGE SERVICE FORM OF INTERRUPTIBLE STORAGE AGREEMENT

EXHIBIT A

Storage Agreement For Interruptible Storage Service Under Rate Schedule IIOS

RATES

Contract No.			
Shipper:			
Commodity Charges	Amount	Per	Dt.
Monthly Inventory Charge			
Injection Charge			
Withdrawal Charge			
Surcharges			

FERC Docket: RP99-506-000

First Revised Sheet No. 594 First Revised Sheet No. 594 : Superseded

Superseding: Original Sheet No. 594

RATE SCHEDULE WS

WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

THIS STORAGE AGREEMENT, made and entered into this day of ______, _____, by and between PANHANDLE EASTERN PIPE LINE COMPANY ("Panhandle") and ("Shipper"),

WITNESSETH:

WHEREAS, Panhandle has available underground Gas storage capacity; and

WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage and transportation service.

NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows:

ARTICLE 1 DELIVERY OF THE STORED VOLUME

1.1 Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and desires to have the volumes hereinafter set forth (Stored Volume) injected into the storage facilities identified below for Shipper's account during the period of April 1 through October 31 (the Summer Period) of each year of the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may have injected into storage during the Summer Period and maintained in storage hereunder at any given time. Stored Volumes are to be withdrawn from storage and delivered to Shipper during the period November 1 through March 31 (Winter Period) of each year pursuant to the terms and conditions of this Storage Agreement, all in accordance with Section 2 of Rate Schedule WS.

			Market Area Storage Field Area Storage	
Maximu	m Store	d Quant	ity	Dt.
Maximu	n Daily	Inject	ion Quantity	Dt./Day
Maximu	n Daily	Withdr	awal Quantity	Dt./Day

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 595 Substitute Original Sheet No. 595: Superseded

RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

- 1.2 In each year of the term of this Storage Agreement, Shipper may elect to retain a portion of its Stored Volume as carry-over volume (Carry-over Volume) to be left in storage during the succeeding Summer Period and withdrawn in the succeeding Winter Period. Shipper may so retain up to 20% of its Maximum Stored Quantity for such carry-over without penalty. In the event the retained Carry-over Volume exceeds 20% of its Maximum Stored Quantity, Shipper shall be subject to an additional percentage retention of Gas as set forth in Section 1.3. The sum of the Carry-over Volume and the volumes delivered by Shipper for storage during any Summer Period shall not exceed Shipper's Maximum Stored Quantity set forth in Section 1.1 of this Storage Agreement.
- 1.3 Shipper agrees to furnish, in addition to volumes delivered into storage, a volume of Gas equal to the percentage set out in effective Tariff Sheet No. 11 of such volumes delivered for injection into storage in each year of the term hereof to be retained by Panhandle for use as compressor fuel, plus an additional volume of Gas equal to the percentage set out in effective Tariff Sheet No. 11 of volumes delivered to Shipper in each year of the term hereof to be retained by Panhandle for use as compressor fuel. No compressor fuel need be furnished for volumes carried over from any Winter Period unless Shipper retains more than 20% of its Maximum Stored Quantity in any year as Carry-over Volumes. In such event, pursuant to Section 1.2, such Carry-over Volumes in excess of 20% of Shipper's Maximum Stored Quantity shall be subject to a .25% additional compressor fuel retention by Panhandle.
- 1.4 Panhandle shall receive from Shipper and inject into storage each Month during the Summer Period Gas as nominated pursuant to the General Terms and Conditions.
- 1.5 Where Shipper has designated as the WS Point a point not located on Panhandle's system, then the maximum quantities for injection and withdrawal on any Day shall be subject to any restrictions or other limitations applicable to the lease or storage agreement between Panhandle and the owner or operator of the storage facility, in addition to the restrictions and limitations of this agreement and Rate Schedule WS.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Sixth Revised Sheet No. 596 Sixth Revised Sheet No. 596: Superseded

Superseding: Sub Fifth Revised Sheet No. 596
RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 2
BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule WS filed with the Commission, as such rates and charges and Rate Schedule WS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule WS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule WS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to ${\tt Quantities}$ below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually injected, withdrawn or stored (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually injected, withdrawn or stored); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit A hereto.

FERC Docket: RS92- 22-005

Original Sheet No. 597 Original Sheet No. 597: Superseded

RATE SCHEDULE WS (Continued) WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

ARTICLE 3 POINTS OF RECEIPT AND DELIVERY

During the Summer Period Gas shall be received by Panhandle from Shipper for injection into storage, and during the Winter Period the Stored Volume withdrawn for the account of Shipper hereunder shall be delivered to Shipper, at the WS Point as defined in the General Terms and Conditions on a firm basis.

ARTICLE 4

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide firm storage service for Shipper pursuant to this Agreement from _____ until _____, when the term of this Agreement shall expire.

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule WS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule WS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RP97- 4-002

Second Revised Sheet No. 598 Second Revised Sheet No. 598: Superseded

Superseding: First Revised Sheet No. 598
RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 6 SUCCESSION AND ASSIGNMENT

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7

Except as herein otherwise provided, any notice, request, demand, statement or bill provided for in this Storage Agreement and in the General Terms and Conditions of the Tariff and the Rate Schedules to which they apply, or any notices which either Panhandle or Shipper may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered mail to the post office address of Panhandle or Shipper, or at such other address as either shall designate by formal written notice. Routine communications, including monthly statements and payments, shall be considered as delivered when mailed by either registered or ordinary mail. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 599 Second Revised Sheet No. 599: Superseded

Superseding: First Revised Sheet No. 599

RATE SCHEDULE WS (Continued) WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose)

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 600 First Revised Sheet No. 600: Superseded Superseding: Original Sheet No. 600

RATE SCHEDULE WS (Continued) WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RP99-506-000

Second Revised Sheet No. 601 Second Revised Sheet No. 601: Superseded

Superseding: First Revised Sheet No. 601
RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT A

Storage Agreement
For
Firm Service
Under Rate Schedule WS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule WS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, and continuing until _____, Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule WS set forth on effective Tariff Sheet No. 11, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule WS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN PIPE LINE COMPANY
BY:	
_	
	(Please type or print name)
SHIPPER:	
BY:	
-	(Please type or print name)
DATED:	
SUPERSEDES	S EXHIBIT A DATED:

Second Revised Sheet No. 602 Second Revised Sheet No. 602: Superseded Superseding: First Revised Sheet No. 602

Second Revised Sheet No. 603 Second Revised Sheet No. 603: Superseded Superseding: First Revised Sheet No. 603

Second Revised Sheet No. 604 second Revised Sheet No. 604 : Superseded Superseding: First Revised Sheet No. 604

FERC Docket: RP97-185-002

Sub Original Sheet No. 604A Sub Original Sheet No. 604A: Superseded

RATE SCHEDULE WS (Continued)
WINTER STORAGE SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT C

Storage Agreement
For
Firm Service
Under Rate Schedule WS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule WS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing ______, 19__ and continuing until ______, 19__. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule WS set forth on effective Tariff Sheet No. 11, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule WS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER: _					
BY:					
DATED:					
SUPERSEDES	EXHIBIT	C DA	ATED:		

FERC Docket: RP99-506-000

First Revised Sheet No. 605 First Revised Sheet No. 605 : Superseded

Superseding: Original Sheet No. 605

RATE SCHEDULE IWS

INTERRUPTIBLE WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

THIS STORAGE AGREEMENT, made and entered into this day of ______, _____, by and between PANHANDLE EASTERN PIPE LINE COMPANY ("Panhandle") and ("Shipper"),

WITNESSETH:

WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage and transportation service on a capacity available basis.

NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows:

ARTICLE 1 DELIVERY OF THE STORED VOLUME

1.1 Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder on an interruptible basis and may nominate for storage hereunder certain volumes of Gas (Stored Volume), up to a Maximum Stored Quantity, set out below. Nominations for injection into the storage facilities identified below may be made for the period of April 1 through October 31 (the Summer Period) of each year of the term hereof, under Transportation Agreements designated by Shipper. Stored Volumes are to be delivered to Shipper during the period November 1 through March 31 (Winter Period) of each year pursuant to the terms and conditions of this Storage Agreement, all in accordance with Section 2 of Rate Schedule IWS.

			Market Area Field Area S		
Maximu	um Stored	d Quant	ity		 Dt.
Maximu	um Daily	Inject	ion Quantity		 Dt./Da
Maximu	um Dailv	Withdr	awal Ouantity	7	Dt./Day

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 606 Substitute Original Sheet No. 606: Superseded

RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

- 1.2 In each year of the term of this Storage Agreement, Shipper may elect to retain a portion of its Stored Volume as carry-over volume (Carry-over Volume) to be left in storage during the succeeding Summer Period and withdrawn in the succeeding Winter Period. Shipper may so retain up to 20% of its Maximum Stored Quantity for such carry-over without penalty. In the event the retained Carry-over Volume exceeds 20% of its Maximum Stored Quantity, Shipper shall be subject to an additional percentage retention of Gas as set forth in Section 1.3. The sum of the Carry-over Volume and the volumes delivered by Shipper for storage during any Summer Period shall not exceed Shipper's Maximum Stored Quantity set forth in Section 1.1 of this Storage Agreement.
- 1.3 Shipper agrees to furnish, in addition to volumes delivered into storage, a volume of Gas equal to the percentage set out in effective Tariff Sheet No. 13 of such volumes injected into storage in each year of the term hereof to be retained by Panhandle for use as compressor fuel, plus an additional volume of Gas equal to the percentage set out in effective Tariff Sheet No. 13 of such volumes delivered to Shipper in each year of the term hereof to be retained by Panhandle for use as compressor fuel. No compressor fuel need be furnished for volumes carried over from any Winter Period unless Shipper retains more than 20% of its Maximum Stored Quantity in any year as Carry-over Volumes. In such event, pursuant to Section 1.2, such Carry-over Volumes in excess of 20% of Shipper's Maximum Stored Quantity shall be subject to a .25% additional compressor fuel retention by Panhandle.
- 1.4 Where Shipper has designated as the WS Point a point not located on Panhandle's system, then the maximum quantities for injection and withdrawal on any Day shall be subject to any restrictions or other limitations applicable to the lease or storage agreement between Panhandle and the owner or operator of the storage facility, in addition to the restrictions and limitations of this agreement and Rate Schedule IWS.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Third Revised Sheet No. 607 Third Revised Sheet No. 607: Superseded

Superseding: Sub Second Revised Sheet No. 607
RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 2 BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule IWS filed with the Commission, as such rates and charges and Rate Schedule IWS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule IWS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule IWS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually stored (i.e., that the rates shall be adjusted in a specified relationship to the actual Stored Quantity); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

First Revised Sheet No. 608 First Revised Sheet No. 608: Superseded

Superseding: Original Sheet No. 608

RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 3 POINTS OF RECEIPT AND DELIVERY

During the Summer Period, Gas shall be received for injection into Storage and, during the Winter Period, the Stored Volume withdrawn for the account of Shipper hereunder shall be delivered at the WS Point, as defined in the General Terms and Conditions on an Interruptible basis.

ARTICLE 4

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide interruptible storage service for Shipper pursuant to this Agreement from _____ until ____ , when the term of this Agreement shall expire.

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule IWS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule IWS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 SUCCESSION AND ASSIGNMENT

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RS92- 22-005

Original Sheet No. 609 Original Sheet No. 609: Superseded

RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 7

Except as herein otherwise provided, any notice, request, demand, statement or bill provided for in this Storage Agreement and in the General Terms and Conditions of the Tariff and the Rate Schedules to which they apply, or any notices which either Panhandle or Shipper may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered mail to the post office address of Panhandle or Shipper, or at such other address as either shall designate by formal written notice. Routine communications, including monthly statements and payments, shall be considered as delivered when mailed by either registered or ordinary mail. The post office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 610 Second Revised Sheet No. 610 : Superseded

Superseding: First Revised Sheet No. 610

RATE SCHEDULE IWS (Continued) INTERRUPTIBLE WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose)

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 611 First Revised Sheet No. 611: Superseded Superseding: Original Sheet No. 611

RATE SCHEDULE IWS (Continued) INTERRUPTIBLE WINTER STORAGE SERVICE FORM OF STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Зу	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Surcharges

 $\textbf{Original Sheet No. 612} \; \texttt{Original Sheet No. 612} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE IWS (Continued)
INTERRUPTIBLE WINTER STORAGE SERVICE
FORM OF STORAGE AGREEMENT

EXHIBIT A

Storage Agreement For Interruptible Storage Service Under Rate Schedule IWS

RATES

Contract No.		=	
Shipper:		-	
Commodity Charges	Amount	Per Dt.	Fuel Reimbursement
Monthly Inventory Charge			
Injection Charge			
Withdrawal Charge			

FERC Docket: RP99-506-000

First Revised Sheet No. 613 First Revised Sheet No. 613: Superseded Superseding: Substitute Original Sheet No. 613

RATE SCHEDULE PS PEAKING STORAGE SERVICE FORM OF STORAGE AGREEMENT

THIS AGREEMENT, made and entered into this ("Shipper"),

WITNESSETH:

WHEREAS, Panhandle has available underground Gas storage capacity; and

WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage service on a firm basis.

NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows:

ARTICLE 1 DELIVERY OF THE STORED VOLUME

1.1 Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and desires to have the volumes hereinafter set forth (Stored Volume) injected into storage for Shipper's account during the period of April 1 through October 31 (the Summer Period) of each year of the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may have injected into storage during the Summer Period and maintained in storage hereunder at any given time. Stored Volumes are to be withdrawn during the period November 1 through March 31 (Winter Period) of each year pursuant to the terms and conditions of this Storage Agreement, all in accordance with Section 2 of Rate Schedule PS.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 614 Substitute Original Sheet No. 614: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

Monthly Stored Volume (Dt.) Average Daily Stored Volume (Dt./Day)

April May June July August September

October

Mont.h

Maximum Stored Quantity

Dt.

In accordance with the above schedule, Shipper agrees that it shall provide and nominate for storage injection hereunder, during each Month of the Summer Period, a minimum of the Monthly Stored Quantity. Consistent with the attainment of the Maximum Stored Quantity prior to November 1 of any year, the parties may mutually agree to alter the Monthly Stored Volumes and Average Daily Stored Volumes from this Storage Agreement. Such alterations shall not result in adjustment to the Maximum Stored Quantity. Such alterations, when made, shall be consistent with the physical and operating capabilities of Panhandle. In the event that Shipper does not provide and nominate for storage injection hereunder, during any Month of the Summer Period, the Monthly Stored Volume set out above, then Panhandle shall have the right to inject Gas into its aggregated storage on Shipper's behalf equal to the shortfall and Shipper shall pay Panhandle as set out in Rate Schedule PS and the General Terms and Conditions.

FERC Docket: RS92- 22-009

Substitute Original Sheet No. 615 Substitute Original Sheet No. 615: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

- 1.2 In each year of the term of this Storage Agreement, Shipper may elect to retain a portion of its Storage Volume as carry-over volume (Carry-over Volume) to be left in storage during the succeeding Summer Period and delivered in the succeeding Winter Period. Shipper may so retain up to 20% of its Maximum Stored Quantity for such carry-over without penalty. In the event the retained Carry-over Volume exceeds 20% of its Maximum Stored Quantity, Shipper shall be subject to an additional percentage retention of Gas as set forth in Section 1.3. The sum of the Carry-over Volume and the volumes delivered by Shipper for storage during any Summer Period shall not exceed Shipper's Maximum Stored Quantity set forth in Section 1.1 of this Agreement.
- 1.3 Shipper agrees to furnish, in addition to volumes delivered for injection into storage, a volume of Gas equal to the percentage set out in effective Tariff Sheet No. 12 of such volumes delivered into storage in each year of the term hereof to be retained by Panhandle for use as compressor fuel, plus an additional volume of Gas equal to the percentage set out in effective Tariff Sheet No. 12 of volumes delivered to Shipper in each year of the term hereof to be retained by Panhandle for use as compressor fuel. No compressor fuel need be furnished for volumes carried over from any Winter Period unless Shipper retains more than 20% of its Maximum Stored Quantity in any year as Carry-over Volumes. In such event, pursuant to Section 1.2, such Carry-over Volumes in excess of 20% of Shipper's Maximum Stored Quantity shall be subject to a .25% additional compressor fuel retention by Panhandle.
- 1.4 Subject to paragraph 1.1, above, Panhandle shall receive from Shipper and inject into storage each Month during the Summer Period the Monthly Stored Volume at flow rates equal to the Average Daily Stored Volumes set out above; provided, however, Shipper and Panhandle may mutually agree to daily delivery volumes other than the Average Daily Stored Volume.
- 1.5 Where Shipper has designated as the PS Point a point not located on Panhandle's system, then the maximum quantities for injection and withdrawal on any Day shall be subject to any restrictions or other limitations applicable to the lease or storage agreement between Panhandle and the owner or operator of the storage facility, in addition to the restrictions and limitations of this Agreement and Rate Schedule PS.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fifth Revised Sheet No. 616 Fifth Revised Sheet No. 616: Superseded

Superseding: Sub Fourth Revised Sheet No. 616
RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 2
BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule PS filed with the Commission, as such rates and charges and Rate Schedule PS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule PS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule PS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually injected, withdrawn or stored (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually injected, withdrawn or stored); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit A hereto.

FERC Docket: RP99-199-000

First Revised Sheet No. 617 First Revised Sheet No. 617: Superseded

Superseding: Original Sheet No. 617

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 3 POINTS OF RECEIPT AND DELIVERY

During the Summer Period, Panhandle shall receive Gas for injection into Storage and, during the Winter Period, the Stored Volume withdrawn for the account of Shipper hereunder shall be delivered at the PS Point as defined in the General Terms and Conditions.

ARTICLE 4

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide firm storage service for Shipper pursuant to this Agreement from _____ until ____ when the term of this $\overline{\text{Agreement shall expire}}$.

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule PS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule PS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RP97- 4-002

Second Revised Sheet No. 618 Second Revised Sheet No. 618: Superseded

Superseding: First Revised Sheet No. 618
RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 6 SUCCESSION AND ASSIGNMENT

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7

7.1 Except as otherwise specified, any notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as either shall designate by formal written notice. Routine communications and bills, (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 619 Second Revised Sheet No. 619: Superseded

Superseding: First Revised Sheet No. 619

RATE SCHEDULE PS (Continued) PEAKING STORAGE SERVICE FORM OF STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 620 First Revised Sheet No. 620 : Superseded

Superseding: Original Sheet No. 620

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RP99-506-000

Second Revised Sheet No. 621 Second Revised Sheet No. 621 : Superseded

Superseding: First Revised Sheet No. 621

RATE SCHEDULE PS (Continued)

PEAKING STORAGE SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT A

Storage Agreement
For
Firm Service
Under Rate Schedule PS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 4.9 of Rate Schedule PS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, and continuing until _____, Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule PS set forth on effective Tariff Sheet No. 12, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule PS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN PIPE LINE COMPANY
BY:	
	(Please type or print name)
SHIPPER:	
BY:	
	(Please type or print name)
DATED:	
SUPERSEDE	S EXHIBIT A DATED:

Second Revised Sheet No. 622 Second Revised Sheet No. 622 : Superseded Superseding: First Revised Sheet No. 622

Second Revised Sheet No. 623 Second Revised Sheet No. 623: Superseded Superseding: First Revised Sheet No. 623

Second Revised Sheet No. 624 Second Revised Sheet No. 624: Superseded Superseding: First Revised Sheet No. 624

FERC Docket: RP97-185-002

Sub Original Sheet No. 624A Sub Original Sheet No. 624A: Superseded

RATE SCHEDULE PS (Continued)
PEAKING STORAGE SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT C

Storage Agreement
For
Firm Service
Under Rate Schedule PS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 4.9 of Rate Schedule PS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing ______, 19___ and continuing until ______, 19___. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule PS set forth on effective Tariff Sheet No. 12, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule PS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER:					
BY:					
DATED:					
SUPERSEDES	EXHIBIT	r c da	ATED:		

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000
First Revised Sheet No. 625 First Revised Sheet No. 625: Superseded Superseding: Original Sheet No. 625 RATE SCHEDULE GDS GENERAL DELIVERY SERVICE FORM OF SERVICE AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows:
ARTICLE 1 SERVICE
1.1 Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point of Delivery, on a firm basis adjusted for fuel, quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)

FERC Docket: RP94-160-000

First Revised Sheet No. 626 First Revised Sheet No. 626: Superseded

Superseding: Original Sheet No. 626

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

1.2 Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and may desire to have certain volumes (Stored Volume) delivered into storage for Shipper's account during the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may maintain in storage hereunder at any given time. Injections shall be made in accordance with the monthly operating plan described hereinbelow, but Panhandle shall not be required to inject Gas into Storage at a rate in excess of 1/200th of Shipper's Maximum Stored Quantity on any Day. At no time shall Shipper's Stored Volume exceed its Maximum Stored Quantity. Stored Volumes are to be withdrawn and delivered pursuant to the terms and conditions of this Agreement, and in accordance with Section 2 of Rate Schedule IOS.

Maximum Stored Quantity

Dt.

- 1.3 Exhibit A hereto states the Point(s) of Receipt and the Point of Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the Primary Point(s) of Receipt set out on Exhibit A. Such assignment may be changed, subject to the availability of capacity, in accordance with the General Terms and Conditions.
- $1.4\,$ Exhibit B hereto sets out a description of the Rate Schedule EFT, SCT and IOS, and, if applicable WS, PS and FS Agreements which underlie the service to be provided hereunder.
- 1.5 Shipper recognizes that no Point(s) of Receipt or Point of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering System, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point of Delivery hereunder which is only available as a Gathering Point(s) of Receipt or Gathering Point of Delivery, as those terms are defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

FERC Docket: RP94-160-000

First Revised Sheet No. 627 First Revised Sheet No. 627: Superseded

Superseding: Original Sheet No. 627

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 2 TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm service for Shipper pursuant to until the earlier of; the first this Agreement from Day of the Month following six months after notification in writing from Shipper to Panhandle that the underlying transportation and storage capacity governed by this Agreement are to be administered pursuant to the Rate Schedules EFT, SCT and IOS (and WS, PS and/or FS) agreements under which Shipper's firm capacity on Panhandle's system was contracted (the underlying Service Agreements), or the date either the underlying Transportation or Storage Agreement governed by Rate Schedule EFT, SCT or IOS expires or is terminated. It is the intention of the parties that while this Agreement is in effect it shall be construed, wherever possible, to operate in conjunction with the underlying Service Agreements. In cases of conflict the terms of this Agreement shall govern.

B. Notwithstanding the above provisions of this Article 2, Panhandle shall have the right to terminate service hereunder, pursuant to any effective provisions for termination of this Agreement or any of the underlying Service Agreements by Panhandle as stated in Rate Schedules GDS, EFT, IOS or the General Terms and Conditions.

ARTICLE 3 RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule GDS filled with the Commission, as such rates and charges and Rate Schedule GDS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule GDS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

FERC Docket: RS92- 22-005

Original Sheet No. 628 Original Sheet No. 628: Superseded

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

For the information of Shipper, Exhibit B hereto states the rates and charges anticipated to be effective and applicable to the service agreed to hereunder as of the date first stated above; but, the terms of Rate Schedule GDS, other applicable provisions of Panhandle's Tariff, and the Regulations and Orders of the Commission shall control in the event of any conflict with the rates and charges stated on Exhibit B hereto.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule GDS and subject to the Regulations and Orders of the Commission. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

ARTICLE 4 FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage as stated on Exhibit A hereto (or succeeding effective percentage) of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule GDS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule GDS, Rate Schedules applicable to the underlying services, and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 629 Original Sheet No. 629: Superseded

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 6 SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7

- $7.1\,$ Periodic scheduling of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions.
- 7.2 For each Month during the term of this Agreement, Panhandle and Shipper shall develop a monthly operating plan wherein Shipper shall provide Panhandle with Shipper's nomination of Gas to be received by Panhandle in accordance with Shipper's underlying Rate Schedule EFT or SCT Agreement(s) and the Gas to be injected into or withdrawn from storage in accordance with its underlying Rate Schedule IOS Agreement.
- If Shipper has delivered sufficient Gas to Panhandle (either as EFT (or SCT) receipts or as Stored Volume) Shipper shall be entitled to take each Day, at the Point of Delivery, the Maximum Daily Contract Quantity set out above (as adjusted according to Article 8, below). As a part of the monthly operating plan described above, Panhandle shall advise Shipper each Month as to the daily minimum take that would be required to maintain Shipper's Rate Schedule EFT (or SCT) service in balance. As a part of the monthly operating plan, Shipper shall provide Panhandle with a storage use priority scheme (the order in which to apply delivery volumes to the component services). If Shipper fails to provide Panhandle with a storage use priority scheme prior to five Days before the beginning of a Month Panhandle shall use the following schemes:

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Fourth Revised Sheet No. 630 Fourth Revised Sheet No. 630: Superseded

Superseding: Third Revised Sheet No. 630

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

November through March -

- 1) EFT or SCT, up to receipt nominations
- 2) WS withdrawals (if applicable)
- 3) PS withdrawals (if applicable)
- 4) FS withdrawals (if applicable)
- 5) IOS or eligible FS withdrawals/injections

April through October -

- 1) PS injection (if applicable)
- 2) FS injection (if applicable)
- 3) WS injection (if applicable)
- 4) EFT or SCT deliveries
- 5) IOS or eligible FS injections/withdrawals

7.3 During each Month hereunder Panhandle shall be entitled to inject Gas into IOS or eligible FS storage (if Shipper has delivered sufficient Gas to Panhandle) and/or withdraw Gas from IOS or eligible FS storage for Shipper's account, in accordance with the operating plan, without further nominations from Shipper. In the event that imbalances or overruns occur due to insufficient confirmed receipts, insufficient IOS or eligible FS Stored Volume, insufficient available storage capacity or any other reason, the balancing and/or Overrun Charges and Unauthorized Overrun Penalties set out in the underlying agreements, in Rate Schedule GDS and the General Terms and Conditions shall be applicable and shall be paid by Shipper.

ARTICLE 8 - OTHER SERVICES

Shipper shall be entitled to include in the services governed by this Agreement storage services for which Shipper shall have contracted with Panhandle pursuant to Rate Schedules WS, PS and/or FS, along with Seasonal Storage Related Transportation service under Rate Schedule EFT (or SCT). Any such inclusion shall be made effective the date first written above or effective April 1 of any year wherein the stated term of this Agreement shall have at least one year remaining. If such services are included herein all obligations of Shipper under the underlying Rate Schedules WS, PS and/or FS (and EFT or SCT) agreements shall remain in force, consistent with Panhandle's authorization to withdraw Gas from such storage during the period November through March without further nominations from Shipper and to inject Gas during the period April through October, including the use of the related transportation service, in order to keep Shipper's takes from Panhandle balanced with Shipper's deliveries to Panhandle.

FERC Docket: RP94-160-000

First Revised Sheet No. 631 First Revised Sheet No. 631: Superseded

Superseding: Original Sheet No. 631

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

Where such services have been included herein they shall be taken into account in the monthly operating plan and the maximum quantity Shipper may take shall be adjusted, for the period November through March of any Contract Year, to reflect storage related deliverability, and for the period April through October of any Contract Year to reflect storage injection capability of Rate Schedules WS and FS service and injection requirements related to Rate Schedule PS storage injections.

ARTICLE 9 - NOTICES

Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 632 Second Revised Sheet No. 632 : Superseded

Superseding: First Revised Sheet No. 632

RATE SCHEDULE GDS (Continued) GENERAL DELIVERY SERVICE FORM OF SERVICE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control

any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 633 First Revised Sheet No. 633 : Superseded Superseding: Original Sheet No. 633

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE FORM OF SERVICE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Effective Date: 05/01/1993 Status: Effective FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 634} \; \texttt{Original Sheet No. 634} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE FORM OF SERVICE AGREEMENT

EXHIBIT A

Agreement Between Panhandle Eastern Pipe Line Company Firm Service For Under Rate Schedule GDS

and	
Contract No.	
Maximum Daily Contract Quantity (Dt./Day)	
Maximum Stored Quantity	_ Dt.

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 635} \; \texttt{Original Sheet No. 635} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT A

Agreement
Between
Panhandle Eastern Pipe Line Company
For
Form Sarvice

Firm Service Under Rate Schedule GDS

and				
Contract	No.			

Point of Delivery

Functionalization/

Seq. No. Redelivered To Mileage County State Meter No.

Description of Facilities

Atmos.

Existing/ Operated and Pres.
Seq. No. Proposed Installed By Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 636 First Revised Sheet No. 636 : Superseded Superseding: Original Sheet No. 636

RATE SCHEDULE GDS (Continued) GENERAL DELIVERY SERVICE FORM OF SERVICE AGREEMENT

> SUPERSEDING EXHIBIT A

Agreement Between Panhandle Eastern Pipe Line Company For Firm Service

Under Rate Schedule GDS

and	
Contract No.	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,,	EXECUTED,

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 637} \; \texttt{Original Sheet No. 637} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE
FORM OF SERVICE AGREEMENT

SUPERSEDING

EXHIBIT A

Agreement
Between
Panhandle Eastern Pipe Line Company
For
Firm Service
Under Rate Schedule GDS

and			
Contract	No.	 	

Point of Delivery

Functionalization/

Seq. No. Redelivered To Mileage County State Meter No.

Description of Facilities

Atmos. Existing/ Operated and Pres. Seq. No. Proposed Installed by Maintained by (Psia)

FERC Docket: RP94-160-000

First Revised Sheet No. 638 First Revised Sheet No. 638 : Superseded

Superseding: Original Sheet No. 638
RATE SCHEDULE GDS (Continued)
GENERAL DELIVERY SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT B

Agreement
Between
Panhandle Eastern Pipe Line Company
For
Firm Service
Under Rate Schedule GDS

and	
Contract	No.

[Description of Contract EFT]

[Description of Contract SCT]

[Description of Contract IOS]

[Description of Contract PS and related transportation]

[Description of Contract WS and related transportation]

[Description of Contract FS and related transportation]

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Second Revised Sheet No. 639 Second Revised Sheet No. 639 : Superseded

Superseding: First Revised Sheet No. 639

RATE SCHEDULE TBS
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)

In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows:

ARTICLE 1 SERVICE

1.1 Panhandle and Shipper are parties to the Storage Agreement set forth on Exhibit A hereto, pursuant to Rate Schedule IOS or FS, as applicable, of Panhandle's FERC Gas Tariff (the "Storage Agreement"). In order for Shipper or other Shippers to receive service under designated Transportation Agreements in a manner calculated to maintain nominations, receipts and deliveries in balance to the extent permitted by Shipper's capacity and Stored Volume under the Storage Agreement, Panhandle and Shipper agree as follows:

FERC Docket: RS92- 22-005

Original Sheet No. 640 Original Sheet No. 640: Superseded

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

- (a) On any Day wherein Gas is nominated and scheduled for Transportation under any of the designated Transportation Agreements and the Quantities of Gas delivered vary from the Quantities of Gas nominated by Shippers under the designated Transportation Agreements, the Tolerance level applicable to each such contract shall be increased by a sufficient quantity for each such Transportation Agreement to be within Tolerance, up to the quantity afforded by the MDWQ, MDIQ and the available Stored Volume and available Maximum Stored Quantity of the Storage Agreement, without further notice or nominations from Shipper.
- (b) For any Month wherein Gas is nominated and scheduled for Transportation under the designated Transportation Agreements and the Quantities of Gas delivered vary from the Quantities of Gas received by Panhandle for the account of Shippers under the designated Transportation Agreements, Panhandle shall schedule withdrawals and or injections of Gas pursuant to the Storage Agreement in quantities sufficient to bring delivered and received Quantities of Gas within the MAIQ of such Shippers' firm Transportation Agreements or the MMIQ of such Shippers' interruptible Transportation Agreements (as defined in Section 12.11 of the General Terms and Conditions of Panhandle's Tariff), without further notice or nominations from Shipper.

Effective Date: 02/01/2002 Status: Effective

FERC Docket: RP00-395-003

Second Revised Sheet No. 641 second Revised Sheet No. 641 : Superseded

Superseding: First Revised Sheet No. 641

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

1.2 All of the activity pursuant to the Storage Agreement to be effected by Panhandle for balancing purposes hereunder shall be subject to Shipper's; Stored Volume, available Maximum Stored Quantity, MDIQ and MDWQ. This agreement does not supersede any of the provisions of the designated Transportation Agreements or the Rate Schedules and General Terms and Conditions of Panhandle's Tariff with respect to balancing. In the event that imbalances or overruns occur due to insufficient confirmed receipts, insufficient IOS Stored Volume, insufficient available storage capacity or any other reason, the balancing and/or Overrun Charges and Unauthorized Overrun Penalties set out in the underlying agreements and the General Terms and Conditions shall be applicable and shall be paid by Shippers under the designated Transportation Agreements.

ARTICLE 2

- A. This Agreement shall be effective from the date first stated above. Panhandle shall provide transportation balancing service for Shipper pursuant to this Agreement from _______ until the earlier of; the first Day of the Month following six months after notification of cancellation in writing from Shipper to Panhandle, or the date the Storage Agreement expires or is terminated.
- B. Notwithstanding the above provisions of this Article 2, Panhandle shall have the right to terminate service hereunder, pursuant to any effective provisions for termination of this Agreement or of the underlying Service Agreement by Panhandle as stated in Rate Schedule IOS or FS, as applicable, or the General Terms and Conditions.

FERC Docket: RS92- 22-005

Original Sheet No. 642 Original Sheet No. 642: Superseded

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 3 RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule TBS filed with the Commission, as such rates and charges and Rate Schedule TBS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule TBS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

ARTICLE 4 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule TBS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule TBS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RS92- 22-005

Original Sheet No. 643 Original Sheet No. 643: Superseded

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 5 SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 6 - NOTICES

Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

ARTICLE 7 - DESIGNATION

No later than 5 Business Days prior to the beginning of each Month during the term hereof, Shipper shall designate to Panhandle, in writing, the Transportation Agreements to which balancing services hereunder shall apply for such Month.

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 644 Second Revised Sheet No. 644: Superseded

Superseding: First Revised Sheet No. 644

RATE SCHEDULE TBS (Continued) TRANSPORTATION BALANCING SERVICE FORM OF SERVICE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967

Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 645 First Revised Sheet No. 645 : Superseded

Superseding: Original Sheet No. 645

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Зу	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RS92- 22-005

 $\textbf{Original Sheet No. 646} \; \texttt{Original Sheet No. 646} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE TBS (Continued)
TRANSPORTATION BALANCING SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT A

Agreement
Between
Panhandle Eastern Pipe Line Company
For
Transportation Balancing Service
Under Rate Schedule TBS

	and	
	Contract No	
Storage Agreement:		

FERC Docket: RP99-482-000

First Revised Sheet No. 647 First Revised Sheet No. 647: Superseded

Superseding: Original Sheet No. 647

FORM OF SERVICE AGREEMENT FOR MESSENGERb

THIS AGREEMENT, executed this day of , by and between

("Messengerb Subscriber"), and Panhandle Eastern Pipe Line Company and Trunkline Gas Company ("Pipeline"), witnesseth that for and in consideration of the mutual covenants and provisions herein contained and subject to all of the terms, provisions and conditions herein set forth, Messengerp Subscriber and Pipeline do hereby agree as follows:

ARTICLE I

SCOPE OF AGREEMENT

Pipeline shall make available for use by Messengerb Subscriber Pipeline's computerized electronic communication service, the Messengerb system. Messengerb Subscriber may at its option use the Messengerb system to: (1) request new services under applicable Rate Schedules; (2) request and execute amendment of existing Service Agreements; and (3) nominate quantities for receipt and delivery by Pipeline pursuant to an existing Service Agreement under any of Pipeline's rate schedules. Messengerb Subscriber may also use the Messengerb system to request and receive from Pipeline such other information as Pipeline may from time to time make available to Messengerb Subscriber through the Messengerb system.

ARTICLE II

TERMS AND CONDITIONS

Messengerb Subscriber's use of the Messengerb system shall be in accordance with and subject to Pipeline's currently effective FERC Gas Tariff, including any and all applicable provisions of the General Terms and Conditions (the "General Terms and Conditions") and the terms and conditions of any relevant rate schedules which appear in Pipeline's currently effective FERC Gas Tariff, all of which terms and conditions are expressly incorporated by reference herein.

FERC Docket: RP99-482-000

First Revised Sheet No. 648 First Revised Sheet No. 648: Superseded

Superseding: Original Sheet No. 648

FORM OF SERVICE AGREEMENT (Continued)

FOR MESSENGERÞ

Authorized Use of Messengerp: Confidentiality

Use of the Messengerb system by Messengerb Subscriber shall be limited only to those persons who have been duly authorized by Messengerb Subscriber to use the Messengerb system (the "Authorized Personnel"), as indicated in the completed Computer Access Request Form to be submitted by Messengerb Subscriber to Pipeline, a copy of which is attached to the Messengerb Agreement. Pipeline will provide each of Messengerb Subscriber's Authorized Personnel with an individualized user identification code ("USERID") and password. The number and type of USERIDs and passwords to be issued by Pipeline shall be subject to the sole discretion of Pipeline.

Messengerb Subscriber shall also designate one or more Authorized Personnel who are additionally authorized: (1) to transmit information to Pipeline through the Messengerb system, and/or (2) to perform the contracting function and thereby legally bind Messengerb Subscriber to any service agreement or amended service agreement entered into with Pipeline (collectively, the "Designated Personnel"). Pipeline shall be entitled to rely upon Messengerb Subscriber's representation that any and all Designated Personnel authorized to perform the Messengerb system contracting function have been duly authorized by Messengerb Subscriber to enter into one or more service agreements or amended service agreements for all purposes of this Messengerb Agreement.

FERC Docket: RP99-482-000

First Revised Sheet No. 649 First Revised Sheet No. 649: Superseded

Superseding: Original Sheet No. 649
FORM OF SERVICE AGREEMENT

FORM OF SERVICE AGREEMENT (Continued)

FOR MESSENGERÞ

Messengerp Subscriber shall not disclose to persons other than Authorized Personnel, and shall otherwise keep completely confidential, all USERIDs and passwords issued to Messengerp Subscriber by Pipeline. In addition, each Authorized Personnel shall likewise not disclose to any other person, and shall otherwise keep completely confidential, the individualized USERID and password issued to such Authorized Personnel.

Messengerp Subscriber shall immediately inform Pipeline in writing whenever Messengerp Subscriber desires to terminate access to the Messengerp system previously granted to any Authorized Personnel, or desires to terminate the status of any Designated Personnel. Messengerp Subscriber shall be solely responsible for any and all unauthorized or otherwise improper use of USERIDs and passwords issued to Messengerp Subscriber by Pipeline, including but not limited to the use of such USERIDs and passwords by Authorized Personnel who at some point are no longer within Messengerp Subscriber's employment or control. Pipeline reserves the right, at any time and in Pipeline's sole discretion, to invalidate without prior notice any password or USERID issued to Messengerp Subscriber.

Messengerp Subscriber shall defend and indemnify Pipeline from and against any and all claims, demands, and actions, and any resulting loss, costs, damages, and expenses (including court costs and reasonable attorney's fees) of any nature whatsoever, which may be asserted against or imposed upon Pipeline by any party as a result of the unauthorized or otherwise improper use of any USERID or password issued by Pipeline to Messengerb Subscriber.

FERC Docket: RP99-482-000

First Revised Sheet No. 650 First Revised Sheet No. 650 : Superseded

Superseding: Substitute Original Sheet No. 650
FORM OF SERVICE AGREEMENT (Continued)
FOR MESSENGERP

Release and Disclaimer of Liability: Indemnification

Pipeline expressly disclaims any and all liability for damages to Messengerb Subscriber, except for damages directly attributable to the negligence, bad faith, fraud or willful misconduct of Pipeline, or to any third parties associated with Messengerp Subscriber's use of the Messengerb system arising out of or in any manner related to use by Messengerh Subscriber or Pipeline of the Messengerp system, including but not limited to any damages resulting from any one or more of the following: (1) any acts of God or force majeure, including but not limited to sabotage, war, riot, lightning, vermin, fire, floods, electrical storms and/or power outages, or any other natural or public calamity; (2) any defects in computer software or hardware or defective programming; (3) any interruption in or malfunction of electronic communication or transmission; or (4) any improper or erroneous service which may occur as a result of use of the Messengerb system. Pipeline's liability, if any, shall be limited to direct damages only incurred by Messengerp Subscriber and shall not extend to consequential, indirect or punitive damages incurred by Messengerb Subscriber or any damages, of any nature whatsoever incurred by third parties.

PIPELINE HEREBY DISCLAIMS ANY AND ALL EXPRESS AND IMPLIED WARRANTIES OF ANY NATURE WHATSOEVER IN CONNECTION WITH MESSENGERD SUBSCRIBER'S USE OF THE MESSENGERD SYSTEM, INCLUDING BUT NOT LIMITED TO ANY AND ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND FITNESS FOR ANY PARTICULAR PURPOSE.

Messengerb Subscriber agrees to protect, defend, indemnify, and hold Pipeline harmless against any and all loss, costs, damages, and expenses of any nature whatsoever (including court costs and reasonable attorney's fees), resulting from or otherwise related to any claim, demand, or action asserted against Pipeline, arising from or in any manner directly or indirectly connected with Messengerb Subscriber's use of the Messengerb system except for the negligence, bad faith, fraud or willful misconduct of Pipeline.

FERC Docket: RP99-482-000

First Revised Sheet No. 651 First Revised Sheet No. 651: Superseded

Superseding: Substitute Original Sheet No. 651
FORM OF SERVICE AGREEMENT (Continued)
FOR MESSENGERD

Other Terms and Conditions

Use of the Messengerb system is expressly subject to all of the General Terms and Conditions, as well as all of the terms and conditions of any Rate Schedule and Service Agreement which may be applicable to any transaction performed by Messengerb Subscriber and Pipeline by means of the Messengerb system, all of which are set forth in Pipeline's currently effective FERC Gas Tariff as amended from time to time and are hereby incorporated by reference. In the event of a conflict between the terms and conditions of this service agreement and any other applicable terms and conditions set forth in Pipeline's currently effective FERC Gas Tariff which are incorporated by referenced herein, such other terms and conditions shall govern Messengerb Subscriber's use of the Messengerb system unless otherwise specifically provided herein.

ARTICLE III

TERM

This Agreement shall be and continue in full force and effect from the date of execution hereof until terminated only for due cause or at Messengerp Subscriber's request. Messengerp Subscriber agrees that Pipeline shall have the unilateral right to file with the appropriate regulatory authority and make changes effective in (a) the rates and charges applicable to service pursuant to this Service Agreement or (b) any provision of the General Terms and Conditions applicable to this Service Agreement. Pipeline agrees that Messengerp Subscriber may protest or contest the aforementioned filings, and Messengerp Subscriber does not waive any rights it may have with respect to such filings.

FERC Docket: RP99-482-000

First Revised Sheet No. 652 First Revised Sheet No. 652: Superseded Superseding: Substitute Original Sheet No. 652

Superseding: Substitute Original Sheet No. 652 FORM OF SERVICE AGREEMENT (Continued) FOR MESSENGERÞ

ARTICLE IV

ADDRESSES

Except as herein otherwise provided or as provided in the General Terms and Conditions of Pipeline's FERC Gas Tariff, any notice, request, demand, statement, bill or payment provided for in this Agreement, or any notice which any party may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered, certified, or regular mail to the post office address of the parties hereto, as the case may be, as follows:

(a) Pipeline: Marketing Department

P. O. Box 4967

Houston, TX 77210-4967

(b) Customer:

or such other address as either party shall designate by formal written notice.

ARTICLE V

INTERPRETATION

The Interpretation and performance of this Agreement shall be in accordance with the laws of the State of Texas without recourse to the law regarding the conflict of laws.

This Agreement and the obligations of the parties are subject to all present and future valid laws with respect to the subject matter, either State or Federal, and to all valid present and future orders, rules, and regulations of duly constituted authorities having jurisdiction.

FERC Docket: RP99-482-000

First Revised Sheet No. 653 First Revised Sheet No. 653 : Superseded Superseding: Original Sheet No. 653

FORM OF SERVICE AGREEMENT (Continued)

FOR MESSENGERÞ

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be signed by their respective Presidents, Vice Presidents, or other duly authorized agents the day and year first above written.

> PANHANDLE EASTERN PIPE LINE COMPANY TRUNKLINE GAS COMPANY

ву:

(Please print or type name)

MESSENGERÞ SUBSCRIBER:

By:

(Please print or type name)

FERC Docket: RP99-506-000

Second Revised Sheet No. 654 Second Revised Sheet No. 654: Superseded

Superseding: First Revised Sheet No. 654

RATE SCHEDULE FS

FLEXIBLE STORAGE SERVICE

FORM OF STORAGE AGREEMENT

THIS STORAGE AGREEMENT, made and entered into this day of _____, ____, by and between PANHANDLE EASTERN PIPE LINE COMPANY ("Panhandle") and ("Shipper"),

WITNESSETH:

WHEREAS, Panhandle has available underground Gas storage capacity; and

WHEREAS, Shipper desires to purchase and Panhandle desires to provide certain storage and transportation service.

NOW, THEREFORE, Panhandle and Shipper do mutually agree as follows:

ARTICLE 1 DELIVERY OF THE STORED VOLUME

1.1 Shipper hereby agrees to purchase the storage service offered by Panhandle hereunder and desires to have the volumes hereinafter set forth (Stored Volume) injected into the storage facilities identified below for Shipper's account during the term hereof, up to the Maximum Stored Quantity, set out below, which shall be the maximum inventory Shipper may have injected into storage during the Injection Period as defined below and maintained in storage hereunder at any given time. Stored Volumes are to be withdrawn from storage and delivered to Shipper during the Withdrawal Period as defined below pursuant to the terms and conditions of this Storage Agreement, all in accordance with Section 2 of Rate Schedule FS and as restricted by the volumes and percentages set out below.

This contract is for Market Area Storage This contract is for Field Area Storage

Injection Period

Withdrawal Period

Maximum Stored Quantity Dt.

Maximum Daily Injection Quantity Dt./Day

Maximum Daily Withdrawal Quantity Dt./Day

FERC Docket: RP94-160-000

Original Sheet No. 655 Original Sheet No. 655: Superseded

RATE SCHEDULE FS (Continued)
FLEXIBLE STORAGE SERVICE
FORM OF STORAGE AGREEMENT

Percentage of Stored Volume to the Maximum Stored Quantity

Available Withdrawal Quantities

100% - % 100% of MDWQ
< % - % 6 MDWQ
< % - 0% 8 of MDWQ
% of MDWQ

- 1.2 In each year of the term of this Storage Agreement, Shipper may elect to retain a portion of its Stored Volume as carry-over volume (Carry-over Volume) to be left in storage during the succeeding Summer Period and withdrawn in the succeeding Winter Period. Shipper may so retain up to 20% of its Maximum Stored Quantity for such carry-over without penalty. In the event the retained Carry-over Volume exceeds 20% of its Maximum Stored Quantity, Shipper will be subject to an additional percentage retention of Gas as set forth in Section 1.3. The sum of the Carry-over Volume and the volumes delivered by Shipper for storage during any Summer Period shall not exceed Shipper's Maximum Stored Quantity set forth in Section 1.1 of this Storage Agreement.
- 1.3 Shipper agrees to furnish, in addition to volumes delivered into storage, a volume of Gas equal to the percentage set out in effective Tariff Sheet No. 14 of such volumes delivered for injection into storage in each year of the term hereof to be retained by Panhandle for use as compressor fuel, plus an additional volume of Gas equal to the percentage set out in effective Tariff Sheet No. 14 of volumes delivered to Shipper in each year of the term hereof to be retained by Panhandle for use as compressor fuel. No compressor fuel need be furnished for volumes carried over from any Winter Period unless Shipper retains more than 20% of its Maximum Stored Quantity in any year as Carry-over Volumes. In such event, pursuant to Section 1.2, such Carry-over Volumes in excess of 20% of Shipper's Maximum Stored Quantity shall be subject to a .25% additional compressor fuel retention by Panhandle.
- $1.4\,$ Panhandle shall receive from Shipper and inject into storage each month during the Summer Period Gas as nominated pursuant to the General Terms and Conditions.
- 1.5 Where Shipper has designated as the WS Point a point not located on Panhandle's system, then the maximum quantities for injection and withdrawal on any Day shall be subject to any restrictions or other limitations applicable to the lease or storage

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fifth Revised Sheet No. 656 Fifth Revised Sheet No. 656: Superseded

Superseding: Sub Fourth Revised Sheet No. 656
RATE SCHEDULE FS (Continued)
FLEXIBLE STORAGE SERVICE
FORM OF STORAGE AGREEMENT

agreement between Panhandle and the owner or operator of the storage facility, in addition to the restrictions and limitations of this agreement and Rate Schedule FS.

ARTICLE 2 BILLING AND PAYMENT

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule FS filed with the Commission, as such rates and charges and Rate Schedule FS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule FS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule FS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually injected, withdrawn or stored (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually injected, withdrawn or stored); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Second Revised Sheet No. 657 Second Revised Sheet No. 657: Superseded

Superseding: First Revised Sheet No. 657
RATE SCHEDULE FS (Continued)
FLEXIBLE STORAGE SERVICE
FORM OF STORAGE AGREEMENT

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit A hereto.

ARTICLE 3 POINTS OF RECEIPT AND DELIVERY

During the Injection Period Gas shall be received by Panhandle from Shipper for injection into storage and during the Withdrawal Period the Stored Volume withdrawn for the account of Shipper hereunder shall be delivered to Shipper, at the WS Point as defined in the General Terms and Conditions on a firm basis.

ARTICLE 4 TERM

4.1 This Agreement shall be effective from the date first stated above. Panhandle shall provide firm storage service for Shipper pursuant to this Agreement from until , when the term of this Agreement shall expire.

ARTICLE 5 GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule FS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule FS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RP97- 4-002

Second Revised Sheet No. 658 Second Revised Sheet No. 658: Superseded

Superseding: First Revised Sheet No. 658

RATE SCHEDULE FS (Continued)
FLEXIBLE STORAGE SERVICE
FORM OF STORAGE AGREEMENT

ARTICLE 6 SUCCESSION AND ASSIGNMENT

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7

Except as herein otherwise provided, any notice, request, demand, statement or bill provided for in this Storage Agreement and in the General Terms and Conditions of the Tariff and the Rate Schedules to which they apply, or any notices which either Panhandle or Shipper may desire to give to the other, shall be in writing and shall be considered as duly delivered when mailed by registered mail to the post office address of Panhandle or Shipper, or at such other address as either shall designate by formal written notice. Routine communications, including monthly statements and payments, shall be considered as delivered when mailed by either registered or ordinary mail. The Post Office addresses of both Panhandle and Shipper are as follows:

Effective Date: 06/15/2003 Status: Effective

FERC Docket: RP03-479-000

Second Revised Sheet No. 659 Second Revised Sheet No. 659: Superseded

Superseding: First Revised Sheet No. 659

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE FORM OF STORAGE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for any other purpose)

Attn: Gas Control P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 660 First Revised Sheet No. 660 : Superseded

Superseding: Original Sheet No. 660

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE FORM OF STORAGE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000 Second Revised Sheet No. 661 Second Revised Sheet No. 661 : Superseded Superseding: First Revised Sheet No. 661 RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE FORM OF SERVICE AGREEMENT EXHIBIT A Storage Agreement For Firm Service Under Rate Schedule FS Contract No. NEGOTIATED RATE AGREEMENT Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule FS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing , and continuing until
_____, ____. Shipper acknowledges that this election is an
alternative to the billing of charges for Rate Schedule FS set forth
on effective Tariff Sheet No. 14, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule FS. Specification of Negotiated Rate: PANHANDLE EASTERN PIPE LINE COMPANY BY: __ (Please type or print name) SHIPPER: __ BY:

(Please type or print name)

SUPERSEDES EXHIBIT A DATED:

DATED: __

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 662 Second Revised Sheet No. 662: Superseded Superseding: First Revised Sheet No. 662

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 663 Second Revised Sheet No. 663: Superseded Superseding: First Revised Sheet No. 663

RESERVED FOR FUTURE USE.

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

Second Revised Sheet No. 664 Second Revised Sheet No. 664 : Superseded Superseding: First Revised Sheet No. 664

RESERVED FOR FUTURE USE.

FERC Docket: RP97-185-002

Sub Original Sheet No. 664A Sub Original Sheet No. 664A: Superseded

RATE SCHEDULE FS (Continued) FLEXIBLE STORAGE SERVICE FORM OF SERVICE AGREEMENT

EXHIBIT C

Storage Agreement
For
Firm Service
Under Rate Schedule FS
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.9 of Rate Schedule FS and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _______, 19___ and continuing until ________, 19___. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule FS set forth on effective Tariff Sheet No. 14, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule FS.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY
BY:				
SHIPPER:				
BY:				
DATED:				
SUPERSEDES	EXHIBIT	r c da	ATED:	

Effective Date: 11/01/1999 Status: Effective FERC Docket: RP99-506-000
First Revised Sheet No. 665 First Revised Sheet No. 665: Superseded Superseding: Substitute Original Sheet No. 665 RATE SCHEDULE LFT LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT
CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"),
Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies)
Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
This contract is Firm Seasonal Storage Related
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery quantities of Natural Gas up to the following daily quantities (Dt./Day), which shall constitute the Maximum Daily Contract Quantity:
(Dt./Day)
Service hereunder is provided on a firm basis subject to Panhandle's right to not schedule service in whole or in part on any Day, but not more than ten (10) Days in each Month.

FERC Docket: RP95-397-003

Substitute Original Sheet No. 666 Substitute Original Sheet No. 666: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule LFT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the primary Point(s) of Receipt set out on Exhibit A, as well as among the primary Point(s) of Delivery set out on Exhibit A.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as are defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide Transportation service for Shipper pursuant to this Agreement from _____ until ____, when this Agreement shall expire.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Fourth Revised Sheet No. 667 Fourth Revised Sheet No. 667: Superseded

Superseding: Sub Third Revised Sheet No. 667
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule LFT filed with the Commission, as such rates and charges and Rate Schedule LFT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule LFT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule LFT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

Effective Date: 01/15/2000 Status: Effective

FERC Docket: RP00-137-000

Third Revised Sheet No. 668 Third Revised Sheet No. 668: Superseded

Superseding: Second Revised Sheet No. 668
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit C hereto.

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule LFT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule LFT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule LFT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RP95-397-003

Substitute Original Sheet No. 669 Substitute Original Sheet No. 669: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

FERC Docket: RP03-479-000

Second Revised Sheet No. 670 Second Revised Sheet No. 670: Superseded

Superseding: First Revised Sheet No. 670

RATE SCHEDULE LFT (Continued) LIMITED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967

Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 671 First Revised Sheet No. 671 : Superseded Superseding: Substitute Original Sheet No. 671

Superseding: Substitute Original Sheet No. 6/1
RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Зу	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

Substitute Original Sheet No. 672 Substitute Original Sheet No. 672: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Service Under
Rate Schedule LFT

and

Contract No.

Effective Date: 01/25/1996 Status: Effective

Maximum Daily Contract Quantity (Dt./Day)

FERC Docket: RP95-397-003

Substitute Original Sheet No. 673 Substitute Original Sheet No. 673: Superseded

RATE SCHEDULE LFT (Continued) LIMITED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Service Under Rate Schedule LFT

Primary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Atmos. Operated and Pres.

Existing/ Installed by Maintained by (Psia) Proposed

Secondary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Atmos. Existing/ Operated and Pres.

Proposed Installed by Maintained by (Psia)

FERC Docket: RP95-397-003

Substitute Original Sheet No. 674 Substitute Original Sheet No. 674: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
For
Service Under
Rate Schedule LFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

First Revised Sheet No. 675 First Revised Sheet No. 675 : Superseded Superseding: Substitute Original Sheet No. 675

Superseding: Substitute Original Sheet No. 675

RATE SCHEDULE LFT (Continued)

LIMITED FIRM TRANSPORTATION SERVICE

FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Service Under
Rate Schedule LFT

and	
Contract No	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RP95-397-003

Substitute Original Sheet No. 676 Substitute Original Sheet No. 676: Superseded

RATE SCHEDULE LFT (Continued) LIMITED FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement For Service Under Rate Schedule LFT

Primary Point(s) of Delivery

Functionalization/

Delivered To Seq. No. Mileage County State Meter No. Quantity

Description of Facilities

Atmos. Existing/ Operated and Pres.

Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Function-

alization/

County State Meter No. Quantity Seq. No. Delivered To Mileage

Description of Facilities

Atmos. Existing/ Operated and Pres.

Maintained by (Psia) Proposed Installed by

FERC Docket: RP95-397-003

Substitute Original Sheet No. 677 Substitute Original Sheet No. 677: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Service Under
Rate Schedule LFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

FERC Docket: RP99-506-000

Third Revised Sheet No. 678 Third Revised Sheet No. 678: Superseded

Superseding: Second Revised Sheet No. 678

RATE SCHEDULE LFT (Continued) LIMITED FIRM TRANSPORTATION SERVICE FORM OF SERVICE AGREEMENT

EXHIBIT B

Transportation Agreement For Firm Service Under Rate Schedule LFT Contract No.

CUSTOMIZED RESERVATION PATTERN'D ELECTION

(Shipper) hereby elects the Customized
Reservation Patternb option in accordance with Section 3.10
of Panhandle's Rate Schedule LFT and hereby notifies Panhandle
that it desires to be billed, and agrees to pay, the Reservation
Charges elected from time to time for the period commencing
November 1, through October 31, .
·
Shipper acknowledges that this election is an alternative to the
uniform monthly billing of reservation charges for Rate Schedule
LFT set forth on effective Tariff Sheet No. 15, as revised from
time to time, and that Shipper's election shall constitute its
agreement to pay the otherwise applicable charges according to t
payment schedule set forth in its election including such

the additional charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming

through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE	EASTERN	1 PIPE	. بل ک	INE CON	1PANY	
BY:						
	(Please	type	or	print	name)	
SHIPPER:						
BY:						
	(Please	type	or	print	name)	
DATED:						
SUPERSEDES EXHIBIT B DATED.						

FERC Docket: RP99-506-000

Third Revised Sheet No. 679 Third Revised Sheet No. 679: Superseded

Superseding: Second Revised Sheet No. 679

RATE SCHEDULE LFT (Continued)

LIMITED FIRM TRANSPORTATION SERVICE

FORM OF SERVICE AGREEMENT

EXHIBIT C

Transportation Agreement
For
Firm Service
Under Rate Schedule LFT
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule LFT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, and continuing until _____, ___. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule LFT set forth on effective Tariff Sheet No. 15, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule LFT.

Specification of Negotiated Rate:

PANHANDLE	EASTERN PIPE LINE COMPANY
BY:	
_	
	(Please type or print name)
SHIPPER:	
BY:	
- -	(Please type or print name)
DATED:	
SUPERSEDE	S EXHIBIT C DATED:

Effective Date: 04/01/1997 Status: Effective FERC Docket: RP97- 4-002

First Revised Sheet No. 680 First Revised Sheet No. 680: Superseded Superseding: Substitute Original Sheet No. 680

RESERVED FOR FUTURE USE.

FERC Docket: RP96-224-000

Original Sheet No. 680A Original Sheet No. 680A: Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT D

Transportation Agreement
For
Firm Service
Under Rate Schedule LFT
Contract No.

CUSTOMIZED RESERVATION PATTERND ELECTION

(Shipper) hereby elects the Customized Reservation Patternp option in accordance with Section 3.10 of Panhandle's Rate Schedule LFT and hereby notifies Panhandle that it desires to be billed, and agrees to pay, the Reservation Charges elected from time to time for the period commencing November 1, 199_ through October 31, 199_.

Shipper acknowledges that this election is an alternative to the uniform monthly billing of reservation charges for Rate Schedule LFT set forth on effective Tariff Sheet No. 15, as revised from time to time, and that Shipper's election shall constitute its agreement to pay the otherwise applicable charges according to the payment schedule set forth in its election including such additional charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER:					
BY:					
DATED:					
SUPERSEDES	SEXHIBIT	מ חיז	ATED:		

FERC Docket: RP97-185-002

Sub Original Sheet No. 680B Sub Original Sheet No. 680B : Superseded

RATE SCHEDULE LFT (Continued)
LIMITED FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT E

Transportation Agreement
For
Firm Service
Under Rate Schedule LFT
Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.11 of Rate Schedule LFT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing ______, 19__ and continuing until ______, 19__. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule LFT set forth on effective Tariff Sheet No. 15, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule LFT.

Specification of Negotiated Rate:

PANHANDLE	EASTERN	PIPE	LINE	COMPANY	
BY:					
SHIPPER: _					_
BY:					
DATED:					
SUPERSEDES	EXHIBI:	r e da	ATED:		

Effective Date: 02/01/2002 Status: Effective FERC Docket: RP00-395-003 Second Revised Sheet No. 681 Second Revised Sheet No. 681: Superseded Superseding: First Revised Sheet No. 681 RATE SCHEDULE GPS GAS PARKING SERVICE FORM OF SERVICE AGREEMENT CONTRACT NO. THIS AGREEMENT, made and entered into as of the _____ day of _____, ____, by and between: PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation, (hereinafter called "Shipper"), Shipper represents and warrants that Shipper conforms to the requirements of 18 C.F.R. Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) _____ Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others) In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Shipper covenant and agree as follows: ARTICLE 1 - SERVICE 1.1 Panhandle agrees to receive Gas for the Parking Account of Shipper and park, on an interruptible and capacity available basis, quantities of Natural Gas at the specified Parking point up to the Maximum Parked Quantity as follows: Maximum Parked Quantity Description of Parking Point: _____ Pooling Transportation Agreement, if Shipper elects for Panhandle to manage its parking service at a Pool Point:

At no time shall Shipper's parked quantity exceed its Maximum Parked Quantity. Parked Quantities are to be delivered in accordance with Shipper's nominations pursuant to the terms and conditions of this agreement and Section 2 of Rate Schedule GPS.

FERC Docket: RP99-199-000

Second Revised Sheet No. 682 Second Revised Sheet No. 682: Superseded

Superseding: First Revised Original Sheet No. 682 RATE SCHEDULE GPS (Continued)

ATE SCHEDULE GPS (Continued GAS PARKING SERVICE FORM OF SERVICE AGREEMENT

1.2 Shipper shall have the right to nominate for delivery to it at the Parking Point, as described in Rate Schedule GPS, on an interruptible basis, any portion of its Parked Quantity, up to the Maximum Parked Quantity, on any Day during the term hereof. In no event shall quantities delivered hereunder to Shipper ever exceed Shipper's Parked Quantity as defined in Section 1.1. Panhandle shall not be required to receive or deliver quantities of Gas on any Day for which there is insufficient available capacity under Shipper's Transportation Agreements to deliver Gas to or receive Gas from the Parking Point.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide interruptible Parking service for Shipper pursuant to this Agreement from _____ until ____ , when the term of this $\overline{\text{Agreement shall expire}}$.

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule GPS filed with the Commission, as such rates and charges and Rate Schedule GPS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule GPS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule GPS and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; or (d) in a specified relationship to the Quantities actually parked (i.e.,

FERC Docket: RP99-199-000

Original Sheet No. 682A <code>Original Sheet No. 682A</code> : Superseded

RATE SCHEDULE GPS (Continued)
GENERAL PARKING SERVICE
FORM OF SERVICE AGREEMENT

that the rates shall be adjusted in a specified relationship to Quantities actually parked); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

FERC Docket: RP96- 51-000

First Revised Original Sheet No. 683 First Revised Original Sheet No. 683: Superseded

RATE SCHEDULE GPS (Continued)
GENERAL PARKING SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 4 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule GPS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule GPS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 5 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 6 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

FERC Docket: RP03-479-000

Second Revised Sheet No. 684 Second Revised Sheet No. 684: Superseded

Superseding: First Revised Sheet No. 684

RATE SCHEDULE GPS (Continued) GAS PARKING SERVICE FORM OF SERVICE AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Attn: Marketing Operations P. O. Box 4967 Scheduling:

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for any other purpose)

Attn: Gas Control P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 685 First Revised Sheet No. 685: Superseded Superseding: First Revised Original Sheet No. 685

Superseding: First Revised Original Sheet No. 685
RATE SCHEDULE GPS (Continued)
GAS PARKING SERVICE
FORM OF SERVICE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,

FERC Docket: RP99-506-000

First Revised Sheet No. 686 First Revised Sheet No. 686: Superseded

Superseding: 1st Rev Original Sheet No. 686
CAPACITY RELEASE
FORM OF SERVICE AGREEMENT

CONTRACT NO.

THIS AGREEMENT is made effective as of the ____ day of ____, by and between:

PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,

and

(hereinafter called "Replacement Shipper").

In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and Replacement Shipper covenant and agree as follows:

ARTICLE 1 - SERVICE

For each occasion that Replacement Shipper obtains an assignment of service rights from a Releasing Shipper, an Addendum in the form of Exhibit A, attached hereto will be made a part hereof. The specific terms and conditions of each release shall be reflected in each Addendum, which shall be incorporated and made a part of this Agreement, and which together shall constitute the terms and conditions of Panhandle's service for each release.

Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a firm basis, Quantities of Natural Gas up to the Maximum Daily Contract Quantity and/or Maximum Stored Quantity obtained from the Releasing Shipper. The Maximum Daily Contract Quantity is stated in delivered Quantities, for which received Quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedules applicable to the Releasing Shipper.

FERC Docket: RP00-162-000

First Revised Sheet No. 687 First Revised Sheet No. 687: Superseded

Superseding: 1st Rev Original Sheet No. 687
CAPACITY RELEASE (Continued)
FORM OF SERVICE AGREEMENT

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above until , when this Agreement shall expire. Service shall commence and remain effective for a term coincidental for the term of each assignment of service rights identified in each Addendum

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder (including any Addendum hereunder), Replacement Shipper agrees to pay Panhandle the maximum rate or the agreed upon rate for the assignment of service rights from the Releasing Shipper under Panhandle's Rate Schedule FT, EFT, LFT, IOS, WS, PS, FS or HFT filed with the Commission, as such rates and charges and Rate Schedules FT, EFT, LFT, IOS, WS, PS or FS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule FT, EFT, LFT, IOS, WS, PS, FS or HFT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

FERC Docket: RP99-506-000

First Revised Sheet No. 688 First Revised Sheet No. 688: Superseded

Superseding: 1st Rev Original Sheet No. 688

CAPACITY RELEASE (Continued) FORM OF SERVICE AGREEMENT

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3 (including each Addendum hereunder), Panhandle shall retain, as Fuel Reimbursement, the percentage of the Quantities received from Replacement Shipper hereunder, for reimbursement in kind from Replacement Shipper for fuel usage and unaccounted-for line loss as provided by the applicable Rate Schedule.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of the applicable Rate Schedule and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of the applicable Rate Schedules and the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - NOTICES

The Post Office addresses of both Panhandle and Shipper are as follows:

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

FERC Docket: RP03-479-000

Second Revised Sheet No. 689 Second Revised Sheet No. 689: Superseded

Superseding: First Revised Sheet No. 689

CAPACITY RELEASE (Continued) FORM OF SERVICE AGREEMENT

PANHANDLE

Nomination and Panhandle Eastern Pipe Line Company

Attn: Marketing Operations P. O. Box 4967 Scheduling:

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control P. O. Box 4967 any other purpose)

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-4272 or 1-800-275-7375

Facsimile: (713) 989-1178

1st Rev Origi	nal Sheet I	No. 690 1st	Rev Original	l Sheet	No.	690 :	Superseded
	CAPACITY FORM OF	RELEASE (Co: SERVICE AG	ntinued) REEMENT				
		REPLACEMENT	SHIPPER				
Billing:							
Nomination an Scheduling: (
All Other:							

(1) Please provide street address in addition to mailing address.

FERC Docket: RP99-506-000

First Revised Sheet No. 691 First Revised Sheet No. 691 : Superseded Superseding: 1st Rev Original Sheet No. 691

Superseding: 1st Rev Original Sheet No. 691
CAPACITY RELEASE (Continued)
FORM OF SERVICE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Replacement Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so.

REPLACEMENT	SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву		Ву
(Please	type or print name)	(Please type or print name)
Title		Title
EXECUTED	(Date)	EXECUTED,

Effective Date: 11/01/1999 Status: Effective
FERC Docket: RP99-506-000

First Revised Sheet No. 692 First Revised Sheet No. 692: Superseded Superseding: 1st Rev Original Sheet No. 692
CAPACITY RELEASE (Continued)
FORM OF SERVICE AGREEMENT

EXHIBIT A

Capacity Release Agreement No. ______
Addendum No. _____

Addendum No. ____
Capacity Release
Service Agreement
Between
Panhandle Eastern Pipe Line Company
and

Releasing Shipper Releasing Shipper Rate Schedule Contract No. Original Releasing Shipper Contract No. _____ Regulation _____ Replacement Shipper's Maximum Daily Quantity (Dt) ___ Permanent or Temporary Release _____ Conditions of Recall Term of Release Begin End _____ Reservation Rate (including applicable surcharges) Other Conditions

FERC Docket: RP99-506-000

First Revised Sheet No. 693 First Revised Sheet No. 693 : Superseded Superseding: 1st Rev Original Sheet No. 693

CAPACITY RELEASE (Continued) FORM OF SERVICE AGREEMENT

EXHIBIT A

Capacity Release Agreement No. ____ Addendum No. ___

> CAPACITY RIGHTS Primary Points of Delivery

Function-

Delivered alization/ Meter Seq. No. County St Quantity No. To Mileage

Description of Facilities

Atmos. Operated and Installed by Existing/ Pres. Maintained by (Psia) Proposed

FERC Docket: RP99-506-000

First Revised Sheet No. 694 First Revised Sheet No. 694: Superseded Superseding: 1st Rev Original Sheet No. 694

CAPACITY RELEASE (Continued) FORM OF SERVICE AGREEMENT

EXHIBIT A

Capacity Release Agreement No. ____ Addendum No. ___

> CAPACITY RIGHTS Primary Points of Receipt

Function-Received alization/

Meter County St Quantity No. Seq. No. From Mileage

Description of Facilities

Atmos. Existing/ Operated and Installed by Pres. Maintained by (Psia) Proposed

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 695 1st Rev Original Sheet No. 695 : Superseded

RATE SCHEDULE HFT
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

FORT OF INANDIONIATION AGNEEMENT
CONTRACT NO.
THIS AGREEMENT, made and entered into as of the day of,, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "Shipper"), Shipper represents and warrants that Shipper conforms to the
requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
ARTICLE 1 - SERVICE
Panhandle agrees to receive at the Point(s) of Receipt and deliver at the Point(s) of Delivery, on a firm basis, quantities of Natural Gas up to the following Maximum Daily Contract Quantity over the Hourly Delivery Period selected by Shipper:

Maximum Daily Contract Quantity _____ (Dt./Day)

FERC Docket: RP00-162-000

1st Rev Original Sheet No. 696 1st Rev Original Sheet No. 696: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

The Maximum Daily Contract Quantity is stated in delivered quantities, for which received quantities must be adjusted for fuel usage and lost or unaccounted for Gas as set out in the then-effective, applicable rates and charges under Panhandle's Rate Schedule HFT.

Exhibit A hereto states the Point(s) of Receipt and Delivery. Exhibit A may be revised from time to time by written agreement between Panhandle and Shipper and, as may be revised, is by this reference incorporated in its entirety into this Agreement and made an integral part hereof. Shipper's Maximum Daily Contract Quantity shall be assigned among the primary Point(s) of Receipt set out on Exhibit A, as well as among the primary Point(s) of Delivery set out on Exhibit A. Such assignment may be changed, subject to the availability of capacity, in accordance with the General Terms and Conditions and Rate Shedule HFT.

Shipper recognizes that no Point(s) of Receipt or Point(s) of Delivery hereunder shall include a Point(s) of Receipt or Point(s) of Delivery on Panhandle's Gathering facilities, unless Shipper agrees to accept a completely separate Gathering service. By designating a firm Point(s) of Receipt or firm Point(s) of Delivery hereunder which is located only on Panhandle's Gathering facilities as defined in the General Terms and Conditions of Panhandle's FERC Gas Tariff, First Revised Volume No. 1, Shipper shall be deemed to have requested such firm Gathering service.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm Transportation service for Shipper pursuant to this Agreement from _____ until _____, when this Agreement shall expire.

FERC Docket: RP00-162-000

Original Sheet No. 697 Original Sheet No. 697: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, Shipper agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule HFT filed with the Commission, as such rates and charges and Rate Schedule HFT may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule HFT and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and Shipper may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule HFT and subject to the Regulations and Orders of the Commission. For example, Panhandle and Shipper may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Points of Receipt, Points of Delivery, mainline segments, transportation paths or defined geographical areas; (e) only to production reserves committed by the Shipper; or (f) in a specified relationship to the Quantities actually transported (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually transported); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and Shipper.

FERC Docket: RP00-162-000

Original Sheet No. 698 Original Sheet No. 698: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit C hereto.

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule HFT of the quantities received from Shipper hereunder, for reimbursement in kind from Shipper for fuel usage and unaccounted for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule HFT and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule HFT and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or Shipper by merger, consolidation or acquisition. Either Panhandle or Shipper may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, except as provided in Section 15 of the General Terms and Conditions, neither Panhandle nor Shipper shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

FERC Docket: RP00-162-000

Original Sheet No. 699 Original Sheet No. 699: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

ARTICLE 7 - NOMINATIONS AND NOTICES

Periodic nominations of the quantities of service requested hereunder must be provided by Shipper to Panhandle, at the times and according to the procedures specified in the General Terms and Conditions and Rate Schedule HFT. Except as otherwise specified, any other notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or Shipper may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and Shipper are as follows:

FERC Docket: RP03-479-000

First Revised Sheet No. 700 First Revised Sheet No. 700: Superseded

Superseding: Original Sheet No. 700

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(And nominations Attn: Gas Control

pursuant to Section 2.4 P. O. Box 4967

of Rate Schedule HFT) Houston, Texas 77210-4967

Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 (713) 627-4272 or Phone: 1-800-275-7375

Facsimile: (713) 989-1178

SHIPPER

Billing:

Nomination and Scheduling: (1)

All Other:

(1) Please provide street address in addition to mailing address.

FERC Docket: RP00-162-000

 $\textbf{Original Sheet No. 701} \; \texttt{Original Sheet No. 701: Superseded}$

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

IN WITNESS WHEREOF, both Panhandle and Shipper have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,,

FERC Docket: RP00-162-000

 $\textbf{Original Sheet No. 702} \; \texttt{Original Sheet No. 702} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Firm Service
Under Rate Schedule HFT

and	
Contract No	
Maximum Daily Contract Quantity	(Dt./Day)
Hourly Delivery Period	(hours/Day)
Hourly Delivery Quantity	(Dt./hour)

FERC Docket: RP00-162-000

Original Sheet No. 703 Original Sheet No. 703: Superseded

RATE SCHEDULE HFT (Continued) HOURLY FIRM TRANSPORTATION SERVICE FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule HFT

Primary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Atmos.

Existing/ Operated and Pres. Installed by Maintained by (Psia) Proposed

Secondary Point(s) of Delivery

Function-

alization/
Mileage County State Meter No. Quantity Seq. No. Delivered To

Description of Facilities

Atmos.

Existing/ Operated and Pres. Proposed Installed by Maintained by (Psia)

FERC Docket: RP00-162-000

Original Sheet No. 704 Original Sheet No. 704: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

EXHIBIT A

Transportation Agreement For Firm Service Under Rate Schedule HFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Meter No. Quantity

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

(Date)

FERC Docket: RP00-162-000

 $\textbf{Original Sheet No. 705} \; \texttt{Original Sheet No. 705} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
Between
Panhandle Eastern Pipe Line Company
For
Firm Service
Under Rate Schedule HFT

and	
Contract No.	
Effective Date	
Supersedes Exhibit A dated	
Maximum Daily Contract Quantity	(Dt./Day)
Hourly Delivery Period	(hours/Day)
Hourly Delivery Quantity	(Dt./hour)
SHIPPER:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED .	EXECUTED

(Date)

FERC Docket: RP00-162-000

Original Sheet No. 706 Original Sheet No. 706: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule HFT

Primary Point(s) of Delivery

Function-

alization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Delivery

Function-

alization/

Seq. No. Delivered To Mileage County State Meter No. Quantity

Description of Facilities

Atmos.

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

FERC Docket: RP00-162-000

Original Sheet No. 707 Original Sheet No. 707: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF TRANSPORTATION AGREEMENT

SUPERSEDING

EXHIBIT A

Transportation Agreement
For
Firm Service
Under Rate Schedule HFT

Primary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

Secondary Point(s) of Receipt

Functionalization/

Seq. No. Received From Mileage County State Quantity Meter No.

Description of Facilities

Existing/ Operated and Pres.
Proposed Installed by Maintained by (Psia)

FERC Docket: RP00-162-000

Original Sheet No. 708 Original Sheet No. 708: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT B

Transportation Agreement
For
Hourly Firm Service
Under Rate Schedule HFT
Contract No.

CUSTOMIZED RESERVATION PATTERND ELECTION

(Shipper) hereby elects the Customized
Reservation Patternp option in accordance with Section 3.10
of Panhandle's Rate Schedule HFT and hereby notifies Panhandle
that it desires to be billed, and agrees to pay, the Reservation
Charges elected from time to time for the period commencing
November 1, through October 31,
Shipper acknowledges that this election is an alternative to the
uniform monthly billing of reservation charges for Rate Schedule
HFT set forth on effective Tariff Sheet No. 17, as revised from time
to time, and that Shipper's election shall constitute its agreement
to pay the otherwise applicable charges according to the payment
schedule set forth in its election including such additional charges as may be required to ensure that the total applicable
annual Reservation Charge obligations are paid in full. Shipper's
annual Reservation Charge Obligations are paid in full. Shipper's

charges as may be required to ensure that the total applicable annual Reservation Charge obligations are paid in full. Shipper's election constitutes its waiver of reliance on uniform monthly billing requirements otherwise applicable for the affected months. Shipper hereby indemnifies Panhandle against and holds Panhandle harmless for any claims of Shipper or any person claiming through Shipper as to rights to billing different from those elected by Shipper.

PANHANDLE EASTERN PIPE LINE COMPANY

PANHANDLE	EASTERN PIPE LINE COMPANY	
BY:		
	(Please type or print name)	
SHIPPER:		
BY:		
	(Please type or print name)	
DATED:		
SUPERSEDES EXHIBIT B DATED:		

FERC Docket: RP00-162-000

Original Sheet No. 709 Original Sheet No. 709: Superseded

RATE SCHEDULE HFT (Continued)
HOURLY FIRM TRANSPORTATION SERVICE
FORM OF SERVICE AGREEMENT

EXHIBIT C

Transportation Agreement For Hourly Firm Service Under Rate Schedule HFT Contract No.

NEGOTIATED RATE AGREEMENT

Shipper agrees to the Negotiated Rate option in accordance with Section 3.12 of Rate Schedule HFT and notifies Panhandle that it desires to be billed, and agrees to pay, the charges specified below for the period commencing _____, and continuing until _____, ____. Shipper acknowledges that this election is an alternative to the billing of charges for Rate Schedule HFT set forth on effective Tariff Sheet No. 17, as revised from to time. Shipper also acknowledges that its election constitutes waiver of its reliance on and its right to use the recourse rates which are available to it under the Rate Schedule HFT.

Specification of Negotiated Rate:

PANHANDLE	E EASTERN PIPE LINE COMPANY
BY:	
	(Please type or print name)
SHIPPER:	
BY:	
	(Please type or print name)
DATED:	
SUPERSEDE	S EXHIBIT C DATED:

Effective Date: 02/01/2002 Status: Effective FERC Docket: RP00-395-003

 $\textbf{Original Sheet No. 710} \; \texttt{Original Sheet No. 710} \; : \quad \texttt{Superseded}$

RATE SCHEDULE DVS DELIVERY VARIANCE SERVICE FORM OF SERVICE AGREEMENT

CONTRACT NO
THIS AGREEMENT, made and entered into as of the day of, by and between:
PANHANDLE EASTERN PIPE LINE COMPANY, (hereinafter called "Panhandle"), a Delaware Corporation,
and
(hereinafter called "DVS Party"),
DVS Party represents and warrants that DVS Party conforms to the requirements of 18 C.F.R.
Section 284.102 (284B - Intrastate Pipelines or
Local Distribution Companies) Section 284.222 (284G - Interstate Pipelines) Section 284.223 (284G - Others)
DVS Party - Shipper (Article 1.1(a),(b) and (c) apply) DVS Party - Delivery Point operator (Article 1.1 (a) and (b) apply)
In consideration of the mutual covenants and agreements as herein set forth, both Panhandle and DVS Party covenant and agree as follows:
ARTICLE 1 - SERVICE
1.1 DVS Party agrees to purchase the Delivery Variance Service offered by Panhandle and desires to have Panhandle manage on a firm basis the daily scheduling variances at a single Point of Delivery up to the Maximum Daily Variance Quantity (MDVQ) as follows:
(a) Maximum Daily Variance Quantity (MDVQ) Dt.
(b) Description of Point of Delivery:
(c) Underlying Transportation Service Agreement(s):

FERC Docket: RP00-395-003

Original Sheet No. 711 Original Sheet No. 711: Superseded

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE
FORM OF SERVICE AGREEMENT

1.2 The MDVQ shall not entitle the Shipper to take Quantities in excess of the MDCQ of the underlying Transportation Agreement(s).

- 1.3 If the DVS Party is a Shipper and the daily scheduling variance at the Point of Delivery exceeds the tolerance level, the MDVQ shall be added to the absolute value of the Shipper's calculated tolerance Quantity for the underlying Transportation Agreement(s) as provided under Section 12.11(h) of the General Terms and Conditions. Any variance then remaining shall be allocated back to the underlying Transportation Agreement(s) and shall be subject to the daily scheduling charge.
- 1.4 If the DVS Party operates a Point of Delivery, the MDVQ shall be added to the absolute value of the calculated tolerance Quantity for the delivery meter as provided in Section 12.11(h) of the General Terms and Conditions. Any variance then remaining shall be allocated back to the underlying Transportation Agreement(s) based on the predetermined allocation methodology for the designated Point of Delivery and shall be subject to the daily scheduling charge.

ARTICLE 2 - TERM

This Agreement shall be effective from the date first stated above. Panhandle shall provide firm delivery variance service for the DVS Party pursuant to this Agreement from _____ until ____, when the term of this Agreement shall expire.

ARTICLE 3 - RATES AND CHARGES

For the services provided or contracted for hereunder, DVS Party agrees to pay Panhandle the then-effective, applicable rates and charges under Panhandle's Rate Schedule DVS filed with the Commission, as such rates and charges and Rate Schedule DVS may hereafter be modified, supplemented, superseded, or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the terms or rate levels under Rate Schedule DVS and the applicability thereof, the General Terms and Conditions or any other provisions of Panhandle's Tariff, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder.

From time to time Panhandle and DVS Party may agree in writing, on a level of discount of the otherwise applicable rates and charges hereunder, pursuant to the effective applicable provisions of Rate Schedule DVS and subject to the Regulations and Orders of the Commission. For example, Panhandle and DVS Party may agree that a specified discounted rate shall apply: (a) only to certain Quantities under this Agreement; (b) only if specified Quantity

FERC Docket: RP00-395-003

Original Sheet No. 712 Original Sheet No. 712: Superseded

RATE SCHEDULE DVS (Continued)
DELILVERY VARIANCE SERVICE
FORM OF SERVICE AGREEMENT

levels are actually achieved or only with respect to Quantities below a specified level; (c) only during specified time periods; (d) only to specified Point(s) of Delivery, mainline segments, transportation paths or defined geographical areas; or (e) in a specified relationship to the Quantities actually delivered (i.e., that the rates shall be adjusted in a specified relationship to Quantities actually delivered at the specified Point of Delivery); provided, however, that any such discounted rate set forth above shall be between the Maximum Rate and Minimum Rate applicable to the service provided under this Agreement. In addition, the discount agreement may include a provision that if one rate component which was at or below the applicable Maximum Rate at the time the discount agreement was executed subsequently exceeds the applicable Maximum Rate due to a change in Panhandle's Maximum Rates so that such rate component must be adjusted downward to equal the new applicable Maximum Rate, then other rate components may be adjusted upward to achieve the agreed overall rate, so long as none of the resulting rate components exceed the Maximum Rate applicable to that rate component. Such changes to rate components shall be applied prospectively, commencing with the date a Commission order accepts revised tariff sheets. However, nothing contained herein shall be construed to alter a refund obligation under applicable law for any period during which rates which had been charged under a discount agreement exceeded rates which ultimately are found to be just and reasonable. Any discount(s) shall be effective only on a prospective basis and as specified in the written agreement between Panhandle and DVS Party.

From time to time Panhandle and Shipper may agree to a Negotiated Rate for a specified term for service hereunder. Provisions governing such Negotiated Rate and term shall be set forth on Exhibit A hereto.

ARTICLE 4 - FUEL REIMBURSEMENT

In addition to collection of the rates and charges provided for in Article 3, Panhandle shall retain the percentage provided pursuant to Rate Schedule DVS of the Quantity of Gas applied to the Delivery Variance Service each Day, for reimbursement in kind from DVS Party for fuel usage and unaccounted-for line loss.

ARTICLE 5 - GENERAL TERMS AND CONDITIONS

This Agreement and all terms for service hereunder are subject to the further provisions of Rate Schedule DVS and the General Terms and Conditions of Panhandle's Tariff, as such may be modified, supplemented, superseded or replaced generally or as to the service hereunder. Panhandle reserves the right from time to time to unilaterally file and to make effective any such changes in the provisions of Rate Schedule DVS and/or the General Terms and Conditions, subject to the applicable provisions of the Natural Gas Act and the Commission's Regulations thereunder. Such Rate Schedule and General Terms and Conditions, as may be changed from time to time, are by this reference incorporated in their entirety into this Agreement and made an integral part hereof.

FERC Docket: RP03-479-000

First Revised Sheet No. 713 First Revised Sheet No. 713: Superseded

Superseding: Original Sheet No. 713

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE
FORM OF SERVICE AGREEMENT

ARTICLE 6 - SUCCESSION AND ASSIGNMENTS

This Agreement shall be binding upon and inure to the benefit of any successor(s) to either Panhandle or DVS Party by merger, consolidation or acquisition. Either Panhandle or DVS Party may assign or pledge this Agreement and all rights and obligations hereunder under the provisions of any mortgage, deed of trust, indenture or other instrument which it has executed or may execute hereafter as security for indebtedness; otherwise, neither Panhandle nor DVS Party shall assign this Agreement or its rights hereunder without first having obtained the formal written consent of the other(s).

ARTICLE 7 - NOTICES

Except as otherwise specified, any notice, request, demand, statement or bill provided for in the General Terms and Conditions and any other notice which either Panhandle or DVS Party may desire to give to the other(s), shall be in writing and shall be considered as duly delivered when mailed by registered mail to the Post Office address of the other(s), or at such other address as may be designated by formal written notice. Routine communications and bills (but excluding payments otherwise provided for in the General Terms and Conditions) shall be considered as duly delivered when mailed by registered or ordinary mail or an equivalent. The Post Office addresses of both Panhandle and DVS Party are as follows:

PANHANDLE

Payment: Panhandle Eastern Pipe Line Company

P. O. Box 201202

Houston, Texas 77216-1202

Nomination and Panhandle Eastern Pipe Line Company

Scheduling: Attn: Marketing Operations P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 989-7590 Facsimile: (713) 989-1121

After Hours Pager: (713) 616-3400

Pipeline Emergencies: Panhandle Eastern Pipe Line Company

(Not to be used for Attn: Gas Control any other purpose) P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-5623 Toll Free: 1-800-225-3913

All Other: Panhandle Eastern Pipe Line Company

Attn: Customer Services

P. O. Box 4967

Houston, Texas 77210-4967 Phone: (713) 627-4272 or 1-800-275-7375 Facsimile: (713) 989-1178

FERC Docket: RP00-395-003

Original Sheet No. 714 Original Sheet No. 714 : Superseded

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE
FORM OF SERVICE AGREEMENT

Billing:	DVS	PARTY	
Nomination and Scheduling: (1)			
All Other:			

(1) Please provide street address in addition to mailing address.

FERC Docket: RP00-395-003

 $\textbf{Original Sheet No. 715} \; \texttt{Original Sheet No. 715} \; : \; \; \texttt{Superseded}$

RATE SCHEDULE DVS (Continued)
DELIVERY VARIANCE SERVICE
FORM OF SERVICE AGREEMENT

IN WITNESS WHEREOF, both Panhandle and DVS Party have caused this Agreement to be executed in several counterparts by their respective officers or other persons duly authorized to do so, as of the date first stated above.

DVS PARTY:	PANHANDLE EASTERN PIPE LINE COMPANY:
Ву	Ву
(Please type or print name)	(Please type or print name)
Title	Title
EXECUTED,	EXECUTED,